ResearchOnline@JCU



This file is part of the following work:

Adu, Mary Damilola (2020) *Enhancing diabetes self-management through mobile phone application*. PhD Thesis, James Cook University.

Access to this file is available from: https://doi.org/10.25903/f2b4%2D0c62

Copyright © 2020 Mary Damilola Adu.

The author has certified to JCU that they have made a reasonable effort to gain permission and acknowledge the owners of any third party copyright material included in this document. If you believe that this is not the case, please email researchonline@jcu.edu.au



ENHANCING DIABETES SELF-MANAGEMENT THROUGH MOBILE PHONE APPLICATION

Mary Damilola Adu

BSc Biochemistry (Hons), MSc Epidemiology, Grad Cert Diabetes Education

Submitted in fulfilment of the requirement for the degree of

Doctor of Philosophy

At the College of Medicine and Dentistry

Division of Tropical Health and Medicine

James Cook University

July 2020

Statement of Access

I, Mary Adu, the author of this thesis, understand that this thesis will be made available for use of others. All users consulting this thesis will have to sign the following statement:

In consulting this thesis, I agree not to copy or closely paraphrase it in whole or part without the consent of the author; and to make proper public written acknowledgement for any assistance which I have obtained from it.

Beyond this, I do not wish to place any restriction on access to this thesis.

24th July 2020

.....

Author's signature

Date

Declaration

I declare that this thesis is my own work and has not been submitted in any form for another degree or diploma at any university or institution of tertiary education. Information derived from the published or unpublished work of others has been acknowledged in the text, and a list of references is given.

I declare that the research included in this thesis was ethically conducted, with approval received for each component of the research from the James Cook University Human Research Ethics Committee. The following ethics approval codes relate to the research relevant to this thesis: H7087, H7285, H7716.

24th July 2020

.....

Author's signature

Date

Acknowledgements

To begin with, all glory to God for the grace to start and finish this project. I feel incredibly blessed to have spent these last three years learning and working with my advisors. My sincere gratitude goes to my primary advisor, Associate Professor Bunmi Malau-Aduli for her unrelenting support and guidance in my educational journey prior to and throughout the PhD. Her wealth of experience, knowledge and encouragement are instrumental to the completion of this thesis. The invaluable guidance and support from my secondary advisor, Prof Usman Malabu and advisor mentor, Associate Professor Aduli Malau-Aduli made this project a success. I am deeply grateful for their endless patience and the confidence they have in me to complete this work in a timely manner.

I would also like to thank James Cook University (JCU) and College of Medicine and Dentistry for providing me the financial support to execute this project. I am grateful to Prof Ian Aitkinson and Sarai Viqar of the JCU eResearch Centre for developing the My Care Hub app software. I would also like to thank Professor Helene Marsh for ensuring that I was awarded a fee waiver from JCU for my Graduate Certificate of Diabetes Education. I also appreciate Dr Aaron Drovandi, for his research assistance and editorial support.

I am also indebted to my amiable parents, Presiding Elder David and Deaconess Phebean Adewolu, for their unwavering love, prayers and encouragement over the years. You are the best parents anyone could have. I am sure the completion of this research and award of a PhD will be joyful for you as it is for me. My special thanks goes to my husband, Kayode and sons, Olamiposi and Oluwaferanmi. Without your acceptance, understanding and love, completion of this project would have been difficult to achieve. Kayode, I cherish your encouragement, enthusiasm and indescribable support in ensuring that I achieve this goal. Finally, thanks to my extended family members, in-laws and friends for their support in this journey.

Kayode, Olamiposi and Oluwaferanmi, I dedicate this PhD to you. You all are indeed God's special gifts to me.

Contribution of Others

I would like to thank the following persons and organisations for their support in the following roles, leading to the completion of this thesis.

Supervision

A/Prof Bunmi Malau-Aduli

Prof Usman Malabu

A/Prof Aduli Malau-Aduli

Financial Support

James Cook University International Research Training Program Scholarship

James Cook University College of Medicine and Dentistry Scholarship Top up

James Cook University College of Medicine and Dentistry Open Access Publication and Research Training Funding

A/Prof Bunmi Malau-Aduli, Prof Usman Malabu & A/Prof Aduli Malau-Aduli's Research Funding Contributions

Software Development

Sarai Viqar

Prof Ian Aitkinson

Statistical and Analytical Support

A/Prof Bunmi Malau-Aduli

Dr Aaron Drovandi

Editorial Support

A/Prof Bunmi Malau-Aduli

Prof Usman Malabu

A/Prof Aduli Malau-Aduli

Dr Aaron Drovandi

Dr Kayode Adu

Research Assistance

A/Prof Bunmi Malau-Aduli

Prof Usman Malabu

A/Prof Aduli Malau-Aduli

Dr Aaron Drovandi

Diabetes Australia

British Diabetic Association (Diabetes UK)

National Diabetes Service Scheme, Australia

Abstract

Introduction: Mobile phone applications (apps) offer a convenient, viable and easily accessible resource for patients to access on-going diabetes self-management education and support (DSMES). However, the lack of optimally consistent user engagement remains a significant and challenging limitation of app as tools for diabetes self-management (DSM). These limitations include lack of information on what constitutes standard/essential elements in the development process prior to use as intervention tools, high attrition rates, low engagement levels, sub-optimal educational components and poor consideration of mediating factors that are necessary for longterm behavioural changes. These knowledge gaps are pointers to the need for further improvements on diabetes apps in order to enhance their effectiveness in providing DSMES to patients. Therefore, this thesis aimed to: (1) Develop a framework on the essential elements required in the development of a DSM app, (2) Assess the perceptions of diabetes patients on the mediating factors of skills and self-efficacy for DSM, (3) Assess the perceptions of diabetes patients on the features and educational contents necessary for engagement with apps and ongoing DSM, (4) Develop a novel mobile app for DSMES and assess its usability among diabetes patients, and (5) Pilot-test the new app in order to ascertain its acceptability and feasibility in terms of retention, engagement and efficacy in promoting DSM.

Methods: Participant groups throughout the research reported in this thesis were people with type 1 or type 2 diabetes. The thesis involved three phases of research: i) Inspiration; ii) Development of intervention (an app) and usability testing; and (iii) Pilot testing of the intervention. The first phase comprised three evidence-based studies on the essential elements for developing diabetes apps. The three studies included a systematic review to derive an app development framework; patients' perceptions of their skills and self-efficacy for DSM; and their preferences for features and educational components in diabetes apps with recommendations on how to foster ongoing engagement.

Results from the first phase of the research informed the development of the app and a two-staged usability study, which was conducted to finesse the beta version of the app (second phase). The third phase involved a three-week pilot testing (single-arm repeated measure) of the beta version of the app to assess its feasibility for retention, engagement, efficacy on DSM behavioural change and acceptability by patients.

All primary studies in the three phases were conducted using a sequential explanatory mixedmethods study design, comprising quantitative online surveys and qualitative telephone interviews. All quantitative data were analysed using descriptive and appropriate inferential statistics. Qualitative data were analysed using thematic and concept-driven analyses.

Results and Discussion: The review findings highlighted the lack of consideration of the key elements of app development by app developers and researchers. The key elements identified were: inclusion of health behavioural theory, views of users and clinical experts, data security and privacy as well as pilot testings, for every diabetes app prior to use as an intervention tool in large-scale trials. There was a high correlation between skills set and self-efficacy (r = 0.906). Patients' skills set was a strong predictor ($R^2 = 0.82$) of self-efficacy, a necessary requirement for DSM. Common gaps were evident in identifying and managing the impact of stress on diabetes, exercise planning to avoid hypoglycaemia and interpretation of blood glucose patterns. The most preferred diabetes app features were visual analytics, food nutrient database, blood glucose trackers and personalised education. Recommendations on fostering better engagement with apps were improved functionalities on healthy recipes, actionable goals with reminders, ease of use, data consolidation, customised features, and certified reliable information sources. Specific educational topics of interest to patients were approaches to problem solving and basic guidelines for the management of diabetes.

A novel app (My Care Hub) was then developed following the pragmatic evaluation and use of the findings from the first phase of the research. Using the derived framework, the mediating

components of skills and self-efficacy, patients' preferred app features and educational components as well as their recommendations to improve engagement, My Care Hub aimed to monitor selfmanagement activities, provide access to diabetes educational information and aid diabetes patients' motivation to engage in self-management. Results of the usability test revealed that the app was user-friendly, informative and provided motivation for DSM. Pilot testing of the app demonstrated significant correlations (r=0.835, p=0.03) between enhanced self-management skills and self-efficacy for long-term improved behavioural change and DSMES. Participants also reported high acceptability and retention rates; high levels of engagement with significantly increased self-management activities from pre- to post intervention time periods (p<.01) with an effect size of 0.24. Perceived benefits of the app included increased accountability, clarity of selfmanagement activities/impact and improved awareness of blood glucose levels. Other benefits included highly educational function related to reinforcement of health providers' self-management recommendations, guidance on meal planning and mindfulness of calorie consumption. The inclusion of features and multi-functional components of visual analytics, blood glucose trackers, food nutrient data base and personalised education were highly effective in fostering better engagement with self-management apps, particularly among diabetes patients.

Conclusion: This research identified limitations of existing approaches to the development and effectiveness of apps for diabetes self-management and attempted to proffer a solution by using a systematic approach to explicate the design and development of a novel mobile phone diabetes app (My Care Hub) and evaluate its functionality. The study findings identified shared decision making between patients, clinical experts, researchers and app developers as essential in diabetes app development. Furthermore, consideration of educational reinforcement in the areas of healthy coping with diabetes stress, exercise planning to avoid hypoglycemia and interpreting blood glucose patterns could enhance self-management skills and self-efficacy in patients. In addition, it is important that apps have features to support healthy eating; blood glucose and physical activity monitoring; provide data analytics and educational contents in order to ensure that patients'

preferences are met. Finally, the reported benefits of My Care Hub provide both anecdotal and empirical evidence of support for its multi-feature functionality and comprehensive interventional role in DSMES. This provides a propitious prospect that the constructs of the models and approaches utilised in the development of My Care Hub will be relevant for the development of its future versions and long-term, larger-scale evaluation to further validate its promising results.

Peer – Reviewed Journal Papers with 2020 Impact Factor (IF)

1. Adu MD, Malabu UH, Callander E, Malau-Aduli AEO, Malau-Aduli BS (2018). Considerations for the development of mobile phone applications to support diabetes selfmanagement: A systematic review. *JMIR Mhealth Uhealth*. 6(6):e10115. (IF – 4.301). https://mhealth.jmir.org/2018/6/e10115/

2. Adu MD, Malabu UH, Malau-Aduli AEO, Malau-Aduli BS (2019). Enablers and barriers to diabetes self-management: A multi-national perspective. *PLoS ONE*. 14(6). (IF – 2.870). https://journals.plos.org/plosone/article?id=10.1371/journal.pone.0217771

3. Adu MD, Malabu UH, Malau-Aduli AEO, Malau-Aduli BS (2018). Users' preferences and design recommendations to promote engagement with mobile apps for diabetes self-management: Multi-national perspectives. *PLoS ONE*. 13(12) e0208942. (IF – 2.870). https://journals.plos.org/plosone/article?id=10.1371/journal.pone.0208942

4. Adu MD, Malabu UH, Malau-Aduli AEO, Malau-Aduli BS (2020). The development of My Care Hub mobile phone app to support self-management in Australians with Type 1 or Type 2 diabetes. *Scientific Report*. 10(1):7. (IF – 4.120). https://www.nature.com/articles/s41598-019-56411-0

5. Adu MD, Malabu UH, Malau-Aduli AEO, Drovandi A, Malau-Aduli BS (2020). User retention and engagement with a mobile app intervention to support self-management in Australians with Type 1 or Type 2 diabetes (My Care Hub): Mixed-methods study. Published in *JMIR Mhealth Uhealth*. 8(6):417802. (IF- 4.301). https://mhealth.jmir.org/2020/6/e17802/

6. Adu MD, Malabu UH, Malau-Aduli AEO, Drovandi A, Malau-Aduli BS (2020).
Efficacy and Acceptability of My Care Hub Mobile App to Support Self-Management in Australians with Type 1 or Type 2 Diabetes. *Int J Environ Res Public Health.* 17(7):2573.
(IF – 2.849). <u>https://www.mdpi.com/1660-4601/17/7/2573</u>

Conference Presentations

- Adu MD, Malabu UH, Malau-Aduli AEO, Malau-Aduli BS (2019). Users' Preferences and Design Recommendations to Foster Engagement with Diabetes Mobile Applications. Oral presentation at the James Cook University Research Festival, October, Townsville, QLD, Australia.
- Adu MD, Malabu UH, Malau-Aduli AEO, Malau-Aduli BS (2019). Enablers and Barriers to Diabetes Self-Management. Oral presentation at the Townsville Health Research Showcase, October, Townsville, QLD, Australia.

Publications in Broader Field of Research

1. Adu MD, Malabu UH, Malau-Aduli AEO, Malau-Aduli BS (2019). Mobile application intervention to promote self-management in insulin-requiring type 1 and type 2 diabetes individuals: protocol for a mixed methods study and non-blinded randomised controlled trial. *Diabetes Metab Syndr Obes*.12: 789-800. (IF – 3.319). <u>https://www.dovepress.com/mobile-application-intervention-to-promote-self-management-in-insulin--peer-reviewed-article-DMSO</u>

Table of Contents

Statement of Access	i
Declaration	ii
Acknowledgements	iii
Contribution of Others	iv
Abstract	vi
Thesis Publications	x
Table of Contents	xii
List of Tables	xvii
List of Figures	xviii
List of Abbreviations	xix
Chapter ONE: General Introduction	1
1.1 DIABETES MELLITUS EPIDEMIC	1
1.2 DIABETES SELF-MANAGEMENT	2
1.3 INTERVENTIONS TO IMPROVE DIABETES SELF-MANAGEMENT	2
1.3.1. Ongoing Self-Management Education and Support	3
1.3.2 Mobile Phone Applications for Consistent DSMES	5
1.3.3 Shortcomings of Mobile Apps as Self-Management Education / Support Tool Solutions	
1.4 THESIS AIMS, RESEARCH QUESTIONS AND HYPOTHESES	8
1.4.1 Aims	8
1.4.2 Research Questions	9
1.4.3 Hypotheses	9
1. 5 THEORETICAL FRAMEWORK	
1.5.1 Social Cognitive Theory (SCT)	
1.5.2 Information-Motivation-Behavioural Skills (IMBS) Model	
1.5.3 SCT and IMBS Blended Model	
1.6 METHODOLOGY	14
1.6.1 Research Design	14
1.6.2 Key Participant Groups and Study Phases	16
1.6.3 Analytical Techniques	17
1.6.4 Interpretation of Results	
1.7 CONCEPTUAL FRAMEWORK OF THESIS	19
1.7.1 Thesis Outline	

Chapter	TWO: Literature Review	27
	Review of Literature on Diabetes Mellitus, its Complications and Current Interventio ove Self-Management	
2.1	DIABETES MELLITUS	27
2.2 SE	ELF-MANAGEMENT	31
2.2.	1 Diabetes Self-Management	32
2.2	2 Diabetes Self-Management Education and Support (DSME)	34
	ECHNOLOGY FOR DIABETES SELF-MANAGEMENT EDUCATION AND ORT (DSMES)	38
2.3.	1. mHealth	39
2.3.	2 Limitations in Mobile Apps for Diabetes Self-Management and How to Address Them	40
	Considerations for the Development of Mobile Phone Apps to Support Diabetes Self- ment: A Systematic Review	45
2.4 AF	BSTRACT	45
2.5 BA	ACKGROUND	46
2.6 M	ETHODS	50
2.6.	1 Data Sources and Search Strategy	50
2.6.	2 Selection Criteria	51
2.6.	3 Data Extraction	52
2.6.	4 Quality Assessment	52
2.7 RF	ESULTS	53
2.7.	1 Selection of Studies	53
2.7.	2 Methodological Quality Assessment	55
2.7.	3 Characteristics of Included Studies	55
2.8 DI	SCUSSION	64
2.9 CC	ONCLUSION	72
_	THREE: Enablers and Barriers to Effective Diabetes Self-Management: A Multi- Investigation	74
3.1 AF	- BSTRACT	74
3.2 BA	ACKGROUND	75
3.3 M	ETHODS	79
3.3.	1 Recruitment Procedure	79
3.3.	2. Study Design	80
3.3.	3 Data Analyses	83
3.4 RF	ESULTS	84
3.4.	1 Socio-Demographic and Health Characteristics	84
3.4.	2 Diabetes Self-Management Skills and Self-Efficacy (Confidence)	85

3.4.3 Other Enablers and Barriers to Diabetes Self-Management	91
3.4.4 Barriers to Self-Management	94
3.5 DISCUSSION	97
3.6 CONCLUSION	106
Chapter FOUR: Users' Preferences and Design Recommendations to Promote Enga with Mobile Apps for Diabetes Self-Management: Multi-national Perspectives	
4. 1 ABSTRACT	
4. 1 ADSTRACT	
4.2 DACKGROUND	
4.3.1 Study Design	
4.3.2 Data Analyses	
4.4 RESULTS	
4.4.1 Participant Characteristics	
4.4.2 Preferences for app features and perceived usefulness	
4.4.3 Recommendations to Foster Long-Term Engagement with Diabetes Apps	
4.4.4. Specific Educational Topics Desired in Future Apps to Support Diabetes Self- Management	
4.5 DISCUSSION	129
4.6 CONCLUSION	137
Chapter FIVE: The Development of My Care Hub Mobile-Phone App to Support So Management in Australians with Type 1 or Type 2 Diabetes	
5.1 ABSTRACT	138
5.1 BACKGROUND	
5.2 METHODS	
5.1.1 Stage 1: Pre-development	
5.2.2. Stage 2: Design and Translation of Pre-Development Output into a Smartpho	ne App 145
5.3.3. Stage 3: Usability Testing	153
5.3 DISCUSSION	156
5.4 CONCLUSION	
Chapter SIX: User Retention and Engagement with a Mobile App Intervention to S Management in Australians with Type 1 or Type 2 Diabetes (My Care Hub): Mixed	
Study	162
6.1 Abstract	162
6.2 INTRODUCTION	163
6.3 METHODS	167
6.3.1 Study Design and Sample Size	167
6.3.2 Recruitment and Eligibility	168

6.3.4 Enrolment and Ap	pp Orientation	168
6.3.5 Intervention Over	rview	169
6.3.6. Post-Intervention	1 Data Collection	170
6.3.7 Measures		170
6.3.8 Interviews		172
6.3.9 Data Analysis		172
6.4 RESULTS		173
6.4.1 Participant Chard	acteristics	173
6.4.2 Retention		174
6.4.3 App Engagement.		176
6.4.4 Interview Results.		180
6.5 DISCUSSION		188
6.6 CONCLUSION		196
Chapter SEVEN: Efficacy	and Acceptability of My Care Hub Mobile App to Support Self	-
0	ns with Type 1 or Type 2 Diabetes	
7.1 ABSTRACT		197
7.2 INTRODUCTION		198
7.3 MATERIALS AND	METHODS	200
7.3.1 Study Design		201
7.3.2 Intervention		202
7.3.4 Instrument and D	Pata Collection	206
7.3.5 Analysis		207
7.4 RESULTS		208
7.4.1 Demographics an	nd Health Statistics	208
7.4.2 Outcomes		209
7.4.3 Relationship betw	veen Mediating Factors and Diabetes Self-Management	210
7.4.4. Acceptability		210
7.4.5. My Care Hub's I	Impact on Diabetes Self-Management and Interest in Future Use	212
7.5 DISCUSSION		214
7.6 CONCLUSION		219
Chapter EIGHT: General	Discussion	220
8.1 Chapter Overview		220
8.2 Research Findings an	nd Novel Contributions	220
	evelopments of the Essential Elements Required in Diabetes App	220
8.2.2. Understanding th	he Perceptions of People Living With Type 1 or Type 2 Diabetes on	the
Mediating Factors of S	kill and Self-Efficacy for Diabetes Self-Management	223

8.2.3 Understanding the Perceptions of People Living with Type 1 or Type 2 Features and Educational Contents Necessary for Engagement with Apps and Self-Management.	d ongoing Diabetes
8.2.4 Development of a Novel Mobile App for Diabetes Self-Management Edu and Assessment of its Usability among Type 1 and Type 2 Diabetes Patients.	
8.2.5 Understanding the Potential of the New App for Improved Self-Manage Acceptability	
8.2.6 Theoretical Underpinning Models	
8.2.7 Strengths and Limitations of the Research	
8.2.8 Suggestions for Future Improvement of My Care Hub and Future Resea	arch Directions. 242
Chapter NINE: Conclusions and Recommendations	
9.1 Chapter Overview	
9.1.2 Summary of the Study Findings	
9.2 RECOMMENDATIONS.	
9.2.1 Patients	
9.2.2. Health Professionals	
9.2.3 Researchers and App Developers	
REFERENCES	
APPENDICES	
Appendix 2.1: Some of the developmental considerations in the reviewed st	udies in Chapter
2b	
Appendix 2.2: Other developmental considerations and key clinical outcom	
hemoglobin in the reviewed studies in Chapter	
Appendix 3.1: Interview guide utilised in Chapter 3	
Appendix 4.1: The quantitative instrument utilised in Chapter 4	
Appendix 4.2: Interview guide utilised in Chapter 4	
Appendix 5.1: Usability instrument utilised in Chapter 5	
Appendix 6.1: Other Screen Shots for My Care Hub that were published in	Chapter 6 320
Appendix 6.2: Semi-structured interview guide utilised in Chapter 6	

List of Tables

Table 1.1 Thesis outline, chapter details and publication status of chapters
Table 2.1 General diagnostic criteria for diabetes mellitus
Table 2.2 Study characteristics of the systematic review (n =11)
Table 3.1 Participants skills and self-efficacy ratings to perform diabetes self-
management
Table 3.2 Summary of t-test or ANOVA and Post Hoc results on mean scores by
participants' characteristics
Table 4.1 Demographic and health characteristics of survey respondents ($n = 217$)118
Table 4.2 App use, preferences and perceived usefulness of app features to support diabetes
self-management
Table 5.1 Examples of app messages
Table 6.1 Participant characteristics (n=50)175
Table 6.2 My Care Hub sections and engagement indices
Table 6.3 Summary of Behavioural Intervention Technology (BIT) model as adapted to My
Care Hub intervention
Table 7.1 Observed mean and standard deviations for the outcome measures
Table 7.2 Influence of mediating variables on diabetes self-management
Table 7.3 Participant acceptability ratings of My Care Hub
Table 8.1 Major findings of each chapter and their contribution to thesis

List of Figures

Figure 1.1 SCT and IMBS blended elements for improving behavioural outcomes in diabetes
self-management13
Figure 1.2 Illustration of the conceptual framework of the research reported in this thesis20
Figure 2.1 Flow diagram of study selection process for the systematic review
Figure 5.1 My Care Hub screen shots
Figure 5.2 My Care Hub educational modules screen examples
Figure 7.1 Conceptual framework of My Care Hub based on blended IMBS and SCT
Models

List of Abbreviations

AD	Aaron Drovandi
AEOMA:	Aduli EO Malau-Aduli
ANOVA:	Analysis of Variance
App:	Application
BG:	Blood Glucose
BGL:	Blood glucose levels
BMA:	Bunmi Malau-Aduli
BP:	Blood pressure
CASP:	Critical Appraisal Skills Program
CHO:	Carbohydrate
COREQ:	Consolidated criteria for reporting qualitative research
DM:	Diabetes Mellitus
DSM:	Diabetes self-management
DSME:	Diabetes self-management education
DSMS:	Diabetes self-management support
DSMES:	Diabetes self-management education and support
EC:	Emily Callander
HCP:	Health Care Provider
IA:	Ian Aitkinson

ID:	Identification Number
IMBS:	Information Motivation Behavioural Skills Model
JBI:	Joanna Briggs Institute
JCU:	James Cook University
MCH:	My Care Hub Application
MDA:	Mary Damilola Adu
mHealth:	Mobile health
PRISMA:	Preferred Reporting Items for Systematic Reviews and Meta-
	Analyses
RQ:	Research Question
SD:	Standard Deviation
SV:	Sarai Viqar
T1D:	Type 1 Diabetes
T2D:	Type 2 Diabetes
TTM:	Transtheoretical Model
UM:	Usman Malabu
UK:	United Kingdom
USA:	United States of America
WHO:	World Health Organisation

Chapter ONE: General Introduction

1.1 DIABETES MELLITUS EPIDEMIC

Diabetes mellitus is considered as an epidemic because of its exponentially increasing prevalence (Bassett, 2005; Zimmet, 2017). International Diabetes Federation (IDF) estimates in 2019, indicated a global prevalence of 463 million adults living with type 1 or type 2 diabetes and predicted to reach 578 million in 2030 in the absence of any intervention. About 163 million adults with type 1 or 2 diabetes live in the Western pacific region which includes Australia (International Diabetes Federation, 2019), making it the fastest growing chronic condition across all ages in Australia. In 2014-2015, approximately 1.2 million (5.1%) Australian adults (18 years and above) had type 1 or type 2 diabetes equivalent to a 1.5 % increase over the 3.6% reported in 2004-2005. The majority (85%) of these cases were type 2 diabetes (Sainsbury et al., 2018; Australian Institute of Health and Welfare, 2019a). Specifically, adults living in areas with the greatest socio-economic disadvantage in rural/remote areas of Australia have three times the rate of diabetes as those living in areas of socioeconomic affluence (9% and 3% respectively) (National Rural Health Alliance International, 2011; Australian Institute of Health and Welfare, 2014). In 2011, diabetes was recognised as the 12th largest disease burden in Queensland, Australia, accounting for 2.4% of the total burden (Department of Health, Queensland Government, 2017). In 2018, there were 242, 061 registered cases of diabetes in Queensland, which equates to 4.8% of the population (Department of Health, Queensland Government, 2018).

People with diabetes are predisposed to major complications, resulting in hospitalisation, financial burden and could eventually lead to death. In Australia, diabetes contributed to 11% of all deaths in 2017 (Australian Institute of Health and Welfare, 2019b) and the health-care system attributed \$2.7 billion of its total disease expenditure to diabetes care in 2015-16 (Australian Institute of Health and Welfare, 2015). This represented 97% increase from \$1,507

million in 2008-09 (Australian Institute of Health and Welfare, 2008). Given the increasing prevalence of diabetes and its impact in Australia, there is an urgent need for improved methods to promote and facilitate proactive diabetes self-management necessary to prevent the development of complications and promote overall good health outcomes for patients.

1.2 DIABETES SELF-MANAGEMENT

Self-management of diabetes relied heavily on lifelong daily activities, which include behavioural changes and adherence to healthy lifestyles that are necessary for the reduction or prevention of complications (Shrivastava, Shrivastava & Ramasay, 2013). Patients self-manage by making choices and decisions to engage in therapeutic regimen behaviours. These include: healthy eating, physical exercise, monitoring of blood glucose, complying with medications, good problem-solving and healthy coping. Other steps include risk-reduction in the areas of smoking cessation, keeping medical appointments and limiting alcohol consumption (Shrivastava, Shrivastava & Ramasay, 2013). The absolute frequency or consistency in performing regimen behaviours relates to the level of self-management and is directly correlated with glycaemic control and improved health outcomes (Shrivastava, Shrivastava & Ramasay, 2013; Jarvis et al., 2010; Norris et al., 2002).

1.3 INTERVENTIONS TO IMPROVE DIABETES SELF-MANAGEMENT

Providing interventions to improve diabetes self-management is an integral component of IDF's global plan on managing diabetes (The International Diabetes Federation, 2003). Many countries including Australia have adopted the IDF's recommendations to improve patients' behavioural changes for self-management, including the imperative provision of Diabetes Self-Management Education (DSME) to patients (International Diabetes Federation, 2010; Australian Government Department of Health, 2015). The major aim of DSME is to develop the knowledge base and skills of patients on targeted self-management behaviours necessary for diabetes care (Mensing et al., 2006). The ultimate goal is an eventual reduction of the onset

and/or advancement of complications (Jarvis et al., 2010) with consequential improvement in health status and quality of life. Although DSME had been previously shown to have positive effects on patient self-management behaviours, training had been generally infrequent (Siminerio, 2006) and the benefits declined one to three months after the intervention ceased (Norris et al., 2002). The unsustained benefits suggest that DSME does not in itself, guarantee long-term and sustainable self-management behaviours among patients. Therefore, patients require consistent and ongoing support to sustain their diabetes self-management behaviours in the long-term.

1.3.1. Ongoing Self-Management Education and Support

DSME promotes knowledge and awareness of the importance of self-management, however, it is typically provided as a formal but irregular program in an outpatient service conducted at a hospital/health facility (Powers et al., 2017). Conventionally, provision of DSME to patients occurs at diagnosis, annually, when a new complication sets in or during transition in care (Power et al., 2017). It is a short-term program with or without some degree of follow-up. This irregular approach is unable to sustain the level of self-management needed on a long-term basis due to the chronic nature of diabetes. Effective long-term management of diabetes requires programs that support the continued enhancement of self-management skills, behavioural strategies and metabolic improvement following DSME sessions. Therefore, it has been advocated that diabetes self-management support (DSMS) should be provided in addition to DSME on an ongoing basis (Powers et al., 2017). This ongoing support for patients sustains the self-management gains made during the DSME process. DSMS builds the resilience needed to overcome barriers, sustains coping skills for ongoing self-management demands and facilitates required behavioural changes for life transitions (Shrivastava, Shrivastava & Ramasay, 2013; Powers et al., 2017; Coyle, Francis & Chapman, 2013). DSMS can be delivered anywhere, at any time beyond and outside of formal health care facility. The type of support provided can be behavioural, educational, psychosocial or clinical (Hass et al., 2012) and could be delivered by a trained supporter: educator, pharmacist, peer, primary care practice staff etc. (Linda et al., 2013).

National standards for DSME and DSMS have been amalgamated and renamed diabetes selfmanagement education and support (DSMES) to reflect the value of ongoing support and multiple services, which are essential for improved health outcomes in patients (Powers et al., 2017; Sherifali et al., 2016). DSMES is an ongoing process of facilitating knowledge, skills and ability necessary for diabetes self-care, as well as activities that assist a person in implementing and sustaining behaviours needed to manage his or her condition on an ongoing basis, beyond or outside of formal self-management educational training (Beck et al., 2019). DSMES has been recommended as a critical element of care for all people with diabetes (Powers et al., 2017). However, adherence to the guidelines has not been possible. For example, the guideline stipulate that patients should be offered structured DSME program at the time of diagnosis with a minimum of an annual update and review (Craig et al., 2011; Royal Australian College of General Practitioners, 2016). However, a national survey conducted in 2011 revealed that approximately half of Australian adults with type 1 or 2 diabetes had never received this intervention, nor ongoing DSMES (Peight et al., 2011). A similar survey was repeated in 2016 and reported no improvement (Ventura et al., 2016). Barriers to ongoing diabetes support in Australia have been attributed to under-resourced diabetes education (Kennedy & Dunning, 2017) and limited access to speciality care services and community resources (Speight et al., 2011; Manski-Nankervis et al., 2014; Dao et al., 2019). Barriers related to financial limitations, distance to health care and inadequate resourcing of speciality providers are particularly prevalent in rural/remote communities (Speight et al., 2011; Overland, Yue & Mira, 2001; Wan et al., 2008).

These barriers limit the necessary care required by patients and therefore pose as enormous challenges to policy makers on how to ensure ongoing support to all patients including those living in rural and remote areas. One potential avenue that has been explored in addressing the problem of accessibility of ongoing DSMES, is the use of mobile Health (mHealth) technology. mHealth is particularly feasible and important for reaching people in resource-poor settings with limited access to speciality providers and care but have connectivity to mobile phone networks (Aranda-Jan, Mohutsiwa-Dibe & Loukanova, 2014; Källander et al., 2013).

1.3.2 Mobile Phone Applications for Consistent DSMES

mHealth interventions are typically behavioural change strategies operationalised for delivery using "*computer devices that are intended to be always on or carried on the person throughout the day*" (Riley et al., 2011). A major example of mHealth is the use of mobile phone application (app). This is a software program embedded in smartphones, with persuasive attributes intended to change users' attitudes or behaviours in a pre-determined way (Fogg 2002). App interventions have leveraged on the added-value that mobile phones are ubiquitous and valued by users (Patrick et al., 2008; Miller, 2012; Boschen & Casey, 2008), making it possible to deliver helpful behavioural change content to individuals as they go about their normal daily activities (Miller, 2012; Boschen & Casey, 2008; Preziosa et al., 2009; Kitsiou et al., 2017). Apps offer a convenient, viable and easily accessible resource for patients to access continuous DSMES outside the clinic environment. They are effective in improving patients' self-management behaviours and health outcomes in many part of the world (Kitsou et al., 2017; El-Gayar, 2013; Hou et al., 2016), including Australia (Kirwan et al., 2013).

1.3.3 Shortcomings of Mobile Apps as Self-Management Education / Support Tools and Possible Solutions

The lack of optimal user engagement with apps remains a significant challenge (Short et al., 2015; Eysenbach, 2005) because a significant proportion of app users disengage from repeat

use, thus limiting apps as an effective DSMES tool. Quinn et al., 2011, utilised a mobile app intervention among people with type 2 diabetes and recorded 23.4% attrition rate at 12 months. A similar observation was made by Kirwan et al., 2013 in their evaluation of mobile app use in patients with type 1 diabetes, where 26.3% of the participants did not complete the study. Additionally, many of the currently available diabetes apps have little or no self-management educational component. A study found only 20% of apps had an educational component, of which only one-fifth delivered personal feedback (Chomutare, 2011). This denotes that fully harnessing the capabilities of mobile phones to deliver real-time feedback and diabetes education is largely under-explored. When a health tool is unable to meet patients' essential requirement, it is at risk of abandonment (Bickmore, Schulman & Yin, 2010), and can contribute to the high attrition and low engagement levels reported in diabetes apps usage. Therefore, further research is required to explore the main reasons for the high attrition rates and low engagement levels and to identity possible solutions.

Furthermore, most diabetes app interventions have only reported clinical efficacy or behavioural outcomes of the intervention without any explanation of the mediating factors underlying such results (Kirwan et al., 2013; Quinn et al., 2011; Rossi et al., 2010a; Berndt et al., 2014; Holmen et al., 2014). Nearly all the results of these studies range from none to low sustained effects of the intervention, which may be due to poor exploration of the necessary mediating factors. Constructs in behavioural theories are essential considerations in designing interventions because they can help developers and researchers to understand the causal effect of such behavioural interventions (Baranowski, Anderson & Carmack, 1998). This knowledge gap in the efficacy of existing diabetes apps intervention signifies the need for critical appraisal of the limitations and development of effective innovative changes that may result in improved ongoing self-management behaviours and sustained health outcomes for diabetes patients.

The limitations of apps as a tool for DSMES are related to lack of information on what constitutes standard/essential elements in the development process prior to use. Interventions rooted in poor consideration of mediating factors that are necessary for long-term behavioural changes can lead to limited success. Furthermore, high attrition rates and low engagement levels (Eysenbach, 2005) as well as sub-optimal educational components are common in most apps targeting diabetes self-management (Chomutare et al., 2011; Liang et al., 2011).

Despite the theoretical emphasis on the development of guidelines and adoption of evidencebased practice in the advancement of an effective suite of behavioural health interventions (Kohatsu, Robinson & Torner, 2004; Lhachimi, Bala & Vanagas, 2016), and to the best of our current knowledge, no guideline is available for diabetes apps development. The lack of standards and evidenced-based principles are responsible for the often reported non-replicable and low-quality health interventions (Glasgow & Emmons, 2007). This gap in the literature necessitates research that provides a framework on minimum standard requirements for the development of diabetes apps that serve as effective tools for DSMES.

The development of effective DSMES app interventions requires creative innovations that draw upon theoretical and evidence-based best practice. These are in addition to utilising systematic processes to help combat attrition and incorporate major mediating components (skills and self-efficacy) to foster improved patient engagement and behavioural health outcomes. In order to adequately self-manage diabetes, patients must acquire the necessary skills resulting from knowledge of the disease and adequate understanding of the interrelationships between various self-management activities and impacts on health outcomes (Persell et al., 2004). Skills is an essential pre-condition to self-efficacy, which is a prominent concept in self-management because successful self-management of any disease stems from a sense of confidence (self-efficacy) in one's self-management abilities (Sarkar, Fisher & Schillinger, 2006). A diabetic patient who has adequate sense of self-efficacy has better

7

confidence in his/her ability to evaluate self-management performance. This could be in terms of diet adjustment, hypoglycaemia prevention or engaging in appropriate physical exercise (Beckerle & Lavin 2013; Yao, 2019). Therefore, incorporation of these strategies may provide remarkable improvements over existing diabetes app interventions with tangible long-term behavioural health outcomes for diabetic patients.

1.4 THESIS AIMS, RESEARCH QUESTIONS AND HYPOTHESES

1.4.1 Aims

Due to lack of guidelines for the systematic development of diabetes apps, high attrition rates associated with previous diabetes apps, low engagement of patients with ongoing selfmanagement and inadequate educational components in diabetes app intervention, this thesis aims to:

1. Develop a framework/guideline on the essential elements required in a diabetes app development process prior to use, as an intervention for self-management in patients with type 1 or type 2 diabetes;

2. Assess the perceptions of people living with type 1 or type 2 diabetes on the mediating factors of skills and self-efficacy for diabetes self-management;

3. Assess the perceptions of people living with type 1 or type 2 diabetes on: (i) features and (ii) educational contents necessary for engagement with apps and ongoing diabetes self-management.

4. Develop a novel mobile app for diabetes self-management education and support and assess its usability among patients with type 1 or type 2 diabetes. 5. Pilot-test the new mobile app in order to ascertain its feasibility in terms of retention, engagement and efficacy in diabetes self-management as well as acceptability among patients with type 1 and type 2 diabetes.

1.4.2 Research Questions

Four research questions (RQ) were asked in order to address the thesis' aims. These are based on the premise that developing a new mobile diabetes app following a standard framework for an effective DSMES tool in addition to considering user needs on app features/functionalities and educational components will improve user' engagement and facilitate behavioural change in diabetes self-management. This is in the light of identifying, establishing and using mediators, which could foster improved patient' participation in self-management using the app.

RQ1: What are the necessary elements in an app development process to foster an effective diabetes self-management education and support intervention?

RQ2: To what extent do skills and self-efficacy stimulate behavioural change for selfmanagement in patients with type 1 and type 2 diabetes?

RQ3: What are the preferences of patients with type 1 and type 2 diabetes as regards features and educational contents in apps that will foster retention and engagement with diabetes self-management activities?

RQ4: To what extent would a newly developed diabetes app stimulate improved diabetes selfmanagement and acceptability among patients with type 1 and type 2 diabetes?

1.4.3 Hypotheses

In line with the afore-stated aims and research questions, it was hypothesised that:

- 1. A systematic review and evaluation of evidence from current literature on the impact of diabetes app intervention on patient health outcomes will assist in the establishment of guidelines that outline important considerations required for the development of effective mobile phone apps to support diabetes self-management.
- 2. Skills and self-efficacy are important mediating factors that stimulate behavioural change and improved self-management in people with type 1 and type 2 diabetes.
- 3. Needs analysis among patients with type 1 and type 2 diabetes will elucidate essential features and educational contents in apps to foster retention and engagement with diabetes self-management activities.
- 4. The newly developed mobile phone app will be an effective and acceptable diabetes self-management education and support tool.

1.5 THEORETICAL FRAMEWORK

To adequately test the stated hypotheses, it is important to hinge the study on an appropriate health behavioural theory. Health-related behavioural change interventions may be more effective if grounded in an appropriate theory (Craig et al., 2008; Campbell et al., 2007). Theoretical models provide insights into underlying principles by identifying target mediators that are antecedents of behaviours, causal determinants of change and their mechanisms of action in an intervention (Hardeman et al., 2005). Theoretical models that are regularly utilised for planning and evaluating public health behavioural change interventions include the following: Social Ecological Model (SEM), Theory of Reasoned Action/Planned Behaviour (TPB), Health Belief Model (HBM), Transtheoretical Model (TTM), Social Cognitive Theory (SCT) (Glanz, Rimer & Lewis, 2008) and Information-Motivation-Behavioural Skills (IMBS) models (Fisher, Fisher & Harman, 2003). Each of these theoretical models specifies various concepts that can be influenced by behavioural change in an individual or population. In this thesis, the self-efficacy construct of the SCT model and the information and motivation

constructs of IMBS model were blended to guide the development of the mobile app intervention.

1.5.1 Social Cognitive Theory (SCT)

The SCT is one of the most widely used conceptual frameworks for understanding and modifying health-related behaviours targeting the management of chronic diseases (Glanz, Rimer, Lewis, 2008; Newman, Steed, & Mullingan, 2004). As the corner stone of effective disease self-management interventions, the SCT has demonstrated positive behavioural changes resulting in improved health outcomes (Lyons et al., 2014; Cotter et al., 2014). It started as the Social Learning Theory where it was described as the link between behaviourism and cognitive approach. Unlike many other theories of behavioural change in health promotion, the SCT considers the unique ways in which individuals acquire and maintain behaviour. The current version (1998) of SCT posits a multifaceted causal structure in which self-efficacy beliefs operate together with knowledge of health risks and benefits, goals, outcome expectations, social and structural impediments to change and perceived facilitators of behavioural change (Bandura, 1998). Self-efficacy occupies a pivotal regulatory role in the causal structure of SCT and it is a core belief that directly leads to behavioural change (Bandura, 1998).

1.5.1.1. Self-efficacy

Self-efficacy refers to an individual's thoughts and beliefs in his or her ability to perform specific actions required to attain preferred goals. The choice of what to do, effort to invest in such activities and how long to persevere in the face of barriers and experiences of failure are influenced by people's judgement of their self-efficacy (Bandura, 1999). Self-efficacy is the belief in what one can do with whatever skills one has (Bandura, 1999) and it is one of the most widely used key constructs in health psychology for modifying health-related behaviours (Ashford, Edmunds & French, 2010). Improved self-efficacy is expected to increase an

individual's move to a higher stage of behavioural change, decreasing the risk of relapse and it is a prerequisite for self-management (Bandura, 1999). In the context of diabetes, self-efficacy is a prominent concept with positive correlation with self-management. Self-efficacy is a key determinant for the adoption and maintenance of health related-behaviours and subsequent health outcomes when used as a guide to develop interventions for the management of chronic diseases including diabetes (Cotter et al., 2014; Avery et al., 2015; Tougas et al., 2015; Steel et al., 2014; Barlow et al., 2002). It has been shown to have a strong positive correlation with self-management of diabetes (Mishali, Omer & Haymann, 2011). Given the precedence of selfefficacy in diabetes related research, self-efficacy construct was utilised as an underlying mechanism to influence behavioural change effort for diabetes self-management in this study.

1.5.2 Information-Motivation-Behavioural Skills (IMBS) Model

Information-Motivation-Behavioural Skills (IMBS) model promotes evidence-based and usercentered approach of knowledge implementation in health related situations (Fisher, Fisher & Harman, 2003; Barlow et al., 2002; Fisher & Fisher, 1992; Fisher et al., 1996). It was originally developed in response to the HIV epidemic to predict HIV prevention behavior (Fisher et al., 1996; Fisher, Fisher & Harman, 2003). It was applied successfully in designing interventions that improved and predicted adherence to medication among individuals living with HIV (Fisher et al., 1996; Horvath, Smolenski & Amico, 2014) and diabetes (Mayberry & Osborn, 2014). The IMBS has a broad application potential in health promotion practice where it provides a framework for understanding and promoting disease preventive behaviours among populations (Fisher, Fisher & Harman, 2003; Fisher & Fisher, 1992). The model focuses on a set of informational, motivational and behavioural skill constructs (factors) associated with disease management. The model specifies that behavioural change occurs mainly as a result of changes in behavioural skills sequel to the effect of information and motivation interventions (Fisher, Fisher & Harman, 2003). Information and motivation influence behaviours independently, and in large part, indirectly through the model's third construct – the behavioural *skills* needed to perform self-management behaviours. IMBS has been used in diabetes cases where diabetes-educators and health promotion researchers have consistently found its strong correlation with behavioural outcomes. Information and motivation influences behavioural skills in diabetes patients and ensures that they have strategies necessary to perform the expected behaviours. This ultimately builds confidence (self-efficacy) in a patient's ability to perform his/her self-management behaviours (Gao et al., 2013; Jeon & Park 2018; Osborn et al., 2010). In this study, in addition to the self-efficacy construct of SCT, information, motivation and skills constructs of the IMBS model were also utilised.

1.5.3 SCT and IMBS Blended Model

In this research, the self-efficacy construct of SCT model (Bandura, 1999) and the information and motivation constructs of IMBS model (Fisher, Fisher & Harman, 2003) were blended to guide the development of the intervention. This is due to their precedence of use in technologybased behavioural change research (Poddar et al., 2010) and most importantly because they both identify significant constructs (mediators - skills and self-efficacy) needed for successful long-term maintenance of self-management behaviours required for improved health outcomes in people with chronic diseases including diabetes (Deakin et al., 2005; World Health organization, 2003).

Figure 1.1 shows a diagrammatic presentation of the SCT and IMBS blended elements for improving behavioural outcomes in diabetes self-management. DSMES intervention components related to information and motivation were utilised to enhance patients' skills and self-efficacy and consequently increase their self-management behaviours.



Figure 1.1: Framework for the blended SCT and IMBS elements for improving diabetes selfmanagement behavioural outcomes.

1.6 METHODOLOGY

1.6.1 Research Design

This study utilised a sequential explanatory mixed-methods design approach using quantitative and qualitative data collection methods and analysis to address the research aims, questions and to test the postulated hypotheses. The results of both methods were triangulated to strengthen the research outcomes (Kennedy, 2009). Quantitative studies such as online surveys have the capacity to produce objective and statistically significant results. These can guide an effective intervention, although they lack the capacity to explore deeper underlying meanings and explanations of research findings due to limited responses (O'Leary, 2014). Furthermore, they overlook respondent experiences and perspectives and are unable to assess how respondents interpret questions asked (O'Leary 2014). Qualitative studies such as interviews
are able to overcome these quantitative research limitations, particularly in gaining detailed understanding of respondents' opinions and perceptions about the effectiveness or ineffectiveness of an intervention. These studies however, are not focused on generalisability due to their small sample sizes and subjective nature of data generated (O'Leary, 2014). Triangulating results from both quantitative and qualitative approaches overcomes weaknesses presented by each research method and fosters improved understanding of the phenomenon of interest (Kennedy, 2009).

The mixed-methods approach was chosen in order to understand not only what patients preferred in an app to support their management, but also their perceptions on mediating factors for self-management. The outcomes of these studies were used to develop a new diabetes app as an intervention tool for DSMES. Thereafter, the efficacy of the intervention was assessed using a mixed method approach.

This research was conducted in three phases: Inspiration, app development and pilot testing. In all phases, quantitative data collection used online surveys using different validated tools to gather the perceptions of participants as appropriate to the aims of each phase of the study. Qualitative interviews were also conducted to build upon survey findings. Within each of the primary studies in each phase, a triangulation design was utilised to independently crossvalidate internal findings between the online quantitative survey and qualitative interview components. This approach is widely used because it offers a comprehensive understanding through interpretations of convergent and divergent findings (Heale & Forbes, 2013). In addition, the qualitative components of the research were carefully designed based on the findings from the quantitative data, to explain the reason behind the observation from the quantitative data, thus deepening the understanding of the underlying phenomenon. The research phases were sequenced and results of the preceding phase integrated into the next. The main limitation of the sequential method is the longer time lag and tendency for unexpected contradictory or divergence between the results in each phase. Nevertheless, divergent findings can provide a new and better explanation for the research question under investigation (Morse, 2003).

1.6.2 Key Participant Groups and Study Phases

There were three participant groups in this research: People without diabetes and diabetic patients with type 1 or type 2 diabetes.

Inspiration Phase: Participants involved in the first phase of the research were a multinational audience (USA, Asia, Australia and the United Kingdom) with type 1 or type 2 diabetes. This enabled the collection of a relatively generalizable dataset relating to patients' perspectives on preferences for app features that support diabetes self-management, recommendations for promoting engagements with such apps as well as specific educational information desired in apps. In addition, the data set inquired about common gaps in skills and self-efficacy as mediating factors for self-management in diabetic patients.

Development of Intervention and Usability Testing Phase: This phase involved the development of a new mobile phone app for DSMES and its usability testing. There were two sub-group of participants in the usability-testing phase - Australians who do not have diabetes and those who have type 1 or type 2 diabetes. Australian residents participated in this phase because one of the features in the newly developed intervention was aimed at reinforcing Australian clinical best practices related to the recommended blood glucose levels for patients with type 1 and type 2 diabetes. Participants who did not have diabetes were involved in the first stage of the usability testing because it assessed user interface and navigation issues of the newly developed intervention. Patients with diabetes took part in the stage 2 of the usability

testing which included assessment of the app's value as a tool for supporting diabetes selfmanagement.

Pilot Testing Phase: Phase 3 participants included people with type 1 or type 2 diabetes living in regional, rural and remote Northern Queensland, Australia. Participants in this phase were restricted to only Australian residents in rural and remote areas because of the higher prevalence of diabetes (6.7% compared to 4.7%) (Australian Bureau of Statistics, 2015), and shortage of health workers, hence poorer service availability for ongoing DSMES (Struber, 2004; Hussain et al., 2015). Therefore, rural and remote residents are the most needy of the new intervention for providing DSMES services. The preliminary efficacy of the intervention for diabetes self-management activities and acceptability by participants as a tool for DSMES were elucidated in this phase.

1.6.3 Analytical Techniques

Four primary methods were utilised in analysing the quantitative data in this research. Descriptive statistics including frequencies, percentages, means and standard deviations were used to analyse the demographic/health characteristics of participants and all dependent variables. Parametric tests including Pearson's Chi-Squared test, independent sample t-test and Analysis of Variance (ANOVA), were used to compute the relationships between dependent and independent variables within each study. Paired sample t-test for parametric repeated measures analyses and Wilcoxon Signed Rank test for non-parametric repeated measures analyses) were used to evaluate changes in pre and post intervention efficacy outcomes. Furthermore, Pearson's correlation analysis was used to measure the strength of association between two continuous variables, while multiple regression analysis was a predictive tool for estimating the contributions of different continuous independent variables on dependent variables in each study. All quantitative data were analysed using SPSS statistical software version 24 (IBM SPSS Statistics, 2018).

The primary method for analysing the qualitative data was inductive thematic analysis. This method was performed in line with Braun and Clarke's (Braun & Clarke, 2006) methodology of familiarization with the data, generating initial codes, organisation of codes into themes, reviewing themes, combination of themes into overarching themes that accurately depict the data and production of a report. This method allowed for easy approach to analysis while providing a rich and detailed account of the data (Braun & Clarke, 2006). Content analysis through objective, systematic and quantitative description of the manifest content of textual data (Hsieh & Shannon, 2005) was used for the open-ended survey in phase 2 of the research. In Phase 3, concept-driven analysis using a theoretical framework was utilised for the qualitative data from interviews. Concept-driven analysis involved iterative stages of labelling, classifying and organising data into main themes, concepts and categories in a theoretical framework (Ritchie et al., 2003). This method fostered understanding of participants' views and experiences in relation to the use of the new intervention. Qualitative data analyses were preformed using both Nvivo (version 11) and a Microsoft word processor (Microsoft word for Windows 2016).

1.6.4 Interpretation of Results

Triangulation of the overall dataset by integrating both quantitative and qualitative findings within every primary study in this research provided meaningful interpretation of the research results as a whole. Phase 1 study findings were integrated into the development of the new mobile app in Phase 2 of this research. Usability testing results of the new app were incorporated into re-designing and improving the quality of the app before moving onto Phase 3, where the resulting output from Phase 2 was pilot tested. Findings from Phase 3 were compared with previous findings from Phase 1 to identify both consistent patterns and unique results, especially relating to perceived factors for patients' engagement with an app and

essential features necessary in an app to foster participation in self-management behavioural activities.

1.7 CONCEPTUAL FRAMEWORK OF THESIS

Figure 1.2 presents the conceptual framework of this research. It highlights the research questions and how they were answered by the 3 phases of the research.

Phase 1: Inspiration

As a first step of this research, a comprehensive literature review was conducted to provide a detailed background of type 1 and type 2 diabetes mellitus overview, the DSMES concept and the emergence of mobile apps as key tools for DSMES (Chapter 2-Part A). This was followed by a systematic review of literature to establish evidence regarding important factors and considerations in a mobile app development process to ensure its effectiveness as a DSMES tool (Chapter 2-Part B). The review focused on answering the first research question to know the necessary elements in an app development process in order to foster effective DSMES intervention.

Furthermore, the inspiration phase elucidated information on the understanding the mediating factors (skills and self-efficacy) that are important for improving diabetes self-management (Chapter 3). Prior to the commencement of the research reported in this thesis, a number of cross-sectional research reporting on skills and self-efficacy as mediating or enabling factors for fostering self-management in patients were identified (Persell, 2004; Aljasem et al., 2001; Johnston-Brooks, Lewis & Garg, 2002; Byers et al., 2016). However, most of these studies considered only a few aspects of diabetes self-management, type 1 (Johnston-Brooks, Lewis & Garg, 2002) or type 2 (Aljasem et al., 2001; Byers et al., 2016) diabetes alone, and rarely had an international audience. Therefore, this chapter contributed to answering the second research question aimed at establishing the relationship between skills and self-efficacy and

knowing the extent to which they stimulate behavioural change for self-management in an international audience of patients with type 1 and type 2 diabetes. In addition, the inspiration phase answered the third research question aimed at exploring users' needs and preferences in app intervention for DSMES, offering recommendations to promote engagements with such apps and specific educational information desired in apps (Chapter 4).

Phase 2: Development of intervention

Findings from the inspiration phase were used to develop a new mobile phone app for DSMES (Chapter 5). In mapping information from the inspiration phase, it was found that a deliberate strategy used by health care givers to increase patients' self-efficacy involved self-monitoring of clinical and health behavioural activities (Steed et al., 2014; Hernandez-Tejada et al., 2012). Intervention targeting self-monitoring enhances self-efficacy, which in turn, positively impacts decision making for self-management behaviours (Cotter et al., 2014; Mishali, Omer & Heymann, 2011; Lalić et al., 2017; Cummings et al., 2011; Lorig et al., 2009; Gleeson-Kreig, 2006). This emphasises the importance of including self-monitoring elements in a behavioural health intervention.

From the systematic review (Chapter 2-part B), the use of health behavioural theory, user and clinical expert involvement, data security and privacy, and pilot testing were perceived as the most important factors to be included in a diabetes app development process to foster an effective intervention. Therefore, these considerations were followed in the second phase.

Phase 3: Pilot testing

The developed app was evaluated in subsequent chapters of this thesis (Chapters 6 and 7) to assess its capacity to retain and engage users. The efficacy of the intervention on diabetes self-management activities and acceptability by participants as a tool for DSMES were also elucidated.

Overall, both phases 2 and 3 of the study were aimed at answering the fourth research question on assessing the extent to which a newly developed intervention could stimulate improved diabetes self-management and its acceptability among end users with type 1 and type 2 diabetes.

Conceptual Framework of Thesis



RQ (Research Questions); IMBS (Information motivation behavioural skills); SCT(Social cognitive theory)

Figure 1.2. Illustration of the flow chart of research activities reported in this thesis

1.7.1 Thesis Outline

This thesis comprises of nine chapters, which taken together, aim to address the research aims, hypotheses and questions stated previously.

Chapter 1 – General introduction gives an introductory background to prevalence of diabetes mellitus epidemic, existing knowledge gaps in mobile phone apps for diabetes self-management support interventions and the need for better user engagement. This chapter also discusses strengths and limitations of currently available apps for diabetes self-management, the theoretical conceptual framework for the development of a new diabetes app intervention, thesis aims, research questions, hypotheses, research design and methodology justification.

- Chapter 2 –Literature Review is divided into two parts: Part A provides an overview of diabetes mellitus, its complications and interventions for self-management. It discusses the evolution of mobile apps as health intervention tools, highlighting barriers to successful use and improvement opportunities for self-management education and support in type 1 and type 2 diabetes. Part B is a systematic review that provides evidence of key sets of essential requirements in the process of developing an effective mobile phone app for DSMES. This chapter also answered RQ1.
- Chapter 3 reports the findings of an online survey and interviews of an international audience of type 1 or type 2 diabetes. It also answered part of RQ2 to establish empirical evidence for the relationships between patients' skills and self-efficacy including their mediating influence on diabetes self-management.
- Chapter 4 reports the findings of a combination of online survey and interviews of an international audience of type 1 or type 2 diabetes to gain in-depth understanding of patients' preferences of app features and educational content for inclusion in diabetes apps to foster engagement and improved self-management. The chapter answers RQ3. The data findings from this chapter informed the content development of a new mobile diabetes app presented in Chapter 5.
- Chapter 5 describes the process for the development and usability testing of a novel mobile app called 'My Care Hub', for self-management education and support of people with type 1 and type 2 diabetes. This study answered part of RQ4.
- Chapter 6 evaluates retention and participants' engagement with the newly developed app through pilot testing and contributes answers to RQ4.

- Chapter 7 assesses the preliminary efficacy of the new app for diabetes selfmanagement and its patient acceptability as a DSMES tool. The findings contribute to answering RQ4.
- Chapter 8 discusses the overall findings of this research and implications of the findings for patients, health practitioners, patients and app developers.
- Chapter 9 draws an overall conclusion and makes recommendations for future research in the development and implementation of diabetes apps as self-management education and support tools.

1.8 SUMMARY

This chapter outlines the prevalence of diabetes mellitus and the need for ongoing selfmanagement education and support in patients. It enumerates the current barriers to patients' ongoing access to DSMES. Furthermore, the potential use of mobile apps as tools for supporting patients in their self-management activities and the limitations of its effectiveness and adoption are highlighted. Theoretical models (Social Cognitive and Information Motivation Behavioural Skill Models) that underpin the current research, methodological framework and analytical techniques are also presented. The reasoning behind the design of a new intervention aimed at DSMES is also detailed. Finally, an outline of the thesis chapters and how they address the research questions is presented.

Chapter (Abbreviated Titles)	Chapter Contents	Author Contributions	Submission Status
ONE: General Introduction	A brief introduction to the prevalence of diabetes mellitus in Australia, and the status of mobile phone apps intervention aimed at improving diabetes self-management. This chapter also discusses the strengths and limitations of current diabetes app interventions, and the theoretical background for the app intervention being evaluated in this thesis.	MDA wrote the introductory chapter, with UM, AEOMA and BMA reviewing each draft before approving the final version.	N/A
TWO Part A: Literature review Part B: Systematic Review of Mobile Phone Apps for Diabetes Self- Management	Part A provides an overview of diabetes mellitus and current literature on the evolution of mobile apps as diabetes self- management tools. It also highlights the barriers to the successful use and opportunities for improvement of diabetes apps. Part B is a systematic review investigating the essential requirements in developing mobile apps for effective diabetes self-management and education.	MDA wrote the part A of the literature review, with UM, AEOMA and BMA reviewing each draft before approving the final version. MDA designed and carried out the systematic search and quality appraisal, and wrote the manuscript drafts. BMA conducted quality appraisal checks. EC, UM, AEOMA and BMA reviewed each draft, and approved the final version.	Part A – N/A Part B - Published in Journal of Medical Internet Research uHealth and mHealth
THREE: Enablers and Barriers to Diabetes Self- Management	A mixed-methods study of an international audience on their perception of the enablers and barriers to diabetes self- management. Specific areas of skills and self-efficacy as enablers for diabetes self-management were evaluated to identify aspects, which require ongoing support.	MDA designed the study and wrote the ethics application. UM, AEOMA and BMA provided support on the study design and assisted in preparation of the ethics application. MDA and BMA analysed the data. MDA wrote the	Published in Plos One

Table 1.1 Thesis Outline Chapter Details and Publication Status of Chapters*

FOUR: Users Preferences and Design Recommendations to Improve Engagement with Apps	A mixed-methods study of an international audience on their preferences for app features, educational contents and recommendations for improved engagement with apps.	 manuscript, while UM, AEOMA and BMA reviewed the final manuscript. MDA designed the study and wrote the ethics application. UM, AEOMA and BMA provided support on the study design and assisted in preparation of the ethics application. MDA and BMA analysed the data. MDA wrote the manuscript, while UM, AEOMA and BMA reviewed the final manuscript. 	Published in Plos One
FIVE: The Development of Diabetes My Care Hub App to Support Diabetes Self-Management	A full report of the development process of a novel app called My Care Hub, aimed at the self-management education and support of people with type 1 or type 2 diabetes. This chapter also includes a two-staged usability testing of the app among Australian general population and diabetes patients.	MDA designed the app's concept. UM, AEOMA and BMA assisted with intellectual content in the app design. IA and SV developed the app's software. MDA wrote the manuscript, while UM, AEOMA and BMA reviewed the final manuscript.	Published in Scientific Report
SIX: User Retention and Engagement with My Care Hub App	A mixed-methods pilot testing of the novel app among type 1 and type 2 diabetes patients living in rural and remote areas of Australia. This is to evaluate participant's retention and engagement with the app.	MDA designed the study and wrote the ethics application. UM, AEOMA and BMA provided support on the study design and assisted in preparation of the ethics application. MDA, AD and BMA analysed the data. MDA wrote the manuscript, while UM, AEOMA, AD and BMA reviewed the final manuscript.	Published in Journal of Medical Internet Research mHealth and UHealth

SEVEN: Efficacy and Acceptability of My Care Hub App	A mixed-method pilot testing of the novel app among type 1 and type 2 diabetes patients living in a rural and remote area of Australia. This is to evaluate the preliminary efficacy of the app on diabetes self-management and its acceptability among the target audience.	MDA designed the study and wrote the ethics application. UM, AEOMA and BMA provided support on the study design and assisted in preparation of the ethics application. MDA, AD and BMA analysed the data. MDA wrote the manuscript, and UM, AEOMA, AD and BMA reviewed the final manuscript.	Published in International Journal of Environmental Research and Public Health
EIGHT: General Discussion	A discussion of the key findings of the research, including its novel contribution and practical implications within a broader context of diabetes self-management. It also outlines prospects for future research to enhance self-management education and support for patients with type 1 or type 2 diabetes.	MDA wrote the discussion chapter, with UM, AEOMA, and BMA reviewing each draft before approving the final version.	N/A
NINE: Conclusions and Recommendations	Final conclusions of the research and recommendations for patients, health professionals, researchers and app developers.	MDA wrote the conclusions and recommendations chapter, with UM, AEOMA, and BMA reviewing each draft before approving the final version.	N/A

*Chapters 2b – 7 are publication-based chapters in this thesis

Chapter TWO: Literature Review

Chapter Overview

This chapter has two sections (Parts A and B). Part A presents the epidemiological background of type 1 and type 2 diabetes mellitus. It also discusses the importance of DSMES and how technological devices especially apps play a key role in providing DSMES. Knowledge gaps in the use of apps as DSMES tools were identified and suggestions on filling these gaps provided.

Part B comprises the methodology and findings of a systematic review investigating developmental considerations adopted in randomised controlled trials that engaged mobile phone applications (apps) for diabetes self-management. This section of the chapter aimed to develop a framework/guideline on the essential elements required in a diabetes app development process prior to use, as an intervention for self-management in patients with type 1 or type 2 diabetes.

Part A: Review of Literature on Diabetes Mellitus, its Complications and Current Interventions to Improve Self-Management

2.1 DIABETES MELLITUS

According to the World Health Organisation (WHO), diabetes is "a metabolic disorder of multiple etiology characterised by chronic hyperglycaemia with disturbances of carbohydrate, fat and protein metabolism resulting from defects in insulin secretion, insulin action, or both." (World Health Organisation, 1999). Insulin is produced in the pancreas and facilitates glucose transport from the blood into cells for energy production. An absolute or partial deficiency in insulin secretion and/or ineffective utilisation by body cells over a prolonged period of time

causes hyperglycemia - high blood glucose levels (World Health Organisation, 1999). There are mainly two type of diabetes: Type 1 and 2 diabetes.

2.1.2 Types 1 and 2 diabetes

Type 1 diabetes mellitus results from an auto-immune destruction of insulin-producing beta (β) cells by the immune system, eventually resulting in absolute deficiency in insulin secretion (World Health Organisation, 1999). The cause of auto-immune reaction has been attributed to genetic disposition triggered by changes in environmental risk factors (Vojdani, 2014). Type 1 diabetes requires lifelong treatment with insulin and is mostly diagnosed during childhood or adolescence (American Diabetes Association, 2017a), although in recent times, it is also diagnosed in adulthood due to late onset (Thomas et al., 2018).

Type 2 diabetes is characterised by relative insulin deficiency/malfunction in addition to insulin resistance, leading to hyperglycaemia. Insulin resistance induces increased insulin production by the β cells in an attempt by the body to maintain normal blood glucose balance, but consequently diminishes β cell mass and insulin (Kahn, Cooper, Del Prato & 2014). Type 2 diabetes, which is the most common type (accounting for 90% of all cases of diabetes) is progressive in nature and has a strong genetic propensity (Kahn, Cooper, Del Prato & 2014).

Generally, symptoms of marked hyperglycemia vary from polydipsia (excessive thirst), polyuria (excessive urine production), blurred vision to unexplained weight loss etc. (World Health Organisation, 1999). In some cases, patients presents with growth impairment and increased susceptibility to certain infections (Powers et al., 2015a). The severity of symptoms depends on the type of diabetes and duration of the disease.

2.1.3 Diagnostic criteria

Diagnostic criteria for both types 1 and 2 diabetes are the same; based on Fasting Plasma Glucose and the 75g Oral Glucose Tolerance Test with a cut point of 7.0mmol/L or 11.1mmol/L respectively (American Diabetes Association, 2017b), as shown in **Table 2.1**. In 2009, HbA1c with a cut-off point of 6.5% was introduced as a preferred diagnostic measure by the International Expert Committee, due to its strong evidence as a marker of chronic hyperglycaemia which reflects average blood glucose levels (BGL) over a 2- to 3- month period (The International Expert Committee, 2009). The measure was endorsed by WHO in 2011 (World Health Organisation, 2011a) and accepted by the Australian Diabetes Society in 2012 (d'Emden et al., 2012).

Table 2.1: General diagnostic criteria for diabetes mellitus (American Diabetes Association, 2017b).

Test	Readings
Fasting Plasma Glucose (no calorie intake for at least 8	\geq 126mg/dl (7.0mmol/L).
hours)	
Oral Glucose Tolerance (2 hours post load of 75g	>200mg/dl (11.1mmol/L)
anhydrous glucose	
Glycated hemoglobin (HbA1c)	≥ 6.5% (48mmol/mil)

2.1.4 Complications of diabetes

Patients with diabetes risk developing acute and chronic complications when the condition is not well managed (American Diabetes Association, 2017a). Acute, life threatening consequences of uncontrolled diabetes are hypoglycaemia, diabetic ketoacidosis and hyperglycaemic hyperosmolar state (Fishbein & Palumbo, 1995; International Diabetes Federation, 2017). Chronic complications include damage to vital organs such as the eyes (retinopathy), kidneys (nephropathy) and nerves (neuropathy) as well as increased risk of heart disease, stroke and poor blood supply to the limbs (neuropathy) (Alberti & Zimmet, 1998; Papatheodorous et al., 2015). Globally, retinopathy affects 35% of people with diabetes and may result in severe visual loss. Also, patients have two to three fold increased rates of cardiovascular diseases (Papatheodorous et al., 2015). In Australia, a study which reported data on 2,731 adults with type 2 diabetes between 2000-2002 reported 13.8-16.5% of these populations had a foot complication (Harris et al., 2006). In a study from Australia's rural and remote Northern Territory, patients with diabetes were five times more likely to have foot infections and poorer diabetes control than other Australians (Commons et al., 2015). Diabetes complications are the leading cause of poor quality of life, pre-mature death and preventable hospitalisations. Diabetes complications accounted for 9% of all potentially preventable hospitalisations in Australia in 2013-14 (Australian Institute of Health and Welfare, 2015). Data from other rural and remote areas of Australia such as Queensland, indicate that in 2015-16, there were 184, 000 cases of potentially preventable hospitalisations with diabetes as the principal diagnosis and 25% of these cases were due to complications (Queensland Health Report, 2018). The hospitalisations were about 20% higher in inner and outer rural areas and 2.2 times higher in remote and very remote areas of Queensland (Queensland Health Report, 2018).

Complications from diabetes impose substantial financial burden on health care systems and national economies. In 2019, an annual global health expenditure of USD760 billion was spent on diabetes with a proposed estimate of USD825 billion by 2030 (International Diabetes Federation, 2019). In Australia, there is a direct annual cost of \$9645 compared with \$4025 among individuals with type 2 diabetes without complications (Colagiuri et al., 2002) and \$16,698 compared to \$3,468 among type 1 diabetes individuals without complications (Colagiuri et al., 2009). Besides the burden on nations and health care systems, diabetes

complications pose significant economic impacts on patients and their families in terms of higher out-of-pocket health care payments, reduced productivity and increased absence from work (World Health Organisation, 2005).

Reducing the risk of complications, hospitalisations and associated financial implications are preventable and could be attained through optimal control of BGL. Specifically, Diabetes Australia guideline recommends a pre-prandial (fasting) BGL of 4 to 8 mmol/L and postprandial levels of <10 mmol/L in those with type 1 diabetes. For those with type 2 diabetes, 6 to 8 mmol/L and 6 to 10 mmol/L fasting/before meals and two hours after starting meals respectively, are recommended (Diabetes Australia, 2015; Deed et al., 2016). These targets may be adjusted for each patient based on the history of hypoglycaemia, age and onset of late complications. These recommended BGL could only be attained and maintained through regular self-management (UK Prospective Diabetes Study Group, 1998; Gæde et al., 2008).

2.2 SELF-MANAGEMENT

The concept of self-management was introduced into the healthcare system in the 1960s (Creer, Renne & Christian, 1976; Lorig & Holman, 2003), however, there exists heterogeneity in its application and understanding as well as a lack of consensus on definition (Ryan &Sawin, 2009; Jonkman et al., 2016). Self-management is often used interchangeably with self-care (Riegel et al., 2009), patient-centred care (Catalyst, 2017; Kuntz et al., 2014), self-regulation and patient activation (Kinney et al., 2015; Hibbard & Greene, 2013). Adding to the ambiguity of the concept is the use of the term self-management intervention programs, the process of self-management and the description of outcomes gained by engaging in self-management practices (Ryan & Sawin, 2009). Although, the concepts are related, definitions relevant to the current research are those delineating the uptake of responsibility for one's own behaviour and well-being, emphasising patient responsibility and acting in concert with health providers (Holman & Lorig, 2000; Bodenheimer et al., 2002). Self-management denotes patientcenteredness, where individuals are active participants in the management of their own disease for improved health outcomes and quality of life (Barlow et al., 2002). Self-management has the potential of an effective paradigm across the public health spectrum for disease prevention (primary), by establishing a pattern for healthy living early in life and providing strategies for disease mitigations and management in later life (secondary and tertiary). Therefore, the concept of self-management is integral to both the maintenance of wellness (Perera & Agboola, 2019) and the management of illness (Starfield et al., 2008; Grady & Gough, 2014).

2.2.1 Diabetes Self-Management

Traditionally, self-management of diabetes has been defined as a set of activities or actions related to healthy eating, being physically active, adhering to medications, self-monitoring of blood glucose (SMBG), foot inspection, problem solving, healthy coping and reducing risks such as smoking cessation and limiting alcohol consumption. Having a healthy lifestyle is crucial for people with types 1 and 2 diabetes, just like it is for the general population. The Australia Diabetes Guidelines targeting lifestyle for good health in diabetes (adopted from the American Diabetes Association, Standard of Medical Care) (American Diabetes Association, 2017c), emphasised increased physical activity, healthier eating and weight loss for those overweight or obese (Deed et al., 2016). Patients are discouraged from consuming foods containing saturated fat, refined carbohydrates and added sugar as much as possible because of their tendency to unnecessarily increase BGL and strain the metabolic system. Instead, consumption of green vegetables and other foods that are low in glycaemic index is advised (Deed et al., 2016, American Diabetes Association, 2017c). In addition to improved dietary recommendations, a minimum of 30 minutes' physical activity is a key component of patients' daily routine in order to help increase insulin sensitivity (Deed et al., 2016). Also, there is a growing body of research exploring the benefits of frequent breaks in prolonged periods of sedentary lifestyle on postprandial spikes and overall glycaemic control (Colberg, 2012;

Dunstan et al., 2012). Therefore, the importance of avoiding sedentary periods of more than 90 minutes is emphasised to patients (American Diabetes Association, 2017c).

For SMBG, it is essential for patients with type 1 diabetes to perform SMBG at least three times daily (American Diabetes Association, 2010; Manroa & Krupa, 2016) to detect asymptomatic hypoglycaemia and facilitate attainment of blood glucose goals (Deed et al., 2016; American Diabetes Association, 2010). There is debate over optimal frequency of SMBG for diet-treated patients with type 2 diabetes, but daily monitoring is important for those who use insulin or oral agents (Deed et al., 2016; Manroa & Krupa, 2016; American Diabetes Association, 2002). Nevertheless, studies have reported better glycaemic control with increased frequency of SMBG in both groups of patients (Evans et al., 1999, Welschen et al., 2005; Poolsup et al., 2009; Polonsky et al., 2011) due to improved ability to schedule lifestyle behaviours including physical activity, food and medication intake (American Diabetes Association, 2010). SMBG in patients promotes personal responsibility and allows for detection of blood glucose fluctuations. Furthermore, SMBG is an important tool for health providers to titrate blood glucose lowering agents (American Diabetes Association, 2010), problem solving and recommend lifestyle (activity, stress, nutrition) modifications for patients (Klonoff et al., 2008; Kirk & Stegner, 2010). For example, if the result of SMBG shows a consistent pattern of high pre-prandial levels, then medications that target liver output of glucose might be helpful. Postprandial glucose levels (two hours after eating) provide information related to the impact of food intake on blood glucose, hence, diet modification or medications such as mealtime insulin, may be helpful in such cases (Kirk & Stegner, 2010). Furthermore, lack of SMBG predicts hospitalisation for diabetes-related complications (Burge, 2001).

In relation to foot care, patients are recommended to wear footwear that fit, protect and accommodate the shape of their feet in order to prevent ulcerations. Also, patients are advised

to regularly check and ensure that there are no signs of trauma, abnormal pressure or ulceration on their feet (Bergin et al., 2013; van Netten et al., 2018). Foot checking is essential for the prevention and early detection of diabetic foot ulcers caused by repetitive shear and pressure on the foot when peripheral artery disease or neuropathy is present (Bus et al., 2016). In relation to reducing risks, smoking caseation is encouraged in diabetes due to increased risk of cardiovascular diseases among patients who smoke (Haire-Joshu, Glasgow & Tibbs, 1999). Similarly, excessive alcohol consumption is discouraged in patients, especially those with type 1 diabetes, because it further increases the risk of hypoglycaemia and diabetes ketoacidosis (DKA) which are life threatening (Turner et al., 2001; Hermann et al., 2017).

2.2.2 Diabetes Self-Management Education and Support (DSME)

In 2006, the term DSME was defined as "an interactive, collaborative, ongoing process involving persons with diabetes and diabetes educators" (Mensing et al., 2006). This process includes: 1) Assessment of the individual's specific education needs; 2) identification of the individual's specific diabetes self-management goals; 3) education and behavioural interventions directed towards helping the individual achieve identified self-management goals and; 4) evaluation of the individual's attainment of self-management goals" (Mensing et al., 2006). This definition was later revised in 2015 as "the process of facilitating the knowledge, skill and ability necessary for diabetes self-care" (Powers et al., 2015a). The definition was also adopted by the Australian Diabetes Educators who described the goal of DSME as a process to assist diabetes patients to better understand their condition, enable them make informed decision and increase their confidence in their self-management in order to improve their quality of life and reduce the risk of complications (Australian Diabetes Educators, 2015).

Often times, patients do not achieve the recommendations for their self-management due to lack of motivation, difficulty changing lifestyle behaviours and low understanding of the direct impact of lifestyle behaviours on glycaemic control and overall health outcomes (Rosenthal et al., 1998; Peveler et al., 2005; Polonsky et al., 2011; Kalra et al., 2013; Saleh et al., 2014; Bonger et al., 2018). A systematic review reported that the mean number of day patients adhering to their self-management behaviours were as low as 29.9%, 26.7%, 13%, 17.0% for diet, exercise, SMBG and foot care respectively (Mogre et al., 2019). Hence, the recommendation for diabetes self-management support (DSMS) to be provided in addition to DSME. DSMS are activities to support the individual with diabetes in order to implement and sustain coping skills and behaviours needed to self-manage their condition on a regular basis (Powers et al., 2015a). The type of support provided can include educational, behavioural, psychosocial, or clinical (Haas et al., 2012).

The need for DSMS originates from the work of Tang and colleagues (2005), who acknowledged the necessity to create a new generation of diabetes self-management support interventions that would reinforce and enhance self-management benefits that a patient achieves during DSME programs. Furthermore, DSMS is aimed at enhancing and sustaining self-management behaviours on a long term basis. Tang et al's (2005) work added self-management support to DSME to become diabetes self-management education and support (DSMES). This addition emphasises the need for ongoing support for patients in their diabetes self-management when they are outside of formal care programs (Haas et al., 2012). Traditionally, DSME is only provided by health professionals during appointments at clinics or hospitals, whereas ongoing DSMS can be provided within a variety of community-based settings (Power et al., 2015a).

The current Australian guidelines for the management of types 1 and 2 diabetes (Australian Diabetes Educators, 2015), are in line with updated recommendations in the "Position Statement for Diabetes Self-Management Education and Support" (Powers et al., 2017) and "Standards of Medical Care in Diabetes" (American Diabetes Association, 2019). These

guidelines are in agreement with continuous DSMES for people with diabetes in order to increase their knowledge, skills and resilience required for ongoing commitment to self-management. DSMES are designed to address knowledge gaps, health beliefs, health literacy, emotional concerns and other factors that influence patients' ability to meet the challenges of self-management.

Through various modes of delivery, DSMES interventions have proven to improve behavioural change outcomes as evident in several reviews that reported improvements in physical activity (Norris, Engelgau & Narayan, 2001; Heinrich, Schaper & de Vries, 2010; Ricci-Cabello et al., 2014), diet (Newsman, Steed & Mulligan 2004; Norris, Engelgau & Narayan, 2001; Heinrich, Schaper & de Vries, 2010; Ricci-Cabello et al., 2014), blood glucose testing (Ricci-Cabello et al., 2014; Heinrich Schaper & de Vries, 2010) and foot care (Ricci-Cabello et al., 2014). The ultimate impact of those behavioural outcomes is the effect on glycaemic control. Many studies have reported the importance of DSMES for improved glycaemic outcome in the form of glycosylated haemoglobin (HbA1c) (Norris et al., 2002; Gary, 2003; Deakin et al., 2005; Steinsbekk et al., 2012; Tshiananga et al., 2012). Steinsbekk (2012) assessed 21 studies and found a significant decrease in HbA1c after 6-24 months. A meta-analysis of 34 RCT with a combined cohort size of 5993 patients, reported a significant effect on HbA1c with a moderate effect size of 0.51. However, the intervention effect reduced after six months (Norris et al., 2002). On the contrary, a meta-analysis by Gary and colleagues (Gary et al., 2003) reported a similar effect, but the intervention effects were sustained beyond six months. The results of these reviews denote an improvement in glycaemic control. However, discrepancies exist with regards to sustainability of the intervention effect.

The mode of delivery for studies in the above reviews ranged from individuals, in-persons, remotely or combined modes including nurse-led education, group meetings, etc. O'Hara et al., (2017) and Norris, Engelgau & Narayan (2001) found mixed results on HbA1c outcomes from

both group-based versus individual interventions in people with type 1 or type 2 diabetes, respectively, thereby suggesting no clear evidence about the effects of delivery mode. However, a meta-analysis conducted a year later by Norris et al. (2002), found no differences related to the intervention delivery mode and yet, a positive impact on HbA1c, suggesting positive contributions from both group- or individual-based interventions. Similarly, a Cochrane review reported the benefits of individual mode of delivery only on type 2 diabetes participants who had a baseline HbA1c greater than 8%. Overall, the study found no significant difference in glycaemic control between individual- or group-based education (Duke, Colagiuri & Colaguiri 2009), thereby further confirming the equal effects of individual-and group-based interventions.

The most commonly used components of DSMES are education about the condition and its management, lifestyle advice and support, as well as psychological strategies for healthy coping. No component emerged as essential or optimal, but rather, a multi-component self-management strategy was advised (Fan & Sidani, 2009; Schaper & de Vries, 2010; Bolen et al., 2014; Sherifali et al., 2016; McBain et al., 2016). Research evaluating self-management interventions for 14 different chronic conditions highlighted the following core components: Education about the condition, practical self-management support, psychological strategies for life adjustments, self-management adherence and social support (Taylor et al., 2014). Another, meta-analysis found that the inclusion of an educational component in interventions produced statistically significant improvements in clinical outcomes (Ricci-Cabello et al., 2014). Furthermore, two reviews concluded that components that employed cognitive reframing techniques using regular reinforcements aimed at increasing motivation and changing behavioural attitudes, were the most effective; far better than those aimed at enhancing knowledge alone (Norris, Engelgau & Narayan, 2001; Ricci-Cabello et al., 2014). Therefore, education and feedback for motivation and behavioural attitudes have considerable effects on

behavioural change. As such, DSMES approaches have evolved from training to empowerment and from one-dimensional to multi-component approach.

Regardless of the importance of DSMES, research has concluded that the number of persons with type 1 or type 2 diabetes accessing DSMES is low, partly due to economic, geographic and cultural barriers. In addition, patients have poor adherence to clinical recommendations for lifestyle changes, which has been attributed to lack of access to regular self-management education and support (Delamater, 2006). In Australia, a nationwide study found that only 25% and 17% of patients with type 1 or type 2 diabetes adhered to healthy eating plans and regular exercise, respectively (Ventura et al., 2016). Other studies reported that adherence to SMBG (Taylor, Fatima & Solomon, 2017) and foot care were sub-optimal (Perrin, Swerissen & Payne, 2009; van Netten et al., 2019). As a result, most patients are unable to achieve the clinical target for blood glucose control and HbA1c and are thus highly predisposed to developing diabetes-related complications (Speight et al., 2011; Ventura et al., 2016; Henderson et al., 2013). A possible way of overcoming these barriers and its subsequent impacts is through alternative settings such as the use of technology, especially those that could be embedded in the daily life of patients.

2.3 TECHNOLOGY FOR DIABETES SELF-MANAGEMENT EDUCATION AND SUPPORT (DSMES)

Technology can increase access to DSMES. Its ascendancy stems from the ability to provide new information and disease self-management programs to large populations that encourage behavioural change (Krishna, Boren & Balas 2009; Cole-Lewis & Kershaw, 2010; Oldenburg et al., 2015; Taj et al, 2019). Several concepts within the health technology space have emerged including telehealth, mHealth, eHealth, telecare and telemedicine. The terms encompass delivery of health care services at a distance using technologies and methodologies to enable remote care, improved health outcomes, health care services and health research (Wilson & Maeder, 2015; Tuckson et al., 2017). mHealth is the applied concept for this research and has been defined as the practice of health care delivery supported by mobile devices, such as mobile phones, tablets, computers, personal digital assistants (PDA) and other wireless devices (World Health Organisation, 2011b). The concept of mHealth is considered to be a component of eHealth, which is health care delivered electronically, using mobile devices (Van Dyk, 2014).

2.3.1. mHealth

The emergence of mHealth stems from the enthusiasm of using mobile phones and other handheld devices for healthcare. A key strength of mHealth is the ability to capture data and provide feedback to users through their personal devices, which are embedded into their daily routines, thus facilitating habitual self-monitoring (Chomutare et al., 2011; Nundy et al., 2014). Mobile phones are carried by persons, typically turned on, and suitable for bi-directional communication and on-demand access to resources (Proudfoot et al., 2011), hence their ability to facilitate point-of-care resources (e.g. decision support systems, remote consultation). Mobile phones have been further revolutionised into smartphones - mobile phones that can access the internet and run software applications (apps) (Poushter, 2016). Apps could be designed to be used by patients and health providers for behavioural prompts, reminders, continuous disease monitoring and self-management programs that extend beyond the confines of a physical clinic (Harrison et al., 2011; Luxton et al., 2011; Ben-Zeev, 2012). Apps could offer access to monitoring disease symptoms, information and communication with peers or health providers. In 2013, Martinez-Perez and colleagues reported in their review that diabetes led the list of the most common disease condition targeted by apps in scientific literature as well as in commercial stores

The use of apps as tools for DSMES has proliferated, as illustrated by the number of diabetes apps currently available for download on Google Play and iOS Store (Pham, Wiljer, Cefazzo,

2016; Huang et al., 2019; Izahar et al., 2017; Kordonouri & Riddell, 2019; Ubaid Ur Rehman et al., 2019; Arnhold, Quade & Kirch, 2014; Martinez et al., 2017). Therefore, as one of the prominent digital behavioural change interventions, apps could provide DSMES in a myriad of ways. These include: monitoring of blood glucose, physical activity, diet, medications, weight, push notifications and reminders/alerts, provision of education/information, analytics and graphical visualisation of data, goal setting, data export, communication with health professionals, synchronisation with health records, social networking, integration with research and news update (Chomutare, 2011; Arnhold, Quade & Kirch, 2014). Some apps allow for feedback and real-time data combined with predictive analytics of behavioural engagement on clinical outcomes, which are effective in self-discovery of the effect of self-management (Kaufman et al., 2016; Van Calster et al., 2019).

Overall, it is clear that diabetes apps can empower patients with the ability to proactively manage diabetes. However, there exists some limitations with apps as supporting tools for DSMES. These limitations include: 1) Lack of information on what constitutes standard considerations in the development of diabetes apps prior to use in trials; 2) poor exploration of mediating factors in diabetes apps to foster behavioural change and; 3) high attrition rates and low/absence of educational components in most apps.

2.3.2 Limitations in Mobile Apps for Diabetes Self-Management and How to Address Them.

a. Framework on standard consideration for diabetes app development:

While interest in the use of apps for DSMES continues to rise, to the best of our current knowledge, no systemic framework or guidelines on processes critical for an effective diabetes app are available. There is a general lack of explicit explanation among researchers and developers on the systematic elements that guide diabetes app development to foster an effective intervention. This gap suggests that research is needed to provide a framework on

minimum standard considerations for diabetes app development prior to its use as an intervention in patients. Provision of standards will promote vigour, quality control and encourage a transparent and reproducible process of app development (Hoffmann et al., 2017). Such frameworks will provide a clearer understanding of the core elements that underline diabetes app development and inform guidelines that can be used consistently to ensure that such apps are fit for purpose - an effective tool for self-management education and support of patients.

b. Poor exploration of mediating factors in diabetes apps to foster behavioural change:

The role of mediating factors is central to the ability of patients' to desire or participate in selfmanagement. This role had been poorly explored in previous diabetes app interventions. There exists a mediating framework between theoretical constructs and behaviour (Baranowski, 1998; Lewis et al., 2002). In behavioural theory, there is an assumption that interventions can target change in critical mediators of behavioural engagement and this will follow a causal chain resulting in behavioural change. Specifically, the independent variable (e.g. intervention) has an effect on outcomes (e.g. changes in diabetes behavioural management) via the mediator (Cerin & Mackinnon, 2009); MacKinnon et al., 2002).

Behavioural change models such as social cognitive theory (SCT) are frameworks for mediating constructs of behavioural change (Bandura, 1998). SCT has a history of supporting people to stay healthy through good self-management behaviours necessary for healthy living (Lyons et al., 2014). SCT has been reported by many studies to guide the development of complex interventions for self-management support of many chronic conditions including diabetes (Cotter et al., 2014; Avery et al., 2015; Tougas et al., 2015). Steed et al., (2014), noted in their evaluation of SCT and Self Determination Theory, that beliefs in seriousness and treatment effectiveness were not significant mediators of self-management, rather, behaviour

specific self-efficacy for monitoring of blood glucose and physical activity had the most convincing and reliable evidence as a mediator of behavioural change. Likewise, a vast majority of evidence reveals that in patients with types 1 and 2 diabetes, improved self-efficacy improved the diets, levels of physical activity and overall self-management behaviours of such patients (Sarkar, Fisher & Schillinger, 2006; Qiu et al., 2012; Mohebi et al., 2013; Steed et al., 2014; Yao et al., 2019; Jiang et al., 2019; Alvarado-Martel et al., 2019).

Perceived self-efficacy enables an individual to succeed in challenging situations and this is the main basis for action. It begins with the perception of the existence of a problem followed by the belief that desired results can be achieved with one's action, thus creating an incentive to persevere (Bandura, 2004). This can greatly influence a patient's ability or desire to participate in self-management (Bandura, 2004). Self-efficacy underlines a person's ability to overcome temporary setbacks and relapses, and the extent to which a new behaviour is maintained (Bandura, 2004). The stronger the level of self-efficacy, the more likely behavioural change will be successful (Locke, 1997), therefore, it is a significant mediator for improved behavioural change.

While self-efficacy is a mediator for behavioural change, it assumes that if people lack the awareness and knowledge of how their lifestyle habits influence their health, they have little reason to pull themselves out of the bad habits they enjoy. Patients infrequently receive self-management training from health care providers and find it difficult to apply to real world settings (Delamater, 2006). Lack of adherence stems from the inability to understand how each care component or recommendation contributes to overall glycaemic control. For instance, many patients do not relate improved exercise to glycaemic control, and therefore lack motivation to increase their level of physical activity. The ability to problem-solve and make decisions based on BGL can significantly reduce long term complications (Siriwardena et al., 2012). Formation of these skills is through both lifestyle and knowledge modifications

(Hernandez et al., 2012). Poor knowledge and lack of these skills in patients impact their selfefficacy and motivation toward the adoption of behavioural change. Therefore, patient skills create the pre-condition for confidence (self-efficacy) to engage in healthy behaviours.

Apart from skills, self-monitoring is another mechanism that fosters improved self-efficacy in patients, because it enables individuals to capture their behaviours, recognise patterns and provides the motivation for behavioural change (Hernandez et al., 2012; Steed et al., 2014). Self-monitoring of blood glucose only has been found to be insufficient for behavioural change. However, self-monitoring of all essential behavioural activities including physical activities and diet, is the best behavioural change drivers, because it provides context around why and how blood glucose values come about. An addition of feedback to monitoring of behavioural activities and information for corrective actions if required (Annesi, 1998; De Vries et al., 2008).

c. High attrition and low engagement

Although the popularity of mHealth has increased in recent times, the utilisation, sustained engagement and adoption of this tool all remain evasive, with evidence of reduced use over time, although patients reported improvement in self-management (Quinn et al., 2011; Kirwan et al., 2013). Given that the use of diabetes apps for self-management is user-driven, these findings suggest that it is imperative that the design of diabetes apps considers how patients interact with the tool in real world settings. The low level of acceptance of mHealth tools has been attributed to insufficient consideration of its usability requirements and preferences by users (Demidowich et al., 2012). Therefore, a critical solution to attrition and low engagement with app use is that of a shared decision-making between app developers and users, entailing needs analysis to confirm what appeals to patients and how to maximize their engagement with apps.

d. None / low educational component in app

Many of the existing diabetes apps lack self-management educational components (Chomutare 2011). With clinical guidelines emphasising the importance of educational support for patients, failure to provide users with this feature puts current apps at the risk of non-effectiveness instead of being a means to facilitate behavioural change and comprehensive self-management. Educational content in apps especially those targeting management skills and confidence, may consequently improve users' level of self-management necessary for improved health outcomes.

Part B: Considerations for the Development of Mobile Phone Apps to Support Diabetes Self-Management: A Systematic Review 2.4 ABSTRACT

There is increased research interest in the use of mobile phone apps to support diabetes management. However, there are divergent views on what constitute the minimum standards for inclusion in the development of mobile phone apps. Mobile phone apps require an evidence-based approach to development which will consequently impact on their effectiveness. Therefore, comprehensive information on developmental considerations could help designers and researchers to develop innovative and effective patient-centered selfmanagement mobile apps for diabetes patients. This systematic review examined the developmental considerations adopted in trials that engaged mobile phone apps for diabetes self-management. A comprehensive search strategy was implemented across 5 electronic databases; Medline, Scopus, Social Science Citation Index, the Cochrane Central Register of Controlled Trials and Cumulative Index of Nursing and Allied Health Literature (CINALHL) and supplemented by reference list from identified studies. Study quality was evaluated using the Joanna Briggs critical appraisal checklist for trials. Information on developmental factors (health behavioural theory, features/functionality, pilot testing, user and clinical expert involvements, data privacy and app security) were assessed across experimental studies using a template developed for the review. A total of 11 studies (10 randomised controlled trials and 1 A quasi-experimental trial) that fitted the inclusion criteria were identified. All the included studies had the functionality for self-monitoring of blood glucose. However, only some of them included functions for data analytics (7/11, 63.6%), education (6/11, 54.5%) and reminder (6/11, 54.5%). There were 5/11 (45.5%) studies with significantly improved glycosylated hemoglobin in the intervention groups where educational functionality was present in the apps used in the 5 trials. Only 1 (1/11, 9.1%) study considered health behavioural theory and user

involvement, while 2 (2/11, 18.1%) other studies reported the involvement of clinical experts in the development of their apps. There were 4 (4/11, 36.4%) studies which referred to data security and privacy considerations during their app development while 7 (7/11, 63.6%) studies provided information on pilot testing of apps before use in the full trial. Overall, none of the studies provided information on all developmental factors assessed in the review. There is a lack of elaborate and detailed information in the literature regarding the factors considered in the development of apps used as interventions for diabetes self-management. Documentation and inclusion of such vital information will foster a transparent and shared decision-making process that will ultimately lead to the development of effective and user-friendly selfmanagement apps that can enhance the health outcomes of diabetes patients.

2.5 BACKGROUND

Mobile apps refer to software installed on smart mobile devices that support medical and public health practices (Kay, Sanjo & Takane, 2011). These applications can deliver healthcare anywhere, subduing geographical and organisational barriers as well as time constraints (Akter & Ray, 2010; Thakkar et al., 2016). Their intended use is for diagnosis, self-management, mitigation, treatment or prevention of diseases such as diabetes (Silva et al., 2015). Self-management of blood glucose minimises the risk and health complications associated with the insidious and chronic nature of diabetes (UK Prospective Diabetes Study Group, 1998; ADVANCE Collaborative Group, 2008). Diabetes self-management (DSM) includes monitoring of glucose level, lifestyle modifications, medication management, prevention of complications and psychosocial care (American Diabetes Association, 2016). As a standard, diabetes self-management education is usually provided during outpatient visits; but it has been advocated that most patients require ongoing support to encourage and sustain behaviour at the level that can maintain good health (Funnell et al., 2009; Haas et al., 2014). Hence, the necessity

for a regularly accessible form of diabetes self-management education and support (DSMES); which can be achieved with the use of mobile apps.

Although, mobile app is a field that has continually attracted the interest of researchers and has excellent prospects, both for the improvement of healthcare and economic interest (Free et al., 2013; de la Torre Diez et al., 2015), comprehensive information on its developmental considerations seem somewhat limited. Studies have reported gaps in the understanding of formal standards and evidence-based approaches employed in the development and evaluation of the mobile apps (Buijink, Visser & Marshall, 2013; Misra, Lewis & Aungst, 2013).

Considerations in Mobile Phone App Development

Presently, knowledge about the standard recommended practice for mobile app development for chronic disease management seems divergent and inconclusive. Some studies have reported the benefits of developing mobile app based on health behaviour and communication change theories (Webb et al., 2010; Riley et al., 2011). The main reason for using these theories is to adopt techniques and strategies to help patients embrace healthier lifestyles. Existing models/theories include transtheoretical model (Prochaska & Velicer, 1997), social cognitive theory (Bandura, 2004), self-determination theory (Ryan & Deci, 2000), social ecological theory (Bandura, 2004) and motivational interviewing (Miller & Rollnick, 2013). These theories have served as guards in designing diabetes management interventions.

Some authors are of the opinion that the development of healthcare tools for patient groups such as those with diabetes requires an understanding of current challenges and barriers to self-management (Arsand et al., 2010). This approach serves as an avenue for exploring users' needs at a specific time and envisaging what may evolve with time; which can help in visualising the use of the app as users' demands change (Yardley et al., 2015; Petersen & Hempler, 2017).

Chomutare et al., (2011) emphasised in their systematic review that good practice in designing mobile apps requires that inclusion of functionalities be anchored on evidence-based recommendations for the target groups. Furthermore, pilot testing with a target audience and incorporating feedbacks will aid identification of barriers to the usage of mobile apps and enhance the evaluation of its reliability, accuracy, usability, acceptability, and patient adherence (Thakkar et al., 2016). Ensuring the incorporation of evidence-based recommendations and pilot testing into app development for diabetes care will allow for accurate interfaces, interpretations, and evaluation of the effectiveness of the mobile app.

Data privacy and security whereby the users' information is securely managed is another major developmental consideration (Thakkar et al., 2016; Al-Taee et al., 2016). Emphasising the use of 'privacy by design' approach such as encryption and protocols for anonymous communication and authentication helps to deter unauthorised users from gaining access to patients' medical data (Gürses, Troncoso & Diaz, 2011; van Rest et al., 2014). Furthermore, it has been recommended that involvement of clinical experts and multidisciplinary health teams should be an integral part of the developmental and testing process of diabetes mobile apps to ensure that medical guidelines and clinical best practices are followed in the management of diabetes (Brandell & Ford, 2013).

The various views described above can be labeled as shared decision-making approach to the development of mobile app. Diabetes care and support using this approach in which patients, health care providers, researchers and app developers make health care decision together; taking into account specific evidence as well as specific needs and preferences of patients, has been recommended by various studies. This is seen to produce effective health outcomes (Kinmonth et al., 1998; American Diabetes Association, 2015; Inzucchi et al., 2015). Such an approach focuses on patient empowerment, ensuring a transition from a state where patients are only seen as the recipients of care to a position where they also have their opinion

considered, and are allowed to make choices, thereby actively contributing to the decisionmaking process. Given that the organisational structure within the healthcare sector now recognises the patients' greater role in their healthcare, this trend should also result in a shift in the process involved in the development of mobile apps. Patient engagement strategies in app development may not necessarily refer to their involvement in the algorithm design but rather in the incorporation of procedures that meet patients' expectations through the consideration of their experiences, needs, reasons for engagement and satisfaction with the usage of the app.

Mobile apps have been proven to be a useful lifestyle modification tool for providing ongoing individual support for DSM and facilitating regular monitoring for improved health outcomes (Liang et al., 2011; Holtz & Lauchner, 2012; Nundy et al., 2014; Heinrich, Schaper & de Vries, 2015; Hou et al., 2016; Whitehead & Seaton, 2016). However, previous reviews have focused mainly on assessing the effectiveness of mobile apps to support DSM (Free et al., 2013; Hou et al., 2016; Liang et al., 2011; Whitehead & Seaton, 2016; Cui et al., 2016). A mixture of shared decision-making approaches that include developmental considerations such as health behavioural theories, user and clinical expert involvement, pilot testing and data security are essential. These approaches may help solve the problems of poor engagement experience and ineffectiveness of mobile apps (Yu et al., 2014).

The inclusion of robust, reliable and repeatable system design that involves end users early in the developmental consideration process will enhance ongoing support, which is crucial to sustaining progress made by patients in their DSM (Arsand et al., 2008). To the best of our knowledge, no other study has collated evidence on the factors taken into consideration in the development of such apps. This evidence will further aid the advancement of evidence-based development and evaluation of mobile apps for effective diabetes management. This systematic review aims to evaluate the factors taken into consideration in the development of mobile phone-based apps used as self-management interventions in experimental trials of adults with diabetes. Also, the review compares these mobile app developmental factors with their impact on the key clinical outcome variable glycosylated hemoglobin (HbA1c). For this study, the developmental factors considered are categorised into the following: (1) Health behavioural change theory, (2) Features/Functionality (comprising documentation, analytics, reminder, and education), (3) Users involvement, (4) Clinical expert involvement, (5) Data security and privacy consideration, and (6) Pilot testing. These factors were considered based on extensive literature search and ingeminate brainstorming sessions among co-authors, with a focus to provide a guide on factors to consider in the development process of mobile app for diabetes self-management precluding the use of such apps in a full trial.

2.6 METHODS

This systematic review was conducted following the Preferred Reporting Items for Systematic Review and Meta-Analyses (PRISMA) statement (Shamseer et al., 2015). Assessed developmental considerations are based solely on author reported descriptions directly available in the selected studies or referenced in another published article. For this review, we defined mobile phone applications as apps that are downloadable to mobile phones and take data inputs from users with a focus on improving one or more aspects of diabetes self-management domains.

2.6.1 Data Sources and Search Strategy

Published literature sources were identified by searching Medline, Cummulative Index to Nursing and Allied Health (CINAHL, EBSCOhost), Scopus, Social Science Citation Index and Cochrane Register of Controlled Trials (CENTRAL) databases. In order for search results to have the maximum possible coverage, the combination of the following terms and medical subject headings (MESH) were used during the search: ("Type 1 diabetes mellitus" OR "Type
2 diabetes mellitus'' OR diabet* OR IDDM OR NIDDM) AND ("Mobile applications'' OR, Smartphone* OR "app" OR "cellular phone" OR "mobile app" OR "portable electronic applications'' OR "portable software application" OR "text messages"). Searches were done between 5th and 29th September 2017. Searches were supplemented by manual searching of reference lists of identified studies.

2.6.2 Selection Criteria

Selected studies were any randomised controlled trial (RCT), quasi-experimental study, or prepost study evaluating the use of mobile apps for self-management in patients (\geq 18 years) with type 1 or 2 diabetes. Studies included were those that used mobile phone-based app intervention which allows real-time interaction between patients and the software. Such interactions include input from the user (which may/may not allow for reinforcement of personalised or general advice), goal setting, data analytics, decision support or reminders to improve DSM. Strict inclusion criteria were applied to streamline and capture only diabetes interventional studies. Therefore, to ensure review of fully functional apps used as an intervention for diabetes management, only trials that evaluated at least one glycaemia index of glycosylated hemoglobin (HbA1_c) or blood glucose levels as primary outcome were included. Selected studies were those published in the English language but not restricted to patients of any particular race.

Exclusion criteria included: (1) technological interventions not including mobile phone based app, for example systems which require patients to input data into a web-based server for review by clinician or researcher, (2) systematic reviews, meta-analyses, conference papers or letters, (3) pre-diabetes, gestational and secondary diabetes, (4) obesity, (5) software solutions mainly for insulin pumps only, (6) studies on mixed populations of adults and children, and (7) studies still ongoing that presented interim results only.

2.6.3 Data Extraction

The titles and abstracts of all identified references were reviewed by MDA. References that did not meet all of the inclusion criteria were excluded. The full-text article of all relevant references was retrieved and assessed. Data were extracted from each selected studies using an electronic form purposely developed for this review. All authors checked the extracted data for consistency. Discrepancies were resolved through discussion.

2.6.4 Quality Assessment

Assessment of study quality was performed by MDA in consultation with BMA. The quality was evaluated using Joanna Briggs Institute's pre-designed standardised critical appraisal tools (Aromataris et al., 2015). For the RCTs, the following criteria were considered: (1) true randomization of assignments, (2) allocation concealment, (3) blinding of outcome assessors, (4) intention-to-treat analysis, and (5) appropriateness of trial design. Criteria considered for the quasi-experimental trial included (1) clear description of cause and effect, (2) presence of a control group, and (3) pre- and post-intervention outcome measurements were assessed. For all studies, criteria included (1) details of similarity in baseline characteristics, (2) identical treatment for groups with the exception of intervention of interest, (3) degree and description of follow up, (4) similarities in group outcome measurements, (5) reliability of outcome (primary outcome of HbA1_c or blood glucose levels), and (6) suitability of statistical analysis were evaluated. Blinding of participants and personnel were part of the quality criteria in the tools but were omitted and termed non-applicable since the nature of the intervention under study makes it difficult to achieve blinding. All criteria on the tools were scored on a 2 point scale: Yes (1 point) or no or unclear (0 points). When adding all quality criteria, the maximum obtainable scores was 11 for the RCTs and 9 for the quasi-controlled trials. Depending on the number of criteria met by each study, the quality of each study was graded as High (≥ 7 points),

moderate (4-6 points) or low (\leq 3 points). Disagreement were resolved through discussion among authors.

2.7 RESULTS

2.7.1 Selection of Studies

The initial search from the 5 databases identified 1203 articles which included 116 duplicates that were removed. Based on the review of the titles and abstracts, 53 articles were potentially relevant. The full text of these articles was retrieved for further examination, and their references were manually screened to identify articles that were not included in the original search. This process yielded 4 additional articles. After reading the full articles, 12 studies met the set inclusion criteria. The study by Quinn et al. (2011) and Quinn et al. (2016) reported on the same study population, with different group classifications The study by Rossi et al. (2009) and Rossi et al (2013) engaged the same app but in different study populations. Therefore, 11 RCTs and 1 quasi-experimental study were eventually included. An adapted PRISMA (Preferred Reporting Items for Systematic Reviews and Meta-Analyses) flow-chart of study selection is shown in **Figure 2.1**.



Figure 2.1: Flow diagram of the study selection process

2.7.2 Methodological Quality Assessment

There were 7/12 (58.3.6%) true randomisation trials (Istepanian et al., 2009; Rossi et al., 2010a; Charpentier et al., 2011; Rossi et al., 2013; Kirwan et al., 2013; Orsama et al., 2013; Waki et al., 2014). There were 2/12 (16.7%) trials with unclear evidence for their randomisation method as there was insufficient detail to make a judgment (Quinn et al, 2011; Quin et al., 2016). Allocation concealment was documented in only 1/12 (8.3%) study (Kirwan et al., 2013).

A total of 7/12 (58.3%) studies reported an intention-to-treat analysis of their data (Quinn et al., 2011; Quinn et al., 2016; Rossi et al., 2010a; Rossi et al., 2013; Charpentier et al., 2011; Istepanian et al., 2009; Waki et al., 2014; Holmen et al., 2014). There was 1/12 (8.3%) study that reported the use of a linear mixed methodology, which allowed the inclusion of all randomised participants (Kirwan et al., 2013). A total of 9/12 (75.0%) studies had details of attrition with reasons for drop out balanced across groups (Quinn et al., 2016; Quinn et al., 2011; Rossi et al., 2010a; Rossi et al., 2013; Charpentier et al., 2011; Kirwan et al., 2013; Orsama et al., 2013; Waki et al., 2014; Kim et al., 2014).

All studies had similar and reliable HbA1_c measure. All studies, except 1/12 (8.3%) by Istepanian et al (2009), were judged to be appropriate in their statistical analyses and trial designs. Overall, 10/12 (83.3%) studies were graded as high quality because they met 7-9 criteria of the grading tool, 1/12 (8.3%) study met 6 of the criteria and was graded as moderate (Istepanian et al., 2009), and the last study (1/12, 8.3%) met only 2 quality criteria (Takenga et al., 2014), was graded as poor and removed from the review.

2.7.3 Characteristics of Included Studies

The 11 studies selected evaluated 9 mobile apps and were published between 2009 and 2016. A total of 10/11 (91.1%) studies were RCTs, while 1/11 (9.1%) was a quasi-experimental study (Kim et al., 2014). Participant numbers ranged from 54 (Waki et al., 2014) to 213 (Quinn et al., 2011). There were 4/11 (36.4%) studies which focused on type 1 diabetes (Rossi et al., 2010a; Rossi et al., 2013; Charpentier et al., 2011; Kirwan et al., 2013), 6/11 (54.5%) studies were specific to type 2 diabetes while 1/11 (9.1%) study (Istepanian et al., 2009) involved both type 1 and 2 diabetes patients. Intervention duration for 8/11 (72.7%) studies ranged from 2 to 10 months, while the remaining 3/11 (27.3%) studies (Quinn et al., 2016; Quinn et al., 2011; Holmen et al., 2014) had their follow up period extended to 1 year. Study locations were from four geographic regions including Europe (6/11, 54.5%), Oceania (1/11, 9.1%), Asia (2/11, 18.2%) and America (2/11, 18.2%).

All studies had major interventions using a mobile app. A total of 2/11 (18.2%) studies had 2 intervention groups (Charpentier et al., 2011; Holmen et al., 2014) and another 2 studies had 3 intervention groups (Quinn et al., 2011; Quin et al., 2016). HbA1c was the primary outcome measure in all trials. A total of 5/11 (45.4%) studies reported a positive and statistically significant improvement in HbA1c in the intervention group (Quinn et al., 2011; Charpentier et al., 2011; Kirwan et al., 2013; Orsama et al., 2013; Waki et al., 2014). A total of 5/11 (45.4%) studies had HbA1c reduction in both the intervention and control groups (Quinn et al., 2016; Rossi et al., 2010a; Rossi et al., 2013; Holmen et al., 2014; Kim et al., 2014). While in 1/11 (9.1%) study, HbA1c remained unchanged between the intervention and control groups (Istepanian et al., 2009). A summary of these characteristics are shown in **Table 2.1**

Author (s), Year	Study Design, Location, Type of diabetes, Total number of participants (N), Duration (in months), Attrition rate	Stated aim	Number of intervention group and details	Summary of HbA1c result	Scores/ Quality
Kirwan, 2013	RCT; <i>Australia</i> ; Type 1 DM; N = 72; Duration = 6; Attrition = 19 (26.3%)	To examine the effectiveness of a freely available smartphone app combined with text message feedback from a certified diabetes educator to improve glycaemic control and other diabetes related outcomes in adults with type 1 diabetes	1 intervention group who received: (i) a freely available mobile app that allows users to manually enter BG levels, insulin dosages, other medications, diet and physical activities. (ii) One personalised SMS per week. <i>Comparator</i> : Usual care	Significant HbA1c improvement in the intervention group [from mean 9.8% (SD 1.18%) to mean 7.8% (SD 0.75%)] compared to control [from mean 8.47% (0.86%) to mean 8.55 (SD 1.16)] after 9 month follow up. (p =.001)	10/ High
Rossi, 2013	Parallel group RCT; <i>Italy</i> ; Type 1; N = 127; <i>Duration:</i> 6 months; <i>Attrition</i> : 15 (11.8%)	To compare the efficacy of the diabetes interactive diary versus usual care on metabolic control, hypoglycaemia and quality of life	1 intervention group who received: (i) a mobile app and attended a course on the usage. App was used to estimate CHO content in meals and prandial insulin doses were adjusted based on app algorithm. (ii) Education on hypoglycaemia (iii) BG meter	Non-statistically HbA1c reduction in both groups. Intervention [decrease of mean 0.41(SD 0.11)] and control [decrease of mean 0.48 (SD 0.11) (<i>p</i> =.73)	9/ High

Table 2.2 Study participants, methodological characteristics and quality (arranged from highest to lowest)

Kim, 2014	Quasi experimental; <i>Korea</i> ; <i>Type of DM:</i> Type 1; <i>N</i> = 73; <i>Duration</i> : 3 months; <i>Attrition</i> : 3(4.1%)	To assess the efficacy of the smartphone-based health app for glucose control and patient satisfaction with the mobile network system used for glucose self- monitoring	1 intervention group who received: (i) a mobile app to log BG data, which were automatically sent to the medical team for analysis to provide recommendations at least once a week. (ii) in the events of hypoglycaemia in participants or failure to record BG measurements, the medical team called such participant to change insulin dose or recommend an early visit to the hospital. <i>Comparator</i> : Usual care	Non-significant improvement in HbA1c in the intervention group [from mean 7.7% (SD 0.7%) to mean 7.5% (SD 0.7%) (<i>p</i> =.077) and the control groups [from mean 7.7% (SD 0.5%) to mean 7.7% (SD 0.7%)] (<i>p</i> =.093)	9/ High
Charpentier, 2011	Parallel group RCT; <i>France</i> ; Type 1; N = 180; Duration: 6 months; <i>Attrition</i> : 7 (3.9%)	To evaluate the efficacy of mobile app in improving metabolic control in poorly controlled patients with type 1 diabetes	2 intervention groups. (A). Both groups were provided with smartphone loaded with app, which can provide (a). bolus insulin dose based on logged, pre-meal BG, CHO, premeal BG and anticipated physical activity. (b). Plasma glucose targets. (c). Algorithm for the adjustment of CHO ratio and basal inulin doses. (B) Group 1 had a quarterly hospital visit. (c). Group 2 received teleconsultation every 2 weeks and a follow-up visit after 6- month. <i>Comparator:</i> Usual care plus the use of paper log book and attendance of two follow-up hospital visits.	The intervention group 1 (app only) had 0.67% improvement in HbA1c over the control group (p =.002) while the intervention group 2(app + teleconsultation) has a significant 0.91% improvement in HbA1 _c over the control group. (p =.002)	9/ High

Rossi, 2010 ^a	Parallel group RCT; <i>Italy, England and</i> <i>Spain</i> ; Type 1; N = 130; <i>Duration</i> = 6 months; <i>Attrition</i> = 11(8.5%)	To evaluate whether the use of mobile software intervention could be effective in improving metabolic control in type 1DM, while avoiding weight gain and reducing time devoted to education. Also, to investigate to what extent the mobile software could affect quality of life	1 intervention group who received: (i) a mobile app to log BG data and insulin dose. (ii) With logged information on physical activity, glycaemic target and specific events, the app can calculate and suggest most appropriate insulin dose and daily CHO intake. <i>Comparator</i> : Usual care	Similar 0.5% reduction in HbA1c in both groups [Intervention: from mean 8.2% (SD 0.8%) to mean 7.8% (0.8) to mean 7.8% (SD 0.8) and control: from mean 8.4% (SD 0.7%) to mean 7.9% (SD 1.1)]. p=.68	9 / High
Waki, 2014	Parallel group RCT; <i>Tokyo</i> ; Type 2; N = 54; <i>Duration</i> = 3 months; Attrition = 5 (9.3%)	To access the usability of a remote health monitoring system and especially its impact on modifying patients lifestyle and clinical outcomes	1 intervention group who received: (1) A smart phone with app. (ii) Glucometer, Bluetooth-enabled BP monitor, pedometer and scale. All devices were paired with a unique communicator that transmit the measured data through wireless network to a mobile app server. <i>Comparator</i> : Usual care	Intervention group had significant improvement (0.4%) in HbA1c [from mean 7.1 (SD 1.0%) to mean 6.7 (SD 0.7%)] compared to 0.1% increase in HbA1c in the control group [Mean 7.0% (SD 0.9) to mean 7.19% (SD 1.1%)] (<i>p</i> =.019).	8 / High
Holmen, 2014	3 arms prospective RCT; <i>Norway</i> ; Type 2; <i>N</i> =151; <i>Duration</i> = 12 months; <i>Attrition</i> = 31(21%)	To test whether the use of mobile phone based self- management system used for 1 year with or without health counselling by a diabetes specialist nurse for the first 4 months could improve HbA1 ^c , self-management and health-related quality of life with usual care	2 intervention groups. (A). One group received a mobile phone app only. (b) The other received an app plus health counselling via telephone by a diabetes nurse. <i>Comparator:</i> Usual care	HbA1c decreased in all three groups but no difference between groups. No specific data was reported	8 / High

Orsama, 2013	Parallel group RCT; <i>Finland</i> ; Type 2; <i>N</i> =56; <i>Duration</i> =10 months; <i>Attrition</i> = 8 (14.3%)	This research involve the development and evaluation of a mobile telephone based remote patient reporting and automated telephone feedback system, guided by health behaviour change theory, aimed at improving self- management and health status in individuals with type 2 DM.	1 intervention group who received: (i) a mobile app for reporting BP, weight and physical activities. In specific cases BG meters was provided to those with high HbA1c values. (ii) Measuring device for BP, weight and physical activity (pedometer). (iii) Automatically generated health promotion information, motivation and behavioural skills feedback messages linked to patient remote self-reported health parameters. (iv) Participants had access to personal health records such a medication, laboratory data and personal care plan. <i>Comparator</i> : Usual care	A significant 0.4% improvement in mean HbA1c in the intervention group [from -0.67% to -0.14% compared to 0.036% in the control group [from -0.23% to 0.30%] (<i>p</i> =0.03).	8 / High
Quinn, 2016	Cluster group RCT; USA; Type 2; N =118; Duration = 12 months; Attrition = Not reported; this was a subset analysis of Quinn 2011	To determine if there were differences in the impact of mobile intervention in younger adults (<55 years old) versus older adults (≥55 years) within the below 64 years old patient population. We were interested in understanding if there were difference for the cohort nearing 65 years old.	1 intervention group (with older and younger subgroup) who received: (i) a mobile app and HCP decision support. App allowed participants to log BG, CHO and medications. (ii) Automated real time education, behavioural and motivational messages in response to logged data. (iii) Logged data were intermittently reviewed by a diabetes educator with whom patients are allowed to have telephone conversation. (iv). Web portal to receive supplemental messages, access personal health records and communicate with HCP. (v) A mobile phone, 1 year unlimited data / service plan and the study app. (vi)Received electronic action plan on diabetes management every 2.5 month and pre-visit summary for physician office visit: <i>Comparator</i> : Usual care	Decreased mean HbA1c in both groups. The intervention arm [older and younger patients] had HbA1c decline of 1.8% and 0.2% respectively while the control group [older and younger patients had HbA1c decline of 0.3% and 0.1% respectively]. (<i>p</i> =.0001)	7 / High

Quinn, 2011	Cluster RCT; USA; Type 2; N = 213; Duration: 12 months; Attrition: 50 (23.4%)	To test whether adding mobile app coaching and patient/provider Web portals to community primary care compared with standard diabetes management would reduce HbA1 _c in patients with type 2 diabetes	3 intervention groups: (A). Who all received: (i) a mobile phone, app, BG meter, one-year unlimited mobile phone data/service plan and web based portal. App allowed participants to enter BG, CHO consumed and medications. (ii) In response to logged data, participants received automated real time educational, behavioural and motivational messages. (iii) Received text messages on their portals as a supplement to automated messages. (iv) Feedbacks summaries of entered data and self-management action plan every 2.5 months, which also serves as pre-visit summary for patient's next visit to HCP. (B). In addition to the above; group 2 HCPs have access to a portal if they choose to review patients unanalysed data. (C). Group 3 HCPs have access to review patients analysed data on the web portal. <i>Comparator:</i> Usual care	Mean HbA1c decline in maximal treatment group of 1.9% and 0.7% mean HbA1c decline in the maximal intervention and control groups respectively. Significant HbA1c improvement was observed between the maximal intervention and control group only (<i>p</i> =.001). No significant changes between other groups	7 / High
Istepanian, 2009	Parallel group RCT; <i>London</i> ; Type 1&2; <i>N</i> = 137; <i>Duration</i> = 9 months; <i>Attrition</i> = 50 (36.5%)	To evaluate m-health system against usual care	1 intervention group who received: (i) a mobile phone app (ii) 2 hours education on general diabetes care and self-BG monitoring (iii) BG meter. <i>Comparator</i> : Usual care and received 2 hours education on general diabetes care and self-BG monitoring	There were no differences in mean HbA1c between the intervention and control groups: 7.9% and 8.2% respectively (p =.17).	6 / Moderate

BP: Blood pressure; BG: Blood glucose; CHO: carbohydrate; HbA1c: Glycosylated hemoglobin; HCP: Health Care Providers; RCT: randomised controlled trial

Details of the developmental factors considered in each of the reviewed studies and the resulting key clinical outcome (HbA1_c) are available in **Appendix 2.1** and **Appendix 2.2**.

<u>Health Behavioral Theories</u>: Only 1/11 (9.1%) study (Orsama et al., 2013) reported on health behavioural theories. Specifically, motivation behavioural skills model was used for the formulation of an automated personalised feedback message content of the mobile app.

Functions of mobile apps: It was apparent from the review that features of the mobile apps were diverse. However, documentation for self-monitoring of blood glucose (BG) either manually or through wireless transmission from BG meter was present in all studies. A total of 8/11 (72.2%) studies had mobile apps with capacity for diet management (Quinn et al, 2011; Quin et al., 2016; Rossi et al., 2010a; Rossi et al., 2013; Charpentier et al., 2011; Kirwan et al., 2013; Waki et al., 2014; Holmen et al., 2014). Three studies incorporated blood pressure features in their mobile apps (Orsama et al., 2013; Waki et al., 2014; Kim et al., 2014). There were 7/11 (63.6%) studies which had a physical activity feature (Rossi et al., 2010a; Rossi et al., 2013; Charpentier et al., 2011; Kirwan et al., 2013; Orsama et al., 2013; Waki et al., 2014) and 2/11 (18.2%) studies incorporated weight tracking feature (Orsama et al., 2013; Waki et al., 2014). There were specific features to log or calculate insulin dosages in mobile apps employed in the 4/11 (36.3%) studies with type 1 diabetes participants (Rossi et al., 2010a; Rossi et al., 2014). There were specific features to log or calculate insulin dosages in mobile apps employed in the 4/11 (36.3%) studies with type 1 diabetes participants (Rossi et al., 2010a; Rossi et al., 2010a; Rossi et al., 2010a; Rossi et al., 2013; Charpentier et al., 2011; Kirwan et al., 2013). A total of 2/11 (18.2%) studies reported a general medication log feature in their mobile apps (Quinn et al., 2011, Quinn et al., 2016).

With the exception of 4/11 (36.4%) studies, all others (7/11, 63.6%) had capacity for mobile apps to allow patients to analyse logged data. These 4 studies had their logged data transferred to a web/cloud storage and analysed by either the researcher or the health provider (Quinn et al., 2016; Quinn et al., 2011; Istepanian et al., 2009; Kim et al., 2014).

There were 6/11 (54.5%) studies that utilised mobile apps with an educational function. Half 3/6 (50.0%) of the studies provided education as a personalised real-time automated educational feedback specific to logged data (Quinn et al., 2011, Quinn et al., 2015; Orsama et al., 2013), while the other 3 provided a general information page (Rossi et al., 2010; Rossi et al., 2013; Holmen et al., 2014).

A total of 6/11 (54.5%) studies utilised a mobile app with reminder/alert function (Rossi et al., 2010a; Rossi et al., 2013; Istepanian et al., 2009; Kirwan et al., 2013; Orsama et al., 2013; Holmen et al., 2014).

<u>Users' Involvement:</u> There was only 1/11 (9.1%) study (Holmen et al., 2014) that clearly described users' involvement in the design of its mobile app. It reported an iterative design process involving 12-15 diabetes patients using the approach of focus group meetings, semi-structured interviews, usability testing, questionnaires and paper prototyping. This approach generated the design requirements and answers to research questions (Arsand et al., 2010).

<u>*Clinical Expert Involvement*</u>: There were 2/11 (18.2%) studies (Quinn et al., 2011; Quinn et al., 2016) which used the same mobile app and engaged the opinions of clinical experts in the field of diabetes during its development and design. The studies reported that the mobile app development involved an Endocrinologist and a Credentialed Diabetes Educator (Quinn et al., 2008).

Data Security and Privacy Considerations: Report on data security and privacy varied among the studies with limited elucidation of information in most cases. In 2/11 (18.2%) studies (Quinn et al., 2011, Quinn et al., 2016) the authors reported a real time capturing of self-monitored blood glucose data into a Health Insurance Portability and Accountability Act-compliant secured Web-based system (Quinn et al., 2008). In 1/11 (9.1%) study, measured data from participants were transmitted to a server. With each new measurement, the patient profile

was updated allowing controlled access to patients' data and record history (Waki et al., 2014). Transfer of mobile app data into a secured central server was the only information provided by Charpentier et al., (2011).

Pilot Testing of Mobile Apps: A total of 7/11 (63.6%) studies provided information with regards to pilot testing. Of these, 2/11 (18.2%) (Quinn et al., 2011; Quinn et al., 2016) reported three months test running of the mobile app on 30 patients with type 2 diabetes with the aim of evaluating the impact on HbA1_c and satisfaction of patients with the technology (Quinn et al., 2008). Likewise, 1/11 (9.1%) study (Charpentier et al., 2011) reported a 4-month open label observational pilot study on 35 type 1 diabetic patients with the aim of confirming if the use of the mobile app resulted in good control of post prandial blood glucose readings (Franc et al., 2009). Only 1/11 (9.1%) study (Waki et al., 2014) reported a one-month piloting on 11 type 2 diabetes patients to assess usability and impact of the mobile app on HbA1c outcomes and home blood pressure monitoring (Waki et al., 2012). In 2/11 (18.1%) studies (Rossi et al, 2010a; Rossi et al., 2013), 2 pilot programs were reported through a citation in another article. The first was with the use of a questionnaire to assess the feasibility and acceptability of the mobile app. The second was a 9-months follow up of 41 patients using the mobile app under routine clinical practice condition with the aim of investigating its effectiveness on metabolic control (Rossi et al., 2009). Lastly, 1/11 (9.1%) study (Holmen et al., 2014) reported a 12month pilot testing on 12 persons with type 2 diabetes (Arsand et al., 2010).

2.8 DISCUSSION

Theoretical basis

Our review shows that most of the studies did not discuss consideration for health behaviour theories in their mobile app development. The lack of report on theoretical basis may be as a result of reliance on evidence-based guidelines that relates to the essential self-management activities in people with diabetes to predict good outcomes American Diabetes Association, 2008). While it is necessary for mobile apps to be guided by health behavioural theories, the current theories appear incapable of answering most of the questions likely to arise when mobile apps are employed as health interventions (Riley et al., 2011). Dunton and Atienza (2009) reported that current health behaviour theories have not been able to incorporate within-person differences, which allow for intra-individual tailoring of interventions. Borsboom and co-authors (2003) noted that between people theories do not imply, test or support causal factors valid at the individual level. Therefore, there is a need for more research into intra-individual non-static regulatory models, which can be incorporated in the development of mobile technology-based health behavioural interventions.

Functionalities of mobile apps

All the 11 trials reviewed in this study included mobile apps with documentation/monitoring component, where self-documentation of blood glucose readings was the most common. Only 3 studies used mobile apps that offer automated direct data transfer of blood glucose values from the glucometer or data from other measuring devices (Istepanian et al., 2009; Waki et al., 2014; Kim et al., 2014). This corroborates the report by Demidowich et al., (2012) where only four of the 42 mobile apps studied offered direct data input from glucometer. Data entry is often perceived as a persistent burden in chronic disease management (Arsand et al., 2008). Therefore, it is imperative that data entry in mobile apps be as spontaneous as possible, requiring little time and effort to use (Jensen & Larsen, 2007). Mobile App developers should prospectively consider including an interface between the app and biomarker measuring devices, which allow users to automatically log measurements. Such interface may include Bluetooth, which enables portable electronic devices to connect and communicate wirelessly (Haartsen, 2017). The success of using this interface was demonstrated in the studies by Waki et al., (2014) and Holmen et al., (2014).

Data analytics as an app feature was included in only 7/11 (63.3%) studies. A consumerdirected software such as mobile app is better incorporated with functions that enable users to enter, analyse their health parameters and view graph trends and statistics. This can improve the patient's ability to observe the impact of their lifestyle and behaviour on health indicators, access trends and even predict health outcome measures (Winters-Miner, 2014). Additionally, decision-making and problem-solving skills of patients can be improved when mobile apps include visualisation techniques such as colour-coded charts or graphs, which indicate when biomarkers, food carbohydrate component and physical activity are out of recommended range (Breland, Yeh & Yu, 2013). It is essential that analytic functions be dynamic, easily accessible and able to project trends to predict individual improvement in self-care activities, which may invariably lead to better health outcomes (Preuveneers & Berbers, 2008; Li & Fernando, 2016).

Despite the emphasis by published guidelines for the need for ongoing patient education (American Diabetes Association, 2016), very few studies used mobile apps that have education as a functionality. This finding is corroborated by another review where the authors confirmed personalised education as an underrepresented feature in diabetes mobile apps (Chomutare et al., 2011). Patients may have difficulty consulting with their diabetes educators or other health care professionals, due to lack of time, financial constraints, and other limitations. Hence, an app with an educational component can supplement healthcare provider diabetes education and reinforce information about the importance of self-management and complication prevention. This can serve as an avenue for continual patient empowerment to successfully deal with the disease. However, it is essential that the personalised educational feedback and advice provided in mobile apps are accurate. This is especially true for those that are automatically generated because monitoring mobile apps pose serious harm to the patients if they fail to function as intended (Barton, 2012).

A total of 6/11 (54.5%) studies reported using mobile apps with reminder function either in the form of prompting to measure missed blood glucose readings or alerts for appointments scheduled for the assessment of complication Quinn et al., 2011, Quinn et al. 2016; Istepanian et al., 2009; Kirwan et al., 2013; Waki et al., 2014; Holmen et al., 2014). They are sometimes referred to as 'push technology'; which enables messages to be delivered without any effort on the part of the recipient (Klasnja & Pratt, 2012). Such reminders can be in the form of text message, alarm, email, automated voice call or image message. Other review has illustrated the benefits of an alarm reminding patients to carry out their health activities (Benferdia & Zakaria, 2014). Another study revealed improvement in treatment adherence as patients get fascinated using reminders to handle their healthcare activities (Wohlers et al., 2009).

Users' involvement

Similar to an earlier review by El-Gayar et al., (2013) on the adoption of user-centered design principles in mobile apps, only one study (Holmen et al., 2014) documented inquiry into users' expectations and perceived needs in the app developmental phase. Users' involvement in design process increases the success rate of computerised system usability (Abras et al., 2004), as it is essential to understand the reasons for use and user requirements (Bevan, 2009; Goldberg et al., 2011). In contrast, a design process lacking the involvement of users in the design loop will fail to recognise the particular odds and problems in the use of the intervention (Höök, 2004). Design processes can use research tools such as questionnaires, focus group discussions and personal interviews. These help to seek users' requirements, preferences, understand current challenges and barriers to self-management and subsequently incorporate the findings into the design process. Incorporation of feedback during app design process can help in producing a more user-friendly application and encourage long-term user engagement.

Clinical experts' involvement

Many of the apps reported in the studies reviewed were designed without the involvement of healthcare professionals, and this observation is supported by an earlier review (Arnhold, Quade & Kirch 2014). Involvement of health professionals in diabetes mobile app development can assure the quality of health information and support provided by such apps (Boulos et al., 2014). This is especially important in mobile apps involving advice on insulin dosing. It has to be mentioned that the 3/11 (27.2%) studies in this review which used mobile apps to assist participants in calculating insulin dosage failed to report whether clinical experts were involved in the development of these apps, even though HbA1_c levels in the intervention groups were not significantly lowered compared to the control groups (Rossi et al., 2010a; Rossi et al., 2013; Charpentier et al., 2011). This finding highlights possible issues with the effectiveness, efficiency, and relevance of these mobile apps to users' health security. Insulin overdose in people with diabetes can result to severe hypoglycemia and coma while under-dose can cause diabetes ketoacidosis; both can have fatal consequences (Seaquist et al., 2013; Wolfsdorf et al., 2013). Participation of health professionals in the development of diabetes mobile apps may decrease the likelihood of such fatal occurrences and protect consumers from incorrect and misleading information. Furthermore, clinical expert involvement in diabetes mobile app development will foster avoidance of legal implications surrounding noncompliance to regulatory and medical standards that relate to digital health services especially those which empower people to track, manage and make decisions about their health (United State Department of Health and Human Services, 2000; Australian Government, Federal Register of Legislation, 2016).

Data security and privacy

Information on data security and privacy considerations in mobile app development were lacking in many of the trials in this review. Late consideration of privacy and security are app developers' errors that cannot be underestimated. Medical data breaches resulting from failed security attract huge financial implications (such as costs associated with a pecuniary penalty, potential liability claim, lost brand value, responding to lawsuits, negative press statements and essentially loss of patients' and health care providers' trust) for non-compliant organisations (United State Department of Health and Human Services, 2000; Australian Government Federal Register of Legislation, 2016). Studies have revealed that some users are concerned about the privacy of their personal health information stored on an electronic device (Chhanabhai & Holt, 2007; Zurita, Nohr & Medinfo, 2004). Procedures to maintain health data privacy and security to avoid data breaches must, therefore, be considered during mobile app design. Encrypted storage, which ensures logged data are protected against malicious attack is a security approach to protecting health data on mobile apps (Kumar & Lee, 2012). Furthermore, the privacy of users' information can be ensured through user authentication or enforcement of password requirements (Kumar & Lee, 2012), and this can protect users' health data in case of mobile phone loss.

Pilot testing

There were 5/11 (45.5%) studies that failed to report on pilot testing of their apps before use in the trial. A previous study also reported that most health apps do not offer patients ample opportunity for feedback on the level of satisfaction and usability of the product (Arsand et al., 2012). The importance of pilot testing mobile apps cannot be overemphasised. Apart from serving as an avenue for testing the impact of the app on behavioural outcome or glycemic control, pilot testing can assess its user-friendly capacity and adherence for use as a self-management tool.

Developmental factors considered in mobile apps and the key clinical outcome (Glycosylated Hemoglobin)

Appendix 2.1 and Appendix 2.2 show an overall evaluation of the developmental factors considered in the design of the mobile apps used in the reviewed studies and the resulting clinical outcome (ie, glycosylated hemoglobin-HbA1_c). Appendix 2.2 highlighted 5/11(45.5%) studies that had intervention groups with significantly improved HbA1c. A comparison of these 5 studies showed that educational functionality was present in all. For example, 3/5 (60.0%) studies provided the educational information directly through the mobile app (Quinn et al., 2011; Orsama et al., 2013; Waki et al., 2014) while 2/5 (40%) provided additional text messaging or teleconsultation (Charpentier et al, 2011; Kirwan et al., 2013). It is likely that the similar outcomes observed in these studies were partly due to similitude in the provision of self-management education to participants, as digital tools with decision support features such as education have been proven to have the capacity to enhance self-management outcomes (Greenwood et al., 2017). This finding demonstrates the importance of consistent and ongoing provision of self-management education to people with diabetes. Diabetes education and diabetes management are inseparable because every patient would benefit from education in self-management. Therefore, in addition to other essential functionalities in mobile apps that support diabetes care, the inclusion of education functionality will provide the recommended ongoing support to promote the importance of self-management, build patient skills and confidence for behavioural change, increase motivation for self-management and ultimately improve glycemic control (Powers et al., 2015a; d Brunisholz et al., 2014).

Furthermore, 3/5 (60%) studies with significant improvement in HbA1_c reported on pilot testing of their mobile apps before use in the full trial (Quinn et al., 2011; Charpentier et al., 2011; Waki et al., 2012). It is possible that excellent efficacy observed in these studies was due to pilot testing. Among other reasons, an essential aim of pilot testing a technology is to

establish its usability. Usability testing of a mobile app examines end users' satisfaction and has been identified as one of the factors that determine its efficacy and success of users' engagement with it (Hornbak & Law, 2007).

Implications for practice and future research

Much work is needed to address challenges limiting the documentation and the implementation of developmental factors in the design of mobile apps for diabetes management. The use of mobile phone interventions in which the developmental design are not explicitly documented is likely to result in a non-replicable app with significant levels of wasted resources. Therefore, future work is required to promote the development of evidence-based apps research and clinical use. These mobile apps should focus on integrating functions to core diabetes selfmanagement practices and primarily with the provision of self-management education. Additionally, integrating theories of health behavioural change, users, and clinical experts' involvement while ensuring data privacy and security are essential factors to be considered in the development of future mobile apps.

Limitations of this review

There are limitations to be considered when interpreting and extrapolating the findings of this systematic review. The results of this review were dependent on the terms used in the search strategy and the efficiency of the search engines used. An attempt to overcome this limitation was ensured by choosing common terms and combination of terms usually used in the literature review on mobile health applications. This review considered only trials that were reported in the English language with strict inclusion criteria and so the number of articles that met the study criteria was small, and this limits the ability to generalise the findings. Also, the process of extracting the data presented some risk of error and uncertainty because some studies were not explicit about their developmental considerations, and it is easy to miss or misunderstand some development description either reported directly within the article or referenced.

However, to avoid this occurrence, the authors ensured that the assessment process involved independent verification and all pitfalls that might invalidate the findings were avoided. Despite these limitations, this review provides valuable information to future researchers and developers of mobile apps for DSM on the necessary factors to consider during app development.

2.9 CONCLUSION

This systematic review has presented the crucial steps that need to be taken in mobile app development to support effective self-management for people with diabetes. Most of the studies in this review offer a limited and non-expository degree of information on the factors considered in the development of the apps employed.

The main stakeholder in diabetes management is the patient. Shared decision-making between diabetes patients, researchers, health-care professionals, and app developers can result in improved management. Therefore, this should be the basis for the development of mobile apps for diabetes support. Shared decision-making can be achieved through the process of patient and clinical expert involvement, ensuring data security and privacy, pilot testing and integration of core functions that support all aspects of the activities as indicated by evidence-based guidelines. Continual integration of these processes during app development (before actual use in clinical trials) will ensure that specific needs of diabetic patients are met in the finally developed app, and this will ultimately improve diabetes support, self-management and clinical outcomes for the patients.

Overall, the systematic review in this chapter addressed the first aim of the thesis by providing a framework/guideline on the essential elements required in a diabetes app development process prior to use as an intervention for self-management by type 1 or type 2 diabetes patients. These findings highlight the need for primary studies that explore health behavioural change mediators for self-management, diabetes patients' educational needs, app feature preferences and opinions on how to improve ongoing engagement with apps. These were addressed in Chapter 3 and 4.

Chapter THREE: Enablers and Barriers to Effective Diabetes Self-Management: A Multi-national Investigation

3.1 ABSTRACT

The study aimed to identify the common gaps in skills and self-efficacy for diabetes selfmanagement and explore other factors, which serve as enablers of, and barriers to, achieving optimal diabetes self-management. The information gathered could provide health professionals with valuable insights to achieving better health outcomes with self-management education and support for diabetes patients. International online survey and telephone interviews were conducted on adults who have type 1 or type 2 diabetes. The survey inquired about their skills and self-efficacy in diabetes self-management, while the interviews assessed other enablers of, and barriers to, diabetes self-management. Surveys were analysed using descriptive and inferential statistics. Interviews were analysed using inductive thematic analysis. Survey participants (N=217) had type 1 diabetes (38.2%) or type 2 diabetes (61.8%), with a mean age of 44.56 SD 11.51 and were from 4 continents (Europe, Australia, Asia, America). Identified gaps in diabetes self-management skills included the ability to: recognise and manage the impact of stress on diabetes, exercise planning to avoid hypoglycemia and interpreting blood glucose pattern levels. Self-efficacy for healthy coping with stress and adjusting medications or food intake to reach ideal blood glucose levels were minimal. Sixteen participants were interviewed. Common enablers of diabetes self-management included: (i) the will to prevent the development of diabetes complications and (ii) the use of technological devices. Issues regarding: (i) frustration due to dynamic and chronic nature of diabetes (ii) financial constraints (iii) unrealistic expectations and (iv) work and environment-related factors limited patients' effective self-management of diabetes. Educational reinforcement using technological devices such as mobile application has been highlighted as an enabler of diabetes self-management and it could be employed as an intervention to alleviate identified gaps in diabetes self-management. Furthermore, improved approaches that address financial burden, work and environment-related factors as well as diabetes distress are essential for enhancing diabetes self-management.

3.2 BACKGROUND

Diabetes mellitus is a major public health problem with rapidly increasing prevalence. In 2017, the global prevalence of diabetes among people aged 20-79 years was 425 million, mainly comprising type 1 or type 2 (International Diabetes Federation, 2017). Diabetes is one of the top 10 global causes of mortality. In 2015, it was responsible for 1.6 million deaths, indicating a 60% increase in 15 years from less than 1 million in 2000 (World Health Organisation, 2018). International audits have found that regimen adherence is less than optimal in both types 1 and 2 diabetes patients (Peyrot et al., 2005). As a consequence, the majority of these patients are at risk of serious health complications that endanger life (International Diabetes Federation, 2017; Papatheodorou et al., 2015) and impose great economic burden on affected individuals and the health care system (International Diabetes Federation, 2017).

Consistent engagement in diabetes self-management (DSM) has been found to be correlated with the attainment of health outcomes. These are in relation to good blood glucose control, fewer complications (International Diabetes Federation, 2017;Viswanathan et al., 2005) improved quality of life (Povey & Clark-Carter, 2007; Chen et al., 2015) and reduction in diabetes-related death risks (UK Prospective Diabetes Study Group, 1998). The term "self-management" refers to day to day activities or actions an individual must undertake to control or reduce the impact of disease on their health and wellbeing (Clark et al., 1991) in order to prevent further illness (Barlow et al., 2002). DSM actions involve engagement in recommended behavioural activities such as healthy eating, medication adherence, being active, monitoring, reducing risks, problem-solving and healthy coping, which are all necessary for the successful management of the disease (Tomky et al., 2008). Level of adherence to DSM differs in patients,

which implies that decision-making processes for self-management are influenced by various factors, which could serve either as enablers or barriers.

Enablers of Self-Management

Enablers of self-management are mechanisms or factors that foster the ability of patients to undertake their recommended self-management regimen. Such factors are diverse and they include effective social support with assistance and encouragement from family members (Maillet et al., 1996; Chlebowy et al., 2010) or peers who have diabetes or close relative familiar with its management (Fisher et al., 2012). Likewise, individual resolution to prevent or reduce the risk of developing diabetes complications (Maillet et al., 1996; Cagle et al., 2002) helps with the determination to engage in self-management. Studies have also noted positive decision making about DSM as a result of effective health care provider-patient communication (Nagelkerk, Reick & Meengs, 2006), characterised by trust, respect and shared decisionmaking in planning health goals (Paterson & Thorne, 2000; Cooper, Booth & Gill, 2003). In addition, patient support with the use of health technological interventions such as mobile phone applications (Hunt et al., 2015) and self-management education (Atak, Gurkan & Kose, 2008; Haas et al., 2012) facilitate effective diabetes management. Individual factors, particularly higher educational level (Chlebowy, Hood & Lajoie, 2013; Al-Rasheedi, 2014) and gender (Carter, 1998) also contribute to patients' ability to care for their diabetes.

More importantly, adequate self-management skills (Persell et al., 2004) and self-efficacy (confidence) (Aljasem et al., 2001) to perform these skills are major enabling factors for engagement in DSM. This is because skills and self-efficacy operate in tandem to foster full engagement with self-management. Self-management skills result from knowledge about the disease (Persell et al., 2004), and understanding the interrelationships between different self-management activities and their impact on health outcomes (Herschbach et al., 1997). On the

other hand, self-efficacy refers to "one's belief in his/her own innate ability to perform specific tasks required to reach a desired goal" (Bandura, 1999). Unless people believe they can produce desired effects by their action, they have little incentive to act (Bandura, 1998), regardless of other enabling factors which may be available to them. In diabetes management, patients' level of self-efficacy is influenced by their level of skills for self-management. Hence, patients with adequate skills and efficacy have more likelihood to adhere to prescribed behavioural regimen necessary to attain optimal health (Persell et al., 2004; Morowati-Sharifabad et al., 2007; Johnston-Brooks, Lewis & Grag, 2002; Aronson et al., 2018). Acquiring skills and efficacy for DSM are ongoing learning processes (Persell et al., 2004; Haas et al., 2012). While some skills and efficacy are easily acquired, others are often difficult to attain. Further research is therefore needed to adequately identify gaps in diabetes patients' skills set and self-efficacy levels for self-management of their health issues. Information on identified gaps will guide health care providers in their development of educational support programs that foster self-management among diabetes patients.

Barriers to Self-Management

Non-adherence to recommended DSM regimen is influenced by barriers encountered by patients. These barriers make managing the disease more difficult. Only few studies have examined patients' perceived barriers to general DSM from a global perspective. An international study identified diabetes related distress as a major factor responsible for poor adherence to self-management in patients (Peyrot et al., 2005). Local studies reported that difficulty in making lifestyle changes (Byers et al., 2016) and inadequate health care system communication interface (Jones et al., 2014) were related to poor DSM. In addition, financial constraints resulted in patients' inability to access diabetes clinical supplies and eat in line with appropriate dietary recommendations (Hunt, Pugh & Valenzuela, 1998; Schoenberg & Drungle, 2001; Campbell et al., 2017). Other studies have examined barriers to some specific

areas of DSM. Nagelkerk et al., (2006) and Ghimire (2017) reported that patients' lack of knowledge of a specific diet plan and perceived belief in social unacceptability of healthy behaviours hindered healthy eating and participation in physical exercise. Furthermore, depressive symptoms and personal belief about medication were observed to be associated with lower adherence to diabetes medications (Chao et al., 2005).

The empirical and conceptual research findings mentioned above are not exhaustive because only a few have an international focus (Peyrot et al., 2005). Additionally, the studies are mostly focused on barriers to self-management in patients with type 2 diabetes only (Hunt et al., 1998; Schoenberg & Drungle, 2001; Jones et al., 2014; Cheng et al., 2016; Ghimire, 2017; Aronson et al., 2018), older populations (Schoenberg & Drungle, 2001), those from low income background without indicating the type of diabetes the respondents had (Gazmararian, Ziemer & Barnes, 2009) or few areas of DSM (Ghimire, 2017; Chao et al., 2005; Cheng et al., 2016). The above limitations in previous studies emphasise the need for further and detailed exploration of factors serving as barriers to self-management in both types 1 and 2 diabetes patients. This will provide strategies that adequately address such challenges and foster better adherence to self-management for better health outcomes in both patient groups.

Study Aims

There is diversity in the level of self-management between patients. The ability to self-manage diabetes is influenced by various factors that can either serve as enablers or barriers. However, to the best our knowledge, global perspectives on the crucial enablers of self-management in terms of skills and self-efficacy, among types 1 and 2 diabetes patients is relatively scarce. Likewise, studies on other enablers and potential barriers to general DSM as perceived by these patient groups is scanty in the published literature. There is special interest in elucidating this information from an international perspective because issues encountered in self-management

by both patient groups are likely to include common experiences and challenges. Identifying these commonalities could provide health professionals with an in-depth understanding of patients' experiences and help guide the development and enhancement of intervention strategies to improve patients' self-management of diabetes. Therefore, this study aimed to: i) identify the common gaps in skills and self-efficacy for self-management among individuals with type 1 or type 2 diabetes; ii) examine factors associated with self-management skills and self-efficacy; iii) explore other factors which serve as enablers of, and barriers to, achieving optimum DSM.

3.3 METHODS

3.3.1 Recruitment Procedure

A maximum variation purposive sampling technique was employed in recruiting participants aged ≥ 18 years who have type 1 or type 2 diabetes. Participants were recruited globally using diverse recruitment strategies. The aim of this sampling method was to obtain a mix of participants with diverse experiences and identify common patterns that cut across the population sample with regards to the subject of interest (Palinkas et al., 2015). Officially approved advertisement for the study was placed on various health organizations' websites. These websites included Diabetes UK and Diabetes Australia. In addition, the advertisement was placed in local digital newspapers, Twitter and Facebook pages focusing on diabetes support. Data collection was conducted between November 2017 and June 2018. There was no limit to sample size in order to capture the maximum number of people with type 1 or type 2 diabetes. The study requested participants' socio demographic characteristics of age, gender, educational level and geographic location. Details of the recruitment strategy and participants' characteristics have been fully described in our previous publication (*currently chapter 4 of this thesis*) (Adu et al., 2018b).

3.3.2. Study Design

A sequential mixed methods approach was used; comprising quantitative and qualitative data collection methods (Creswell et al., 2003). The quantitative phase of the study involved a cross sectional survey and data analysis. This was followed by qualitative telephone interviews of a subsample of the participants in order to provide a more complete and comprehensive understanding of the results which were integrated into the data interpretative phase (Creswell et al., 2003). Quantitative data were obtained through an online survey that focused on assessing participants' self-reported skills and self-efficacy (confidence) as part of the factors that enable DSM. Qualitative data were collected through individual telephone interviews, which further explored additional factors that serve as enablers and barriers to DSM.

Quantitative measures – Survey

The survey questions were divided into two parts. First, the following health characteristics which were likely to influence skills and self-efficacy for DSM were assessed: type of diabetes, duration of diagnosis and whether participants had recently received (within the previous 12 months) diabetes self-management education (DSME) from a member of their health care team.

Second, novel LMC Skills, Confidence and Preparedness Index (SCPI) tool was used to assess skills and self-efficacy in core behaviours central to DSM such as healthy eating, blood glucose monitoring, being active, healthy coping, medication adherence, problem solving and reducing risk (Tomky et al., 2008; Mbuagbaw et al., 2017). The SCPI tool had been previously validated, where its construct validity for different ages, ethnicity, gender and level of education was established (Aronson et al., 2018). Additionally, the validity of the tool for use in different settings is established by the fact that, as a new tool, the questions reflect the currently recommended self-management regimen for diabetes patients, and this has not been fully

explored by previous tools (Mbuagbaw et al., 2017). It has excellent readability and reliability. Permission was obtained to use the tool. The SCPI tool consists of three subscales: skills, confidence and preparedness. The skills subscale was used to assess perceived ability to perform the self-management activities mentioned above. The confidence subscale was used to assess self-efficacy in being able to perform the skills. The preparedness scale was not used in this study because this subscale assesses the readiness of patients to implement behavioural changes following an educational session; which was not applicable in the present study.

The skills and confidence domains consist of nine (9) and eight (8) items respectively. Two of these items focus on skills and confidence to use insulin. These skills were adapted to accommodate participants who have type 2 diabetes but do not use insulin/other medications as part of their treatment regimen. All items were rated using a visual analogue scale, with scores between 1 and 10. Each of the items in the domains produced its own score out of 10. The total score was the mean score in each of the subscales, where higher scores denoted better skills and confidence. The scoring process is not affected by demographic factors such as age, gender, level of education or ethnicity (Mbuagbaw et al., 2017), hence, its' applicability for use in study populations with diverse social and health characteristics. The instrument was administered in English Language.

Qualitative measures – Phone interviews

Through the online survey, all participants were invited to participate in an individual telephone interview. They were requested to indicate interest by providing their best contact number and availability. A single independent resource person (male) who is an experienced researcher in qualitative studies conducted all interviews. The interviewer was trained on the aims of the study and the interview guide by MDA. The guide (see **Appendix 3.1**) was then pilot tested between the interviewer (AD) and MDA before actual use. Additionally, MDA was present in

the first three interviews to ensure appropriateness of data collection. While the interviews were used to reflect on the interview guide, no changes were made to the guide afterwards. There was no interaction or previous relationship between MDA and the participants. The interviewer was located in a private office at James Cook University, Townsville, Australia. Prior to the commencement of the interview, each respondent was asked if they were located in a comfortable place for an interview, and were briefly presented with the general idea of the study and key diabetes self-management activities. The interviewer did not have prior relationship with the participants. Each interview was audio recorded and lasted between 7 and 20 minutes, with an average duration of 12 minutes. Data saturation was achieved through recurring explicit ideas (Guest, Brunce & Johnson, 2006) after completing the 14th interview. However, the interview was conducted for the remaining two participants who had indicated interest in order to ensure that no main idea was unintentionally discarded. Repeat interviews were not required and due to the remoteness of the study participants, there was no post interview debriefing. The semi-structured interview guide was developed by the research team. Topics covered in the interview included open ended questions and probes to facilitate discussion.

Ethics and Consent

The study procedures (registration number: H7087) were approved by James Cook University's human research ethics committee. The protocol detailed information on the ethical obligations of researchers toward participants engaging in online research activities. Essentially, these obligations included confidentiality, anonymity, scientific value, maximising benefits, minimizing harms, and informed consent (Hewson & Buchanan, 2017). All these obligations were strictly adhered to during the research process. Furthermore, as part of the application process for advertisement of the study on the website of health organisations, the ethics approval document was made available to the appropriate and designated officials of these organisations. All prospective study participants were provided with the study information along with the privacy policy prior to the survey. Therefore, participants were informed about the use of their answers for analysis under anonymity. Informed consent was implied by submission of the online survey, while all telephone interviewees provided verbal consent.

3.3.3 Data Analyses

IBM SPSS (Version 23) was used for quantitative data analysis. Cronbach's alpha of the subscales of measure used in this study was acceptable (0.92 and 0.91 for skills and confidence scales respectively). Participants' demographics and health variables were presented using descriptive statistics. Items in the skills and self-efficacy domains were reported as means and standard deviations (SD). For the purpose of explaining and discussing the results, scores were graded as high (\geq 7), moderate (4-6) or poor (\leq 3). Mean scores were calculated for demographic and health variable subgroups. Bivariate analyses were performed using Independent sample t-test and Analysis of Variance (ANOVA) to test the relationship between participants' subgroups and level of skills and confidence. Specifically, t-test was used for variables with two categories (i.e. type of diabetes, received DSME or not, gender) while ANOVA was used for variables with three or more categories (i.e. educational status, duration of diagnosis, geographic location, age range). Effect sizes were calculated using Eta squared values to show the magnitude of difference in mean scores between categories within each variable. Pearson correlation coefficients were used to estimate the strength of association between skills and self-efficacy scores. Additionally, multiple regression analysis was used to estimate the contributions of the different independent variables to participants' reported skills levels. Significant variables in the bivariate analysis were included in the regression. In all statistical analysis, values were considered statistically significant at p < 0.05 (two tailed).

For qualitative data analysis, audio recordings were transcribed verbatim by an independent professional transcriber and reviewed by MDA for accuracy. The transcripts were uploaded into a qualitative data analysis software (QSR Nvivo 11). Emerging themes were identified using in-depth inductive thematic analysis (Braun & Clarke, 2006) undertaken in six steps. These include: (i) Re-reading of data line by line to ensure familiarization (ii) identification of patterns within data and organization into codes (iii) grouping of initial codes through constant comparison to identify emerging themes (iv) grouping and review of identified themes into general themes (v) refining themes and (vi) selection of representative quotes to support themes (Braun & Clarke, 2006). The first coding and generation of themes was done by MDA. In order to enhance result credibility and validity, raw data transcripts, coded data and themes were independently reviewed by BMA. Data were cross-checked in a consensus meeting and there was 90% degree of congruence between both authors' coding, themes and classifications. Discrepancies were resolved through discussion and mutual agreement. Both MDA and BMA have experience in qualitative research methods. UM and AEOMA checked the quotes and themes to ensure consistency. Key themes were reported along with relevant quotes affixed with an assigned number code and the type of diabetes the respondent has (for instance P3, T2D). The final manuscript was subjected to COREQ checklist for consolidated criteria for reporting qualitative research (Tong, Sainsbury & Craig, 2007).

3.4 RESULTS

3.4.1 Socio-Demographic and Health Characteristics

A total of 217 complete responses to the online survey was received. Respondents were located in four geographic regions; namely, Europe (35%), Australia (34.6%), Asia (29.5%) and America (0.9%). The mean age of respondents was 44.65 ± 14.0 years (range 18-76 years) and 56.7% of them were females. More than half of the respondents had type 2 diabetes (61.8%) and had received DSME in the previous 12 months prior to the study (64.1%). About half of them were diagnosed in the last 5 years (52.5%) while 20.3 % were diagnosed 6-10 years ago and the remaining 27.2% over 10 years. Over half of the respondents (56.2%) reported having a minimum of bachelor's degree, 20.3% completed high school, while 18.9% completed technical college and 4.6% attained other forms of education.

A total of 31 respondents (14.3%) expressed interest to participate in the telephone interview. However, about half of them declined at time of interview or never responded to phone calls, leaving a final respondent number of 16 individuals who were interviewed. The participants were mostly males; 56.2% (9/16), had type 1 diabetes; 62.5% (10/16) and lived in Australia; 87.5% (14/16), with age ranging from 26 to 61 years [mean age of 44.56 (SD 11.51).

3.4.2 Diabetes Self-Management Skills and Self-Efficacy (Confidence)

Table 3.1 shows the mean scores for each of the items across the skills and self-efficacy domains. Scores were highest in the skills for knowing the appropriate time to check blood glucose levels in order to reflect either the impact of meals consumed ($\bar{x} = 7.81 \pm 2.33$) or medications/physical activities ($\bar{x} = 7.47 \pm 2.37$). In addition, participants possessed a high ability to recognise the effect of missed physical activity or excess carbohydrate consumption on their health ($\bar{x} = 7.35 \pm 2.35$). The lowest scores were in the areas of skills for: identifying and managing the impact of stress on diabetes ($\bar{x} = 6.88 \pm 2.43$), exercise planning to avoid hypoglycemia ($\bar{x} = 6.88 \pm 2.48$), and interpreting blood glucose patterns ($\bar{x} = 6.84 \pm 2.58$).

In relation to participants' self-efficacy levels, the highest scores were in confidence to reduce risk by preventing and monitoring diabetes complications ($\bar{x} = 8.08 \pm 1.85$), and using blood glucose results to plan for meal intake ($\bar{x} = 7.22 \pm 2.06$). Participants scored lowest in their confidence for healthy coping with stress ($\bar{x} = 6.72 \pm 2.28$) and adjusting medications or food intake to reach targeted blood glucose levels ($\bar{x} = 6.87 \pm 2.62$). There was a strong positive correlation between the scores in the two domains, r = 0.906, p < 0.001, where higher levels of perceived skills were associated with higher levels of perceived self-efficacy. Coefficient of determination (R²) indicates that level of skills explained 82% of the variation in respondents' scores on self-efficacy.

Table 3.1: Participants Skills and Self-efficacy (confidence) Ratings to Perform Diabetes Self-

Management

SKILLS	Mean	SD
I am able to portion out and choose foods that have the minimal balance between carbohydrates, proteins and vegetables to keep my blood sugar in target	7.23	1.97
I know how my diabetes insulin and medication works in my body and at what time of the day I should check my blood sugar(BS) to make sure my dose is correct (<i>For T2D not controlling with insulin and medication</i> : I know how my diet and physical activities impact my BS and at what time of the day to check my BS to make sure they are in target)	7.47	2.37
If I eat too much carbohydrate, or do not engage in my regular physical exercise, I know how my body will react and the steps to take to get it back on track	7.35	2.35
When I am planning to exercise, I know what changes I need to make to avoid low blood sugar before, during and after exercise	6.88	2.48
I know when to check my blood sugar if I wanted to see how my body reacted to a meal	7.81	2.33
When I am sick, I know what to do differently with my medications, fluids intake, food intake, blood sugar testing and when to go to the hospital	6.91	2.67
I know how to identify stress in my life and how it can impact my diabetes management and overall health	6.88	2.43
When I look at my blood sugar in my meter or in my log book in a given week, I could explain to my diabetes educator or doctor what my blood sugar pattern is	6.84	2.58
I know what the ABCs (A1c, Blood Pressure and Cholesterol) of diabetes are, what my targets are and how they impact my diabetes	7.00	2.54
Average score on skills	7.15	1.97
SELF-EFFICACY		
I feel confident that I can plan meals and snacks effectively in a way that it will not raise my blood sugar unnecessarily above my targets	7.22	2.06
I am confident that I can implement stress management techniques in my lifestyles	6.72	2.28
I am confident that at the next time I am eating out in my home, I will be able to plan and select the foods that best keep my blood sugar under control	7.06	2.34
--	------	------
I am confident that I can plan ahead for what to do and how to react either before, during or after exercise to avoid a low blood sugar	6.92	2.4
I am confident that I can choose a healthy physical activity for myself and include it in my schedule	7.16	2.26
I am confident that I can adjust my insulin or medication doses on my own, to reach the target blood sugar levels (<i>For T2D not controlling with insulin and medication</i> : I am confident that I can adjust my meals and levels of physical activities on my own to reach the target blood sugar levels)	6.87	2.62
I am confident that I can commit to preventing and monitoring my diabetes complications such as seeing my eyes doctor at least once in a year and checking my feet on daily basis	8.08	1.85
I am confident that I can use my blood sugar results to make changes to my diet and/or insulin to help keep my blood sugar in target	7.00	2.54
Average score on confidence	7.17	1.81

A1c: Glycosylated hemoglobin; T2D: Type 2 diabetes mellitus

Relationship between participants' characteristics and levels of skills and self-efficacy

Table 3.2 shows the relationship between demographic and health characteristics and the levels of skills and self-efficacy for DSM in participants. All demographic characteristics except geographic location, gender and age, were significantly associated with perceived skills and self-efficacy.

Participants who had type 1 diabetes had higher levels of skills compared to those with type 2 diabetes, t(215) = 17.41, p < 0.001, eta squared = 0.123. Additionally, receiving DSME within the past 12 months prior to participating in the study had a moderate but significant association with level of skills, t(215) = 2.01, p = 0.045, eta squared = 0.018. There was a significant difference in duration of diabetes diagnosis, F(4, 215) = 5.59, p < 0.001, eta squared = 0.095. Skill scores were significantly higher in the >15 years (M=8.28, SD=1.22) when compared to <1 year (M=6.28, SD=1.82), 1-5 years (M=6.98, SD=2.08) and 6-10 years (M=6.97, SD=2.14) of diabetes diagnosis. There was no significant difference for those with 10-15 years of diagnosis (M = 7.00, SD = 1.58). In addition, level of educational qualification significantly

influenced the level of skills, F(4, 215) = 7.87, p<0.001, eta squared = 0.132. Skill scores were significantly higher among postgraduate degree holders (M = 7.76, SD = 1.12) in comparison to high school (M=6.13, SD = 2.21) and technical school (M = 6.43, SD = 2.25) certificate holders. No significant difference was observed when compared to those with bachelor's degree (M = 7.76, SD = 1.53).

For self-efficacy (confidence), type 1 diabetes participants had higher confidence levels compared to their type 2 counterparts, t (215) = 5.46, p = 0.02, eta squared = 0.051. Furthermore, confidence score was significantly associated with duration of diagnosis, F (4, 215) = 3.23, p = 0.013, eta squared = 0.057. Confidence was significantly higher in the >15 years (M=7.95, SD=1.30) when compared to <1 year (M=6.50, SD=1.68) only. Furthermore, level of educational qualification significantly influenced confidence level, F (4, 215) = 6.77, p <0.001, eta squared = 0.11. Participants with postgraduate degree had significantly higher confidence (M = 7.71, SD =1.55) in comparison to those with high school (M = 6.42, SD = 1.98) and technical school (M = 6.47, SD = 2.11) certificates. No significant difference was observed for those with bachelor's degree (M = 7.48, SD =1.39).

Multiple regression analysis identified the simultaneous contributions of time since diagnosis, type of diabetes, educational qualification and receiving DSME within 12 months prior to the study on participants' level of skills. These variables predicted 22% of the variation in level of skills F (2, 216) =14.815, p<0.001, R^2 =.218. All variables, except receiving DSME, were statistical significant at p <.05.

	Skills			Self-Efficacy			
Variables	mean ± SD	F or t statistics, p-value	Effect size	mean ± SD	F or t statistics, p-value	Effect size	
Type of Diabetes		t(215)= 17.41,p<0.001	0.123		t(215)=5.46, p=0.02	0.051	
Type 1	7.95 ± 1.35			7.66 ± 1.47			
Type 2	6.66 ± 2.12			6.87 ± 1.93			
Duration of diagnosis (years)		<i>F</i> (4) = 5.59, <i>p</i> <0.001	0.095		<i>F</i> (4) = 3.23, <i>p</i> =0.013	0.057	
<1	6.28 ± 1.82			6.50 ± 1.68			
1-5	6.98 ± 2.08			7.12 ± 1.86			
6-10	6.97 ± 2.14			6.96 ± 2.15			
10-15	7.00 ± 1.58			7.20 ± 1.27			
>15	8.28 ± 1.22			7.95 ± 1.30			
Received DSME in the previous 12 months		t(215)= 1.89, <i>p</i> =0.045	0.018		t(215)=1.48, p=0.141	0.01	
Yes	7.35 ± 1.77			7.31 ± 1.66			
No	6.79 ± 2.23			6.93 ± 2.03			
Educational status		<i>F</i> (4) = 7.87, <i>p</i> <0.001	0.132		F(4) = 6.77, p < 0.001	0.113	
High School	6.13 ± 2.21			6.42 ± 1.98			
Technical College	6.53 ± 2.25			6.47 ± 2.11			
Bachelor Degree	$7.47 \hspace{0.1in} \pm \hspace{0.1in} 1.65 \hspace{0.1in}$			7.48 ± 1.39			
PG degree ^b	$7.76\ \pm 1.53$			7.71 ± 1.55			
Others ^c	$8.40\ \pm 1.12$			8.18 ± 1.46			
Gender		t(215)=-1.18,p=0.238	0.006		t(215)=-0.43, <i>p</i> =0.665	0.001	
Male	$6.97 \hspace{0.1in} \pm 1.93 \hspace{0.1in}$			7.11 ± 1.76			
Female	7.28 ± 1.99			7.22 ± 1.85			

Table 3.2: Summary of t-test or ANOVA and Post Hoc results on mean scores by participants' characteristics

Geographic location		<i>F</i> (3) = 3.14, <i>p</i> =0.124	0.005		<i>F</i> (3) = 3.22, <i>p</i> =0.145	0.004
Australia	7.11 ± 2.01			6.96 ± 1.87		
Europe	6.73 ± 2.40			6.91 ± 1.94		
America	$6.99 \hspace{0.1 cm} \pm \hspace{0.1 cm} 2.11 \hspace{0.1 cm}$			6.77 ± 2.10		
Asia	$7.16\ \pm 2.10$			6.76 ± 2.13		
Age		F (4) = 1.46, <i>p</i> =0.215	0.027		F (4) = 1.48, <i>p</i> =0.211	0.027
18-29	$6.95 \hspace{0.1cm} \pm \hspace{0.1cm} 1.78$			6.88 ± 1.80		
30-39	7.60 ± 1.52			7.52 ± 1.58		
40-49	7.06 ± 2.16			7.01 ± 2.02		
50-59	$6.74 \hspace{0.1cm} \pm \hspace{0.1cm} 1.98 \hspace{0.1cm}$			6.81 ± 1.64		
60-69	7.12 ± 2.42			7.39 ± 2.05		

^aDMSE: Diabetes Self-Management Education

^bPost Graduate

°Others: Professional qualifications, graduate diploma

3.4.3 Other Enablers and Barriers to Diabetes Self-Management Other enablers of self-management

Two major themes were identified as factors, which could facilitate diabetes self-management. These were patients' determination to prevent the development of complications and the use of health technological devices or software.

Theme 1 - Determination to prevent diabetes complication: The decision to regularly engage in self-management was fostered by participants' resolution to prevent the development of diabetes complications. Participants ensured that they engaged in the necessary lifestyle behavioural activities due to their determination to maintain better quality of life and thereby avoid what was observed in their peers who had already developed some form of diabetes complications:

"I see a lot of other people who already have diabetes talking about their diabetes on social media. Looking at others who are worse off than me and the problems they struggle with, I guess is keeping me in check saying, hell no, I'm not going down that path''. [P6, T2D]

Furthermore, the determination to prevent diabetes complication was expressed by refusal to purchase certain foods which participants believed could increase the risk of progressing type 2 diabetes management into requiring the use of insulin injection:

"It's just the fact that I don't want to get to the stage of having injections many times a day.... I have to remind myself of that always. I'm quite happy to walk past some chocolate....knowing fully well that whilst I might enjoy a ** (name of a chocolate brand), then I get an injection at the end, which I don't want, which mean I will leave (name of a chocolate brand) alone". [P3, T2D] Respondents acknowledged that having good knowledge and problem solving skills in diabetes has proven useful to aid their self-management. Awareness of how foods impact their health was reported as highly essential:

"I think having knowledge of the foods and the type of foods and diet and portion sizes are very important. Also, I found understanding what hypo or hyper, and understanding how my body reacts and how I can resolve that has been very useful in managing my diabetes." [P15, T1D]

Theme 2 - Use of health technological software and devices: Participants mentioned the use of mobile technological devices specifically, smart phone application (apps), insulin pump and continuous glucose monitors (CGM) as supporting tools which have enhanced their DSM.

<u>2. 1 Apps</u>: Some of the participants use smart phone apps to record their blood glucose data. They noted that having access to such previously stored data on their phones gave them insight into the best self-management strategy, which had assisted in adequate glycemic control:

"I have been diagnosed for a long time and back in the days I used to write it on a note book. But in these days, I record it using a smart phone app, which allows me to search. So it allows me to access the data quickly and make a sort of best guess for now based on what happened in the past. If it's not working, as it has done recently, I can go back to strategies that I might have been using years ago, that seems to work then ". [P14, T1D]

Reminder feature in apps were found useful to give alert for recurring tasks such as taking medications thereby improving medication-taking behaviour especially during busy schedules:

"I'm only kind of new to this (newly diagnosed), so I am actually looking for ways to remind myself of the tablets (medication) am meant to be taking. When I get really busy I forgot..so my app pings at me a certain time of the day...just to kind of prompt me" [P3, T2D] Also, motivations and encouragements were received through the use of app especially in the event of unstable blood glucose control. Participants stated that whenever their blood glucose level fluctuated and differed from the prescribed limits despite all efforts to stabilise it, looking at good data previously stored in apps provided an assurance that their blood glucose levels will not always be unstable:

"Sometimes it is simple as realizing it's not all terrible. Being able to flip back on my smart phone. If you've had a rough four or five days, it can feel like it's a long time since you've seen numbers that felt like relatively stable or in range. You can get disheartened but if you can just check back and see, actually no, it's fine because two weeks ago it was all right, so I'll be able to get back to that again. So having access to that sort of information storage allows me to be a little bit more relaxed when inevitable things start to wobble and go adrift again''. [P14, T1D]

<u>2. 2 Insulin pump and CGM</u>: Participants with type 1 diabetes reported the use of insulin pump or continuous glucose monitor (CGM) as external aids which made it easier for them to manage and effectively monitor their health. In this regard, one participant stated that:

"With the insulin pump, I find it easier to manage. Also, I've got the CGM and I can see what my sugar is on the screen all the time....you know that changed my life". [P1, T1D]

Participants also indicated that use of insulin pump provided additional support and relief from pains experienced while using needles:

"I'm quite a thin bloke...... and have no body fat so inserting needles really hurt. My insulin pump definitely helps. So the best way I've managed my diabetes is through the insulin pump". [P12, T1D]

3.4.4 Barriers to Self-Management

In spite of the factors that foster effective self-management of diabetes, the key themes that emerged from the interview indicated that people with diabetes encountered diverse challenges in performing their self-management due to the: i) dynamic and chronic nature of diabetes; ii) financial constraints iii) work and environment related factors; and (iv) unrealistic expectations

Theme 1 - Dynamic and chronic nature of diabetes: The most common complaint reported by participants was the dynamic and chronic nature of diabetes and how these attributes make DSM require multiple needs. Participants felt there were many reasons including environmental conditions, which may demand an adjustment in their self-management even within short time periods. They believed the constant requirement to modify needs of the condition denoted certain things they were not doing right in their self-management and they always had to put in great effort to meet up with their health requirements:

"Because I live with type 1 diabetes I have to do a complete insulin replacement, which involves balancing for activity, ambient temperature, stress levels, insulin sensitivity of my body. It could be so much easier if you could just work out what your insulin to carb sensitivity portion is, work out how to behave around exercise, work out correction factors and that would be all. But no, my experience is that that's it for a week and then your basal requirement would have changed. Then the weather get warmer, you may need to re-evaluate your insulin sensitivity and carb ratio. So it's just- you are never getting it right and you're just always constantly trying to play catch up''. [P14 T1D]

Likewise, the effects of self-management on diabetes outcome was referred to as a system which could not be automatically controlled. Participants described how similar behavioural activity such as eating the same diet over time could impact their health differently.

94

"It is a dynamic disease. I mean what works today doesn't work tomorrow. You can eat something today and you can be okay, eat something tomorrow and it can be completely different. So you can never just put it on a cruise control and away you go". [P2, T1D]

The weariness about the never-ending need for self-management because diabetes is a lifetime disease was expressed:

"The biggest thing that fazes me is just the fact that it's something that you have to do 24 hours a day, seven days a week and nothing ever going to change that". [P4, T1D]

Participants were sometimes unwilling to undertake their self-management because they felt it is not a permanent cure for the disease, diabetes is chronic, so what is the point?

"...Probably my mind frame, in just getting yourself down to the fact that it's never going to...I'm always going to have it. So you sort of question what's the point (of management)? It's hard to comprehend". [P11, T1D]

The presence of other diabetes related complications or health problems such as neuropathy and depression in some participants limited their ability to actively engage in behavioural activities especially physical exercise or healthy eating:

"Physical exercise is difficult... Yeah, I have peripheral neuropathy of the leg, a collapse in the foot and yeah, problems with the other foot". [P10, T2D]

"Nutrition is something that is hard to keep on top of. I suffer from a major depressive disorder, so I have a lot more trouble following my optimum diet". [P7, T1D]

Theme 2 - Financial burden: The difficulty in meeting the financial cost for some diabetes medical tests and other treatment requirements was also identified as a barrier. Participants voiced out the financial burden they experienced by citing the need to pay for some clinical tests and diabetes supplies which are not covered by their health insurance such as the

glycosylated hemoglobin (HbA1c) test and continuous glucose monitor. They expressed the desire to receive more support from the government:

"I manage my diabetes fairly closely and I pay for HbA1c, you know ... the financial cost is quite large. In Australia, our health system's pretty good but you still have to pay for a lot of equipment which the government doesn't seem to agree necessarily. Continuous Glucose Monitor should be government funded for over 21s for Christ sake". [P2, T1D]

Another participant based in the United Kingdom (UK) stated:

"I don't have unimpeded access to Continuous Glucose Monitor (CGM). I mean..the situation of health care in UK is that it's (CGM) not often funded by National Health Service (NHS) apart from people that are in quite profound need. I don't get that assistance... So that's a challenge and access issue". [P14, T1D]

Theme 3: Work and environment-related conditions

<u>3.1 Occupation</u>: Job requirements especially those involving a lot of travelling serves as deterrent to maintaining a healthy diet. Participants stated that the inability to get healthy choices of foods in most restaurants or public places when unavoidably required to eat out due to travelling long distances to fulfill their job requirements:

"My work requires a lot of travelling. If you are actually going to eat something that is actually not good and could put you in the circumstance where you know... Like I had a 16 hour travelling the other day and everywhere I turned, I couldn't touch any of it. I had some but I had to acknowledge that it was not what I really needed to eat" [P3, T2D]

Work related stress was also reported as a hindrance to attaining optimal blood sugar levels:

"With me personally, it's stress. I'm an electrician, and I'm full time employed, so stress gets me. When I get stressed, my blood sugar level goes downhill" [P13, T1D] <u>3.2 Weather condition</u>: Participants find it difficult to engage in physical exercise in hot weather conditions:

"...Exercise is something I have trouble getting around to doing. Like during the summer, the heat hits me big time. So I'm loving the cooler weather we're starting to have because I can start to work a bit more, but during the heat, I cannot do it". [P2, T1D]

Theme 4- Unrealistic demands: Unrealistic expectations and advice about self-management from family or friends especially those not diagnosed with diabetes could be a hindrance to effective care. Participants' found such wrong advice irritating as evident in the following comment:

"You know I don't think a lot of non-diabetic actually get to know how much it can take to actually manage a high or a low (Blood sugar) potentially. You know, you get comments from people that you're low and they know you are diabetic saying, oh, should you be eating that? Well, I'm going to say this nicely, you want me to die now or not or to go into coma? Because I need to eat this. They go oh, you didn't need to say it like that. You go well, stop asking a stupid question that you don't know anything about". [P4, T1D]

Additionally, discrepancy between patients and their health professionals' (HP) perception of care could be a barrier to self-management. Participants felt that some recommendations from HPs were contrary to their opinions on what their DSM should entail:

"My doctor doesn't feel I need to be using a glucose meter to monitor my sugar levels and the diabetes educator doesn't think I need to be on any sort of diet, even though I've had increases in diabetes medications". [P10, T2D]

3.5 DISCUSSION

To the best of our knowledge, this is the first mixed-methods study that has investigated the enablers and barriers to general self-management among a multinational audience of people who have type 1 or type 2 diabetes. Most importantly, our findings emphasise the consequential impact of currency of exposure to DSME (within the previous 12 months), duration of diagnosis, level of educational qualification and use of technological devices on self-management skills and self-efficacy, regardless of geographical location or ethnicity. This implies that provision of ongoing self-management education/support through the use of mobile phones may help address the various difficulties (including time/financial constraint, diabetes distress, and limited access to care providers) encountered by patients and foster adherence to recommended self-management activities, which are necessary to prevent the risk of developing diabetes complications. Furthermore, this study presents in-depth understanding of the experiences of diabetes patients and provides useful insights to health professionals and researchers on how to improve the frequency and quality of self-management support provided to diabetes patients to achieve better health outcomes.

Skills and self-efficacy for diabetes self-management

The overall skills score was found to be high and many participants reported good level of ability for self-management. This is specifically in the area of accurate monitoring to assess the impact of diet, medication or physical activities on blood glucose levels. Similar findings were observed in a previous study (Persell et al., 2004). Accurate monitoring of blood glucose in relation to foods consumed and physical activities are important because they predict good outcomes in diabetes management (Shrivastava, Shrivastava & Ramasay, 2013).

Even though, the participants in this study scored high in their ability to monitor blood glucose, their capacity to interpret their blood glucose patterns over time was only moderate. Although, self-monitoring of blood glucose is important to assess glycemic pattern, accurate interpretation of these patterns is highly important to ensure effective management of glycaemia related problems encountered in diabetes management (Glasgow et al., 2007). More emphasis should be laid on glucose pattern management during DSM educational sessions in

order to expatiate patients' skills on effective monitoring and interpretation of blood glucose data and the resulting health implications.

Participants in this study possessed lower skills related to planning for physical exercise in order to avoid hypoglycemia and adjusting medication to reach targeted blood glucose levels. This result corroborates previous findings (Basu et al., 2014). Ability to manage and make appropriate adjustment to multiple regimens often determine success with other core areas of DSM and glycemic control (Glasgow et al., 2007). For instance, studies have reported that due to the fear of hypoglycemia, patients have resorted to unhealthy behaviours (such as reducing or eliminating medication dose, inappropriate food choices and /or avoidance of physical activities) that increase glucose levels (Perlmuter et al., 2008). Diabetes patients have an increased risk of developing hypoglycaemia particularly when treated with insulin or insulin secretagogues (Permulter et al., 2008). Hence, they should be provided with regular refresher courses and continuous training on blood glucose levels awareness and strategies to balance exercise, which could promote glycemic control and adherence to self-management.

Healthy coping strategies to identify and manage the impact of stress on diabetes management may be a difficult aspect of diabetes care because the participants in this study scored lowest in this area for both the skills and self-efficacy domains. All forms of stress either physical or mental, negatively impact blood glucose levels in those with diabetes (Pouwer, Kupper & Adriaanse, 2010) and it is a potential obstacle to attaining effective self-management and optimal health outcomes (McEwen, 1998). Patients' understanding of dimensions of diabetes related stress is a clinically important factor and forms of stress that are potentially modifiable should be prioritised to guide clinical and educational interventions. This can include regular educational information on the impact of stress on health of diabetes patients and suggestions to reduce it. Contrary to the findings of a previous study (Toljamo & Hentinen, 2001) that reported people with type 1 diabetes as having poorer self-management; our study participants who had type 1 diabetes scored higher than those with type 2 diabetes in skills and self-efficacy to care for their diabetes. Additionally, there was a significant positive relationship between the duration of diabetes and both skills and confidence for self-management. Patients with type 1 diabetes are typically diagnosed at an early age that may correspond to longer duration of diabetes. This pattern might have afforded them prolonged and regular exposure to health education, which is a significant predictor of successful DSM (Haas et al., 2012).

Overall, the strong correlation between the level of skills and self-efficacy found in this study strengthens the body of evidence supporting this link (Aronson et al., 2018). This pattern may be related to high level of education among most of the study respondents as also observed in a previous study (Schillinger et al., 2002). Patients who possess higher skills usually have higher perceived level of efficacy and are most likely to actually engage in their self-management (Persell et al., 2004; Aronson et al., 2018). Building patients' skills and confidence in their ability to self-manage diabetes is therefore imperative. Regular encouragement, which could either be provided verbally or through other means of contact (e.g text messages through phones or emails) could be beneficial to patients (Yoon & Kim, 2008). While for those with limited educational backgrounds, the use of clear and simple communication styles when providing diabetes education to them will be essential to foster their skills and confidence (Schillinger et al., 2002).

Other enablers of self-management

Based on the results of the interviews, the most commonly perceived factor that fostered regular self-management was the will to prevent the development of diabetes complications. This result corroborates previous findings (Maillet et al., 1996; Koch, Kralik & Taylor, 2000) and indicates that the participants in this study took responsibility for their choices and respective

consequences. Discipline and proactive approaches to self-management are essential to reducing or preventing the development of diabetes complications. Regular reinforcement of education and motivation of patients could provide in-depth information about the disease and foster the will to mitigate its' clinical course.

Furthermore, our study findings confirm those of other studies that the use of mobile technologies such as smartphone applications (Hunt, 2015), insulin pump (Ghazanfar et al., 2016) and continuous glucose monitor (Burge et al., 2008) could enhance DSM in patients. Technology interventions have positive impact on diabetes outcomes such as adherence to self-management activities, glycosylated hemoglobin and diabetes self-efficacy (Hunt, 2005). Therefore, health professionals could recommend the use of mobile health technologies to patients who are capable of using them as they benefit from them.

Barriers to self-management

The lack of enthusiasm towards regular self-management due to the chronic and dynamic nature of diabetes is not entirely unexpected. This phenomenon could be referred to as diabetes distress which is the emotional stress resulting from living with diabetes and the "burden of relentless management" (Pandit et al., 2014). High diabetes distress results in sub-optimal diabetes management and compromised quality of life (Peyrot et al., 2005; Balfe et al., 2013). Diabetes distress is common among patients and impacts on their self-management and health outcomes. Therefore, the importance of providing appropriate regular support to all patients in this regard cannot be overemphasised. Health professionals could ask patients at every consultation about how they are coping with diabetes, encourage them to express their diabetes-related issues causing distress and offer encouragement and suggestions on ways to deal with it on a daily basis.

For many of the respondents in this study, the need to meet up with job requirements especially frequent travelling makes adherence to healthy eating difficult. Additionally, work related

stress impacts greatly on their blood glucose levels. These findings echo the results by Chao et al. (2005) Recommendations to patients to engage in creative planning and social support are strategies to help address this barrier. Social support from families are essential. Families should be encouraged to attend educational training sessions with patients so as to offer appropriate support which can assist patients to make healthy food choices and decisions regarding their diabetes management (Maillet et al., 1996; Chlebowy, Hood & Lajoie, 2010).

Furthermore, financial burden associated with diabetes could be a hindrance to selfmanagement especially those associated with out-of-pocket expenditure for medical needs. Campbell et al., (2017) observed that the predominant area of management where patients experience financial burdens are medications, diabetes supply and healthy food. People with diabetes require regular self-management and clinical monitoring to prevent the development of complications and foster optimal health outcomes; hence the associated financial demand. Health care providers could inform patients about resources available to them to buffer financial constraints that limit adherence to treatment plans. Such resources may include referring patients to specific social programs or compassionate relief programs to support financial burdens and enable easier access to necessary services.

Differences in patients' and health care professionals' (HCP) views of what constitutes the best approach to care was also identified as a barrier to DSM. This may be due to gaps in the way treatment recommendations were communicated to patients. Often times, HCPs' view of good care are based on adhering to stipulated biomedical care model, structured communication and central decision making (Van Keer et al., 2015), whereas patients perceived quality health care is how the scientific knowledge of HCPs' aligns with their own experiential knowledge and personal preferences (Pomey et al., 2015). Therefore, patients are always seeking exhaustive information about their diagnosis and treatment (Pomey et al., 2015). There is responsibility on the part of HCP's to advice and educate their patients on different treatment options and the reasons they are placed on a particular option and not the other. This patient centered-approach will empower patients and foster their health outcomes.

Integration of findings and recommendations for future interventions

The survey results show that many patients have limited capacity for healthy coping strategies to identify and manage the impact of diabetes related stress. This finding was confirmed in the interviews where diabetes distress was reported as a major barrier to self-management. Given that stress is a potential contributor to chronic elevated blood glucose levels, it is essential for health care professionals to assist patients with identifying approaches to reducing diabetes distress. Additionally, increased access to providers through expanded clinic hours could be a means of easing the burden of diabetes diagnosis (Gazmararian, Ziemer & Barnes, 2009).

The quantitative data also showed that higher educational level was the strongest predictor of better self-management skills in patients and this was affirmed by the highly skilled interviewees who identified the use of technological devices as an enabler to their DSM. This corroborates that higher educational level is a good predictor of eHealth usage (Kontos et al., 2014). In addition, in accordance with previous literature (Lim et al., 2011), good overall self-efficacy level observed in the survey might have influenced the positive report on the usefulness of technology in diabetes management. Therefore, given that use of health technologies provides both short and long term health improvements in diabetes patients (Kaufman et al., 2016), active use should be encouraged where necessary especially among patients who are educated and have the ability to engage with them. Furthermore, it is important to device avenues to improve patients' self-efficacy in their ability to manage the disease as this could increase their likelihood of engaging with technology for their self-management (Dou et al., 2017).

The interviews revealed that determination to prevent the development of complications is one of the major enablers to DSM. This might explain the overall high score in skills and self-

efficacy observed in the survey. Therefore, we suggest that educators could focus on improving patients' skills and self-efficacy for DSM thereby raising patients' awareness of the negative effect of diabetes. This approach could in turn stimulate the patients' determination to engage in their self-management and thereby reduce the risk of developing complications.

A unique perspective from the qualitative results revealed that patients and HCPs have divergent views/opinions about what should constitute patient care. It is therefore, imperative that HCPs ensure that patients understand the reasons for the recommended treatments and engage them in shared decision making which is essential for patients' satisfaction and engagement in self-management practices (Heisler et al., 2002).

Lastly, it has been advocated that people with diabetes should receive self-management education and support in an ongoing and consistent manner (Powers et al., 2017), but the reality of facilitating face-to-face diabetes education between patients and HCPs on an ongoing basis is low due to limited human and organisational resources. Health behavioural treatments and therapies such as DSMES could be provided to patients on an on-going daily basis outside the clinical setting through the use of Ecological momentary interventions such as mobile technologies (Heron & Symth, 2010). Apart from the fact that apps were opined by patients to enable self-management in this study, the World Health Organisation (WHO) also confirmed that the use of mobile technologies (such as apps) can support the attainment of health outcomes which could transform health service delivery globally (World Health Organisation, 2011b). Considering that apps are cost effective avenues for providing ongoing delivery of care to patients outside the clinical environment (Iribarren et al., 2017), DSM educational messages could be developed and integrated into apps for patients. Such information should be targeted at improving patients' skills and self-efficacy capacity for effective self-management.

Strengths and limitations

The strength of this work is that it provides a multinational picture of skills and confidence for self-management in people with type 1 or type 2 diabetes. Such an elaborate and international approach to assessing the capacity and confidence levels for self-management is scanty in the literature. In addition, the data identified a number of factors serving as enablers and barriers to DSM emanated from patients' perspectives and their lived experiences. Therefore, the results are tenable to provide immense insights into improved strategies for supporting patients in their self-management.

There are some limitations to this study. Firstly, the reliability and validity of the quantitative tool used have not been previously demonstrated at multinational/multicultural levels, and this may limit the interpretation of our findings. Although, in a previous study (Aronson et al., 2018), the construct validity of the scale was tested among type 1 and type 2 diabetes patients who were from different ethnic backgrounds (Asians, Caribbeans, Caucasians etc.) but living in the same regional location. The study reported that the scale was not influenced by ethnicity. Secondly, the small sample size/groups for the survey, which mainly comprised of participants from three continents may limit the generalisation of our findings to other settings. Thirdly, the quantitative data was self-reported and therefore susceptible to bias, which may not reflect participants' actual skills and confidence levels for self-management. Hence, under or over reporting could result in inaccurate identification of common gaps in skills and confidence requiring intervention. Nevertheless, self-report can be made more reliable when questions are asked in non-judgmental manner as obtains in the SCPI tool used in this study. Lastly, the small number of interview participants is also acknowledged and the interview sessions were brief because additional compensation was not offered to interviewees. Short interview duration was utilised to foster increased participant numbers as long interview may not be justifiable for participants' time involvement in the study. Nevertheless, literature has shown that the

anonymity of telephone interview reduces the interviewer bias, which makes the interviewing setting more calming and forthcoming, thus fostering a more accurate and truthful data collection (Musselwhite et al., 2007).

3.6 CONCLUSION

This study identified the common gaps in the skills and self-efficacy of people with type 1 or type 2 diabetes mellitus as well as other perceived enablers of, and barriers to, self-management in this population. Diabetes health care stakeholders may consider strategies for regular educational reinforcement in patients in order to foster healthy coping with diabetes stress, exercise planning to avoid hypoglycemia, interpreting blood glucose patterns and adjusting medications or foods to reach the targeted blood glucose levels. Furthermore, designing of interventions that capitalise on how to improve patients' desire to reduce the progression of diabetes and the use of relevant technological devices could enhance DSM. Improved approaches to address diabetes distress, financial burden, discrepancy between patients and their health professionals' perception of care as well as work and environment related factors are essential to foster improved self-management in patients. Finally, attention should be paid to type of diabetes, level of education and duration of diagnosis when counselling patients on diabetes self-management. Consideration of these areas of educational reinforcement and interventions could enhance self-management in patients and consequently improve their health outcomes.

This chapter addressed the second aim of the thesis and provided empirical evidence of skills and self-efficacy as mediating factors for diabetes self-management. The findings from this study guided the development of the DSM interventional app that is presented in Chapter 5.

Chapter FOUR: Users' Preferences and Design Recommendations to Promote Engagement with Mobile Apps for Diabetes Self-Management: Multi-national Perspectives

4.1 ABSTRACT

Mobile phone applications (apps) offer motivation and support for self-management of diabetes mellitus (diabetes), but their use is limited by high attrition due to insufficient consideration of end-users perspectives and usability requirements. This study aimed to examine app usage and feature preferences among people with diabetes, and explore their recommendations for future inclusions to foster engagement with diabetes apps. The study was conducted internationally on adults with type 1 or type 2 diabetes using online questionnaire (quantitative) to investigate usage and preferences for app features that support diabetes self-management (DSM) and semi structured telephone interview (qualitative) to explore suggestions on fostering engagement and specific educational information for inclusion into diabetes apps. Survey and interview data were analysed using descriptive / inferential statistics and inductive thematic analysis respectively. A total of 217 respondents with type 1 diabetes (38.25%) or type 2 diabetes (61.8%), from 4 continents (Australia, Europe, Asia and America) participated in the survey. About half of the respondents (48%) use apps, mainly with features for tracking blood glucose (56.6%), blood pressure (51.9%) and food calories (48.1%). Preferred features in future apps include nutrient values of foods (56.7%), blood glucose (54.8%), physical exercise tracker (47%), health data analytics (42.9%) and education on DSM (40.6%). Irrespective of the type of diabetes, participants proposed future apps that are user friendly, support healthy eating, provide actionable reminders and consolidate data across peripheral health devices. Participants with type 1 diabetes recommended customised features with news update on developments in the field of diabetes. Nominated specific educational topics included tips on

problem solving, use of insulin pump therapy and basic guideline for the management of diabetes. The study has highlighted patients' perspectives on essential components for inclusion in diabetes apps to promote engagement and foster better health outcomes.

4.2 BACKGROUND

Mobile phone applications (apps), are extensively used to provide support for diabetes selfmanagement (DSM) (Chomutare et al., 2011, Arnhold, Quade & Kirch, 2014). These apps include features for tracking blood glucose, calories in diet, body weight, as well as reminders for medication intake or health appointments (Rao et al., 2010; Chomutare et al., 2011; Tran, Tran & White, 2012). There is strong evidence suggesting that the use of apps encourages adherence to management therapy, improves glycaemic control, which subsequently prevents or delays the onset of diabetes complications and enhances patients' quality of life (Liang et al., 2011; Goyal & Cafazzo, 2013; Scheibe et al., 2015). However, despite the proven effectiveness and investments into the technological processes of app development, their use as an intervention for diabetes mellitus (diabetes) management, have been limited by high attrition rate (Istepanian et al., 2009; Quinn et al., 2011; Holmen et al., 2014), evidenced by reduced user engagement over time. The low level of adoption and use of apps has been attributed to insufficient consideration of end users' preferences (Chomutare et al., 2011; Holtz & Lauckner, 2012) and the factors for engagement. (Arsand & Demiris, 2008).

Trends in users' preferences for features and engagement with apps to support diabetes management

There are limited studies that have explored patients' use, feature preferences and recommendations that could improve engagement with diabetes apps. Recent surveys found that although many people with diabetes own a smartphone, only a few of them used apps to manage their diabetes (Kayyali et al., 2017; Dobson et al., 2017). The major reason for non-

use was that patients were unaware of the existence of apps that could support their care (Dobson et al., 2017). Additionally, there are diverse patients' views about the essential features in apps to support diabetes management. A survey found that features for data recording, social coaching, reminder and remote collaboration with health care professionals were appealing to people with diabetes (Kayyali et al., 2017). Other studies have reported carbohydrate counter, blood glucose and physical activity tracking as the most commonly preferred app features by diabetes patients (Lithgow, Edwards & Rabi, 2017; Boyle et al., 2017). However, most apps were unable to meet up with patients' needs because diabetes education which is a crucial and evidence-based requirement for diabetes management is often lacking in the currently available apps (Chomutare et al., 2011; Boyle et al., 2017).

Moreover, to be an effective self-management support tool, app must continuously capture the attention of users and stimulate users' interest to actively engage with it. Engagement indicates the degree of interaction a user has with the technology within a given time span or the overall length of time from the onset use of a technology to when the user totally lost interest in the usage (Bickmore, Schulman & Yin, 2010). Engagement with health technologies including apps is a dynamic process comprising of different stages, namely: point of engagement, period of engagement, disengagement and reengagement. Therefore, users' engagement is multifaceted in nature and may change within a short or long periods of time (O'Brien & Toms, 2008). User's engagement can either be measured during a short session or long-term use of a technology. It is important to note that the ultimate goal of mobile apps usage for patients with chronic diseases is to foster their ongoing and regular participation in their self-management activities. Where participation in those activities may be reflected in the frequency of tracking or monitoring of the activities using an app hence denoting the extent of their engagement with the app. Nevertheless, patients may participate in management activities and not track with an app (non-usage) (Kayyali et al., 2017) even when apps are present on their molecular denotion.

which may be due to several reasons. Since health technologies are voluntary use systems, the extent of engagement with them is determined by the users' perceived quality of experience, ongoing benefit of usage (Lalmas, O'Brien & Yom-Tov, 2014), consideration of viable alternatives to using the technology and decrease in perceived costs (Bickmore, Schulman & Yin, 2010). Furthermore, Studies have reported that mobile health apps which are able to adequately meet patients' self-management needs and have clear evidence of data privacy will motivate users engagement with the technology (Woldaregay et al., 2018; Torous et al., 2018).

Low engagement is not unique to diabetes apps alone, but rather to all computerised behavioural therapeutic interventions that support the management of chronic diseases or health promotion (Dennison et al., 2013; Bender et al., 2014; Conway et al., 2016; Taki et al., 2017). Engagement with mobile apps is negatively affected by factors such as lack of motivation or commitment to change health behaviours (Dennison et al., 2013) and knowledge in managing the targeted behaviour. Once knowledge is attained, it is likely that participants' interest in app use will reduce (Guertler et al., 2015; Taki et al., 2017). Moreover, sub-optimal usability has been stated as a reason for low engagement with apps (Mallenius, Rossi & Tuunaunen, 2007; Demidowich et al., 2012). Optimal usability in apps could be achieved through simplicity, reduced time consumption and customised users' experience (Juarascio et al., 2015). To date, there is little or no clear evidence as to why engagement with diabetes apps remains low. Tatara et al., (2013), in their study on long-term engagement with a mobile self-management system for people with type 2 diabetes reported perceived sense of mastery over diabetes and experiences of problems with the app as factors for declining motivation to continuous use of app.

The few studies described above have shown that diabetes patients have different needs and requests for health care technology development (Holmen et al., 2017) and there is limited understanding of the factors which could foster engagement with apps to support diabetes.

Moreover, these studies have been limited to single countries and current app users alone. A critical step to knowing what appeals to end users and maximising app engagement is shared decision-making, whereby app developers involve targeted end users in the process of developing the content and features of apps (Arsand & Demiris, 2008; McCurdie et al., 2012). Hence, it is important to build upon previous research by examining the perception of a diverse range of people with diabetes; that is both current users and non-users, residing in diverse locations, about the usability and functionality of diabetes apps to support their health-care and factors for usage over time.

Study Aims

Thus, the aim of this study was to involve individuals with type 1 and type 2 diabetes in guiding the development of future mobile apps for DSM. The specific objectives were: (1) to assess the use and preferences for potential features in apps, which could support the health management needs of individuals with type 1 or type diabetes. (2) Seek recommendations on components and motivators that could foster long term use of apps (3) Assess specific educational topics desired in apps to support DSM. We hypothesised that diverse multinational respondents comprising of adults with type 1 or 2 diabetes (both users and non-users of apps) would recommend the necessary design components and strategies to maximise engagement with apps to support the self-management of diabetes. Getting this insight would provide an effective approach to the design and development of an evidence-based and highly functional app that best meets the identified needs of people with diabetes.

4.3 METHODS

4.3.1 Study Design

The study employed a mixed methods design, combining quantitative and qualitative data to explore patients' perceptions and the results were integrated in the data interpretation phase (Creswell et al., 2003). Information was sought globally from adults aged \geq 18 years who have type 1 or type 2 diabetes. Users and non-users of apps were included in the study to obtain balanced and un-biased findings.

Quantitative approach

Data were collected through an online survey conducted between November 2017 and June 2018. The study was advertised using multiple outlets, including Townsville Bulletin digital newspaper, Diabetes Australia and Diabetes UK websites as well as various diabetes support groups on Facebook and Twitter. Through these advertisements, a link was provided where potential participants were directed to the Survey Monkey page containing information about the study. Consenting participants could subsequently click on the survey link and submit their responses. As an incentive to encourage participation, a chance to enter a draw to win one of 6 US\$50 e-gifts card was offered to respondents. This recruitment procedure yielded an inadequate response rate particularly from Asia and Europe, hence the need for another approach. More targeted respondents were recruited from Asia and Europe, and were offered an additional incentive (\$US 5) for their participation.

Instrument

Through a systematic review, an overview of the frequent features currently available in mobile apps to support DSM as well as the gaps in literature were obtained (Adu et al., 2018a). This review guided the development of the study questionnaire, which comprised two sections (see **Appendix 4.1**). The first section inquired about basic demographic and health characteristics of the respondents. The second section comprised 3 questions aimed at elucidating preferences and perceived importance of various features in apps that could be utilised by patients with type 1 and type 2 diabetes, regardless of the use of insulin as part of the management therapy. The first question that was on a dichotomous scale (Yes or No) inquired about the current use

of apps to manage diabetes. Those who indicated 'No' were requested to specify reasons for non-use while those who answered in the affirmative were asked to specify the features available in such apps. The second question assessed participants' preferences on a list of app features and allowed multiple responses. The third question evaluated the perceived usefulness of various app features to support diabetes management. These features were rated on a 10point scale with "1=least useful and 10=most useful", so that higher rating indicates greater perceived usefulness. Descriptors for intermediate points within the range were excluded to avoid clustering of scores around a preferred descriptor. App was defined as any program downloadable to a smart phone, which is used to support any aspect of diabetes management to foster improved health outcomes for the user. It could either be lifestyle oriented such as those for tracking diet and exercise or patient oriented for blood glucose or blood pressure monitoring. The instrument was reviewed among a diverse team of researchers with expertise in survey development in order to ascertain its readability and ease of understanding, and was revised based on feedback obtained. The instrument readability was further accessed with the Flesch Kincaid readability test (Flesch, 1948) producing a reading score of 61.8, indicating the instrument is well comprehensible, consistent with standard English that is easily understood by an eighth or ninth grade level student or one who has attained 13 years of age or above (Flesch, 1948). The instrument was administered in English Language.

Qualitative approach

Semi-structured telephone interviews were conducted with a subset of participants who responded to the questionnaire. To recruit participants for the interviews, a final question was added to the questionnaire asking for interest in a telephone interview aimed at further exploration of opinions on the use of apps to support DSM. Interest was indicated by providing a phone number and suitable contact time. No additional incentive was offered to interview participants. A standardised text message was sent to all phone numbers provided to confirm

interview schedules. All interviews were conducted in June, 2018 by a resource person who was independent of the research team (designated as the interviewer, who is a male researcher with prior experiences in qualitative research and interview facilitator). The interviewer (AD) was trained on the aims of the study and the interview guide by MDA. Additionally, a pilot session of the interview was conducted between AD and MDA. The interviewer had no prior relationship with the participants and the interview sessions was conducted in a secure, private room at the James Cook University, Australia. Prior to the commencement of each interview session, participants were asked if they were in a location that was convenient for the interview. Interviews were conducted in English and MDA was present during the first three interviews to listen to the interactions between the participants and the interviewer and to ensure appropriate data acquisition. There was no previous relationship or interaction between MDA and the participants. Data saturation was achieved at the end of the 12th interview, that is, the information was deemed sufficient as there were no new response patterns identified by the interviewer at this point. However, interview sessions with all consenting respondents (16) were completed in order to allow rich documentation and to ensure no point was accidentally missed. Repeat interviews were not required. Given that participants' location to the research site was remote, we were unable to return data transcripts to participants for comments.

The interview questions (See **Appendix 4.2**) were developed and iteratively reviewed by the research team. The notion behind the development of the questions was to seek recommendations on components and factors which patients considered important to continually stimulate their interest in the regular use of apps. Based on the fact that health technologies are voluntary use systems, therefore, the extent of engagement with them is determined by users' perceived quality of experience, ongoing benefit of usage (Lalmas, O'Brien & Yom-Tov, 2014), and consideration of viable alternatives to using the technology (Bickmore, Schulman & Yin, 2010). Furthermore, the development of questions on specific

educational topics which could be incorporated into diabetes apps was informed by these reasons: (i) gaps in the literature which shows that most of the currently available diabetes apps are lacking in educational component (Chomutare et al., 2011; Izhar et al., 2017, Adu et al., 2018a) and (ii) the ability of apps to meet self-management needs (diabetes education fosters self-care); which could subsequently motivate users' engagement with apps (Woldaregay at al., 2018).

Ethical considerations and consent

Approval for the study was granted by the Human Research Ethics Committee of the James Cook University (H7087). Submission of the survey responses implied consent to participate in the quantitative phase of the study while verbal consents were obtained for the telephone interviews and recordings.

4.3.2 Data Analyses

Quantitative data analysis was done using SPSS version 23 (IBM, Armonk, New York, US). Continuous variables were reported as means and standard deviations (SD) while categorical variables were reported as percentages. Chi squared test for independence and ANOVA were used for group comparisons involving categorical and continuous variables respectively. Statistical significance was set at p < 0.05

For qualitative data analysis, each interview was digitally recorded and the duration ranged from 7 to 20 minutes. Audio files were professionally transcribed by an individual with privacy certification and reviewed for completeness. Raw data files were imported to QSR Nvivo 11 for open coding and analysis. Emerging themes were identified using an inductive thematic analysis approach in order to ensure rich description of the data set (Braun & Clarke, 2006). The thematic data coding followed six phases as described by Braun and Clarke (2006). (1) Familiarising with the data through reading and re-reading the transcripts to make meaning of participants' responses; (2) generation of initial codes; (3) initial coding to identify emerging themes; (4) review of identified themes; (5) refining theme names and (6) documenting the findings and presenting illustrative quotes from participants' responses to support each theme. All phases of the thematic coding were completed by MDA. To establish the trustworthiness and credibility of the data, BMA was involved in phases 2-6 of the thematic analysis. Data were cross-checked in a consensus meeting; there was an 80% agreement between both researchers and discrepancies were resolved through discussion and mutual agreement. Both MDA and BMA have prior methodological training and experiences in conducting qualitative research. The remaining two researchers also checked the codes and themes to ensure consistency. Illustrative quotes were appended with a combination of number code and the type of diabetes the respondent has (for example "P4, T1D"). Quotes which included marketed names of apps were not reported verbatim for commercial reasons. The final manuscript was reported according to the COREQ criteria for reporting qualitative research (Tong, Sainbury & Craig 2007).

4.4 RESULTS

4.4.1 Participant Characteristics

A total of 245 respondents attempted the survey. Twenty-eight responses were excluded due to missing data (19) or ineligibility where the type of diabetes was not stated (9), leaving a total of 217 complete responses for the analyses. Participants' mean age was 44.65 (SD 11.51) and ranged between 18-76 years. They were predominantly type 2 diabetes (61.8%), females (55.7%) and were diagnosed with diabetes within the previous 1-5 years; (40.1%). Further demographic and health details are provided in **Table 4.1**.

	Study Participants (N	=217)
Characteristics	n	%
Gender		
Male	94	43.3
Female	123	56.7
Age (Years)		
18 - 29	27	12.4
30 - 39	64	29.5
40 - 49	41	18.9
50 - 59	46	21.2
60 - 79	39	18
Education		
High School	44	20.3
Technical College	41	18.9
Bachelor Degree	60	27.6
Post Graduate degree	62	28.6
Others	10	4.6
Continent		
Australia	75	34.6
Europe	76	35
Asia	64	29.5
America	2	0.9
Employment Status		
Employed	148	68.2
Unemployed	36	16.6
Retired	33	15.2
Type of Diabetes		
Type 1	83	38.2
Type 2	134	61.8
Duration of Diagnosis (Years)		
<1	27	12.4
1-5	87	40.1
6-10	44	20.3
11-15	16	7.4
> 15	43	19.8

Table 4.1. Demographic and health characteristics of survey respondents

For the telephone interviews, initially 31 respondents indicated interest to participate by providing phone numbers on completion of the survey. However, due to subsequent decline, only 16 respondents participated in the phone interviews. Respondents were predominantly

males; 56.2% (9/16), Australian residents; 87.5% (14/16) and have type 1 diabetes; 62.5% (10/16). Their ages ranged from 26 to 61 with a mean of 44.56 (SD 11.51) years old.

Use and non-use of apps.

Less than half of the participants (106/217, 48.8%) used apps to support their health management, where blood glucose tracker (60/106, 56.6%), blood pressure tracker (55/106, 51.9%) and food calorie counter (51/106, 48.1%) were the mostly used features. The remaining 51.2% (111/217) of the respondents did not use apps to care for their diabetes. Major reasons for non-use were "lack of awareness" (52/111, 46.8%) and "disinterest" (37/111, 33.3%) in the use of apps (See **Table 4.2** for details).

Influence of participant characteristics on the use/non-use of apps.

App use was significantly influenced by demographic/health characteristics. Individuals more likely to use apps tended to be younger (p=0.00), have type 1 diabetes (p=.002), resident in Asia (p=0.00), employed (p=0.001) and have higher educational level (p=0.00) than their counterparts in their respective demographic domains.

4.4.2 Preferences for app features and perceived usefulness

When asked to indicate the features they would prefer in a new app for DSM, major participant inclinations were towards apps with information on nutrient value of foods (56.7%), blood glucose tracker (54.8%), physical activity tracker (47%) and visual analytics to view trends in health status indicators (42.9%). Furthermore, interests were expressed in apps that provide either general education on diabetes self-management (40.6%) or personalised education in response to logged blood glucose data (40.1%).

Participants' ratings of perceived usefulness of app features to support diabetic' health management were highest for blood glucose tracker (mean 8.35 [S.D 2.03]), food nutrient

counter (mean 7.73 [SD 2.32]) and fitness/exercise monitor (mean 7.37 [S.D 2.29]). Apps for social networking among people with diabetes (mean 5.96 [S.D 2.99]) had the lowest rating. (See **Table 4.2** for details).

Table 4.2. App use, preferences and perceived usefulness of app features to support diabetes

 management

Variables	n	%
Usage of app [N=217]		
Yes	106	48.8
No	111	51.2
Features in currently used apps [n =106]		
Blood pressure tracker	55	51.9
Blood glucose tracker	60	56.6
Food calorie counter	51	48.1
Fitness/exercise monitor	50	47.2
Body weight monitor	35	33
Transfer of health data to doctor	20	18.9
Reminder (e.g take medication, BG ^a monitoring)	38	38.6
Others ^b	10	9.4
Reasons for not using app [n=111]		
Not interested	37	33.3
Lack of awareness	52	46.8
Do not have a smart phone	12	10.8
Lack of access to the internet	8	7.2
Expensive	10	9
Missing	10	9
Preferred features / functions if offered a new app [n=217]		
Food nutrient composition	123	56.7
Blood glucose tracker	119	54.8
Body weight tracker	79	36.4
Physical exercise tracker	70	32.3
Task reminder	70	32.3
General education on diabetes self-management &		
complication prevention	88	40.6
Personalised advice /education in response to logged		
BG data	87	40.1
Visual Analytics (view trends of logged data)	93	42.9
Data export to Doctor or other health team members	75	34.6
Social networking among people with diabetes	54	24.9

Apps feature ratings	Mean	SD
Blood pressure tracker	6.64	2.7
Blood glucose tracker	8.35	2.03
Fitness and exercise monitor	7.37	2.29
Body weight monitor	6.93	2.45
Task reminder	6.71	2.56
Export of data to health care team (e.g doctor)	6.91	2.7
Social networking	5.96	2.99
Food nutrient composition	7.73	2.32

^aBlood glucose

^bbolus insulin dose calculator, heart rate monitor, apps for continuous glucose monitors

Influence of participant characteristics on preferences for app features

App feature preferences were not influenced by type of diabetes. Affirmative response rate for each of the suggested app features varied between 25-48/83 (30.1-57.8%) for those with type 1 diabetes and 28-76/134 (20.9-56.7%) for those with type 2 diabetes. The most commonly preferred feature by all participants was food nutrient composition (56.6%, 47/83 for Type 1 diabetes and 56.7%, 76/134 for Type 2 diabetes). Also, 54-123/217 (24.8-56.7%) of all respondents had an inclination towards each of the various suggested features, irrespective of diabetes type.

4.4.3 Recommendations to Foster Long-Term Engagement with Diabetes Apps

Four themes emerged from the interview based on recommendations to foster long-term engagement with diabetes apps. The themes include: (1) improved functionalities (exciting and new recipes, actionable goals with convenient reminder), (2) certified and reliable information sources (social networking, research and news update), (3) consolidated and customised features, (4) ease of use. The themes and subthemes are reported below with representative quotes.

Theme 1: Improved functionalities

Subtheme 1.1 - Exciting and new healthy recipes: Participants commonly advised that essential attributes which could foster their engagement with apps that support healthy eating was for such apps not to offer just suggestions on suitable food recipes for people who have diabetes, but that the recommended recipes are also delectable:

"Some ideas about recipes to brighten up my day, because turning food into just fuel I think is just terrible. Although, we do have to kind of trim off some lavish recipes, but that doesn't mean it has to be eating cardboard". [P3, T2D]

Some participants want healthy eating support by apps to include assistance with shopping:

"I was thinking like- just look up something. If you are in the supermarket and you decide what you're going to buy, if I haven't prepared my menu, it would be nice not just to rely on memory all the time and be able to look up something (in the app), like something different to make". [P8, T2D]

Additionally, many participants felt that when apps provide nutritional information not only on various conventional foods but also on rare foods, this could stimulate their continual use.

"The aspect with just kind of having all available, so more not so common foods and you can virtually put any product in (the app) and it will split out the nutritional table". [P9, T1D]

Subtheme 1.2 - Actionable goals with convenient reminders: Most of the participants said, they would be highly attracted to engage with the use of apps for their self-management if its' reminder feature can be complemented with actionable instructions on how to accomplish behavioural goals of DSM.

"My anxiety and stress levels are high around the time of my menstrual period, so I find my sugar levels difficult to manage around that time...So if an app could connect to my CGM and

sort of warn me about those days. So if those could be tracked and I could be given a reminder thing like, okay, you need to cool down, you need to go for an extra session of meditation, or you need to run a little bit more during this time, that would be helpful". [P11, T1D]

A participant who identifies as a software developer gave specific examples of how apps could provide actionable reminder.

"Actionable knowledge that nudges me on the key items which helps me to succeed. For example, I've been messing around with the idea of building an app that geofences the bad places that you shouldn't eat. I mean first of all, geofence every single [Name of an American fast food company], so right from the moment you walk into them, it pings up and say you don't really want to be here...ordering this food, this is a bad thing, at this time". [P3, T2D]

"It can be listen, I've looked at the number of steps that you walked and you need to do... go for a walk at lunch time otherwise, you're not going to do and you know you that you've got to do something active for at least 30 minutes a day to help your diabetes". [P3, T2D]

Nevertheless, some participants commented that in the event that a user is unable to immediately review a behavioural data prompted from an app, it is necessary that users are able to turn off such prompting to attend to it at a more convenient time.

"Because you don't always want to be nagged by these things. Sometimes that sort of prompting-I mean there are apps that will automatically remind you, if you had a low blood glucose they might go, they might check back in and they might remind you in half an hour, or you might be able to set an optional reminder, to check again and make sure it is resolved. But you don't want any of that functionality to be so hard wired that you can't turn it off, because sometimes it's not convenient just at the moment. You know you might be in a meeting or you just never want to be nagged like that". [P14, T1D]
Theme 2: Certified and reliable information sources.

Subtheme 2.1 - social networking: In multiple instances, participants, particularly those who have type 1 diabetes, emphasised that interfacing of apps with social forums would be appealing. They would like to make social connections using apps, through which they can access role model narratives of stories from other individuals who had overcome salient obstacles in DSM. Participants believed that social networking in apps could serve as a motivator for long-term engagement as well as an opportunity to receive information on problem-solving.

"I think personal experience would be a massive thing. If I could sit down and read someone else's story and how they've gone from what I'm at now, like with minimal control, to being on top of it. It would be awesome to hear from someone that's done it, to give me a belief that I can do it. An app could be the only way to sort of get that connection ". [P12, T1D]

Participants further explained that apps linked to certified information sources would equally serve the similar purpose of providing a platform for accessing problem solving techniques in diabetes management.

"For apps to be able to make connections to other information sources will be good. So if for example, that you-it was a logging app and it could be seen that the frequency of low blood glucose had increased this month...., may be you may be directed out to resources that might help someone think about why that is. You know, either, peer-to-peer resources or respected kind of academic resources on ways to manage whatever the challenge is". [P14, T1D]

Subtheme 2.2 - Research and news update: Many participants with type 1 diabetes requested that apps should serve as a resource for news update on ongoing developments in diabetes management. The participants want an app to function as a platform to seek out information

whether in form of recent research, technology, medication or treatments which could further improve diabetes care.

"Good information on the latest technology or when the next pump is going to be approved. Am constantly looking for something that is going to make it (diabetes management) easier. It can go in there (into an app) and it's going to tell me what's waiting for FDA approval or what's the latest thing". [P1, T1D]

"In particular, the latest research and things they are implementing in recent times. So the latest news on use of medications and even new treatments would be fabulous (in an app)". [P7, T1D]

Theme 3: Consolidated and customised features

Subtheme 3.1 - Data consolidation: Consolidation of data from peripheral health devices used by patients into a single app was also an important consideration. Participants frequently mentioned that getting merged data would provide a unified record view.

"So it's got to be integrated completely into the health data in the phone. It can't be a standalone. It's got to be sharing data with everything else, so that if I look at my consolidated health chart collected by the app, say, how much I drank and may be the total sugar count I've consumed, I kind of want to see the (combined) data collected". [P3, T2D]

Also, participants' felt that when data are consolidated, clear meaningful outputs will be provided which will remove the barrier of deciphering the relationship between the data provided on each of the various health devices.

"Well, it's actually linking all the information together. I'm on a Continuous Glucose Monitor. I'm on a pump and I have the Fitbit. Trying to link all these three things together is impossible. What I would like is for all the information to correlate with each other and be able to see it potentially on a single page". [P4, T1D]

Some participants noted that non-interdependency of data from various health devices is difficult to manage especially for people with low education and may discourage their use of app.

"You know, you've got products like ("name of app") and ("name of app") for booking doctors' appointments. So I've got the ("name of app") managing my CGM, I've got insulin pump managing my insulin injections and telling me how much carbs I've had to how much I need. Then, I've got different system to manage doctors and different systems to manage my carb counting. So, it's complex and if you're not literate enough to be able to use your phone in that way, then it's a big negative. So you could have that integrated solution". [P2, T1D]

Subtheme 3.2 - Customised features: Participants with type 1 diabetes stated that the ability of apps to adapt to changes in users' requirement could encourage adoption and enhance engagement. Participants indicated that they would like to engage directly with specific functionality they desire to use per time without interruption by other functionalities which may be present in the same app.

"I think the thing to bear in mind is management and management strategies vary significantly from person-to-person. If you have an app that's got 15 functions, each individual person might really only use six or seven of these. For example, I've got no interest in calorie at all and some other things and someone might be really interested in keeping track of their blood pressure. Again it's not something am particularly interested in. So the fact that it (the app) supports that is great but for me I'd want to be able to turn that entry thing off on the screen so I haven't got to keep scrolling past stuff am never going to be using. So, it's the tailorability, it's making it flexible really". [P14, T1D] Additionally, inability of apps to adapt to the needs of a user could serve as a barrier to usage.

"Less amount of clicking stuff you've got to do on the app, I suppose would be beneficial. I like getting my tester out and all that stuff. Then having to go into an app where I'd have to go through 30 different things to find what I'm looking for would be more of a hindrance than what it would be to not just use it at all". [P12, T1D]

Theme 4: Ease of use - Several of the participants pointed out that simplicity of usage and limited time consumption by apps are major determinants for their engagement with the app. For example, a participant explained his view on these facts by giving reasons for abandoning a specific app for the other:

"I was formerly using ("name of app"). I just couldn't get it so I stopped using it. So the one I use now ("name of app") as a carb counter is just a bit easier to use and not too time consuming". [P1, T1D]

Apps' ability to save and display previously logged data will also limit the time requirement for use:

"Apps that remembers what you have put it,...because you will probably find that we eat a lot of same foods all the time but having to re-enter that in as a new-entry every single time is a bit of pain".[P2, T1D]

4.4.4. Specific Educational Topics Desired in Future Apps to Support Diabetes Self-Management

Three major topics emerged from participants' responses to the question on areas of diabetes management, which may be embedded as educational information into apps. The topics were approaches to problem solving, basic guidelines for the management of diabetes (transitioning from paediatric to adult care, signs of acute and chronic complications of diabetes) and the use of insulin pump therapy.

Theme 1 - Approaches to problem-solving: Majority of the participants proposed that apps should provide information on specific steps to take in response to challenges posed in the self-management of diabetes. Specifically, respondents expressed their desire for apps to offer education on possible strategies for problem resolution when logged blood sugar data are out of the clinically recommended range.

"I think you need to potentially---if someone puts in low blood sugar or a high blood sugar, you need to give directions on what they need to do...whether you need to have more insulin or if you are low, you need to have 15g of carb.. just pointers like that would potentially help people's management longer term". [P4, T1D]

Furthermore, automating the problem solving suggestions, with the inclusion of information on the health consequences of blood glucose levels per time was recommended by respondents as echoed in the quotes below.

Just putting in whatever the blood sugar reading might be, and it could come back with ways to help me with that". [P8, T2D]

"Let's say I did a pinprick (blood sugar) test and I had a reading that was odd, I would like to see what the implications of that is". [P8, T2D]

Theme 2 - Basic guidelines for the management of diabetes mellitus

Subtheme 2.1- Transitioning from paediatric to adult care: Participants with type 1 diabetes indicated that the inclusion of information to augment their knowledge of self-management during the transitioning phase from paediatric into adult diabetes care would be helpful. Participants find this phase of life to be difficult, thus, they mentioned that providing encouragement/teaching for self-management skills via an app may help to answer some of the mind-boggling questions they have about their management.

"I think mainly, issues when you become a bit older. So, it's just that middle section, I found it quite difficult going through transition, so from probably about 18 until about 30. So information as you leave adolescence into adulthood. Also, going into pregnancy, there are a lot of questions I've kind of like asked out of curiosity to my endocrinologist about pregnancy, which I was quite shocked at the answer". [P9, T1D]

Subtheme 2.2 - Signs of acute and chronic complications of diabetes: Many participants noted that apps having information on the signs of acute and chronic complications of diabetes would further create awareness about the disease and the importance of self-management.

"I think it would definitely be beneficial to have (information) on the problems and complications. Like what problems you will face before you face the major complications, so you sort of know what to look out for and what to be aware about". [P12, T1D]

The importance of the reliability of all provided information in apps was also emphasised.

"From a lot of people that I've spoken to, everyone get different information from different doctors...doctors have got medical view which they must pass across and they've got their own personal view. So I would say, it's got to come from the Australian standard, you know medical view of diabetes". [P6, T1D]

Theme 3 - Use of insulin pump therapy: Provision of guidance on insulin pump therapy through apps was highlighted by many participants with type 1 diabetes. The participants enunciated that having adequate knowledge on how to set a pump to deliver the right amount of insulin (especially bolus rate) could be challenging. Therefore, they expressed that having information on the use of insulin pump therapy in apps would be beneficial to foster the attainment of a stable blood glucose levels.

"Something like (when to deliver boluses on insulin pump). I mean, it's potentially something that is quite hard to do...even when you have a pump before a meal is actually not really good thing to do because your insulin starts working at the same time you are eating. If you dose 15 minutes or 30 minutes before you eat, you then don't get a peak because you've got your insulin working at the same time as the food is being digested. That sort of stuff- long-term for a person with diabetes, I think is critical so you are not going up...you are not doing the rollercoaster". [P4, T1D]

It was further stated that guidance on physical exercise and insulin dose adjustment when using a pump therapy would be helpful. Participants want apps to provide direction on administration of basal insulin relative to glycaemic levels in order to reduce or prevent hypoglycaemic events.

"So my experience about 6 months ago was that I started exercising and after 15 minutes of crunches I hit a 3 from 10 sugar level. I had my pump disconnected at that point. So I've been a little scared to get back to that. So in this scenario, it would be helpful if I sort of then put my sugar level in (an app), when I've exercised last or how am feeling at this point and I'd get at least an idea of whether I should continue to use the pump or disconnect it altogether". [P11, T1D]

4.5 DISCUSSION

A mixed methods study (online survey and telephone interviews) was conducted with a diverse pool of participants who have type 1 or type 2 diabetes. The online survey focused on obtaining information about the use and preferences for app features, which could support DSM. The interviews provided recommendations on how to foster long-term engagement with apps that support diabetes management and the educational topics, which could be included in such apps.

App use and preferences

Statistics on current app uses (48.8%) and mostly used app features (food calorie counter, blood glucose and blood pressure tracker) are in congruence with previous studies (Lithgow, Edwards

& Rabi, 2017; Conway et al., 2016). Monitoring of blood glucose and blood pressure are important to attain diabetes management goals. Therefore inputting these data into apps may encourage personal reflection and reveal out-of-range values in need of urgent treatment.

The current study also echoes the findings of previous surveys which reported that usage of apps was mostly seen among individuals with type 1 diabetes, those of younger age, high educational level (Boyle et al., 2017) employed and resident in Asia (Guertler et al., 2015). These observations confirm that socio-demographic characteristics of individuals especially the age, level of education and income status have a strong influence on their digital health use (Lustria, Smith & Hinnant, 2011; Kontos et al., 2014; Carroll et al., 2017). Additionally, eHealth literacy (the potential to accurately interpret health data obtained from mobile health devices) has a direct correlation with the aforementioned socio-demographic characteristics (Cho, Park & Lee, 2014), thus, may also be responsible for the app usage trends observed among the sub-groups in this study.

Contrary to the findings of other studies on app use among people with diabetes (Shibuta et al., 2017; Kayyali et al., 2017), a larger proportion of the respondents in this study do not use apps, primarily due to lack of interest and awareness. Given that most patients with chronic diseases often regard regular tracking of their health as an additional burden (Ancker et al., 2015), it is unsurprising that lack of interest was a major reason for not using an app. It is therefore important that apps are unambiguous, provide clear information on the specific health benefits for patients and at the same time are intuitive enough to stimulate interest in usage. Furthermore, since the use of apps offer great potential to support DSM, clinicians may promote and create awareness about apps to patients through their unique role of providing health recommendations to their patients (Gagnon et al., 2015).

Interestingly, the type of diabetes participants had did not influence their choice of app features. Preferences and ratings were highest for information on nutrient content of foods and capacity to track blood glucose. This result is expected since healthy eating and regular monitoring of blood glucose are important self-management domains for people with type 1 and type 2 diabetes in order to ensure healthy life (Shrivastava, Shrivastava & Ramasamy, 2013). Due to the direct impact of food on blood glucose levels, monitoring carbohydrate and calorie intake is imperative to maintain optimal glycemic control (Franz et al., 2010). Therefore, people with diabetes may benefit from apps that help with nutritional tracking in order to assess the impact of foods on blood glucose levels. Another app feature highly preferred by the participants was visual analytics that could present behavioural and health data progress in a graphical format. This feature increases the level of awareness and encourages accountability for self-management behaviours (Winters-Miner, 2014; Giroux et al., 2014) and provides support for healthy lifestyle decisions (Garabedian, Ross-Degnan & Wharam, 2015).

Recommendations to foster long-term engagement with Apps

Ye and colleagues (2018) reviewed about 1050 apps from the Google play and iTune stores and observed that more than 70% of these apps were designed to support healthy eating mainly with components for 'carbohydrate count' and 'diet tracking'. The present study shows that participants advocated for more comprehensive diet management features in future apps, comprising of nutrient data base of both the common and rare foods, and recipes for tasty foods which are suitable for people with diabetes. Patients' access to ideas on diverse food varieties via an app, will prevent monotonous dieting and subsequently may foster motivation to maintain a healthy diet and keep track of foods eaten and at the same time boost engagement with the app. As voiced by the participants, future diabetes apps could include reminders for selfmanagement behaviours which are accompanied by advice on realistic actionable steps to attain those behaviours. App developers could ensure that the proposed actions are simple and instinctively written in plain language, because these are the key factors to attaining health literacy (Stableford & Mettger, 2007). Additionally, in such suggested steps, users should be able to find what they need, understand what they find and act on it (United State Government, Plain Language Action and Information Network, 2010).

Participants with type 1 diabetes recommended the inclusion of community/social forum in apps and this is worth considering by app developers. Type 1 diabetes is usually diagnosed at a younger age, its management could be quite complex, and requires lifelong daily health commitments. Providing social support for type 1 diabetes patients may improve treatment adherence by encouraging optimism, which can mitigate the stress of living with the illness and accompanying depression (DiMatteo, 2004). App developers could design apps that offer peer interactions among individuals who share common health experiences (Heisler, 2007). Such social networking features may foster a sense of connection, serve as an essential social support system for patients to be more aware of the importance of DSM and learn new practical and effective ways for health maintenance. Nevertheless, it is important that patients who use social support features in health apps are reminded not to rely heavily on advice from other users or substitute such information for regular check-ups with their health care providers.

Furthermore, consolidation of data from peripheral devices into app (as proposed by participants), if done appropriately eliminates redundancy and provides data portability. Although in reality, meeting this need will be difficult and may face some challenges particularly when data streaming are from health devices developed by different companies. Such challenges will include which developer will be responsible for the accuracy, reliability, security and compliance with privacy regulation of the consolidated data. This may be the

reason why data has remained confined to their respective platforms and integration with other health devices has been very limited (Chiauzzi, Rodarte & DasMahapatra, 2015).

Earlier studies on the use of smartphone apps for health-care have reported ease of use and simplicity as strong determinants of technology acceptance (Kayyali et al., 2017; Crane et al., 2017) and our study corroborates this finding. Hence, in order to maximise patients' uptake and long-term engagement with app resources, app developers need to facilitate easier maneuvering through apps menus as a means of reducing the burden of time consumption on users (Rao et al., 2010).

Customisation of apps in an adaptable way to match the interests, needs or habits of their respective users was another key point raised by the type 1 diabetes participants in this study. This proposition might have been influenced by participants' awareness of the highly individualised demand for self-management of type 1 diabetes (Smith & Harris, 2018). Customising apps features such as personalised alerts modified to users' specific needs and preferences may promote the required behavioural change (Klein et al., 2017) and engagement with the app (Juruascio et al., 2015).

Educational topics in future apps

Problem solving may be a difficult area of diabetes management hence the suggestion by many participants for the inclusion of tips on problem solving in diabetes apps. Problem solving in diabetes management is a learned behavioural process comprising a set of potential solutions to problems, selecting the most appropriate solution, applying the solution and evaluating its effectiveness (Hill-Briggs, 2003). Problem solving is the foundation upon which patient attainment of the remaining self-management behaviours (For example, healthy eating, monitoring, reducing risk) are built (Mulcahy et al., 2003). Also, there exists a direct correlation between problem solving and improvement in glycosylated hemoglobin (HbA1c) level (Hill-

Briggs & Echemendia, 2001; Glasgow et al., 2007), therefore it is important that patients receive additional support in this area. The support will involve development and incorporation of diabetes specific problem-solving measures into apps. Nevertheless, it should be noted that in some situations, apps may not be able to provide effective support to resolve health dilemma in people with diabetes because there may be too many variables in the decision making process needed to address certain challenging areas of self-management, therefore requiring direct interaction between a patient and the clinician or diabetes educator (Ye et al., 2018). Moreover, meeting this particular patients' desire should be handled with caution as diabetes management is uniquely complex, so the use of a single problem solving idea to serve all patients may not be feasible or applicable to some areas of diabetes management.

Personalised or general education is currently available in only a few diabetes apps (Chomutare et al., 2011; Izahar et al., 2017), and participants of the current study, requested for its inclusion in apps. There is ample evidence that patients retain little of the education provided at clinic visits (Kessels, 2003). Since, mobile devices have become ubiquitous in educational settings, they can give expanded opportunities for users to access health information anywhere or double check knowledge (Wallace, Clarke & White, 2012; Payne, Wharrad & Watts, 2012). Embedding apps with education and behaviour change techniques can complement instructions provided face to face by health care team members. However, apps that focus on provision of diabetes education should consider the medical accuracy of their content.

Practical implications

Patients' recommendations observed in this study demand thoughtful considerations from designers and developers to build mobile apps that best meet the identified patient preferences and propositions. Based on the findings of this study, apps with features on documentation (blood glucose and physical exercise log), reminders (with actionable notifications), advisory

(educational information), analytics (view trends in behavioural and health data indicators) and comprehensive nutritional data base are proposed to strengthened the functionality of apps and foster long-term engagement and better health outcomes for people with diabetes. In addition, it is recommended that the educational information in these apps are related to problem solving and basic guideline for DSM.

Meeting the recommendation for future apps to include a data consolidation system across several peripheral health devices will require careful consideration of existing privacy regulations as this is one of the factors contributing to a lag in data interoperability (Chiauzzi, Rodarte & DasMahaptatra, 2015). Furthermore, the concept of providing a link to social support groups also warrants consideration especially if the app is targeted at people with type 1 diabetes although, social support can also promote self-management behaviours in those with type 2 diabetes. Lastly, it is important to ensure ease of use and limited time consumption in the developed app in order to further promote engagement.

It is imperative to evaluate the functionality of developed and operational apps based on experimental pilot studies and large randomised controlled trials (RCTs), which are essential next steps to ascertain patient engagement with the app. This will offer the opportunity to investigate if user engagement can be guaranteed if they (users) are provided with apps that offer the type of features they 'ask' for. These experimental studies should aim to integrate app interventions seamlessly into the daily routine of end-users so that app usage is not perceived as an extra chore.

Strengths and limitations

Firstly, this study was able to quantitatively characterise the social and health demographic profiles of adults with type 1 or type 2 diabetes in relation to their use of apps and feature preferences. Secondly, through a qualitative approach, it provided information on educational

topics which could be included in apps to support DSM, in addition to ways to foster adoption and engagement with such apps. To the best of our knowledge, obtaining the information described above in a single study has not been attempted before. Thus, by using a mixed method design which offers the opportunity to comprehensively address the research questions and provide a clear picture of how to meet the research goal; this study fills important knowledge gaps, adds to the body of science and provides direction for future development of functional apps for DSM. The reliability of the study findings is further strengthened by the fact that participants were asked to indicate their preferences and provide recommendations for a future diabetes app and not on an already developed app. This affords them the convenience to give their candid opinion without the agitation of negatively criticising the work of a developer/investigator.

The findings of this study should be interpreted with the following limitations. Firstly, due to the cross-sectional nature of data collection which was done at one point in time, it is possible that people with diabetes may vary their preferences and recommendations for future diabetes apps over time especially due to evolving nature of digital technology. Secondly, generalisation of the findings to other settings might be limited due to the sample size/groups, which mainly comprised of participants from three continents whose responses might have been influenced by cultural or geographic differences. The inclusion of more participants especially from other continents might have led to the emergence of other results. Thirdly, it is worth noting that the quantitative part of this study did not access the differences in preferences of participants based on the use/non-use of insulin as part of their treatment regimen. Participants' preferences for app features may be influenced by their treatment therapies. Fourthly, more themes and topics may have emerged if the interview duration was longer. Given that no additional compensation was offered to interviewee, short interview duration was utilised to foster increased participant numbers as long interview may not be justifiable for participants' time involvement in the

study''. Nevertheless, literature has also shown that telephone interview sessions are usually shorter than face-face interviews. Additionally, even when data saturation was observed at the 12th interview, all the 16 respondents were interviewed to ensure no point was accidentally missed. Lastly, the survey responses were based on self-report and therefore may be subjective to responder's bias.

4.6 CONCLUSION

This study will serve as a useful guide for researchers and developers of future apps aimed at diabetes self-management because it highlights the importance of involving end users in the process of developing the content and features that appeal to them. Perceived importance and preferences were higher for app features that could support healthy eating, blood glucose and physical activity monitoring; provide data analytics and diabetes education. Recommendations to foster long-term engagement with diabetes apps have been described via four themes: improved functionalities (exciting and new recipes, actionable goals with convenient reminder), certified and reliable information sources (social networking, research and news update), consolidated and customised features, and ease of use. Additionally, patients are interested in apps embedded with educational information on problem solving techniques, use of insulin pump therapy, signs of diabetes complication and prevention as well as transitioning from paediatric into adult care.

Overall, this chapter addressed the third aim of this thesis by highlighting the perceptions of type 1 and type 2 diabetes patients in relation to features and educational contents in apps that will foster retention and engagement with diabetes self-management activities. The results of this chapter provide evidence of the unmet app feature needs and support the development of a novel app that best meets these needs. These results were prioritised and utilised in the design of the novel mobile diabetes app in Chapter 5.

Chapter FIVE: The Development of My Care Hub Mobile-Phone App to Support Self-Management in Australians with Type 1 or Type 2 Diabetes

5.1 ABSTRACT

Non-adherence to self-management poses a serious risk to diabetes complications. Digital behavioural change interventions have the potential to provide education and motivate users to regularly engage with self-management of diabetes. This paper describes the development of My Care Hub mobile phone application (app) aimed at supporting self-management in people with type 1 or type 2 diabetes. The development of My Care Hub involved a comprehensive process of health behavioural change identification, end users' needs, expert consensus, data security and privacy considerations. The app translation was a highly iterative process accompanied by usability testing and design modification. The app development process included: (1) behaviour change strategy selection; (2) users' involvement; (3) expert advisory involvement; (4) data security and privacy considerations; (5) design creation and output translation into a smartphone app and (6) two usability testings of the app prototype version. The app features include self-management activities documentation, analytics, personalised and generalised messages for diabetes self-management (DSM) as well as carbohydrate components of common foods in Australia. Twelve respondents provided feedback on the usability of the app. Initially, a simplification of the documentation features of the app was identified as a need to improve usability. Overall, results indicated good user satisfaction rate.

5.1 BACKGROUND

More than 1.1 million Australians have type 1 (12%) and type 2 (85%) diabetes (Australian Institute of Health and Welfare, 2019c). Poorly controlled diabetes increases the risk of chronic complications (Alberti & Zimmet, 1998). Diabetes self-management education (DSME) and support are effective interventions to assist patients navigate through decision making

processes for participation in self-management activities necessary for improved glycemic control (Henrich, Schaper & de Vries, 2010) and reduce the risk of developing complications (Powers et al., 2015b). Self-management support could be educational, behavioural, psychosocial or clinical (Haas et al., 2012). Providing ongoing support to patients could mitigate disease related distress, improve adherence to recommended self-management activities and consequently enhance health outcomes (Norris et al., 2002; Haas et al., 2012) but this practice has proven difficult in a real-world context (Kennedy et al., 2014). Provision of self-management support through the use of smartphone may help address health system level limiters which impact negatively on the frequency and quality of self-management support patient receives. Such limitations include time constraints, limited access to care providers (Wilkinson, Whitehead & Ritchie, 2014) and partial cost reimbursement by third party health insurance (Piette, 2007).

Smart phone is a ubiquitous technological device with more than 2 billion users worldwide (Statista, 2019a) and over 16% of 6 billion mobile subscriptions are smartphone subscriptions (MobiThinking, 2014). In Australia, the growth in smartphone usage has been sporadic. There is approximately 84% smartphone users among the entire population of mobile phone users in 2018, which is an increase from 74% in 2014 (Statista, 2019b). Most smart phone functionalities are aided by apps (software that are designed to run on smartphones), which could complement highly developed health care technologies and serve as supporting tools in many chronic disease management (Kay, Santos & Takane, 2011). Various apps have been developed to enhance self-management of diabetes (Quinn et al., 2011; Kirwan et al., 2013; Kim et al., 2014; Holmen et al., 2014; Waki et al., 2014). However, most apps have no reference to health behavioral change models or scientific evidence-based theories (Waki et al., 2014; Quinn et al., 2011; Kim et al., 2014). Additionally, considerations of target users' preferences (Waki et al., 2014; Quinn et al., 2014; Quinn et al., 2011) and data privacy input (Norri et al., 2002;

Kim et al., 2014), are minimal or totally absent in apps. Furthermore, some lack educational information, which is a crucial component of diabetes care, to foster coping skills for ongoing self-management in patients and improved health (glycaemic) outcomes (Kim et al., 2014; Holmen et al., 2014).

Objective

Emerging evidence support the use of smartphone apps for DSM (Quinn et al., 2011; Kim et al., 2014; Waki et al., 2014). However, there remains paucity of information in relation to the description of the development processes and design of smartphone apps interventions, leaving unanswered questions about how to productively leverage apps for DSM (Adu et al., 2018a). Adequate description of the development of interventions could decrease waste in health research and enhance better organised synthesis of study results (Hoffmann et al., 2017; Adu et al., 2018a). Vivid explanation of study process will promote wider dissemination of research findings and improve the rigour and quality control of published research (Munafò, 2016). In addition, explicit reporting of intervention development can allow external scrutiny of its plausibility when later used in trials and help evaluators and policy makers decide which context to prioritise for future replication (Moore et al., 2015).

On the other hand, assessing usability occupies a central part of app development (Zaid, Jamaludin & Wafaa, 2012). A well designed app with high usability positively influences users' engagement (re-use rates) (Eysenbach, 2005). Conversely, poor usability is associated with low effectiveness and engagement (Eysenbach, 2005; Whitlock & McLaughlin, 2012). To the best of the authors' knowledge, there is no published full report of the systematic development of a diabetes smartphone app targeting Australian population before use in trials. Therefore, this paper reports a full description of the development and usability testing process of an app named "My Care Hub" prior to use in a pilot trial. My Care Hub was developed as

an app to best meet the needs of type 1 and type 2 diabetes patients in the provision of preferred features and diabetes educational contents which could foster incremental knowledge gain, skills, self-efficacy and motivate patients to actively engage with self-management activities.

5.2 METHODS

An initial systematic review (Adu et al., 2018a) by the authors highlighted the importance of five essential factors that needed consideration in the diabetes app development before actual use in a Randomised Controlled Trial (RCT). Specifically, the factors included involvement of users and clinical experts, health behavioural change theory employed, data security and privacy considerations, and pilot testing (Adu et al., 2018a). In practice, these steps overlap and are iterative. Herein, we describe how these steps were considered in the development of My Care Hub. The process of pilot testing is excluded because it is beyond the scope of the present paper. Additionally, this paper describes the features and functionalities available in My Care Hub and the usability testing of its prototype. All activities conducted during the predevelopment, development and testing stage of the app are described.

Ethical Consideration

This study was approved by the James cook University Human Research Ethics Committee (H7087, H7285). Informed consent was obtained from all participants involved in the usability testing of My Care Hub. All methods were performed in accordance with the relevant guidelines and regulations.

5.1.1 Stage 1: Pre-development

Users' involvement

A user centered design process was utilised in the development of My Care Hub. An iterative process addressing people's needs and the enabling infrastructure to meet those needs

constitute the central focus of a user centered design approach (The International Organisation for Standardisation, 2010). People with type 1 or type 2 diabetes were the primary target users of My Care Hub, hence their involvement in the development process, in order to gather information and ideas for appropriate design and educational content of the app. We performed a need analysis study among a multinational audience of people with type 1 or type 2 diabetes using a mixed methods study design (*currently chapter 4 of this thesis*) (Adu et al., 2018b). The study elucidated diabetes patients' app feature preferences and explored their recommendations for inclusion to promote engagement with My Care Hub diabetes app. The result indicated that both patient groups desired apps that featured documentation (blood glucose and physical exercise), advisory information (educational), analytics (view trends in behavioural and health data indicators), reminders (with actionable notifications) and food nutritional database. Furthermore, patients desired educational information on approaches to problem solving when blood glucose data were out of the clinically recommended range. Basic guideline information for DSM was perceived to be highly beneficial. Full details of the findings of the needs analysis study have been published (Adu et al., 2018b).

Health behavioural change theory

Maximising the potential efficacy of health behavioral change interventions in humans requires an understanding of the theoretical models or mechanisms that underpin such behavioral changes (Davis et al., 2015). An extensive literature search of behavioural change theories (Michie et al., 2008) was adopted to make informed decision in the development of My Care Hub. This provided the opportunity to identify the best interventional techniques (and the underlying theories) that are likely to be effective in a DSM app. The first technique considered was self-monitoring; an important intervention that fosters adherence to self-management and driven by self-efficacy construct of the social cognitive theory (Bandura, 2001; Bandura 1998). Self-efficacy refers to confidence in a person's ability to take action required to implement situation specific behaviours in order to attain specific goals (health outcomes) (Bandura, 1999). Self-efficacy forms the "major basis for action", occupies a pivotal regulatory role in the causal structure of self-management and perseverance to continue even when barriers are encountered (Bandura, 1999. Hence, improving self-efficacy through task related activities such as self-monitoring increases the confidence and persistence towards accomplishing the task (Barling & Beattie, 1988). On this basis, it *was* decided that My Care Hub should provide users with a platform to create daily entries and monitoring of their health behavioral activities.

Although self-monitoring is an important feature in behavioural interventions, patients' adherence to self-monitoring alone is often low, especially when not accompanied by another intervention which could further enhance self-efficacy (Mishali, Omer & Heymann, 2011). High attrition rates when self-monitoring activities are the only intervention features in diabetes apps have been reported (Kirwan et al., 2013; Holmen et al., 2014; Guertler et al., 2015). Therefore, sole-reliance on self-monitoring features in an app may limit its effectiveness and use to only those who are personally motivated to self-monitor regularly. Consequently, it was considered important to include other essential interventional features such as provision of information and feedback in My Care Hub.

Components of educational information and feedback in health behavioural interventions are best developed using Information-Motivation-Behavioural Skills (IMBS) model (Fisher et al., 2011; Fisher, Fisher & Harman, 2003). The IMBS model has several constructs pertaining to patient adherence to recommended health behaviours. Primarily, the self-management information needs to be easily actionable in order to enhance optimal care. In addition, the information should provide basic knowledge about the relevant medical condition and effective management strategies. Users also need to be provided with encouraging and motivational feedback and recommendations in order to boost their confidence and enhance their ability to achieve desired health behavioural changes (Bandura, 1994; de Bruijn, 2010). Reinforcing healthy behaviour significantly improves self-efficacy for participation in self-management activities (Williams & French, 2011). Furthermore, inclusion of behavioural skills which ensure patients are provided specific strategies on how to perform the adherence behaviours (diabetes self-management activities - e.g. choosing appropriate foods, problem solving for low/high blood glucose) are important to enhance effective self-management. Lastly, IMBS model indicates that strategies aimed at increasing patients' knowledge are prerequisites for behavioural change but not sufficient enough to sustain the change (Mazzuca, 1982). Therefore, such information should be coupled with motivation to increase the possibility of adherence.

The IMBS constructs were employed in the development of three educational information modules embedded in My Care Hub. These modules were: general information on diabetes management, automated feedback in response to logged blood glucose and carbohydrate contents of foods. These modules were aimed at improving patients' knowledge, skills, selfefficacy and provide specific directions for fostering patients' adherence to ongoing selfmanagement.

Data security and privacy considerations

Several security features and privacy policies in the design and development of mobile health apps (Morera et al., 2016) were followed. Authorization and authentication of users to assess the app was controlled, whereby the first user interface after the app download is a login screen. The screen requires a user to enter an assigned unique username (email address) and password in order to gain access to the app features. Authentication verifies user session handling for future research and ensures that the server is not vulnerable to injection attacks (Lin & Chen, 2007; Osawaru & Habeeb, 2014). Protection of data transfer from My Care Hub to cloud storage was ensured through encryption that makes the data illegible, unusable and indecipherable to unauthorised persons (Osawaru & Habeeb, 2014). My Care Hub was built on a proprietary personal health (PHR) platform using Firebase (Moroney, 2017) and it has no connection or interaction with third party health information systems. In accordance with the "Australian Privacy Principle" (Office of the Australian Information Commissioner, 2014), the services of an independent legal firm was employed to develop a comprehensive legal document pertaining to legal agreement, terms of use and privacy policy. The document describes how My Care Hub manages personal information, data collection, usage and storage. The legal document was incorporated into My Care Hub app for easy accessibility by users.

Expert advisory group

An expert advisory group comprising diabetes educators, endocrinologists, health researchers and app developers worked as a team to make decisions regarding My Care Hub. Implications of the users' needs analysis study findings (Adu et al., 2018b) and effectiveness of identified health behavioural change interventions were discussed by the team. Results were prioritised, adjusted and refined before a final agreement on the selection of My Care Hub features, design and content in order to meet the requested needs of users as much as possible.

5.2.2. Stage 2: Design and Translation of Pre-Development Output into a Smartphone App

1. Development of My Care Hub Prototype

App developers within the eResearch Team at James Cook University (JCU), Queensland, Australia developed the prototype. The purpose of the app was to serve as a tool for monitoring self-management activities, providing access to information and aiding motivation to engage with DSM. These purpose were adequately met by providing opportunity to users to monitor and track their core health behaviours (weight, physical activities and carbohydrate contents of the foods consumed) and vitals (blood glucose); and have easy access to three streams of diabetes educational information: (i) an overview of diabetes management, (ii) problem solving related and motivational feedback on logged blood glucose data, and (iii) information on carbohydrate contents in foods. To ensure that the app is usable by both types 1 and 2 diabetes patients, the feedback took into consideration the different recommended blood glucose values.

Initially, a visual skeletal framework of the app was created and subsequently used to develop the prototype. The framework comprised features, educational content, lay out, and interfacing of the app functionalities. The app was developed in Java in order to run on Android Platform. The resulting prototype was thoroughly examined and tested to ensure that all specified requirements were incorporated into the app for optimal functionality.

2. Features and Functionalities of My Care Hub

My Care Hub contains the following features: selection of type of diabetes, My Info, documentation, View insights, Carbs in foods and educational tips. Description of each of these features are provided below and a screen shot of the features in My Care Hub are provided in **Figures 5.1a-d and 5.2a-d** below.



Figure 5.1a-d: My Care Hub Screen Shots



Figure 5.2a-d: My Care Hub educational modules screen examples

Type of DM: This feature is displayed after the user has downloaded, installed and registered the app on their smart phone. The user is then required to choose the type of diabetes they have (**see Figure 5.1a**) in order to determine the type of feedback to be received when blood glucose data are logged.

My Info: This feature gives the user the choice to set up a profile to document their treatment details. Users can indicate their treatment specifics such as recommended blood glucose targets (upper and lower), types of medication (oral only, oral plus insulin, insulin only or none) and types of blood glucose meter, insulin pump, continuous glucose meter, if applicable. Profile set up may also include personal details such as gender and height (see **Figure 5.1b**).

Documentation: The home screen of My Care Hub (**Figure 5.1c**), displays all its documentation features for logging of self-management activities including blood glucose, body weight, physical activities and carbohydrate content of foods consumed. The actual date and time of logging is automatically set and users have the option to record the activity location. For the blood glucose (BG) interface, the type of BG data measured (either fasting or post hours post breakfast, lunch or dinner) can be selected from a drop menu and saved.

View insights: This analytic feature provides users with graphical display of all data logged through the documentation features (**Figure 5.1d**). It also takes the average of all data logged into each of the documentation features over a period of seven days and displays the mean on the homepage in form of summary feedback. This enables patients to visualise trends in lifestyle activities (especially physical activity and carbohydrate content of foods) and observe the impact on blood glucose levels over time (Winters-Miner, 2014), thus fostering the ability to adjust their self-management strategy accordingly.

Educational modules: My Care Hub comprises three educational modules:

a. Overview of diabetes management: The app contains textual information on the seven essential self-management activities for people with diabetes. These activities predict good health outcomes: lifestyle modifications (healthy eating and improved physical activity), monitoring of blood sugar, complying with medications, good problem-solving skills, healthy coping skills, and risk-reduction behaviours (such as smoking cessation and reduction in alcohol intake) (Tomky et al., 2008) (see Figure 5.2a). The messages were designed to be action-oriented, create awareness about associated benefits and provide engagement suggestions. The messages were written in plain language to enable users find, interpret and act on findings (Stableford & Mettger, 2007). Table 5.1 depicts examples of app messages under each of the vital self-management activities.

Category	Message
What is	Diabetes results from inherited and/or acquired deficiency in insulin
diabetes?	production or by the ineffectiveness of the insulin produced.
	In a diabetic state, insulin function is impaired, therefore the body needs conscious help to manage blood sugar by eating right, exercising, taking medications appropriately and reducing stress.
Health food choices	Stay motivated to eat healthy. When new foods are eaten, testing of blood sugar before and 2 hours after the first bite is recommended in order to see the effect of the food on the blood sugar.
	Vegetables are rich in fibre, make you comfortably full, have vitamins for healthy immune system and most likely not spike your blood sugar level.
Physical activity	Exercise improves bone strength, keeps the heart and blood vessels healthy and lowers insulin resistance.
	Exercise does not have to involve the whole day. You can split it up into 10 minutes, 3 times a day at a convenient period for you.
	If you are on insulin and planning for prolonged exercise, if your blood sugar level is below 6mmol/L, it is advisable to eat an extra carbohydrate.
Medication usage	Some diabetes medicines can lose effectiveness if they are old or stored improperly. For example, insulin should not be frozen or exposed to extreme heat.

Table 5.1: Exa	mples of	app	messages
----------------	----------	-----	----------

	If you often forget to take your medications, setting a reminder or alert may help you.
	Do not inject insulin on moles or scars because it slows absorption rate and limit insulin effectiveness.
Monitoring blood glucose	Some of the important times to test blood glucose include before breakfast (fasting), 2h after a meal, before rigorous exercise, before bed, when you are not feeling well.
	Regular testing of blood sugar improves confidence to look after diabetes, give a better understanding of the impact of food intake, medication, exercise and other factors such as stress and illness on blood glucose.
Reducing the risk of complications	Keeping appointment with the eye/foot doctor, endocrinologist and other health care team is a great way to detect on time any complication development.
	Smoking increases the risk of developing acute and chronic complications in diabetes mellitus.
	Keep your foot healthy by checking regularly for any changes, washing daily and wear shoes that fit properly. In the occurrence of unexplainable blisters, cuts or openings, it is best to consult your doctor immediately.

b. Feedback: My Care Hub includes an algorithm that provides automated feedback messages in response to logged blood glucose (BG) data. It assesses if the logged BG data are within or outside the clinically recommended range and provides appropriate encouragement or advice. This feature was founded on relevant evidence-based literature and Diabetes Australia recommended BG level guidelines (Diabetes Australia, 2015) for appropriate and accurate feedback. For people with type 1 diabetes; 4-8mmol/L at fasting and < 10 mmol/L 2 hours postprandial BG levels were recommended. Fasting levels of 6-8 mmol/L and 2 hours postprandial levels of 6-10 mmol/L were endorsed for people with type 2 diabetes.

A range of evidence-based, motivational, health promotional and behavioural skills information were developed as feedback on BG levels recommended for both type 1 and type 2 diabetes. Decision based system rules were programmed into the app to ensure that users receive semi-individualised feedback based on their logged data. The system is controlled by the indicated type of diabetes, value of BG (whether within or beyond the clinically recommended range) and the period of BG measurement (either fasting or 2 hours postprandial). Once BG data are logged, the app delivers a brief feedback. If the BG is within the clinically recommended range, the app delivers messages to encourage the patient to continue with regular self-management. Such messages include: "Excellent; BG within target range, continue your medication as prescribed", and "Excellent, BG within target range, continue with healthy eating"). It has been demonstrated that providing motivation and encouragement after attaining a goal can enhance self-efficacy (William & French, 2011) which in turn, facilitates health behavioural changes in disease control (Holloway & Watson, 2002). If the logged BG are outside normal range, the app offers suggestions related to problem solving for low or high BG as deemed appropriate. For instance, if logged BG is less than 4 mmol/L, sample feedback includes: "BG levels seems too low, this may occur when medication is not balanced with food and physical activities", and "you are at risk of hypoglycemia; treat immediately''. If BG values are extremely aberrant to the normal range (over 15 mmol/L), messages such as "if high levels persist and you don't know why, seek medical attention immediately" are triggered. Messages are unidirectional and displayed using colour labels, where green, orange and red indicate "ideal", "not ideal" and "extremely low or high" BG levels, respectively (see Figures 5.2b and 5.2c).

(c) Carbohydrates in foods: This feature comprises textual information about carbohydrate and calorie components of common foods in Australia sourced from the Australian Food, Supplement and Nutrient database (AUSNUT 2011-13) (Asutralian NewZealand Food Standards, 2019). Some commonly available foods were selected for this database and organised under four main groups: fruits and vegetables; eggs and meat; diary; and legumes and grains. Portion sizes and approximate carbohydrate and calorie content were provided for each food item. For example, 1 slice (40g) of whole wheat bread contains 20.56g of

carbohydrates and 111 calories. Foods with low glycaemic index were displayed in green colours as healthier models for consumption by those who have diabetes (Rizkalla et al., 2002) (see **Figure 5.2d**).

Functionalities and features for future research: To facilitate future research, export of the logged data into a cloud storage for all documentation features, date and time of log-ins were enabled for downloading as comma-separated value files into our database for subsequent statistical analysis. Furthermore, the app platform offers analytics of users' logged data and includes cloud and in-app messaging features to allow for push notifications. These features will provide additional educational messages for future research.

5.3.3. Stage 3: Usability Testing

Usability testing of My Care Hub was done in two stages:

1) Members of the public who do not have diabetes were randomly selected for early testing strategy to ascertain the technical performance of the app and to identify any navigation issues when downloaded on various android phones. The app's functionality and aesthetic usability testing by users was over a period of 7 days. Using convenient and snow balling sampling methods, twelve participants (app testers) were individually contacted and provided with information about My Care Hub and the ultimate aim of its development. The primary inclusion criterion was access to an android phone since the app was developed on an android platform only. Specific tasks required of the app testers included app download and registration; daily log-ins of random numbers into the documentation features (no limit was set to the amount of data to be logged into the app) and a read through the educational information about DSM embedded in the app. Tasks further include observation of the automated feedback messages in response to each logged BG, examination of the graphical outputs of all the documentation features and general browsing of the app. Any data crash or lag time in the app response during

launching and usage, screen "swiping", using the slide out keyboard and apps touch screen buttons were also monitored. Each tester was contacted via email and provided with a unique user-name, password and app download instruction. Only testers who signed into and used the app were provided with a link to the online survey questionnaire items adapted from the mobile app rating scale (Stoyanov et al., 2015) (see **Appendix 5.1**). Testers were asked to rate the app's functionality (performance, ease of use, navigation, gestural design) and aesthetics (layout, graphics, visual appeal of the analytic display). In addition, open-text-comment boxes were provided to gather information related to any concern or observations during the testing period. The results from this first stage of usability testing was used to improve the app prototype before the second stage was completed.

2) The second stage of testing recruited participants who had diabetes from the diabetes center of a tertiary hospital in Queensland, Australia, and through snow balling. Participants were asked to use My Care Hub and provide feedback using a questionnaire. In addition to functionality and aesthetics of the app, they also provided feedback on their satisfaction using measures that included perceived usefulness of the app to motivate participation in, and increase awareness of, diabetes self-management, intention to use, perceived ease of use and accuracy of the educational components of the app. Furthermore, they were asked if they would recommend the app to people with type 1 or type 2 diabetes and give an overall rating of 1-5 with "1= one of the worst apps I have ever used" and "5 = one of the best apps I have ever used".

Results

Demographic and health characteristics

Of the 12 testers without diabetes, only eight (8) signed into the app. In the second stage of testing, 6 individuals with type 1 or type 2 agreed to test the app, but only 4 signed in. Time

constraint was cited by the 6 people who withdrew from the testing as reason for nonparticipation. Feedback was provided by the remaining 12 participants (8 non diabetes + 4 with diabetes) who downloaded and used the app. The mean age of participants was 43.08 ± 14.02 years (range 28-76 years) and 58% of them were women. More than 75% were married and had obtained an educational level of first degree or higher. Of the four participants who have diabetes, three of them had type 2, one of them was diagnosed less than 5 years ago, while the remaining 3 had been diagnosed over 5 years. None of them reported hypoglycemia unawareness. Three of the four participants reported that their recommended range of fasting blood glucose was 4 to 7 mmol/L, with a postprandial of 5 to 9 mmol/L. The remaining one participant who had type 2 diabetes did not provide any information on this.

Usability result

In stage one of testing, all participants but one, were able to easily learn the use of the app following instructions provided via email. This one participant expressed an initial difficulty in operating the app by providing free text comment in the survey. Locating and learning how to use the documentation features took time, hence the suggestion to include a video recording of the instructions in addition to the already developed instruction manual for future users. Although, it was only one participant that indicated difficulty in using the app, it meant 13% of the participants had potential usability issues in relation to layout and ease of use. Therefore, based on the results of the first stage usability testing, minor modifications were made. In the prototype, accessing the documentation features required users to tap on a menu bar located on the top left corner of the app's homepage (which may not be apparent to users). The current version provides an improvement with an additional access via the app's homepage through direct taping on the icons of the documentation features.

In the second stage of testing using individuals with type 1 or type 2 diabetes, all the four participants were able to use the app easily. Across all participants (12), majority were satisfied with the performance of the app in terms of how fast the app features and components work (50% chose "perfect/timely response", 41.7% "mostly functional", 8.3% "app work overall, slow at times"). Several participants were able to navigate between app screens and features easily (66.7% "perfectly logical and clear screen flow throughout", 25% "easy to use") and felt that interactions across all tabs in the app were consistent and spontaneous (41.7% "perfectly consistent and spontaneous", 50% "mostly consistent and spontaneous"). Many participants were satisfied with the arrangement and size of icons/contents in the app (66.7% "mostly clear", 33.4% "professional, simple, clear and logically organised") and the quality of graphics was high/very high. With regards to the general visual appeal of the app and the analytic feature, 83.3% participants chose very high/high level of visual appeal for both domains. The app was given a five and four-star rating by 8.3% and 91.7% respondents respectively.

Among the four participants with diabetes, 75% of them strongly agreed that they saw value in the educational content of the app as it was relevant and likely to raise awareness of the importance of DSM. 75% strongly agreed that the app is likely to increase motivation of people with diabetes to engage in self-management activities. Furthermore, 75% noted that they would recommend My Care Hub to people with type 1 or type 2 diabetes and that they can continue to use it if granted continual access to the app.

5.3 DISCUSSION

This paper follows the principle of intervention development study report which details "the rationale, decision making processes, methods and findings from the inception of the intervention to the usability testing prior to full trial or evaluation" (Hoddinott, 2015). To the

best of the authors' knowledge, My Care Hub is arguably the first DSM app aimed at Australian population with type 1 or type 2 diabetes to have reported / documented a systematic and transparent approach to its development based on empirical and theoretical framework of health behavior change theories, involvement of users and clinical experts, data security and privacy considerations (Adu et al., 2018a).

Behavioural theory is critical to the development of health behavioural change interventions (Campbell et al., 2007) because interventions grounded in theory are more effective at modifying behavior (Michie et al., 2008; Craig et al., 2008). Health behavioural theories predict how applying an intervention will drive change in underlying behavioural mechanisms or technique (mediating construct) that will in turn, drive behavioural change (output) (Michie et al., 2008). Majority of currently available diabetes app are lacking in health behavioural content (Cowan et al., 2013; Adu et al., 2018a), which may be an early indication of the low potential of such apps to influence behavioural change: Social cognitive theory (Bandura, 1998) and Information-Motivation-Behavioural Skills model (Fisher, Fisher & Harman, 2003; Fisher et al., 2011). Diverse constructs were employed within these theories; hence, giving My Care Hub the potential for effectiveness when eventually used as an intervention in trials.

Specific features and educational content in My Care Hub were chosen based on users' needs and preferences (Adu et al., 2018b). This is an added strength of the app which is often lacking in many health apps that are developed without considering the needs of the end users or guidelines for the management of such diseases (Chomutare et al., 2011; Holtz & Lauckner, 2012). Development of an intervention to meet the actual needs and demands of targeted users assures the feasibility of the product (International Organisation for Standardisation, 2010). Previous reports show that only few diabetes mobile apps incorporate elements of clinical best practices established by diabetes professionals (Bradell & Ford, 2013). Blood glucose level guidelines recommended by Diabetes Australia were consulted during the development of feedback messages provided in My Care Hub in order to ensure clinically sound information is provided to the patients. Providing relevant feedback is an important strategy to stimulate reflection in patients about their blood glucose goal and to engage them in healthy behaviours necessary for optimal health outcomes (Lie et al., 2018).

Data privacy and security in mobile health interventions are vital to intervention development, relevance and acceptance of such technology (Wilkowska & Ziefle, 2012), therefore, health apps must be protected from security breaches. This was ensured in My Care Hub. The multi-stage usability evaluation method applied to My Care Hub have better propensity to capture the complete usability of a technology in comparison to a single method (Georgsson & Staggers, 2016; Cho et al., 2018). First stage of testing revealed important changes to the app design in order to improve the ease of access to user interface. Similar to our findings, participants in studies assessing the usability of mobile apps for DSM had commented positively on the performance and ease of navigation of the app (Georgsson & Staggers, 2016; Booth et al., 2016). Furthermore, good level of satisfaction reported by our participants in relation to the app graphics, layout and visual appeal has also been reported in previous studies (Georgsson & Staggers, 2016; Schmocker, Zwahlen & Denecke, 2018). End users saw value in the app as a tool to support DSM and expressed interest in continuing to use the app in the future.

Strengths and limitations

The strengths of this study include a full description of the development process of My Care Hub - an app aimed at the Australian population with type 1 or type 2 diabetes as opposed to the reports of only efficacy and effectiveness of diabetes apps within RCTs in the published literature (Adu et al., 2018a). In our systematic review of previous RCTs using a diabetes app
as an intervention, none of the studies provided detailed information about their developmental process (Adu et al., 2018a). The need for full description of the intervention development strategy, including detailed explanation of how important factors of health behavioural theories, users and clinical expert involvement, data security and privacy were used to guide the development process has been met in our current study. Not only is the use of two distinct approaches to usability testing unique in providing effective development guide and strategies for future apps, the results from both phases also allow for identification of pertinent issues from a general and targeted populace of app users. Usability testing of the app at this stage and not later during feasibility testing is essential for identifying improvement needs prior to full studies to secure future engagement with My Care Hub by patients with diabetes.

A limitation of this study is that the participants in the usability study may not be representative of the target population due to low sample size. Nonetheless, studies have shown that large sample sizes are not required in usability studies, as they do not propose inferential results. Therefore, the number of participants involved in the usability testing of My Care Hub in this study, may have been sufficient to identify usability issues in the app, which may occur under conditions of regular use (Macefield, 2009). Another limitation is that only some of those who initially agreed to test the app eventually did so. This may have introduced selection bias wherein only participants interested in the use of an app downloaded and signed into it and provided feedback. It is possible that we could have drawn different conclusion if all the initial participants had used the app. However, in reality, research outcomes are not free of either opinion or bias because they are highly subjective (Six & Macefield, 2016). In addition, although, the app needs analysis results indicated that patients will like an app that reminds them of their self-management activities, currently My Care Hub app does not have this feature. This is because upon prioritisation and consideration, we felt inclusion of a reminder feature might further increase the complexity of the app. In its' present version, the app entails a

variety of features, therefore an addition of reminder feature (which ideally will be needed for each of the diabetes specific tasks) will increase the app's complexity. Studies have reported a significantly negative correlation between number of app features and ease of use (Arnhold, Qualde & Kirch, 2014). Nonetheless, upon ensuring that the app remains user friendly, inclusion of a reminder feature is a possible consideration for our future research. Lastly, given that no data was collected about the specifics of participants' android smartphone configurations, we were unable to assess the influence of such configuration on their responses to My Care Hub usability testing.

Future research

Future research includes pilot testing of My Care Hub as a key step for optimising an intervention before its evaluation in full trials (Eldridge et al., 2016). This pilot testing will provide valuable information on users' engagement, retention, acceptability and preliminary efficacy (Feeley et al., 2009) as well as the development of future versions of the app (if required) and design of future trials (Day, Bench & Griffiths, 2015).

My Care Hub is currently only available for Australians on Android operating system. The decision to make the app Australia-specific was based on the differences in the guidelines for recommended blood glucose levels between countries. There was the need to focus resources on developing a native app that could as much as possible provide a better user experience instead of including additional country specific requirements. My Care Hub presents a proof of concept and can be developed for other nations in order to reach a wider proportion of users, but there will be the need to modify the Australia-specific features of the app before it could be used elsewhere.

5.4 CONCLUSION

Detailed explanation of My Care Hub development was provided for future researchers to learn and understand the development process. My Care Hub has the potential to benefit Australians with types 1 and 2 diabetes in monitoring self-management activities. The app provides easy access to educational information and supporting tools, which could enhance knowledge, skills, self-efficacy and motivate patients to perform self-management activities to improve glycaemic control. An app such as My Care Hub developed based on the needs and preferences of its intended users maximises the potential to enhance self-management.

Essentially, this chapter addressed the fourth aim of this thesis by providing a full description of the novel app (My Care Hub) development. Usability testing results showed the app is userfriendly, intuitive and navigation was simple and easy to use. These results informed the final version of the app in the pilot testing stage (Chapters 6 and 7).

Chapter SIX: User Retention and Engagement with a Mobile App Intervention to Support Self-Management in Australians with Type 1 or

Type 2 Diabetes (My Care Hub): Mixed Methods Study

6.1 Abstract

Mobile health apps are commonly used to support diabetes self-management (DSM). However, there is limited research assessing whether such apps are able to meet the basic requirements of retaining and engaging users. This study aimed to evaluate participants' retention and engagement with My Care Hub, a mobile app for DSM. The study employed an explanatory mixed methods design. Participants were people with type 1 or type 2 diabetes who used the health app intervention for 3 weeks. Retention was measured by completion of the postintervention survey. Engagement was measured using system log indices and interviews. Retention and system log indices were presented using descriptive statistics. Transcripts were analysed using content analysis to develop themes interpreted according to the behavioural intervention technology theory. Of the 50 individuals enrolled, 42 (84%) adhered to the study protocol. System usage data showed multiple and frequent interactions with the app by most of the enrolled participants (42/50, 84%). Two-thirds of participants who inputted data during the first week returned to use the app after week 1 (36/42, 85%) and week 2 (30/42, 71%) of installation. Most daily used features were tracking of blood glucose (BG; 28/42, 68%) and accessing educational information (6/42, 13%). The interview results revealed the app's potential as a behavioural change intervention tool, particularly because it eased participants' self-care efforts and improved their engagement with DSM activities such as BG monitoring, physical exercise and healthy eating. Participants suggested additional functionalities such as extended access to historical analytic data, automated data transmission from the BG meter, and periodic update of meals and corresponding nutrients to further enhance engagement with

the app. The findings of this short-term intervention study suggested acceptable levels of participant retention and engagement with My Care Hub, indicating that it may be a promising tool for extending DSM support and education beyond the confines of a physical clinic.

6.2 INTRODUCTION

Mobile health (mHealth) apps offer a unique opportunity to deliver health promotion interventions to reach any population due to their ubiquitous nature (Payne et al., 2015; Zhao, Freeman & Li, 2016), with some developed specifically to support diabetes management (Chomutare et al., 2011; Arnhold, Quade & Kirch, 2014). However, these mHealth interventions suffer from low participant retention (Eysenbach, 2005; Kirwan et al., 2013) and non-usage attrition (Eysenbach, 2005; Kelders et al., 2012). Therefore, interventions that are more engaging are required to address these concerns (Short et al., 2015; Yardley et al., 2016) through user-centered and iterative approaches that integrate input from users and other relevant stakeholders in app design and development. This approach is necessary to provide interventions that meet user requirements and ensure greater retention, uptake, engagement, and sustainability (Arsand & Demiris, 2008; Yardley et al., 2016).

Retention

Inadequate participant retention is a major methodological challenge experienced by many mHealth app interventions (Murray et al., 2009). Low retention rates and lower statistical power threaten outcome validity (Eysenbach, 2005) and serve as a major reason for premature trial termination (Kasenda et al., 2014); hence, pilot studies are important before conducting large-scale studies. The evaluation of participant retention levels enables researchers to assess the relevancy and tendency for sustainable implementation of intervention ideas (Bowen et al., 2009). In addition, the assessment reveals any required research methodology modification in preparation for future large-scale research (Bowen et al., 2009).

Engagement

An effective mHealth intervention requires not only retention but also continuous and active engagement by users, as lack of engagement leads to study dropout and dampening of the treatment effect (Eysenbach, 2005; Murray et al., 2009). User engagement refers to interaction, experience, perceived usefulness, and desire to use the intervention repeatedly over a long period of time (Lalmas, O'Brien, Yom-Tov, 2014; O'Brien & Cairns, 2016). The degree to which users engage with a health app signifies their willingness to invest time, attention, and emotion into the use of the technology to satisfy and eventually achieve their pragmatic needs (such as self-management) (Lalmas, O'Brien & Yom-Tov, 2014). Measurement of users' engagement can be long- or short-term in nature with short-term measurement reflecting initial adoption of the intervention and the tendency of apps to successfully engage users in the long-term (Lalmas, O'Brien, Yom-Tov, 2014). Although, there are various approaches to measuring engagement with apps, system usage data and user-reported interactions with the system using specific techniques such as questionnaires and interviews are the most relevant in the context of short-term measurement (Lalmas, O'Brien, Yom-Tov, 2014; Wiebe et al., 2014; Yardley et al, 2016).

System usage is measured through the collection of non-invasive data on the frequency of access to the app, push notifications opened, and average time spent per usage (Burby, Brown & Commitee 2007; Perski et al., 2016). This provides information on user participation with specific target behaviours and frequency of access to the corresponding app features (Heerwegh, 2003; Stern, 2008). On the other hand, user-reported approaches reveal users' experiences related to behavioural engagement with the intervention (Lalmas, O'Brien, Yom-Tov, 2014; Weibe et al; 2014). This is necessary to assess intervention tendency to foster achievement of behavioural goals when used over a long period.

Behavioural Engagement Framework

Rate of use alone is not a sufficient indicator of engagement with a mHealth intervention (Yardley et al., 2016). There must also be an assessment of engagement with the behavioural goal of the intervention to ascertain the intervention's potential as an effective tool to support behavioural change. One possible way to achieve this is by assessing users' engagement with the process of achieving behavioural change. Behavioural change is fostered by intervention components that motivate users to achieve a behavioural goal [in this case, diabetes self-management (DSM) behaviours (Yardley et al., 2016)]. Assessing engagement in behavioural change process requires the use of models and frameworks that reveal the relationship between factors in a system for the realisation of a defined (Cole-Lewis, Ezeanochie & Turgiss, 2019).

Within the field of mHealth engagement, models and frameworks provide a richer understanding of the core components that influence user engagement to achieve the behavioural goal of the intervention (Cole-Lewis, Ezeanochie & Turgiss, 2019). The concept of behavioural engagement is complex and includes the extent to which users interact with the intervention. Major considerations include the quality of users' experience with the technology (O'Brien & Tom 2008), and if they have engaged with it as needed (Kelders et al., 2012) or as intended (O'Brien & Tom 2008). The behavioural intervention technology (BIT) model by Mohr et al., (2014) describes the full range of components that must be available in a technology to influence engagement with behavioural change and its potential as an effective intervention to attain a behavioural goal.

The BIT framework (Mohr et al., 2014) was utilised in this study as it describes the theoretical components necessary in the conceptualisation of mHealth and instantiates the necessary components for its implementation. The theoretical level covers the overall goal (*why*) or reason for mHealth development and *how* specific aims related to the goal could be achieved through the required *behavioural change strategies*. Each strategy is instantiated by *elements*:

features (*what*) available in the intervention. In addition, the *characteristics (technic)* of the intervention affect *how* an element is displayed to the users as well as their perception about the intervention. Finally, the *workflow (pattern of use)* describes *when* and under what conditions BIT interventions will be delivered. Therefore, the BIT model explains that achieving an intervention goal is fostered through relationships between the components *of aims, behavioral change strategies, elements, characteristics, and pattern of use* of the intervention (Mohr et al., 2014). We used this model to interpret our qualitative findings, allowing for an open approach to the concept of behavioural engagement, focusing on exploring the tendency of My Care Hub as an intervention tool for diabetes behavioural engagement.

Study Context and Objectives

Owing to poor retention and engagement with previous diabetes apps, we performed an initial study to explore user needs and preferences to foster engagement with a diabetes app (Adu et al., 2018b), which was used to develop a new app called My Care Hub (Adu et al., 2020a). Patients with diabetes who interacted with a prototype of My Care Hub reported that it was easy to use and that the educational contents were valuable in raising awareness about the importance of DSM and increased motivation to engage in self-management activities (Adu et al., 2020a). Although the usability of the app was satisfactory, it was unclear if My Care Hub has the potential to retain and engage users and if its components meet the requirements of a supporting tool to foster engagement with DSM.

Therefore, this study aimed to examine levels of user retention and engagement with My Care Hub in a short-term single-arm pilot trial. Retention was measured through completion of follow-up surveys, and engagement with the app was assessed in 2 areas: (1) system usage data and (2) qualitative feedback from users on behavioural interactions with the intervention. We expect that the app's contents and features (Adu et al., 2020a), which were developed based on results from our previous study on users' needs (Adu et al., 2018b; 2019) would result in high participant retention and greater engagement during the short trial period. Understanding these factors is critical in identifying areas where intervention design may need improvement and inform plans for future trials of mHealth interventions such as My Care Hub.

6.3 METHODS

This study received ethics approval from the James Cook University Human Research Ethics Committee (reference #H7716). Participants were informed about the study aims, and consent was implied by survey submission. Verbal consent was obtained for telephone interviews.

6.3.1 Study Design and Sample Size

This study utilised a sequential explanatory mixed methods design with quantitative surveys and qualitative interviews. This design captures both the engagement with technology and the process of behavioural change by triangulating the results of multiple measures (O'Brien, Cairns & Hall, 2018). This provides information about how users react to the contents and design of the intervention and offers an explanation for why users interact with the intervention in a particular way (Yardley et al., 2016). This study was conducted from August to October 2019, where each participant was given 3-week access to the app. Following this period, participants filled out a survey and were invited to participate in a telephone interview to better understand their interaction with the app.

The study used a maximum variation purposive sampling tailored to recruit participants who showed interest in the study within the time available. This sampling method is appropriate for an implementation feasibility assessment as related to this study (Leon, Davis & Kraemer, 2011). The components of the pilot testing that relate to retention and engagement with the app are presented in this paper.

6.3.2 Recruitment and Eligibility

Participants were recruited through a single invitation email sent to patients registered with the Australian National Diabetes Service Scheme. Email invitations were limited to patients who have type 1 or type 2 diabetes and live in North Queensland, Australia. North Queensland has a relatively high prevalence of diabetes (Queensland Health Report, 2018) and socioeconomic disadvantage, which can affect accessibility to regular diabetes support services (Australian Institute of Health and Welfare, 2020). Therefore, the use of mHealth interventions to provide DSM support may be essential among this population. Other eligibility criteria included ownership of an Android-operating smartphone, having a current recommended blood glucose level (BGL) target of 4 to 10 mmol/L (Diabetes Australia, 2015), and being aged 18 to 65 years. The upper age limit was chosen because of the less stringent glycaemic recommendations for many older adults who are above 65 years. Patients were excluded if pregnant or currently using an app with an educational component to support their diabetes management.

6.3.4 Enrolment and App Orientation

Participants enrolled through the Web by completing an eligibility screening form, providing consent, and completing the baseline survey, which entailed questions regarding socio-health demographics, email address, and residential postcode. Participants were emailed a unique code to enable them to download My Care Hub from Google Play store of any android-powered phone, an app manual, and a 5-min video explaining how to install the app, features, and functionalities. Participants could contact MDA for assistance with technical difficulties or for study clarification. It was emphasised that there was no limit to the frequency of use of My Care Hub as participants could engage with it at a level they considered useful and desired. My Care Hub is intended to be a stand-alone intervention. Therefore, push notifications (aimed at improving patients' awareness about diabetes distress and potential ways to reduce its impact on their self-management) were sent from the app during the first 2 weeks of the intervention

and withheld in the third week to see the achievable level of engagement with the app with or without push notifications. Throughout the study period, no log-in reminders or calls were made from the study researchers to participants.

6.3.5 Intervention Overview

A detailed description of the development of My Care Hub and the methods of usability studies have been previously published (Adu et al., 2020a). In brief, the goal of My Care Hub is to provide support and education that facilitate positive behavioural change in DSM. The app was specifically designed for type 1 diabetes patients with standard Australian BGL recommendations of 4 to 8 mmol/L for fasting and <10 mmol/L 2-hour postprandial, and for type 2 diabetes patients with recommended fasting BGL of 6 to 8 mmol/L and 2-hour postprandial levels of 6 to 10 mmol/L. The app incorporates multiple functions and features to foster engagement with the app within 3 broad categories: *documentation, analytics,* and *education.*

In *documentation features*, users can manually input data for tracking BGL, physical activity, the carbohydrate content of foods eaten, and body weight. *Analytic features* provided a graphical output of each documentation feature, thus offering users the ability to visually inspect their logged data over time. *Education* was provided through four main features. First, users can review a variety of actionable textual information related to healthy food choices, self-monitoring of BGL, medication, reducing risk, healthy coping, problem solving, and physical activities. Second, users can look up information related to carbohydrate and calorie content of common foods in Australia (categorised under fruits and vegetables, egg and meat, dairy, grain and legumes). Third, the BGL feature provided immediate tailored feedback to every inputted data, driven by a decision-based system. The system is controlled by the value of logged BGL (either within or beyond the standard range), type of diabetes, and the indicated period of BGL measurement (either fasting or 2 hours postprandial). Messages were health-

promoting and motivational information aimed at supporting behavioural skills building for self-management practices. Finally, the app provided education through daily push notifications aimed at improving awareness about diabetes distress and encouraging patients to focus on potential ways to reduce its impact on their self-management. Push notifications were terminated at the end of the second week. Sample screenshots are provided in **Appendix 6.1**.

6.3.6. Post-Intervention Data Collection

At the end of the study, participants were sent an email (with 1 reminder email sent to noncompleters), which directed them to the post-intervention survey on the acceptability of the app and its preliminary efficacy (*results will be reported in future publications*). Through this survey, participants were also invited to participate in individual telephone interviews to further understand their perception of the app. Participants who completed the post-intervention survey were awarded an electronic gift (e-gift) card worth Aus \$40 (US \$25.07). All telephone interviewees were contacted within 3 weeks of completing the survey and awarded an additional Aus \$20 (US \$12.53) e-gift card.

6.3.7 Measures

Retention was assessed using the following indicators of study completion per protocol: number of participants enrolled, number of participants who used the app during the intervention period, and completion rate of the post-intervention survey.

Engagement with My Care Hub was measured using participants' app usage log and verbal feedback. App usage data were extracted from the app's activity database. The following time frames were considered: (1) date of log-in into the app to 2 weeks of use when the daily push notification was administered (referred to as week 1 and week 2) and (2) data during the third week (referred to as week 3) after the termination of push notifications. Key metrics collected from the database included app use (number of active users, frequency of daily access to app),

data logs/time spent (for BGL, exercise, food activity, and weight), and number of opened notifications. Metrics were presented using an adapted version of the Frequency, Intensity, Time, and Type (FITT) principle index (Barisic, Leatherdale & Kreiger, 2011; Short et al., 2018). This index explores multidimensional domains of usage data, which provides greater insight into interaction with a mHealth app. Event count in the app was available for active (documentation features) and passive (viewing of educational screen) app features. Data had to be logged/saved in the documentation features before it could be counted as an active event as the app discarded data not logged after 30 min of inactivity. Users had to exit from an educational screen before it was counted as a passive event, and no maximum count per user was stipulated. This implies that the total count of passive events could be higher if a screen was viewed more than once. The FITT index used in this study is as follows:

- 1. Frequency index (F_i): This subindex is an attention proxy that provides information on how often a participant uses the app. It recognises the number of users who return to use the app and active app users in each time period.
- 2. Intensity index (I_i): This subindex denotes the proportion of users who interact with each feature in the app. In total, 2 metrics were used in the assessment of I_i . These are the frequency of daily use of app features (I_{i1}) and number of push notifications opened versus the total sent (n=14) in 2 weeks (I_{i2}). In addition, intensity also measures the proportion of app features used out of the total available features.
- 3. Type index (T_y) : This provides information on the form of engagement based on actions performed by users using the available app features. In this study, the type of action was categorised as *active* denoted as T_{ya} (use of documentation features for selfmonitoring), and *passive* (T_{yp}) , reading information on educational contents in the app).
- 4. Time index (T_i) : This measures the duration of engagement, which signifies attention to the app as a function of daily event duration with each app feature.

6.3.8 Interviews

Interviews were conducted using a semi-structured interview guide (see **Appendix 6.2**) that explored behavioural engagement with the app through questions on patterns of use, perceived ease of use, perceived usefulness of app features enabling motivation for continued engagement with DSM, and recommendations on how the app could be improved.

Interviews were conducted by AD, who is well experienced in qualitative research. The interview guide was pilot tested between MDA and AD before actual use. The interviewer was located in a private office at James Cook University, Australia, while participants were asked if they were in a comfortable location before commencement of the interview. The first 3 interviews were used to reflect on the guide, although there were no resultant changes. Data saturation was achieved as judged by no emerging new information (Guest, Bunce & Johnson, 2006) after completing the 15th (of 17) interview. Interviews were audio recorded, and none of the participants had a previous relationship with any of the authors.

6.3.9 Data Analysis

Descriptive statistics were calculated for all quantitative variables. Baseline characteristics comparison between those who completed the study and those who did not were done using a Pearson chi-square test. All statistical analyses were performed using SPSS version 23.

Interviews were completed in an average of 15 min (range 9-30 min). Participant responses were transcribed verbatim by 1 researcher (AD). In this analysis, a combination of data and a concept-driven strategy was applied. Initially, inspired by the work of Schreier (2012), MDA and AD independently used a data-driven strategy to obtain an overview of the data, and then similar text segments were selected and sorted using coding. Coded segments were grouped to identify recurring themes from the data. Themes were compared between the 2 authors, discussed with BMA and agreement was reached about the main themes. Subsequently, the

themes were analysed by applying a concept-driven strategy in accordance with the BIT framework (Mohr et al., 2014) to assess behavioural engagement with the intervention. The BIT components in the My Care Hub app that could potentially enhance behavioural engagement were identified ad described. These components overlap and diverge within the identified themes, which are presented using representative quotes affixed with an assigned number code and the type of diabetes the respondent has (for instance, respondent 3 with type 1 diabetes; P003, T1D and respondent 4 with type 2 diabetes; P004, T2D). The conduct and reporting of the interviews followed the consolidated criteria for reporting qualitative research (Tong, Sainsbury & Craig, 2007).

6.4 RESULTS

6.4.1 Participant Characteristics

Participant demographics and health characteristics are shown in **Table 6.1**. Participants were predominantly male (31/50, 62%), had type 2 diabetes (36/50, 76%), and aged between 20 and 64 years (mean 49.12, SD 12.34 years). On average, the recommended BGL in enrolees was as follows: for fasting, 4.58 (SD 0.78; range 4-6 mmol/L), and for 2-hour postprandial, 7.01 (SD 1.02; range 6-10 mmol/L). Most participants were diagnosed as having diabetes in the last 5 years (27/50, 54%), and an equal proportion rated their health status as being fair or good (20/50, 40%). Most had a technical college education or higher (39/50, 78%) and were employed (31/50, 62%). Only a few had previously used a health app to manage diabetes in the past (16/50, 32%). The linking of participants' postcode to the Australian Standard Geographical Classification System (Australian Government Department of Agriculture, 2019a) indicates the geographic location of the majority to be rural (37/50, 74%).

Of the 22 participants who indicated an interest in participating in the interview, only 17 were contactable within 3 call attempts. Most were males (12/17, 71%), had type 2 diabetes (13/17, 71%)

77%), and had been diagnosed for an average of 6 years (range 1-17 years). Overall, participants were between the ages of 36 to 64 years (mean 51.58, SD 11.31), except for one who was aged 20 years.

6.4.2 Retention

Of the 4984 patients who were emailed an invitation to participate in the study, 79 (1.59%) completed the eligibility form. Low response rate may be due to lack of access to an android phone to download the app, which is the main eligibility criterion as stated in the email invitation. Subsequently, only 84% (67/79) of those who responded met all the inclusion criteria and were provided access to download the app. Some participants (17/67, 25% of those eligible) failed to log in to the app, resulting in 50 enrolled participants (75% of eligible participants). Most enrolees (43/50, 86%) activated the app within the same day (range 0-5 days) of having access to it. One participant logged out of the app on the second day of installation stating that it did not meet her requirement. At the end of the study period, 41 of the enrolled participants completed the study per protocol by providing feedback about the app using the post-intervention survey (retention rate: 41/50, 82%). Reasons for non-completion of the study protocol were not recorded. In assessing baseline characteristics associated with retention, only employment status emerged as a significant predictor, with those unemployed being less likely to complete the study than those who were employed (50.0% versus 14.7%, respectively; P=.02). The full details of the demographic variables and comparison between those who completed the study and those who did not are shown in **Table 6.1.**

	Baseline $(n - 50)$	Completers	Loss to follow up $(n-0)$	
Characteristics	(n = 50)	(n=41)	(n=9)	
		Frequency (%	Frequency (% within	P
		within group)	group)	value
Gender	21	25 (01)		.75
Male	31	25 (81)	6 (19.4)	
Female	19	16 (84)	3 (15.8)	
Age (Years)				.82
Mean±SD		49.29±12.74	48.67±11.25	
18-29	5	4 (80)	1 (20)	
30-39	6	5 (83)	1 (17)	
40-49	12	10 (83)	2 (17)	
50-59	15	11 (73)	4 (27)	
60-65	12	11 (92)	1 (8)	
Type of diabetes				.81
Type 1	15	12 (80)	3 (20)	
Type 2	35	29 (83)	6 (17)	
Type 2 medications or not (n=35)				.32
None	2	1 (50)	1 (50)	
Oral drugs alone	33	28 (85)	5 (15)	
Oral and insulin	1	1 (100)	0 (0)	
Duration of diagnosis (years)				.92
≤ 5	27	23 (85)	4 (15)	
6-10	10	8 (80)	2 (20)	
11-15	9	6 (67)	3 (33)	
≥16	4	4 (100)	0 (0)	
Education		× ,		.59
High school equivalent	17	12 (71)	5 (29)	
Technical college	10	9 (90)	1 (10)	
First degree	11	10 (91)	1 (9)	
Post graduate	8	7 (88)	1 (13)	
Missing	4	3 (75)	1 (25)	
Ethnicity		- ()	- ()	.87
Caucasian/White	47	38 (81)	9 (19)	
Missing	3	3 (100)	0 (0)	
0	0	2 (100)	· (•)	.02 ^a
Employment	8	4 (50)	4 (50)	.02
Unemployed		4 (50)	4 (50) 5 (15)	
Part/Full time Employed	34	29 (85)	5 (15)	
Retired	8	8 (100)	0 (0)	26
Living Environment	10	12 (02)	1 (0)	.26
Remote	13	12 (92)	1 (8)	
Rural	37	29 (78)	8 (22)	
Usage of Smart Phone (Years)	10	11 (05)	0 (15)	.42
1-5	13	11 (85)	2 (15)	
6-10	28	24 (86)	4 (14)	
> 10	9	6 (67)	3 (33)	
Previous use of health apps to				
manage diabetes				.93

Table 6.1: Participant Characteristics

16 34	13 (81) 28 (82)	3 (19) 6 (18)	
			.38
1	1 (100)	0 (0)	
19	14 (74)	5 (26)	
21	17 (81)	4 (19)	
9	9 (100)	0 (0.)	
	34 1 19 21	34 28 (82) 1 1 (100) 19 14 (74) 21 17 (81)	34 28 (82) 6 (18) 1 1 (100) 0 (0) 19 14 (74) 5 (26) 21 17 (81) 4 (19)

 $^{a}P < .05$

6.4.3 App Engagement

Most (42/50, 84%) enrolled participants logged data into the app at least once (during week 1 of installation) with the frequency index showing that they actively used the app on an average of 11 of the 14 days in the first 2 weeks when push notifications were sent (range 2-14 days; week 1 average: 5.2 days, week 2 average: 4.8 days). This reduced to an average of 4 of 7 days (range 2-5) in week 3: average 3.8 days. Furthermore, all participants who logged in to the app used it during week 1, and most returned to use the app after week 1 (36/42, 85%) and week 2 (30/42, 71%) of installation. With regard to the intensity index related to daily use of each app feature (I_{il}) , most participants used features for tracking their BGL (28/42, 68%) and accessing educational information (6/42, 13%) more frequently. The feature with the least daily use was tracking the carbohydrate content of foods (2/42, 2%). All 14 push notification messages during the first 2 weeks (1 per day) sent were published, and on average, 57% (24/42) of participants opened this notification within 24 hours, after which they were automatically deleted. None of the app features were unused. The type index (T_y) shows active and passive actions with the My Care Hub. The average frequency of BGL data log per participant in week 1 was 10.85 (SD 9.32; range 1-36), which reduced to 6.75 (SD 7.75; range 1-24) in week 2 and 5.67 (SD 6.05; range 0-22) in week 3. Physical activity logs showed a mean of 4.48 (SD 3.64; range 0-15) in week 1 compared with 2.97 (SD 2.93; range 0-11) in week 2 and 1.69 (SD 1.70; range 0-7) in week 3. Average passive engagement per participant on occasions of viewing screens alone in week 1 was 26.5 (SD 2.51; range: 9-32), 17.55 (SD 7.39; range 7-26) in week 2, and

14.4 (SD 6.13; range 6-24) in week 3. The time index (T_i) revealed that, for all events of participants' visit to the app, an average daily time of 3.56 min (range 1.37-7.48 min) was spent. More time was spent on BGL activity (2.2 min) and accessing the educational tips embedded in the app (1.35 min). **Table 6.2** summarises the app functions and features, their purposes, usage and engagement.

			User Engagement	
Functions/Features	Elements	Purpose	Percentage of daily users (I _{i1}) ^a	Average time spent per user per day
Documentation			n (%)	(T _i) ^c
Blood glucose activity $(T_{ya})^d$	 Blood glucose log Type of blood glucose Automatic Feedback (as part of education) 	 Monitoring and tracking of blood glucose values over time Gain knowledge to support self-management practices 	29 (69)	2m 2s
Physical activity $(T_{ya})^d$	 log of time spent on physical activity Calories used Place	• Monitoring of physical activity behaviour over time	4 (10)	0m 7s
Food activity $(T_{ya})^d$	 Record food intake Log of carbohydrate content of food 	• Monitoring and tracking of food intake and their carbohydrate content over time	1 (2)	0m 17s
Weight log $(T_{ya})^d$	• Body weight log	• Body weight assessment over time	2 (5)	0m 22s
Analytics (T _{yb}) ^e	• Graphical display of data log into each documentation feature	• Keeping track of trends in lifestyle activities and observe impact on blood glucose levels over time	3 (6)	0m 20s

Education

Textual screens for management tips and food choices $(T_{yb})^e$	 Information on behaviours in DM management Information on average carb and calorie content of common Australian foods 	 Assess current knowledge on diabetes self-management Review carbohydrate content of foods in order to make healthy choices. 	6 (13)	1m 35s
Push notifications $(T_{yb})^e$ and $(I_{i2})^b$	• Messages on diabetes distress	• Create awareness about diabetes distress and ways to reduce it impact on self-management	24 (57)	Not captured ^f

^aIntensity index for frequency of daily use

^bIntensity index for number of push notifications opened

^cTime index

^dType index for active app use

^eType index for passive app use

^fNot captured due to limitations of system usage tracking data base

6.4.4 Interview Results

Different themes emerged from the data with interconnection among the themes over the course of My Care Hub usage. Overall, the results suggest that the use of the app has the potential to ease the effort in aiming for improved self-management and for better awareness of BGLs. In addition, participants provided their recommendations for extra functionalities that may further enhance engagement with self-management behaviours. We present our findings in relation to themes related to components of the behavioural intervention model (Mohr et al., 2014) used for this study, which are outlined in **Table 6.3**.

	BIT ^a Components		Details in MCH ^b	
Theoretical	Why	Broader goal: Self- management education and support	aims:	
			Improved blood glucose - long term impact Increase physical activity Healthy eating Decreased diabetes stress	
		Behavioural change		
	How	strategies	Elements/Strategies Documentation/Analytics: Accountability Clarity of self-management activities and impact	
			Improved awareness of BG ^c levels Mindfulness of calorie consumption	
			Feedback response: Reinforced HP's ^d recommendation Informative	
			Carbs in Foods: Guidance on meal planning	
			Knowledge provision and reinforcement Educational Tips: Knowledge reinforcement	

 Table 6.3: Summary of Behavioural Intervention Technic (BIT) model as adapted to My Care Hub

 (MCH) Intervention

Instantiation			
	What	Elements (app features)	Documentation (Logs) / Analytics
			Feedback response
			Carbs in Foods
			Educational Tips screen
			Push notifications
	How		
	(Technic)	Characteristics	Aesthetic:
			Beautiful
			Ease of use:
			Simple and straight forward
			Few difficulties
			User defined:
	When	Pattern of use	Type of diabetes
			Established self-management routines
			Frequency:
			Daily
			Partly with reasons
^a BIT: behabioura		technology	

^bMCH: My Care Hub ^cBG: Blood Glucose

^dHP: Health Provider

Pattern of use ('When')

User defined

Patterns of app use depended on users' type of diabetes and self-management routines, with most participants using the app multiple times per day, where those with type 1 diabetes input their BGL any time it was measured:

"I use it multiple times per day, basically any BGL I took I enter it at any time I took it." [P001, T1D]

In contrast, participants with type 2 diabetes described that the frequency of usage depends on the self-management activity carried out on that day.

"I used it at least once a day. if I had done exercise, then I was putting in an exercise and blood test virtually every day. On every second day I was using it to stick in weight but the exercise was done at a different time." [P005, T2D] Conversely, some participants were only able to use the app infrequently because of issues such as limited internet access or multiple competing interests:

"I didn't use it fully, because at the moment I am having a problem with my internet, so I didn't get a chance to watch the video that comes with it." [P012, T2D].

"I used it a few times to start with, but then I stopped pretty much because I was juggling between doing a lot of writing, doing a course, and was having other things to do." [P003, T1D]

Characteristics of the App (How)

Simple and straightforward

Participants described the design as:

"Very well crafted and well put together, really easy to use." [P007, T2D],

and could be used even by the elderly who may not be too proficient in using mobile technology:

"I would even say that like an older person in their 60s or so, once they get an idea of how to use it properly, would have no worries using it if they were in that way inclined." [P012, T2D]

App difficulties

Some participants found a few aspects of the app difficult:

"There was one for the activities you had to put in what calories you might have burned off and I didn't have a clue how I was going to find out that information. [P013, T2D] I had a problem figuring out how to put dates in it, but I think it does it itself, so yeah." [P008, T2D]

Goal (Why), Elements (App Features; What), and Behavioural Change Strategies (How)

The goal of developing My Care Hub was to enhance engagement with DSM activities such as improved BG, increased participation in physical exercise, and healthy eating. Participants identified multiple elements (features) that support this overall goal. They also described the perceived benefits (mechanism of action) of each of the elements that encouraged their interaction with it, and toward achieving an improved DSM. The commonly mentioned features are noted below, as well as reasons why participants found the features engaging.

Documentation/Analytics

Accountability

Participants mentioned that the documentation element strengthened the sense of responsibility to keep up with routines in DSM:

"I liked the activity log, because it gives you accountability, when did you go to the gym, how long were you there, what did you do." [P014, T2D]

Clarity of self-management activities and impact

Participants explained that visualisation of logged data using *analytics* encouraged their interaction with My Care Hub. They noted that the feature provides better clarity on their level of self-care:

"Just the tracking of my fitness, exercise and my blood sugars, it is much better for me seeing it in a graph, makes it really clear how you are going." [P006, T2D]

The feature also hinted at some participants to consult their physician for medication review or consultation if their BGLs were not in the recommended range:

"I liked the graphs...that was what gave me the red flag...maybe I have to see the doctor to have my medications changed." [P010, T2D]

Improved awareness of blood glucose levels

Participants noted that although they have a BG meter that provides BG measurement history, having the graphical output of their BGL in My Care Hub further improved awareness of any fluctuations in BGLs:

"It was quite good to see longitudinal things, obviously on my blood monitor I can see by just hitting the back key what the previous readings are.., But to see it in a graphical linear form was really good. It showed me where my blood sugar was, if I went up and down." [P005, T2D]

Mindfulness of calorie consumption

The analytic feature enabled participants to pay attention to daily calorie intake or carbohydrates consumed:

"I liked that idea of putting it all in and seeing how your graphs went up and down, and it sort of kept you a bit more mindful of how many calories or carbs you are eating during the day." [P013, T2D]

Feedback Response

Reinforced health provider's recommendation

Feedback received in response to logged BGL is an element that reinforced the doctor's recommendation about participants' BGLs. A participant with hypoglycaemia unawareness noted that his doctor suggested continuing using the app to serve as an alert in the event of low BGL:

"It is one thing that made me maintain my BGLs. I tend to be what my doctor calls hypoglycaemia insensitive. So, he suggested that I stick with the app because it reminds me to do regular BGL tests to make sure that I am not dropping too low." [P007, T2D]

Informative

Feedback feature serves as an alert about a potential problem in users' BGLs:

"I got confirmation that was somewhat reassuring. I mean if it was out and higher, it just alerts you to a potential problem that you may or may not be aware of." [P005, T2D]

It aided decision making for improved self-management:

"If my levels were over the target range, it gave me very helpful ways to reduce the blood glucose level back into the range." [P007, T2D]

Carbohydrate Components in Foods

Guidance on meal planning

Participants valued the *carbs in foods* feature as it provided information about the average carbohydrate and calorie contents of foods. Participants perceived they were better supported in their choice of appropriate foods to eat and avoid exceeding their recommended daily amount of carbohydrate intake. It also provided guidance on food planning:

"I try to stay between 20 and 50 grams a day, so the carb counting feature was very useful because then you can make an informed decision on what you are going to put on your plate, and you can plan out your week." [P009, T2D]

Knowledge provision and reinforcement

Participants who had difficulties knowing the carbohydrate content of foods found this feature useful through outlining the best foods for consumption to ensure proper health management:

"I have a lot of trouble with how much carbohydrate is in one food but it (app) sort of gets you to realise okay then I have got to check on that." [P004, T2D]

Furthermore, engaging with the *carbs in foods* feature reinforced knowledge and served as a reminder about carbohydrate content in foods:

"There is so much to take in, like reading labels, it is so much to take in. So I found it (app) quite interesting that it is a bit more set out with carbs and how much is in it, and some of them are low and you thought it would be high. Just reinforcing the information because I just can't remember everything." [P014, T2D]

Educational Tips

Knowledge reinforcement

Educational tips were also acknowledged as a tool for knowledge reinforcement and fostered the use of the app. Participants found information on 7 essential ways to manage diabetes quite useful and reflective:

"It is useful, I have got a couple of books, and there is a lot of information, and whilst I may have read it, I am not sure I can regurgitate it." [P005, T2D]

"It was just interesting to read it and think about it." [P014, T2D]

In addition, participants felt that the element provided more comprehensive information in comparison with the feedback element:

"That (educational tips) was more useful than the little hint things (feedback messages) yeah... I think it probably covered it (all information) fairly thoroughly." [P006, T2D]

Recommendations to further improve engagement with the app

Participants' recommendations were majorly based on extended functionality in the app including automated data transmission, more historical analytic data, feedback on physical activity, and update on food list and corresponding nutrient contents. *a. Automation of data input:* Some participants found the manual recording of BGL, physical activity and carbohydrate content of foods consumed as burdensome and expressed that addition of Bluetooth, which could automatically extract data from BGL meter would not only encourage users' engagement with My Care Hub but also improve BGL monitoring. *"It is just the Bluetooth thing to me encouraged me to do blood samples as well without having to write them down manually."* [P003 T1D]

Furthermore, the desire for My Care Hub to automate tracking of time spent on physical activities and equivalent calorie expended was expressed. *"For exercise, if the app could automatically tell you that you have walked 5km, or you have done 30 minutes of exercise and that equals burning so many calories..that would be great."* [P016, T2D]. In addition, it was recommended that MCH should have features to calculate the calorie content of composite dishes. *"for someone like me being a lazy fella it is just easier if I can just click on apple and select small apple or large apple and it just tells me. And I think okay I had this much wholegrain bread, some meat, and an apple and it will tell me what I have consumed in calories for lunch."* [P016, T2D]

b. More analytic histories: Participants suggested extended historical data access and believed this would provide further opportunity to study patterns in self-management activities and have long-term data that could be reviewed by their health care providers. "If you could scroll back and had a look at everything you put in to see if there is a pattern, or if you could scroll back and go back through those graphs. Because you couldn't scroll back any more than whatever was shown on the screen. Also, if there was an option, where you could print out those graphs and take it to your GP when you go for your next check-up and you could show them what is going on." [P013, T2D]

c. Information update: It was suggested that the "Carb in Foods" feature needed more food list and varieties of composite dishes. *"Probably just a greater level of information in the carbohydrate and food section, a broader range, just more information, fish meals, chicken, beef, you know."* [P007, T2D]. Participants suggested this information could be provided in monthly updates because users' awareness of finding new information in the app on a regular basis could foster fresh interest in using the app. *"if there were more foods that would be good. If some of those changed you know like update once a month…a bit more refreshed, just to make things a bit different once a month like because once you have read this you wouldn't keep reading it every day or once a week, once you have read it."* [P014, T2D]

d. Feedback on physical activities: The idea of providing motivational feedback in the app, especially when users achieve certain levels of physical exercise, was raised. This behavioural change strategy in My Care Hub is presently limited to the BGL documentation, presumably, participants want an extension of it to the physical activity documentation. "Maybe a little bit of an achievement, like if I have exercised maybe it..oh great you have done this much for today, so good job well done. So, some positive reinforcement yeah. Because I know a lot of people struggle, my dad has type 2 diabetes and he struggles with exercise. Like some encouragement yeah, so for older people I guess, good job well done you have done 10 minutes of walking or something like that." [P004, T2D]

6.5 DISCUSSION

The My Care Hub mobile app intervention was intended to encourage ongoing participation in DSM activities. This paper reports the levels of participant retention and engagement (usage and behavioural aspects) with the technology over a 3-week pilot study. The findings of the study revealed an acceptable level of participant retention with the intervention, where the majority completed the study per protocol. Furthermore, participants reported that the intervention eased and improved their effort in participating in self-management activities.

Thus, suggesting the app's potential as a tool for DSM support and education. Nevertheless, a larger sample and longer-term studies are required to establish these claims.

Participant retention

The retention rate was relatively high, with more than three-quarters (82%) of participants completing the study per protocol, which is similar to previous short-term pilot studies of diabetes app interventions (Agarwal et al., 2019a; Koot et al., 2019). This indicates that participants were highly motivated and willing to participate in their self-management activities. However, some other pilot studies on DSM support programs reported higher retention than this study. For example, Dick et al., (2011) reported 0% attrition over 4 weeks, whereas Kim et al., (2016) reported only 3% loss to follow-up over a 3-month pilot testing. Such findings are expected because the studies (Dick et al., 2011; Kim et al., 2016) were conducted in controlled settings where participants' recruitment took place in health care facilities, whereas our study utilised Web-based recruitment. Participants are likely to be more committed to the studies when recruited from their care facility and with the knowledge of their care physician (Newington & Metcalfe, 2014). In contrast, studies such as ours that recruited participants through the Web may experience a quick loss to follow-up due to a less structured environment (Eysenbach, 2005; Murray et al., 2009). Future studies with My Care Hub might consider recruitment from a structured setting as a further strategy to improve participants' retention.

Retention was not influenced by participant characteristics measured, with the exception that unemployed participants were less likely to complete the study, which was contrary to the results of a previous mHealth study (Jahangiry et al., 2014). Reasons for this discrepancy are unclear, although despite this difference, 50% of unemployed participants were retained in this study, which is relatively high for Web-based interventions. Future research with My Care Hub will explore reasons for higher attrition among unemployed participants and the use of empirical strategies to improve their retention rates.

Intervention engagement

Users in our study actively used the app for 11 of 14 days (11/14, 79%) in the first 2 weeks, where they all used the app at least once during the first week and 85% returned to use the app during week 2 and 71% during week 3. To put these rates into perspectives, we refer to studies of Faridi et al., (2008) and Kim et al., (2016) who found that 53% and 38%, respectively, of participants used the app for a portion of the 12 weeks intervention duration, where in some cases, there was up to 33% of completely inactive participants (Faridi et al., 2008). In comparison, our app frequency usage rate can be interpreted as reasonable. However, mobilebased interventions differ widely in terms of population, features, settings, and techniques used to foster engagement. For example, although our intervention was self-directed and we did not utilise reminders for self-management or data entry, the above-mentioned studies used face-toface intervention orientation (Faridi et al., 2008; Kim et al., 2016), automated reminders for diabetes management (Faridi et al., 2008), and physician review of adherence (Kin et al., 2016). These disparities may have been a major influence on usage, thus making direct comparison with other app-based interventions difficult. However, the sharp reduction in app usage during week 3, where only 71% retuned to use the app without the push notifications reveals the role of push notification as a feature that could further stimulate users' engagement with apps (Freyne et al., 2017) especially those with content containing insights into how to overcome barriers to achieving health goals (Bidargaddi et al., 2018) as provided in this study. Nevertheless, some users find push notifications intrusive and annoying, especially when too frequent, thus limiting engagement with the intervention (Danaher et al., 2015). Hence, health apps should be built in ways that patients can customise and review when they see notifications

or adjust the timing to suit the selected period of specific self-management tasks such as physical exercise or blood glucose monitoring.

The intensity of usage showed that participants interacted more with features for monitoring of BGL and physical activities, which are in congruence with previous studies (Waki et al., 2012; Kirwan et al., 2013). This was confirmed in the interviews where participants mentioned that these documentation features improved accountability for their self-management activities. This may be due to patients' understanding of the importance of these self-management activities for optimal health outcomes. Another explanation might be because the documentation features were accompanied by analytics that foster improved awareness of BGLs, accountability, and better clarity of self-management activities, as mentioned in the interview. These behavioural strategies in the documentation and analytic features might have encouraged personal reflection among participants, hence the increased intensity of usage.

The active time spent on the documentation features demonstrated that the duration of app usage necessary to generate consistency is a parameter that depends on individual users (Vehi et al., 2019). This was reflected in the interviews where the pattern of use was denoted by users' decision on sequence and DSM routine. This result reveals the advantage of a multicomponent intervention such as My Care Hub, which offers users the opportunity to embrace it in ways most relevant to their needs (Mohr et al., 2010). A user can bypass a feature that they feel does not apply to them, potentially increasing engagement with more relevant areas in relation to their needs. Therefore, the diverse elements available in My Care Hub represent an advancement over many existing diabetes app interventions that consist of only a single element that requires participants to complete a predefined behavioural program (Guertler et al., 2015). Although the My Care Hub system log recorded participants' passive usage of the education textual screens, there are no standard measures to compare these data with similar diabetes-focused interventions. However, the interviews indicated that participants appreciated this feature as an important element that provided knowledge reinforcement as a behavioral strategy for DSM. Nonetheless, the app system was unable to capture whether participants were actually reading and comprehending the embedded information or simply clicking them. An approach to address this limitation is to incorporate eye-tracking technology (dos Santos et al., 2015) or tailored quizzes (Patel et al., 2009) into My Care Hub to measure cognitive responses and knowledge acquired through engagement with each information screen. These measures would need to determine if success or failure of a user to acquire knowledge is due to the intervention component delivery mode, users' engagement with the information, or some other intrinsic factors exclusive to the user.

Generally, engagement indices were initially high but decreased in subsequent weeks. Previous studies using mHealth interventions over short- and long-term periods have identified similar trends (Guertler et al., 2015; Dennison et al., 2013). This finding was expected, as this study was a real-life pragmatic pilot testing of an app, prone to non-use or infrequent use because users prefer to engage with apps periodically (Dennison et al., 2013). In addition, non-usage attrition with mHealth could be due to other reasons such as lack of self-motivation or commitment to change health behaviours (Dennison et al., 2013) and satisfactory attainment of knowledge or skills in managing the disease (Guerlter et al., 2015).

Participants' perceptions related to behavioural change strategies in My Care Hub derived from documentation, feedback response, calories in foods, and education tips features are consistent with the needs analysis study conducted as part of the predevelopment phase of the app (Adu et al., 2018b). Both type 1 and type 2 diabetes patients expressed a strong interest in these elements because of their ability to not only foster engagement with an app but also provide

benefits for self-management behaviours. This reinforces the notion that benefits derived from an intervention strongly affect users' experience and, hence, engagement with the technology (O'Brien & Toms, 2008). As these elements are targeted toward self-monitoring of behavioural activities and the provision of educational information to support those activities, the perceived behavioural change strategies may be an indicator that the app has the tendency to support users to achieve their behavioural goals. Nevertheless, further long-term studies are required to establish this claim.

Perceived ease of use of mHealth positively affects continuance in intention to use (Cho, 2016). The presentation and characteristics of a technology determine the way users can optimise the elements to achieve their aim and overall behavioural goal (Mohr et al., 2014). If users enjoy their experience in a digital behavioural intervention, exposure to the behavioural change component will be improved and may subsequently influence behaviours (Cole-Lewis, Ezeanochie & Turgiss, 2019). These were reflected in our study as participants expressed their opinion about the simplicity of My Care Hub and perceived it as uncomplicated and effortless to use. Even when engagement is a purposeful choice and evolves from how people choose to obtain value from their experience, it has to be enabled by the technology and, thus, impacts long-term interaction with such technology (Lalmas, O'Brien & Yom-Tov, 2014).

The educational component of the app was informed by our previous study, which shows that information on basic guidelines for the management of diabetes and approaches to problem solving in diabetes were highly desired by both type 1 and type 2 diabetes patients (Adu et al., 2018b. However, once that knowledge is obtained, there is a tendency for a drop in participants' rate of use of the app (Taki et al., 2017). This highlights that apart from developing an app to meet end-user requirements and perceived relevance to diabetes management, mHealth developers need to consider ongoing novel strategies that will keep participants engaged. Novelty is also a main contributor to app engagement because it prevents boredom (O'Brien & Tom, 2008; Webb et al., 2010). The downward trend in engagement indices may be explained by a lack of novelty in the app throughout the study period. Hence, future long-term research with My Care Hub must consider ongoing novel strategies that will keep participants engaged. Such strategies may be achieved by considering the suggestions raised by participants in this study. These include periodic information updates on meals and their corresponding nutrient values. Other suggestions on extended functionality in accessing more historical data, automated data transmission, and feedback on physical activity performance are also potential future improvements of My Care Hub, as they have been proven to have an effect on behaviour (Webb et al., 2010).

Strengths and limitations

A mixed methods study design was used to evaluate patient engagement with My Care Hub, which is a strength of the study compared with previous studies that have arbitrarily classified engagement as high or low based on frequency of use (Guertler et al., 2015) or overall adherence to the intervention (Sieverink et al., 2017). The unique contribution of this paper is threefold. First, retention with My Care Hub indicates its potential as a relevant behavioural change intervention tool for patients with diabetes in rural or remote environments with poorer access to specialist health care services. Second, participants' engagement based on interaction with multiple intervention elements was measured using the FITT metrics. The use of this measure reveals the level of user engagement with each intervention feature, thus providing results that are beneficial to inform future enhancements of My Care Hub. Although FITT is commonly used in physical activity research (Barisic, Leatherdale & Kreiger, 2011) to the best of our knowledge, this is the first study to use this measure engagement with the intervention in this study was possible because behaviour metrics and physical activities were measured. The use of FITT as a measure provided results that could broadly serve as a reference to
evaluate other diabetes mHealth interventions before the execution of a full-scale trial. Third, due to the short intervention period of this study, we employed a theoretical and conceptual framework to confirm the components of BIT present in My Care Hub, as an analogue to measures of behavioural engagement with the app. Therefore, the framework served as a predictive device to evaluate the app's suitability as a behavioural change intervention tool. This approach supports a more comprehensive assessment of engagement than most existing short-term pilot studies, which lack theoretical foundations. The use of this framework provides guidance on aspects of mHealth interventions to ensure the development of a meaningful tool that could improve patient engagement with healthy behaviours (Mohr et al., 2014).

This study has some limitations that should be taken into account when interpreting the findings. The short intervention period is acknowledged. However, 3 weeks is the minimum time required for anyone to form a behavioural habit (Middelweerd et al., 2014), and multiple components as found in our intervention are potentially effective techniques to achieve behaviour change (Vernooij, Willson & Gagliardi, 2015). Furthermore, participant recruitment was restricted to a single source, and the sample size was small, thus limiting the sample diversity and generalisability of the results. In addition, the requirement that eligibility includes access to both an Android smartphone and an active email account may imply that the findings may not be generalisable to all smartphone users. In addition, because of the need for our app to comply with the Australian privacy policy and best practice on users' confidentiality, we were unable to include programming codes within the app that could capture users' personal profiles such as age, gender, browser, connection speed, etc. Having this information could provide an opportunity to assess different levels of engagement between those who completed the study and those who did not. In addition, we would have been able to assess if app use was moderated by users' profile. Despite these limitations, considering the promising results further research with a larger sample and over an extended period of time is necessary.

6.6 CONCLUSION

This study provided a comprehensive understanding of participant retention, technology usage, behavioural change process, and engagement with My Care Hub app during a short trial period. Retention was high, although further strategies may be required to further sustain retention when the app is used in long-term trials. The system log indices of FITT of engagement reveal a reasonable level of technology usage during the intervention period. The BIT model employed to measure behavioural change and engagement suggests that My Care Hub could be a behavioural change intervention tool to support self-management behaviours in people with type 1 or type 2 diabetes. Information obtained through the use of multicomponent measures of engagement in this study provides rich and useful data regarding the strengths and weaknesses of My Care Hub and areas requiring improvement to foster increased engagement, sustainable long-term use, and effective health behavioural intervention.

Overall, this chapter contributed to addressing the fifth aim of this thesis by reporting the pilot testing of the novel mobile app in order to ascertain patients' levels of retention and engagement with the app.

Chapter SEVEN: Efficacy and Acceptability of My Care Hub Mobile App to Support Self-Management in Australians with Type 1 or Type 2 Diabetes

7.1 ABSTRACT

Regular access to support and education is required for ongoing participation in diabetes selfmanagement, which is essential for maintenance and improved health outcomes. Patients living in rural areas are the most disadvantaged in accessing such support and education, making mobile app interventions a potential avenue to target these specific populations because of its relatively low cost and easy accessibility. The aim of this study was to evaluate the preliminary efficacy and user acceptance of My Care Hub mobile app - developed to provide evidencedbased support and education on diabetes self-management (DSM). Using a mixed-methods prepost trial design, efficacy and acceptability of My Care Hub were measured among people with type 1 or type 2 diabetes after 3 weeks of intervention. The primary outcome measure was levels of involvement with DSM while the mediating factors were skills and self-efficacy for DSM. In addition, general user acceptance was measured using a questionnaire, and telephone interviews were conducted to elucidate information on participants' perceptions of the app's impact on their DSM and interest in future use. Paired t-test, Wilcoxon-signed ranked test and regression analysis were applied to quantitative data while qualitative data were thematically analysed. Fourty-one participants completed the pre- and post- study questionnaires and 17 participants were interviewed. Statistically significant improvements were observed between pre and post intervention measures: DSM activities $(4.55 \pm 1.14 \text{ versus } 5.35 \pm 0.84; p = 0.001);$ skills (7.10 \pm 1.99 vs 7.90 \pm 1.67; p =0.04) and self-efficacy (7.33 \pm 1.83 vs 8.07 \pm 1.54; p =0.03). Multivariate analysis showed that self-efficacy had the strongest, though not significant influence on DSM. The ratings of each items in the acceptability-measuring tool were above 3 on a 5-point Likert scale. Interview findings revealed that the app reinforced knowledge and provided motivation to participate in DSM activities. The study suggested a positive impact of My Care Hub on diabetes self-management and acceptability by patients. To confirm these promising results, further large scale and long-term studies are required.

7.2 INTRODUCTION

Diabetes self-management education and support (DSMES) is an ongoing process beyond the formal self-management training, that facilitates the knowledge, skills and ability necessary for lifestyle behaviours that assist patients to manage their condition (Sherifali, Jones & Mullan, 2013; Beck et al., 2018). This is essential to prevent or reduce the risk of developing complications thus fostering improved short- and long-term health outcomes (Shrivastava, Shrivastava & Ramasamy, 2013). Currently, there are complex interplays between barriers and ongoing support for diabetes patients. Barriers include economic, geographical and time constraints for patients, and workforce shortages required to support patients beyond irregular diabetes self-management education classes by health professionals (Brown et al., 2002; Al-Azri et al., 2011). In Australia, patients living in rural and remote areas are more severely impacted by these barriers (Smith, Humphreys & Wilson, 2008), leading to significant gaps in service delivery, accessibility (Thomas, Wakerman & Humphreys, 2014), and lower health outcomes (Phillips, 2009). The Australian Institute of Health and Welfare, (2019d) defines rural as any area outside Australia's major cities, and a significant proportion of Australians with diabetes live in these rural areas (National Rural Health Alliance, 2011). Health system limitations in rural areas highlighted the key role that mobile health (mHealth) interventions such as applications (apps) play in the provision of ongoing DSMES to patients.

Numerous apps aimed at improving self-management activities exist for patients with type 1 or type 2 diabetes. However, many diabetes apps lack explicit description of the development

process and design (Adu et al., 2018a), as well as educational components that enhance patients' knowledge for behavioural change (Chomutare et al., 2011; Adu et al., 2018a). There is also limited consideration of users' preferences, which is necessary to improve usage of the intervention and behavioral engagement in self-management (Arsand & Demiris, 2008; Woldaregay et al., 2018). Furthermore, there has been poor integration of the mediating factors that underpin reported self-management (behavioural) or clinical outcomes in studies using diabetes app interventions (Rossi et al., 2013; Holmen et al., 2014; Berndt et al., 2014; Quinn et al., 2016). Factors including knowledge of DSM skills and self-efficacy (confidence) are important mediators for behavioural change outcomes (Persell et al., 2004; Sarkar, Fisher & Schillinger, 2006). Thus, integration of these factors into interventions could foster patients' engagement with DSM. Adequate self-management skills are provided through knowledge about the disease and understanding of the relationship between various self-management behaviours and resulting health outcomes (Aronson et al., 2018). Self-efficacy, on the other hand develops patients' confidence to perform these behaviours and to overcome barriers that prevent achievement of behavioural goals (Wu et al., 2007).

Previous studies have reported linear positive association between levels of DSM skills and self-efficacy and participation in specific self-management behaviour such as diet control, monitoring of blood glucose (Wang et al., 2012), physical exercise (McAuley et al., 2011; Alemdag, 2018), foot care (Perrin, Swerissen & Payne, 2009) as well as overall self-management behaviours (Wu et al., 2007; Fang, Yu & Tao, 2013; Huang et al., 2014; Yao et al., 2019). This implies that participation in self-management behaviours is an end-product of an individual's management skills and confidence to perform the behaviour. Therefore, a diabetes mHealth intervention aimed at behavioural change should target patients' improvements in the mediating factors of skills and self-efficacy, which could consequently trigger improved diabetes self-management behaviours.

Researchers have also indicated a positive association between acceptability of a technology and improved levels of self-management (Harrison et al., 2014; Hirani et al., 2017). The content and quality of mHealth technology have implications for its acceptability (Sekhon, Cartwright & Francis et al., 2017) which is an antecedent of users engagement and a key consideration for implementation into practice (Kim, 2015).

Drawing on this background, a new diabetes app intervention called My Care Hub was developed to provide evidenced-based support and education to foster self-management behavioural change in Australians with Type 1 or Type 2 diabetes (Adu et al., 2020a). My Care Hub provide multiple features and functions targeting the mediating factors of skills and self-efficacy in patients in order to foster improved behavioural change. These features/functions include: (i) an electronic diary and analytics to self-monitor behavioural activities such as blood glucose, physical activities, food intake and weight; and (ii) various educational modules.

Study Aims

This study reports the preliminary efficacy of My Care Hub. The primary outcome measure is diabetes self-management behaviour pre- and post-intervention. Determinants factors that underpin the process of the primary outcome include changes in diabetes management skills and self-efficacy. In addition, we assessed the acceptability of the app among patients. We posit that the use of My Care Hub in this short-term trial would be acceptable and foster modest improvement in diabetes self-management behaviours due to improvement in skills and self-efficacy.

7.3 MATERIALS AND METHODS

The study procedures were approved by the Human Research Ethics Committee of the James Cook University (reference #H7716). Participants were informed about the study aims and the use of their de-identified data for analysis. Informed consent was implied by submission of an online survey while verbal consent was obtained for all telephone interviews.

7.3.1 Study Design

Details of the study methodology have been fully described in a previous publication (Adu et al., 2020b). In brief, this pilot study (conducted in August to October 2019) employed a mixedmethod sequential explanatory design, where participants accessed the intervention over a three-week period and communicated their perceptions through surveys and interviews. The quantitative phase involved a single-arm repeated measures design entailing assessment of (1) preliminary efficacy of the intervention through measures of DSM activities, skills and selfefficacy where data were collected online before and after the intervention and (2) the app's acceptability. The qualitative phase involved telephone interviews with a subsample of participants and it was aimed at gaining more insight into the role that My Care Hub played in their DSM during the intervention period.

Participants were recruited via an email circulated to people interested in research who were registered with the National Diabetes Service Scheme, Australia. The inclusion criteria were (a) diagnosed with type 1 or type 2 diabetes, (b) aged 18-65 years, (c) live in North Queensland (a rural/regional part of Australia), (d) have a current recommended blood glucose level (BGL) target of 4 -10 mmol/L, (e) not pregnant, (f) able to perform activities of daily living, (g) have an android smartphone, and (h) not currently using an app which provides educational support related to DSM. To minimise response bias, a three-staged selection process was used: (i) all invited prospective participants were provided a link to the study information page containing details of the study focus and eligibility; (ii) those who indicated interest and gave consent were directed to the screening questions to confirm that they met all eligibility criteria; (iii) only those who met all the eligibility criteria were then directed to fill the pre-intervention survey

which examined participants' demographic characteristics and health profile as well as their DSM, skills and self-efficacy levels. A total of 50 participants were enrolled into this study, which is sufficient for a preliminary efficacy study (Lancaster, Dodd & Williamson, 2004).

After filling the pre-study survey, participants were emailed a unique username and password to access the app and its user manual. Participants were provided with technical support to tackle any problems with the app and respond to queries.

7.3.2 Intervention

As outlined in the My Care Hub development protocol (Adu et al., 2020a), the app was specifically designed for those who have type 1 diabetes with recommended BGL of 4-8mmol/L-fasting and <10 mmol/L-2 hours postprandial, and fasting levels of 6-8 mmol/L and 2 hours post prandial levels of 6-10 mmol/L for those with type 2 diabetes. The self-efficacy (confidence) construct of the social cognitive theory (Bandura & National Institute of Mental Health, 1998) and the information, motivation constructs of the Information Motivation Behavioural Skills (IMBS) model (Fisher, Fisher & Harman, 2003) were the two underlying health behavioral change theories employed in the development of My Care Hub. In the context of this study, the blended concepts of the theories provided mediators for behavioural change. We hypothesised that diabetes self-management behavioural change is mediated by an individual's self-efficacy, which is related to their level of skills to undertake specific tasks required to reach a desired goal (diabetes self-management).

In relation to the framework described above, "Documentation" and "Analytic" features to monitor BGLs, physical activities, food intake were provided in My Care Hub as techniques to facilitate self-efficacy and consequently improve DSM in patients. Furthermore, the app's educational modules were developed using the three constructs of IMBS: Information, Motivation and Behavioural Skills. The IMBS model postulates that behavioural change occurs as a results of changes in skills sequel to effect of 'information' and 'motivation' interventions. Features on 'overview of diabetes management' and 'carbs in foods'' provide information on diabetes and its self-management. Specifically, actionable 'information' on lifestyle modifications (healthy eating and physical activity), monitoring of BGL, complying with medications, good problem-solving skills, healthy coping skills, and risk-reduction behaviours (such as smoking cessation, alcohol intake reduction and foot care) for DSM (Shrivastava, Shrivastava & Ramasamy, 2013) were inputted into the app. In addition, the app contains 'information' on approximate equivalent carbohydrate and calorie content of common foods in Australia based on portion sizes of each food.

In relation to 'motivation', this was targeted using the "Feedback" and "Push notification" features in the app. Logged BGL data were automatically evaluated following the Australian Diabetes care guideline's targeted values: optimally for people with type 1 diabetes, BGL 4-8 mmol/L before breakfast and <10 mmol/L 2 hours after each meal; for people with type 2 diabetes, BGL 6-8 mmol/L before breakfast and 6-10 mmol/L after each meal. The feedback feature determines if each data item satisfies guideline requirements or not and then provides feedback in the form of motivational encouragement, advice on lifestyle modifications or reinforcing health behaviours as applicable. Lastly, push notifications were provided to strengthen healthy coping necessary for improved engagement in DSM activities (Hood et al., 2015). Notifications provided messages related to diabetes distress, the importance of acknowledging it if experienced by participants and its' potential impacts. Participants were then advised to identify realistic goals and focus on them in order to alleviate the distress, which consequently impact their DSM and health outcomes. Examples of actionable goals were provided in order to foster comprehension and engagement. Short, simple text notifications were sent at 12:30pm once daily during the intervention period. Long and frequent notifications may be perceived by patients as intrusive and annoying and might limit the opportunity for

engagement with the intervention (Danaher et al., 2015). Although push notifications on apps can provide intervention content to users in a way that can be relatively difficult to ignore (Danaher et al., 2015), we took steps to increase the probability that all participants viewed the messages in order to equalise this intervention dosage. Hence, message sets sent in the first week were reshuffled and resent in the second week. This technique ensured that participants viewed messages - if a particular message was not opened in the first week on a specific day, it is likely it will be opened in the second week when sent on a different day. **Figure 7.1** illustrates the conceptual framework for the development and evaluation of the efficacy of the My Care Hub, which was informed by the mediating constructs of social cognitive and IMBS models.



Legend: Interventions on diabetes self-management information and motivation were provided in order to increase participants' skills for self-management aimed at fostering self-efficacy and consequently improved

diabetes self-management behaviours

Figure 7.1: Conceptual framework for My Care Hub based on blended IMBS and SCT models.

7.3.4 Instrument and Data Collection

Baseline demographic and health characteristics reported by participants include age, gender, employment status, educational level, health care practitioner's recommended fasting and 2-hour post-prandial BGLs, duration since diagnosis and self-perceived health status.

Measures of primary outcome and mediators for preliminary efficacy

The primary outcome was frequency of involvement with DSM activities while mediating factors were diabetes management skills and self-efficacy. Improvement in each of the outcomes was defined by a statistically significant increase between pre- and post- intervention scores. The measuring tool (questionnaire) consisted of two sections, where section one measured the DSM activities using 10 items from the Summary of Diabetes Self-Care Activity (SDSCA) questionnaire (Toobert, Hampson & Glasgow, 2000). The SDSCA items covered five DSM behavioural domains: BGL monitoring (2 items), healthy eating (4 items), regular physical activities (2 items) and foot care (2 items). Participants were asked to recall their activities for the last seven days and state the number of days they performed the behaviours, after which the mean scores across each activity domain was calculated. Section two of the survey collected data on skills and self-efficacy for managing diabetes using the LMC skills, Confidence and Preparedness Index (SCPI) tool (Mbuagbaw et al., 2017). Only 17 items in the SCPI tool which addressed perceived skills (9 items) and self-efficacy (8 items) were relevant and used in this study. Participants were asked how they perceived their ability and confidence to perform diabetes related activities on diet, exercise, taking medications, managing stress, monitoring blood glucose and complications. Items were rated on a 10-point scale where higher values denoted better skills and self-efficacy. The scales have good internal consistency. In the current study, Cronbach alpha for the skills and self-efficacy items were 0.89 and 0.88 respectively.

Measures of acceptability

Post-intervention, participants also rated their experience with the app using a set of 18 relevant items adapted from different tools (Brooke, 1996; Venkatesh & Davis, 2000; Brooke, 1996; Lund, 2001). On a scale of 1 (strongly disagree) to 5 (strongly agree), participants rated their agreement with a series of statements about the app's acceptability. These are related to ease of use, intelligibility, satisfaction, perceived value, intention and behaviour towards recommendation.

Interviews

Within 3 weeks of participants indicating interest, semi-structured interviews were conducted by a single researcher (AD) who is well experienced in qualitative research. A semi-structured interview guide was developed for the study, and elucidated information on participants' perceptions of the app's impact on their DSM and interest in future use. The guide was pilot tested by MDA and AD, and the first three interviews were used to reflect on the guide, which was found appropriate for data collection in its original form. There was no prior relationship between the participants and any of the researchers. Each interview was audio-taped and transcribed. Data saturation was achieved after completing the 14th interview. However, interview sessions with all consenting respondents (17) were completed in order to allow for rich documentation. Repeat interviews were not required and there was no post interview debriefing. The conduct and reporting of the interviews followed the consolidated criteria for reporting qualitative research (COREQ) (Tong, Sainsbury & Craig, 2007).

7.3.5 Analysis

All quantitative data were analysed using SPSS version 23. Descriptive statistics was used to present participants' demographic characteristics. Outcome measures (pre-and post-intervention data) and acceptability of the intervention were reported using means and standard

207

deviations (SD). Paired sample t-test and Wilcoxon signed-rank test were used to evaluate changes in the outcomes over 3 weeks for the normally distributed and non-normally distributed variables respectively. Effect sizes were calculated using Eta squared values to show the magnitude of changes in outcomes pre and post intervention. In addition, multiple regression analysis was used to estimate the contribution of the mediating factors to participants reported overall DSM levels post intervention. All mediating factors which increased (either significantly or not) from pre to post intervention were included in the regression. Two-tailed with p < 0.05 were considered statistically significant.

Audiotaped interviews were transcribed verbatim and coding of text fragments based on content was performed by two researchers MDA and AD independently. Consolidation of codes and grouping into themes was achieved through discussions with BMA. Findings are supported with illustrative quotes.

7.4 RESULTS

7.4.1 Demographics and Health Statistics

Of the 50 participants initially enrolled, 41(82%) completed the study, including filling in the post-study survey. Participants were predominantly male (61%), aged between 20-64 years (mean, 49.29 years [SD 12.74] and were Caucasians (92.7%). Most of the respondents were residents in rural areas of North Queensland (70.7%), had a technical college education or higher (78%) and were employed (70.7%). Most had type 2 diabetes (71%), rated their health status as 'good' or 'better' (63.4%) and were diagnosed with diabetes in the previous 5 years (56.1%). Participants reported their recommended fasting BGL: mean 6.03 \pm 1.35; ranged 4-8 mmol/L and 2 hours postprandial: mean 7.53 \pm 1.23; ranged 6 -10mmol/L.

Table 7.1 shows the total mean scores of the DSM domains and mediators: knowledge of diabetes management skills and self-efficacy. At baseline, self-reported adherence to daily dietary recommendations, engaging in physical exercise and BGL monitoring were generally performed five days a week while foot check was the lowest at about 3 days a week. The total mean score across all DSM domains was 4.55 ± 1.44 . Comparison between pre and post intervention scores shows that adherence to diet, monitoring of BGL and overall DSM significantly improved over time (p=0.04, eta squared = 0.1; p = 0.04, eta squared = 0.2 and p = 0.001, eta squared 0.24 respectively). In relation to skills and self-efficacy, significantly higher scores were observed in both of these factors after the intervention (p < 0.05 for both factors with an overall small effect size of 0.11).

Outcome	Baseline, Mean (SD)	Post intervention, Mean (SD)	Р	Effect size
Diabetes self-management				
Diet	5.13 (1.10)	5.54 (0.90)	0.04*	0.10
Physical activity	4.48 (2.16)	5.35 (2.27)	0.09	0.07
Monitoring of BGL ^a	5.16 (2.81)	6.80 (1.95)	0.04*	0.20
Foot check	2.87 (1.86)	3.51 (1.79)	0.18	0.05
Overall	4.55 (1.14)	5.35 (0.84)	0.001*	0.24
Skills and self-efficacy				
Skills	7.10 (1.99)	7.90 (1.67)	0.04*	0.23
Self-efficacy	7.33 (1.82)	8.07 (1.54)	0.03*	0.25
Overall	7.27 (1.83)	8.00 (1.55)	0.04*	0.11
* 0.05				

Table 7.1: Observed mean and standard deviations for the outcome measures

*p < 0.05

^aBGL: Blood glucose levels

7.4.3 Relationship between Mediating Factors and Diabetes Self-Management

Positively strong significant correlations were found between skills and self-efficacy (r = 0.835, p < 0.001), where those with high level of skills have high self-efficacy. In addition, self-efficacy was weakly correlated with diabetes self-management (r = 0.285, p = 0.07).

Multiple linear regression analysis was conducted to examine the variables that predict DSM. After establishing the assumptions of multiple linear regression, analysis identified the simultaneous contributions of skills and self-efficacy on participants' level of DSM. These variables predicted 8% of the variation of DSM [F (1, 41) = 1.590, p = 0.218, R = 0.08]. While both factors did not have a significant relationship with DSM, the result shows that self-efficacy has the strongest influence on DSM (β = 0.478). Details are shown in **Table 7.2**.

Table 7.2: Influence of mediating variables on diabetes self-management

Determinant Variables	В	SE	Beta	t	Р
Diabetes management skills	-0.15	0.144	0.298	0.1.38	0.306
Self-Efficacy	0.26	0.157	0.478	1.664	0.105
Constant (α) = 4.428					
$\mathbf{p}^2 = 0.050 + 1^2 + 1 \mathbf{p}^2 = 0.00$					

 $R^2 = 0.079$; Adjusted $R^2 = 0.29$

7.4.4. Acceptability

As presented in **Table 7.3**, overall mean ratings for all of the items were above 3 on the 5-point scale; suggesting that participants were satisfied with the app's ease of use and educational content. They noted that the app facilitated improved awareness, and stimulated their interest in DSM activities and assented that My Care Hub could serve as a DSM support tool. Participants also expressed interest in future use of the app if continually available and would recommend it to a friend or family with similar health condition.

Survey item	Mean	SD
Ease of use / intelligibility / satisfaction		0.15
I feel confident using the app	4.2	0.68
I am satisfied with how easy it is to use the app	3.9	0.83
I felt comfortable using the app	4.02	0.76
I found the educational tips embedded in the app easy to understand	4.07	0.65
I found the immediate feedback provided after my BG log easy to understand	4.15	0.73
The messages displayed through push notification were easy to understand	4.17	0.59
Overall, I am satisfied with the app	3.68	1.04
Total	4.02	0.75
Value		
The daily messages (push notifications) increased my awareness of the importance of engaging in my self-management activities	3.59	1.14
The app features could stimulate my interest to continually participate in my self- management and record the activities	3.56	1.16
The app support my self-management such as tracking of BG, provide an idea of the carb content of my food	3.8	1.03
The daily messages (push notifications) motivates me more to pay attention to managing my diabetes	3.46	1.14
I found the immediate feedback received after logging my BG helpful for my self- management	3.61	1.16
The notifications motivates me to do my self-management activities (e.g exercise, healthy eating, BG monitoring)	3.41	1.16
My Care Hub app could serve as a self-management support tool for people with diabetes	4.05	0.87
Total	3.64	1.09
Intention for use and recommendation		
If I have continual access to the app, I will use it frequently	3.46	1.23
I think I would like to use the app more frequently	3.49	1.23
I could recommend the app to family and friends who have my type of diabetes	3.66	1.15
If I were to proceed with the program, I want to receive fewer push notification messages	3.02	1.01
Total	3.4	1.16

Table 7. 3: Participant acceptability ratings of My Care Hub (N = 41)

7.4.5. My Care Hub's Impact on Diabetes Self-Management and Interest in Future Use

An in-depth understanding of the quantitative findings in relation to My Care Hub's impact on participants' DSM abilities during the intervention period was elucidated through the interviews. Three major themes emerged from the interviews: "Reinforced knowledge", "motivation for self-management" and "continuity".

Reinforced knowledge: Participants perceived that the educational messages in My Care Hub reinforced their knowledge about diabetes and self-management of the condition.

"I did have some knowledge as I have been to a dietician. But with the app. It is always good to have that little message to reinforce you each day to watch out for things that you shouldn't have too much of." [P014, T2D]. "It sort of helped me and remind me of watching the diet. High blood glucose level is a reflection of what you consume. It reminded me in that regard to be careful of what I eat." [P009, T2D].

The messages also prompted reflection on how best to handle events/situations. "It clarifies the information I already know because this is a sort of disease that you can't see. It is eating away in the background there and the app lets you look at it from a different way other than just pricking your finger three or four times a day and prick again and it is still high. With the app, I kind of try to keep it under control." [P011, T2D].

A few participants reported that apart from the app reinforcing their knowledge, they also gained new information related to the effect of diabetes distress on blood glucose. "*Some of it was new information. It was quite interesting to know how stress affects diabetes and your sugar. I have a bit of stress every now and then. That information is something I had never thought about.*" [P016, T2D].

When asked about the advantage of promoting this intervention to a larger population, participants reported that the app would particularly improve the knowledge of people newly diagnosed with diabetes as a result of its educational information component: "especially people who are new to diabetes could get a lot of information from it (My Care Hub). It would help them a lot to sort out what they are doing and what is going on." [P008, T2D]. "People who are new to diabetes, like after attending a couple of courses, it could help them to understand a lot more." [P10, T2D].

Motivation for self-management: Participants described that the app provided motivation to care for themselves and encouraged participation in different aspects of their self-management. "Yes, it increased my motivation. I do my blood test and I weigh myself regularly and I was going out to a do a reasonable amount of exercise." [P005, T2D]. Some participants mentioned that My Care Hub gave them some degree of control on managing their condition. "For once it was about doing something for me, giving information to me and giving me I would say a degree of control... what I stick in my mouth." [P005, T2D]. Several features in My Care Hub supported different self-management activities and were perceived to improve easy accessibility to necessary support: "I think just having everything there at your fingertips, the BGL levels, the exercise, your food, your diet, your carbs counting sort of thing. It was all there for you. You know the flexibility of it." [P009, T2D].

Continuity: Participants also expressed strong interest in future use and recommendation of the app to other users. Participants reiterated their intention to continue using the app if accessible: *''If you are serious about looking after yourself and stay within your target blood glucose range, I would definitely say yes to the app.''* [P007, T2D]. In addition, a participant narrated that his doctor was positive about the app's content and willing to recommend it to his other patients: *''I told my doctor I was doing a study and he had a look (at the app) and said yeah, that looks good. He asked what the green things were and I said, the green ones are what*

you should be eating and the other ones are high in carb. He thought that was OK and wanted to know what it is called because if he had other patients, he said he could direct them to downloading that app." [P13, T2D]

7.5 DISCUSSION

This pilot study investigated the preliminary efficacy and acceptability of the My Care Hub app, which was designed to improve participation in DSM in people with type 1 and type 2 diabetes.

Preliminary efficacy

In this study, patients reported improved levels of participation in all domains of measured DSM activities and this may be due to increased motivation to engage in self-management activities through the use of My Care Hub as reported in the interviews. Few short-term research studies have reported on the preliminary efficacy of mobile phone apps either in relation to overall DSM activities or for single self-management activity change (e.g. dietary or physical activity only). Agarwal et al., (2019b) and Faridi et al., (2008) tested the effect of a mobile technology on overall DSM, while others have monitored diet or physical activity (Rossi et al., 2010b) as part of program evaluation for diabetes support. Preliminary efficacy results of these apps vary from none (Agarwal et al., 2019b) to moderate (Faridi et al., (2008); Rossi et al., 2010b) among participants in the intervention settings. Therefore, the significant improvements in DSM observed in our study are unique and could be termed to have clinical significance when viewed in the context of impact on diabetes management.

In reality, continuing health-care provider support for DSM is not always available. Ongoing DSMES for improved self-management is needed to reduce or prevent the risk of developing complications and other poor health outcomes (Shrivastava, Shrivastava & Ramasamy, 2013; Beck et al., 2017), which are particularly prevalent among Australian rural populations

(Wakerman et al., 2008; Phillips, 2009; Australian Institute of Health and Welfare, 2019d). The provision of a potentially highly effective mobile health app such as My Care Hub for improving DSM could be an important supportive measure among this patient population. The My Care Hub intervention provides educational features, documentation features, Analytics and feedback i.e guidance based on information entered by users. The use of multi-component behavioural change strategies and mHealth features as described above have the greatest potential impact on behavioural change in self-management (Middelweerd et al., 2014; Direito et al., 2014; Danaher et al., 2015).

In recent years, several mHealth applications have been developed in order to support selfmanagement in people with diabetes, with these interventions deemed feasible and acceptable, though evidence of improved self-management is either unclear or weak (Schoeppe et al., 2016; Whitehead & Seaton, 2016). This may be due to a lack of proper consideration of the mediating factors necessary to produce improved DSM. Adequate skills and self-efficacy are major pivoting mechanisms for behavioural change in diabetes management (Persell et al., 2004; Sarkar, Fisher & Schillinger, 2006; Wu et al., 2007; Aronson et al., 2018). The consequent impact of these factors to produce improved DSM is therefore expected and confirmed our hypothesis. Skills is the understanding and ideas that patient possesses about a subject (diabetes and its management), potentially with the ability to use it for a specific purpose (selfmanagement) (Boger et al., 2015) and it fosters self-efficacy - the confidence a patient has in his/her self to achieve the purpose (Bandura et al., 1999). Self-efficacy is a prerequisite for informed health decision-making (Tomky et al., 2000; Mulcahy et al., 2003) and greatly influences the probability for behavior initiation, level of applied effort and how long behaviour will be sustained (Bandura et al., 1999). Therefore, the results of this study is a further proofof-concept, supportive of previous literature on the value of improved self-efficacy to promote behavioural change (Holloway & Watson, 2002; Williams & French, 2011). Nevertheless, the non-significant predictive power of self-efficacy on DSM which we found in this study might be an indicator that self-efficacy is not strong enough to make a large effect in a short time frame. As such, it is likely that the strong causal relationship will require more prospective investigations.

Participants in this study emphasised that the intervention motivated them to engage in their DSM as well as reinforced their knowledge of diabetes. My Care Hub educational content serving as just-in-time resource to increase motivation and prompts to action in self-management have also been described in other studies (Webb et al., 2010; Klasnja & Prat, 2012). This result suggests that the app could be a feasible means of augmenting self-management education and support. Furthermore, the app was perceived to be a particularly useful tool for people who are newly diagnosed with diabetes to remind them of many issues discussed during face-to-face diabetes education session with their health providers on the importance of self-management and adherence to it for improved health outcomes.

Acceptability

The acceptability of a mHealth technology is an indication of its value and importance for wider implementation into the healthcare system (Kim, 2015). The result of this study indicates good level of acceptability of My Care Hub as most participants endorsed the app components as useful and supportive of their DSM. Other studies assessing acceptability of mobile apps for diabetes self-management were similarly positive (Kollmann et al., 2007; Torbjørnsen et al., 2018). This result might have also fostered higher levels of DSM reported in the post-intervention, because studies have demonstrated a positive association between higher levels of acceptability of a mHealth and self-management (Harrison et al., 2014; Westergaard et al., 2017). Likewise, perceived ease of use and satisfaction with health apps positively affect continued intention of use (Li et al., 2019). These were reflected in our study, as participants

expressed overall satisfaction with the simplicity of My Care Hub with intentions of continued use and recommendation to others. Nevertheless, acceptability has been described as only 'one piece of a puzzle'; because even with high acceptance levels, uptake and upscale of the intervention may diverge. Hence, the recommendation that healthcare providers who perceive strong benefits of mHealth technology should endeavour to encourage patients' adherence to it (VanAnh, Auroy & Sarradon-Eck, 2019).

Strengths and limitations

The study utilised mixed-methods research design, which allowed for detailed exploration of participants' experiences and perspectives about the app. In addition, the study provided explanation of the preliminary-efficacy of My Care Hub app on diabetes self-management in relation to its mediating factors as targeted in the intervention. Such report is often lacking in many preliminary efficacy studies of mHealth technologies. The use of theory-driven and evidenced based intervention support strategies is also a notable strength of this study.

There are some limitations to the current research. A longer follow-up period would have provided clearer insights into sustenance of the reported behaviour changes. However, the short-term intervention period in this study is comparable to that of other studies with 2-3week intervention period (Herrejon et al., 2009; Heinrich et al., 2012). Furthermore, our study population were patients registered with NDSS and interested in research, potentially implying many participants were already on top of their self-management, as reflected in the high level of DSM at baseline. This reduces to an extent, the generalisability of the study findings to other populations. Also, the tools adapted for measuring acceptability of the intervention were not used in their entirety as items not relevant to the current study were removed. Using only few items from a validated tool might compromise its uniformity. Nevertheless, the selected set of items in each validated tool demonstrated good internal consistencies with Cronbach alpha

from 0.70 to 0.91. We noted that our sample size like many similarly published pilot trials was modest. In addition, measured outcomes were self-reported and thus may be subjected to social desirability and recall bias. Lastly, the lack of control group may limit the conclusions that can be made regarding the beneficial impact of the app. Nevertheless, preliminary work such as this is a useful and necessary precursor to more rigorous examination of the intervention in a large-scale trial with longer-term follow up.

Future Directions

Automatic push notifications

The education component in My Care Hub specifically delivered through the push notification feature requires human coaching whereby a diabetes educator provides daily education through this platform during the intervention period. Intervention that relies on human input requires substantial human resources, which if lacking may limit the scalability of the intervention. Therefore, further improvement of My Care Hub requires automation of the push notification educational components free of human involvement as much as possible. This will lower the cost of operation and improve scalability of the intervention.

Long-term trial

The promising result of this pilot My Care Hub app project which shows preliminary efficacy, acceptability (as reported in this study) as well as good level of retention and engagement with the intervention (Adu et al., 2020b) will require further confirmation using long term controlled trials in the future. An adaptive randomised controlled trial design (Pham, Wiljer & Cefazzo, 2016) may be best suited due to the rapidly evolving nature of mHealth. The design will enhance dynamic adaptation of the app to the advancing field of information technology thus facilitating better understanding of the unique impact of each of the app features, thereby

fostering improvement and long-term utility of the My Care Hub intervention in the support and management of diabetes.

7.6 CONCLUSION

The use of mobile phone application intervention among underserved population represents a novel approach to augmenting self-management education and support. We propose an innovative app–My Care Hub, as a self-management tool for Australians with type 1 or type 2 diabetes. The results of this pilot trial suggest that My Care Hub app can be an acceptable and potentially effective intervention that can be replicated in other contexts to improve diabetes self-management. Future work should employ larger and long-term trials to further establish the efficacy of the app and impact on glycaemic control and other health outcomes.

Overall, this chapter contributed to addressing the fifth aim of this thesis by reporting the pilot testing of the novel mobile app in order to ascertain its efficacy in diabetes self-management as well as level of acceptability among patients with type 1 and type 2 diabetes.

Chapter EIGHT: General Discussion

8.1 Chapter Overview

A discussion of the key findings of this research project in the light of current knowledge is presented herein. Novel contributions of the different components, practical implications within a broader context, strengths and limitations of the research are highlighted. The chapter also outlines prospects for future research to enhance self-management education and support for patients with type 1 or type 2 diabetes.

8.2 Research Findings and Novel Contributions

A systematic process was utilised to develop and examine preliminary efficacy of a mobile phone app to provide DSMES to Australians with type 1 or type 2 diabetes. An over-arching framework (Figure 1.2) was developed to guide the series of studies (each with its own hypothesis) presented as research chapters in this thesis. A summary of the key findings and novel contributions of each chapter is presented in Table 8 and discussed below.

8.2.1 Guidelines and Developments of the Essential Elements Required in Diabetes App Development

Chapter 2b tested the hypothesis that a systematic review and evaluation of evidence from existing literature on the impact of diabetes app intervention on patient health outcomes will assist in the establishment of guidelines that outline important considerations required for the development of effective mobile phone apps to support diabetes self-management. The review revealed a general lack of consideration of the key elements of app development by previous app developers and researchers. The key elements identified as lacking in large-scale trials with every diabetes app prior to use as an intervention tool were: inclusion of health behavioural theory, views of users and clinical experts, data security and privacy as well as pilot testing.

Poor considerations of health behavioural theoretical constructs in the development of mobile diabetes app interventions highlighted a missed opportunity of using digital interventions to inform the refinement of existing theories or developing new suitable diabetes apps. Theoretical components could facilitate health behavioural change; therefore, their inclusion is an important first steps in improving the efficacy of apps targeting the promotion of health behaviours (Noar & Zimmerman, 2005; Glanz & Bishop, 2010).

Insufficient consideration of users' expectations and preferred components in a diabetes app resulted in poor adoption of user-entered design principles. Non-inclusion of the opinions of end users (diabetes patients) in terms of usability requirements and preferences in app features caused low retention and engagement with apps when used as intervention tools (Arsand & Demiris, 2008; Yardley et al., 2016).

The lack of involvement of healthcare professionals (HCP) in developing apps was consistent with previous studies (Arnhold, Quade & Kirch, 2014). Involving HCP in the early phase of mobile app development ensures accuracy, precision and quality of the intervention components. Inaccuracy of intervention content is one of the major barriers to the implementation of technology in clinical care settings (Maguire et al., 2008; Murray et al., 2016), hence, involving HCP in developing apps for patients provides an opportunity to minimise imprecisions and inaccuracies.

Furthermore, most of the apps reviewed lacked information on data security and privacy in line with previous analyses of mobile health and fitness apps (Ackerman, 2013). Both patients and clinicians have indicated that privacy policy and data security inclusion in apps are important for gaining user trust (Fife & Orjuele, 2012; Schueller et al., 2018). As such, security measures to protect patients' information must be initiated at the time of app development and maintained long-term. Providing privacy policies, encrypting personal identifiable information on

cyberspace and data encryption are some of the ways app data security and privacy could be guaranteed.

In most apps reviewed, pilot testing processes prior to use in either randomised controlled trials or quasi-experimental studies were not reported. A pilot study is a necessary step in exploring the feasibility of novel interventions prior to use on a larger scale (Leon, Davis & Kraemer, 2011). Well-conceived and implemented piloting of apps could reduce the risk of several longterm problems in large-scale studies; including the inability to recruit and retain an effective sample size and the subsequent reduction in statistical power. Piloting reveals the need to modify an intervention or protocol prior to use, in order to minimise the percentage of failed trials and maximise the opportunity to efficiently utilise research funds on developing feasible and impactful app interventions.

Most importantly, the review found a significant association between the inclusion of educational components in apps and improved glycaemic control which is the ultimate goal of self-management interventions. This highlighted the need for using self-management education to advance the efficacy of diabetes app as a supporting tool for patients.

Overall, the systematic review filled the gap of providing evidence-based information and clear guidelines on essential elements for app development (Kohatsu, Robinson & Torner, 2004; Ben-Zeev et al., 2015; Lhachimi, Bala & Vanagas, 2016) as a delivery mode for DSMES. The review revealed the need to carry out other preparatory studies that facilitated the identification of these essential elements (Chapter 3) and preferred features / educational content (Chapter 4) in apps that promote DSMES prior to developing the new mobile app (chapter 5), which was then pilot tested for efficacy (Chapter 6) and acceptability (Chapter 7).

8.2.2. Understanding the Perceptions of People Living With Type 1 or Type 2 Diabetes on the Mediating Factors of Skill and Self-Efficacy for Diabetes Self-Management

Chapter 3 tested the hypothesis that *skills and self-efficacy are important mediating factors that stimulate behavioural change and improved self-management in people with type 1 and type 2 diabetes.* A multinational audience across four continents (USA, Australia, Europe and United Kingdom) was utilised to provide a relatively generalisable data set to foster the understanding of the phenomenon of interest. The findings revealed a strong positive correlation between skills and self-efficacy, which could lead to improved DSM due to behavioural change. Patients across all continents reported high levels of skills and self-efficacy to monitor the impact of physical activity, medication and diet on their blood glucose levels. However, they reported lower levels of skills and self-efficacy in identifying and managing the impact of stress on diabetes, exercise planning to avoid hypoglycaemia and interpreting blood glucose patterns. These revealed common areas of low skills and self-efficacy for self-management among diabetes patient could guide health providers and researchers in prioritising areas of need when providing education and support intervention to patients with type 1 or type 2 diabetes.

The study confirms the importance of adequate skills and self-efficacy acquisition in relation to disease management (Johnston-Brooks, Lewis & Garg, 2002; Persell et al., 2004; Aronson et al., 2018). Previous research on diabetes has established that increasing patients' skills and understanding of their illness alone does not lead to actual improvement in health-related behaviour (Atak, Gurkan & Kose, 2008). This phenomenon is known as 'knowledge-behaviour gap' (Sligo & Jameson, 2000). Therefore, in addition to relevant skills acquisition, individuals need to believe (self-efficacy) that they are capable of executing and maintaining a behaviour in order to devote the necessary effort to succeed (Bandura, Freeman & Lightsey, 1999). In addition, this study showed that patients who were more educated had higher levels of skills

and self-efficacy (Bandura, Freeman & Lightsey, 1999), thus suggesting that low educational status in patients is a risk factor for low levels of skills and self-efficacy for DSM. This corroborates several researchers in diabetes and other health conditions (Gazmararian et al., 2003; Wilson et al., 2010). Low health literacy (Kindig, Panzer & Nielsen-Bohlman, 2004) and reduced health prevention behaviours (Bennett et al., 2009) occur as a result of poor knowledge and difficulty with recalling information from consultation (Poon et al., 2004; Wilson et al., 2010). Therefore, there is the need to ensure that educational materials developed as part of this research are written in concise and plain languages in order to improve readability of the intervention.

Taken together, the findings of this study point to the need for inclusion of skills and selfefficacy as key mediators in any intervention targeting behavioural change towards improved self-management among people with type 1 and type 2 diabetes. Therefore, the intervention developed and evaluated in subsequent studies (Chapter 5, 6 and 7), considered features and content targeting skills and self-efficacy in users. This was in addition to educational materials being written in plain language in order to increase the efficacy and acceptability of the intervention and the users' self-management behaviours.

8.2.3 Understanding the Perceptions of People Living with Type 1 or Type 2 Diabetes about the Features and Educational Contents Necessary for Engagement with Apps and ongoing Diabetes Self-Management.

As earlier identified in the systematic review (**Chapter 2b**), it is essential to involve end-users in app development process. Therefore, **Chapter 4** hypothesised that *needs analysis among type 1 and type 2 diabetes patients will elucidate essential features and educational contents in apps to foster retention and engagement with diabetes self-management activities*. **Chapter 4** utilised a multinational target audience of type 1 and type 2 patients from Australia, USA, Asia and UK. The primary aim was to identify the highest priority mobile app features and educational needs that encouraged participation in diabetes self-management. Such empirical evidence with an international focus is rare in the published literature. Firstly, the findings revealed that patients with diabetes valued a multi-component app with three broad features. These include features that supported monitoring of self-management activities (documentation), facilitated visual representation of reported activities (visual analytics) and provided general education and feedback related to specific blood glucose levels per time (personalised education). These findings could guide diabetes researchers to design DSMES interventions that are of more value to patients, compared to intervention designed without direct involvement of target users.

Secondly, this chapter identified patients' preferences for an app that had its contents related to problem-solving and basic guidelines on how to manage diabetes. The patients' information needs reflected their experiences of barriers to information gathering and in-depth understanding of disease management between consultations with their health care providers. Nonetheless, the stated information needs are required for optimal attainment of self-management behaviours, which could consequently prevent acute diabetes complications (Tomky et al., 2008; Lambrinou, Hanson & Beulens, 2019).

Thirdly, this chapter highlighted the conditions under which apps might be embraced by patients. The strategies recommended to mostly engage users were: ease of use, improved functionalities for healthy recipes, actionable goals, data consolidation and reliable information sources for problem solving in DSM. Again, suggestions on reliable information sources for problem solving revealed the difficulties patients had in accessing reliable information outside of consultations. These findings emphasised the importance of adopting evidence-based protocols (such as providing information for problem solving and other recommended strategies) in the development of intervention apps. Although, data consolidation was recommended to promote engagement with apps, it was not implemented in the app

225

intervention developed as part of this thesis. This was because the provision of other monitoring gadgets such as blood glucose monitors and insulin calculators were well beyond the scope of this study. Besides, data streaming from other health devices would have required a much more complex level of guaranteeing data security, privacy and accuracy of consolidated data.

Finally, this study confirmed previous findings that the use of mobile app for self-management appeared to be more common among individuals with type 1 diabetes, young, employed and of higher educational status (Boyle et al., 2017). This finding suggests that app developers and researchers need to consider the demographic characteristics of end-users and their different requirements in an intervention app. This finding further emphasises the importance of developing intervention apps with an information component that is easy to comprehend. This could improve the tendency of the intervention to serve those of lower literacy level, most likely unemployed and within the older age bracket.

Altogether, the findings of this chapter are of particular value to the wider diabetes intervention field as they describe patients' needs and the underlying contexts. Thus, the results were prioritised to inform decisions on the intervention components used in developing a brand new app as reported in Chapter 5.

8.2.4 Development of a Novel Mobile App for Diabetes Self-Management Education and Support and Assessment of its Usability among Type 1 and Type 2 Diabetes Patients.

While Chapters 2b - 4 (the first three aims of this thesis) justified and informed the design of an app intervention for patients with type 1 or type 2 diabetes, **Chapter 5** reported on the systematic process that was used to create the novel app called 'My Care Hub' app, following the guidelines produced in Chapter 2b. This consisted of an iterative development process that included the choice of underlying health behavioural constructs, selection and development of app features and educational contents based on users' preferences. Other aspects of the process included the involvement of clinical experts, data and privacy security considerations and app interphase design. These were followed by a two-stage usability testing with non-patient and patient population groups to inform the final version of the app. The app is purposely targeted to support people with type 1 or type 2 diabetes whose recommended blood glucose (BG) ranges are between 4 and 8mmol/L for type 1 and 6 and 10mmol/L for type 2 diabetes; according to the Australian guidelines (Diabetes Australia, 2015). Following the guideline is essential for accurate feedback to users in response to BG data logged into the app.

Several other smartphone apps for patients with diabetes such as mySugr, Health2Sync and Diabetes Connect currently exist, but the 'My Care Hub' app has several unique advantages: First, the 'My Care Hub' app is evidence-based with underlying theoretical mechanisms to drive behavioural change in patients. This ingrained mechanism maximises its potential to foster behavioural change (Davis et al., 2015). This is in contrast to most of the existing mHealth apps which lack theoretical models for behavior change, resulting in technologies that are not intuitive and neither useful nor evidence-based (Conroy, Yang & Maher, 2014; Chib & Lin, 2018). Second, My Care Hub provides patients with a unique and multiple combination of diabetes self-management supporting tools and education that are all accessible in a single platform. Third, the app offers monitoring documentation features for diet, blood glucose, physical activity and body weight, visual graphic tends of all monitored activities and immediate feedback in response to logged BG values. Fourth, the app also allows patients to troubleshoot key aspects of DSM that are out of the recommended BG range. Fifth, unlike most existing apps (Chomutare et al, 2011), My Care Hub highlights all the significant and fundamental differences between the self-management of type 1 and type 2 diabetes.

Findings from the first stage of the app usability testing highlighted the need to provide additional access to the documentation features of the app. This omission was immediately addressed, highlighting the importance of detecting and solving usability problems (Hoegh et al., 2006; Knuth et al., 2019). The second stage of the usability testing provided evidenced based on the participants' perceptions that the app accomplished its intended goals of being user-friendly, informative and motivational for DSM. The positive results obtained from the usability testing validated the My Care Hub novel system design and indicated that users can achieve specific DSM goals when using the app (Volentine et al., 2017).

8.2.5 Understanding the Potential of the New App for Improved Self-Management and its Acceptability

Chapters 6 and 7 tested the final hypothesis that the developed mobile phone app will be an effective and acceptable diabetes self-management education and support tool. The studies utilised participants with type 1 and type 2 diabetes residing in North Queensland, Australia. This region was chosen because of its rural and remote location, as well as its high prevalence of people living with diabetes (Queensland health report, 2018). As obtainable in other rural and remote areas of Australia, people in North Queensland also have poor accessibility to specialised diabetes support services (Australian Institute of Health and Welfare, 2019e). The studies provided several novel contributions to existing knowledge on retention, engagement, preliminary efficacy on behavioural change and acceptability of the new intervention app among end users. First, a satisfactory and high level of retention rate was demonstrated since most participants returned to use the app in weeks 2 and 3, and completed the study. During the feasibility testing, regular reminders and calls to participants to engage with the app were not used. This is because My Care Hub was designed to be an independent stand-alone app without the need for cajoling end users. This approach is important for planning future studies with My Care Hub because a satisfactory level of study completion minimises the methodological problem of accurately measuring the effectiveness of digital trials (Eysenbach, 2002). In addition, high retention minimises the problem of missing data, which undermines outcome validity and credibility of inferences from trial findings (Eysenbach, 2005); Murray et al., 2009). Therefore, the high usage and retention rate imply that the app was fully used and has the potential to retain users in future studies.

Second, the use of a theoretical framework to assess the potential of the app's components to stimulate behavioural change identified the strategies by which engagement with My Care Hub features leads to behavioural change. Several benefits of using the app were reported and patients found all the features to be useful. The 'Documentation' and 'Analytics' features provided patients with clarity to take better responsibility for their self-management. The 'Feedback' and 'Educational tips' features reinforced health providers' recommendations and general knowledge on DSM. These features also raised alerts to potential problems in BG levels and suggested appropriate solutions. Patients perceived the 'Carbohydrate components in foods' feature to be useful for planning meals, prompting self-awareness of calorie consumption and knowledge about carbohydrate and caloric values of foods. These findings suggested that the app's components supported diverse areas of DSM regimen as intended in the intervention goal.

Interventions like My Care Hub, which furnish users with information based on key elements of skills and self-efficacy in fostering DSM; coupled with provision of feedback on logged BG and graphical display of all logged lifestyle behaviours enabled patients to draw a well-defined association between lifestyle and glycemic control (Caban & Gorz, 2015). It also provided patients with reassurance and motivation for behavior modification to foster improved health outcomes (Quinn et al., 2008). Such interventions that ease monitoring of lifestyle activities improve adherence to self-management (Room et al., 2017). Therefore, the benefits of the app features coupled with the results obtained on engagement with all features during the intervention period, were evidenced-based proofs that support the inclusion of all the tested components in future versions of My Care Hub app.

Other Unique Aspects of My Care Hub

The most frequently used feature in this app was the blood glucose tracker. This observed engagement could have been due to the coupling of the feature with immediate feedback. My Care Hub's approach of providing actionable lifestyle related information and individualised education was essential for behaviour change based on logged data. This in turn, enhanced continued skills acquisition and self-efficacy; a concept closely linked to information literacy and lifelong learning. This was in stark contrast to other existing apps with engagement methods that do not provide tactics for improving skills and self-efficacy on an on-going basis.

In contrast to the findings in **Chapter 4**, indicating that older patients were less likely to use an app compared to younger patients, many of the participants retained in the pilot testing of My Care Hub were over 50 years of age. This implied that a patient's age was only a temporary barrier to the use of new technology, and older patients were willing to learn to use it if they think it will benefit them (Parker et al., 2013; Joe & Demiris, 2013). This was encouraging because it suggested that the novel app could be a useful platform for delivering diabetes interventions that include older patients who may lack experience with new technology. This current study anticipated that patients who had less experience with apps would be able to easily learn to use My Care Hub due to its simple design and user-friendly navigation.

This study highlighted the benefits of promoting the intervention to people newly diagnosed with diabetes in order to reinforce the knowledge gained during face to face consultations with healthcare providers. It is well established that diabetes patients' information needs are highest following diagnosis (Grobosch et al., 2018). Therefore, presenting an app to patients early in their consultations in order to maximise the benefits of the apps and subsequently reduce complications is essential. Nonetheless, the present study found that the app was generally
useful in enhancing improved DSM in all durations of diagnosis including those that had been diagnosed for over 15 years.

The high level of acceptability of My Care Hub was evidenced by the high ratings for ease of use, perceived value for DSM, recommendation to others with diabetes and intention for future use by participants. This finding could be interpreted using the new technology adoption model whose items were used as part of the acceptability assessment of My Care Hub (Davis, 1989; Vankatesh & Davis, 2000). The theory stipulates that when end-users are exposed to a technology and perceive it to be useful and easy to use, they are likely to accept and use it. Thus, these findings were positive indicators of high uptake of the app in the future.

Furthermore, the pilot-testing highlighted only minor suggestions for modifications of the app, indicating an overall patient satisfaction with the app design and its components. It was thought that documenting activities in the app could be made easier by including a 'Bluetooth' interface and the creation of feedback feature specific to logged physical activities data. In addition, some patients suggested information update on a regular basis in the app, believing that this will further improve engagement. Lastly, more analytic histories were suggested in future versions of the app. However, none of the suggested additional features were mentioned in the needs analysis study in Chapter 4, therefore they were either unpopular or not essential for improving self-management and engagement with apps. However, the suggested additional features could be considered in the future version of My Care Hub as long as ease of use is not in any way compromised.

Generally, the app was able to retain and engage users. It also improved behavioural change and acceptability to patients in real life settings. However, the generalisability of these findings were restricted to a particular sub-group of patients - Android smartphone owners, most of whom probably had high motivation for their self-management going by their registration of research interest with the National Diabetes Service Scheme. It is possible that patients included in the pilot testing found the app easier to use and more acceptable than their other counterparts. Therefore, future field-testing with a more generalisable sample of patients is required to further confirm the present findings.

Chapters	Major findings	Contribution to the thesis
2b	 The systematic review found that: Considerations of health behavioural theory, user and clinical expert involvement are crucial in app development. Ensuring data security and privacy in addition to pilot testing are other factors essential in the process of developing a diabetes app. Inclusion of self-management education in an app was found to improve glycaemic control. 	 Chapter 2b addressed the first Research Question – What are the necessary elements in an app development process to foster an effective diabetes self-management education and support intervention? The findings of this review gave an understanding of the essential elements required in an app development process prior to use as an intervention for self-management education and support by patients with type 1 or type 2 diabetes. This stimulated the need for the studies presented in chapters 3 and 4, which were used to gather information in preparation for the development of a novel diabetes app. This chapter also contributed to the questions used in the online survey and study interviews in chapters 3 and 4. In addition, the results of the review were used in the development of the novel diabetes app (Chapter 5).
3	• There were highly positive correlations between skill and self-efficacy in both the preparatory and pilot testing phases of the novel app ($r = 0.906$ and 0.835 respectively). In addition, the scores of skills and self-efficacy during pilot testing were significantly increased among participants, pre- and post-intervention ($P = 0.04$ and 0.03 respectively).	Chapter 3 contributed to Research Question 2 – To what extent do skills and self-efficacy stimulate behavioural change for self-management in patients with type 1 and type 2 diabetes? This first original-research component of the study established the groundwork for behavioural change mediators that were utilised in Chapters 5. In addition, the chapter shed light on the areas to accentuate (in relation to skills and self-efficacy)

Table 8.1: Major findings of each chapter and contributions to the thesis.

	 Patients' skills set was a strong predictor of self- efficacy required for diabetes self-management. Common gaps were highlighted in the areas of identifying and managing the impact of stress on diabetes, exercise planning to avoid hypoglycaemia and interpreting blood glucose patterns 	when providing educational support to patients. Thus, revealing areas to accentuate when providing educational support to patients. The Chapter was key in the development of intervention elements presented in Chapters 5. This chapter further established mHealth as an important facilitator of self- management in patients.
	• The most preferred features in a diabetes app are food nutrient composition, visual analytics, general and personalised education, trackers for blood glucose and body weight	Chapter 4 addressed Research Question 3 - What are the type 1 and type 2 diabetic patients' preferred features and educational contents in apps that will foster retention and engagement with diabetes self-management activities?
4	 Recommendations on fostering better engagement with apps were improved functionalities on healthy recipes, actionable goals with reminders, ease of use, data consolidation, customised features, certified and reliable information sources. Specific educational topics of interest to patients were approaches to problem solving and basic guidelines for the management of diabetes 	This preparatory phase of the research provided information on patients' preferences for features and education contents in apps. The findings of this study largely contributed to the development of the My Care Hub intervention reported in Chapter 5.
	 This study presents a full description of the development of a novel app (My Care Hub) and its usability testing using mixed-methods design. The first stage of usability testing of the app among non-patient populations highlighted minor 	Chapter 5 contributed to answering Research Question 4 - <i>To</i> what extent would a newly developed diabetes app stimulate improved diabetes self-management and acceptability among type 1 and type 2 diabetic patients?
5	modifications to the design which were promptly addressed.	The findings from Chapters $2b - 4$ were utilised in developing the new app. Specifically, the app development was described following the framework derived in Chapter 2b and using the

	• The second stage of the usability testing revealed that patients were satisfied with the app's performance, navigation between screens and features, quality of graphics and general visual appeal. Additionally, the app was deemed by patients to be user-friendly, informative and motivational in inducing and sustaining DSM.	mediating components of skills and self-efficacy (Chapter 3). This was in addition to the type of app features and educational components preferred by patients as well as their recommendations to improve engagement (Chapter 4).
6 and 7	 A mixed-methods pilot testing of the novel app revealed that demonstrated: High acceptability and retention rates. High levels of engagement with significantly increased DSM activities from pre-to post intervention time periods (p<.01). Effect size: 0.24, which corroborated patients' perception of the app's usability. Various perceived benefits of the app, including increased accountability, reinforcement of health providers' recommendation and., highly educational level. Perceived enhanced self-management skills and self-efficacy for long-term improved behavioural change and DSMES 	 Chapters 6 and 7 contributed to answering Research Question 2- To what extent do skills and self-efficacy stimulate behavioural change for self-management in patients with type 1 and type 2 diabetes? And Research Question 4 - To what extent would a newly developed diabetes app stimulate improved diabetes self-management and acceptability among type 1 and type 2 diabetic patients? Potential outcomes of the novel app were identified in terms of retention, engagement, behavioural change and acceptability.

8.2.6 Theoretical Underpinning Models

In chapters 1 and 2, several models and theories were cited as commonly utilised for developing and evaluating public health behavioural change interventions. It was apparent that similar elements permeate the model or theories in terms of providing insight into a disease, its impact and coping mechanisms. The theories also provide insight into the determinants of patients' engagement with care and self-management of their condition. However, major approaches relating more to fostering positive behavioural changes for improved health outcomes include utilisation of frameworks to develop interventional support for chronic diseases (Araújo-Soares et al., 2019). The self-efficacy constructs of the Social Cognitive Theory (SCT) posits that the capacity of being confident influences the thought pattern necessary to perform selfmanagement tasks (Bandura, Freeman & Lightsey, 1999). Thus, individuals could acquire and maintain behaviours through this essential mechanism (Bandura, 1998; Bandura, Freeman & Lightsey, 1999). On the other hand, Information Motivation Behavioural Skill (IMBS) model believes that information and motivation influences behavioral skills in an individual (Fisher, Fisher & Harman, 2003). Adoption of these models in this research project provided insight into patients' preferences for app components (Chapter 4). For example, the recommendation of educational components such as basic steps of managing diabetes and strategies for problemsolving might suggest that patients were attempting to increase their diabetes related information and motivation towards building their skills and self-efficacy to enable effective self-management of their conditions (Atak, Gurkan & Kose, 2008; Hill-Briggs, 2003; Stetson et al., 2010; Mohebi et al., 2013). Similarly, preferences for features to self-monitor blood glucose and behavioural activities might suggest that patients would use this feature as a mechanism to further enhance their self-efficacy (Brackney, 2018).

The self-efficacy construct of the SCT (Bandura, 1999) and information and motivation constructs of IMBS were considered particularly useful during the development of My Care

Hub due to their frameworks of building mechanism to foster successful behavior change initiative towards on-going self-management for chronic diseases (Fisher, Fisher & Harman, 2003; Ashford, Edmunds & French, 2010). Features that could be used to foster these mechanisms were selected for inclusion in the novel app based on patients' recommendations. For example, the use of self-monitoring features could enhance self-efficacy because frequent self-monitoring is a multistage process of observing and recording one's behaviour (Mace, Belfiore & Shea, 1989). The two steps involved in self-monitoring where an individual must initially identify the occurrence of a target behaviour, then self-record some aspect of the target behaviour (Craig & Nemeroff, 2004), are approaches to prevent relapse and motivate lifestyle modification (Gallagher et al., 2013). Similarly, immediate feedback on BG data provides continuous avenue either for reassurance of recommended BG levels or reinforcement of knowledge for problem solving (when needed) (Quinn et al., 2008). This builds patients' skills, motivation and confidence on how best to attain self-management goals. The visual analytic feature provided an opportunity for patients to identify patterns in their BG control in relation to lifestyle activities (physical activities, diet and weight). This provided a clearer guide towards decision making on how to self-manage (Caban & Gotz, 2015). Information on the seven essential ways to manage diabetes (Tomky et al., 2008) may further guide decisionmaking and ultimately improve patient's ability to self-manage.

Findings from the field-testing of the novel app provided further insight into the relevance of the adopted theories to this type of intervention. Reported benefits experienced by patients, including clarity of self-management activities and impact, as well as improved awareness of BG levels suggest that the app increased their skills as reported. This is in addition to reports on the app providing guidance on meal planning, knowledge provision and reinforcement. Furthermore, some patients noted that they experienced accountability, and awareness of calorie consumption as a result of using the app. These findings indicate that the app was useful for improving self-efficacy with the condition, as anticipated by the SCT. However, the development of the app for this thesis was largely investigative, so the studies carried out cannot yet confidently determine the most or least useful theories. Nevertheless, as the app is a multi-featured comprehensive intervention, it is likely that the constructs of these models and approaches discussed throughout this thesis will be relevant for the development of its future versions and long-term evaluation.

8.2.7 Strengths and Limitations of the Research

To the best of my current knowledge, this is the first study to use a systematic process to develop and examine the preliminary efficacy of a mobile phone app to provide DSMES to Australians with type 1 or type 2 diabetes. The development and reporting of My Care Hub were transparent and detailed. This transparent reporting is in accordance with 'Open Science', which encourages making materials, data, results and publications freely available for efficient scientific progress (Munafo, 2016). The app development process right from all preparatory studies, details of the design and content as well as results of pilot testing are available in open access journals. This may provide a good illustration to other researchers developing health-related behavioural change apps.

This research filled the gap on existing needs to understand the important elements required in an app development process to increase its potential as an effective self-management tool. The developed guideline for app development could provide a standard approach to the design of effective diabetes care apps and inform policy, practice and research.

The incorporation of users' preferences into My Care Hub allowed improved user engagement with the app, as reflected in the app usage data and interview with participants. While assessing common gaps in skills and self-efficacy, establishing the relationship between these variables and targeting them as mediators in the intervention led to improved health behaviours.

238

Furthermore, the pilot testing provided preliminary findings in the context of a diabetes mobile app intervention, that there is support for the role of multi-component features in an app. These features for documenting lifestyle and blood glucose, feedback and analytics and educational content enhanced skills and self-efficacy in improving diabetes self-management. Therefore, the development of the app with initial studies at the preparatory phase prior to actual development provided an evidence-based approach on which intervention components could be effective.

Another strength of the research is that assessing behavioural engagement with My Care Hub using a theoretical model provided an understanding of the usefulness and impact of all available app features. The use of this method gave insights to other diabetes app developers and researchers on how to assess their intervention components. Moreover, practice-based evidence derived from programmes implemented in real-life settings was a more suitable source of evidence for inspiring and guiding public health intervention (Ng & de Colombani, 2015). Thus, the use of field trial to pilot test the potential of the novel app is a strength of this research. Publication of the present research studies (see List of Publications) introduced a comprehensive body of internationally relevant knowledge to researchers and app developers on how to foster the potential effectiveness of a diabetes app intervention.

However, there were some limitations within this research. Acknowledgement of such limitations would assist readers in clearly understanding the scope of this research. The main limitations were time and resource constraints inherent in conducting a PhD research. Therefore, pragmatic decisions on categories and number of participants, type and time available for data collection were paramount. More pertinently, some of the processes in a diabetes pilot intervention studies were waived. For example, it would have been more ideal to include a control or comparator group in the trial to get an idea of whether the overall improvement in DSM was a 'natural improvement' or a regression to the mean - observations

that differ substantially from the true mean tend to align with observations closer to the true mean (Barnett, Van Der Pols & Dobson, 2005). Nevertheless, the choice of a comparator group when evaluating a digital behavior change intervention is an issue with multiple options such as intervention and wait list control groups or pre- and post- trials as used in the current research. No ideal option has been stipulated because each of the options has its own advantages and disadvantages (West & Michie, 2016). Furthermore, due to time and resource constraints, it was impossible to continue the pilot testing of the app for data collection of glycosylated hemoglobin (HbA1c) as an outcome measure after a minimum period of 3 months of the app use. Measuring HbA1c would have provided the opportunity to evaluate the efficacy of the intervention on glycaemic control. Although, the study did not evaluate clinical outcomes, it did a comprehensive evaluation of the app in terms of usage, behavioural engagement strategies and self-management outcomes. This facilitated a thorough understanding of how the app component engaged patients and improved self-management, which consequently foster improved glycaemic control.

The study in Chapter 3 had another section that identified some barriers to DSM. This included lack of enthusiasm towards self-management due to the chronic nature of diabetes, financial burden, work and environment related conditions and unrealistic demands from friends and family. Further studies that were carried out after that chapter were unable to make use of most of the identified barriers because of non-alignment with the overall goal of the research. However, My Care Hub components provide motivation for improved self-management and could ameliorate the issue of low interest in DSM due to its chronic nature.

A further limitation of the study is that samples of the pilot testing were drawn from the database of registrants with the National Diabetes Service Scheme Australia. Information on the key characteristics of those who declined to participate was not collected, although this may be due to not having an Android phone, which was a main recruitment criterion. The samples

240

included in the pilot study may therefore not be truly representative of the general population. The sample included in the pilot testing, apart from having an android phone, may be those who have a favourable perception of an app than those who choose not to participate. It is therefore possible that this limitation artificially inflated the positive findings in the app's pilot testing. Also, participants in the pilot testing already owned or had access to smart phones. These individuals were likely to be more familiar with the use of a smartphone than those who did not own such technology. Those who declined to participate and who did not have access to smart technology might not have found the app as easy to use as those included in the trial.

A further limitation of the pilot testing of the app is that about 95% of participants identified as White Caucasian. It is possible that other Australian ethnic groups such as the Aboriginal and Torres Strait Islanders, may have different experiences of diabetes as well as needs and perceptions regarding an app. Further testing of the app may be required with other population samples in order to support the present findings. Some studies had suggested little evidence for a digital divide by race or ethnicity (Kontos et al., 2014).

It is possible that participants' use of the app was influenced by the awareness that their activities were being recorded. Thus, the app use data may have been subjected to the Hawthorne effect, where participants alter their behaviour due to being observed (Merrett, 2006). However, in order to minimise this potential bias, participants were instructed at the start of the trial to use the app as often as they wished and to the extent to which they found it useful.

Furthermore, the lower sample size in the pilot trialing of the intervention (phase 3) in comparison to studies in its pre-development stage (phase 1) is an additional limitation. The requirement that eligibility for participation in the pilot trialing includes access to an Android smartphone could be a reason for low numbers of participants showing interest in the study in

241

relation to phase 1 of the project, which does not require the use of any intervention. However, the sample size in phase 3 was moderate, well characterized with both participants with type 1 and type 2 diabetes and similar to published pilot trials.

8.2.8 Suggestions for Future Improvement of My Care Hub and Future Research Directions

During the evaluation of the app, users expressed their concerns about the limitations in the variety of foods listed in the app. Due to the amount of time for this research, available foods with information on calories and carbohydrate contents were kept as simple as possible. It is suggested that in future, more research on other foods, calories and carbohydrate contents could be carried out in order to make the 'carbs in foods' section of the app more comprehensive. It will also be useful for the food choices to be updated on a regularly basis as users suggested.

Although My Care Hub was developed after the emergence of the using Bluetooth with app, this interface was not included in the My Care Hub due to limited funding available for the research. Future versions of My Care Hub may consider an incorporation of Bluetooth interface as recommended by participants. This will enable users to wirelessly upload their BG readings or other behavioural activities into the app, thereby reducing the burden on patients when reporting their self-monitoring activities. However, it should be noted that apart from this improvement requiring extra funding, most Bluetooth interfaces are only compatible with specific monitoring devices and corresponding mobile phones housing the app. The disadvantage of providing participants with study-provided mobile phones rather than installing an app directly on their own personal devices is that it may add an unanticipated burden on the subject with the need to use an extra mobile phone within the study period (Ben-Zeev et al, 2014; Lenhart et al., 2015). This may defeat the concept of embedding health interventions into the daily routines of patients, which is a key strength of mobile health. Therefore, the deployment of My Care Hub app directly onto personal mobile phones in future

studies is still recommended because it is the best method to improve usage and facilitate seamless integration into daily life (Goyal et al., 2017).

The results of the pilot trial demonstrated rigorous evaluation of mHealth apps for understanding user engagement, impact on behavioural outcomes and acceptability among end users. While traditionally, the next stage of evaluation of My Care Hub will be a randomised controlled trial (RCT), which has been considered as a 'gold standard' for the evaluation of interventions, a review by Pham et al (2016) emphasised that RCT may not be best suited for evaluating rapidly evolving softwares such as mHealth. Traditionally, RCT are costly, time consuming (average 5.5 years from enrolment to publication) and follow a rigid protocol, which in the context of apps might restrict the intervention to a static design. This prevents the dynamic adaptation of the app to fast emerging technological developments. Future evaluation of My Care Hub and other mHealth apps may consider the use of adaptive RCT designs. For example, a study using a sequential multiple assignment RCT would facilitate a better understanding of the impact of each app feature on an individual by evaluating outcomes at pre-determined time intervals. Future participants can be allocated to specific combinations of My Care Hub features where a re-allocation may occur based on the outcomes of individuals at a specific time point or anticipated trajectories. This adaptive study design can enable the rigorous evaluation of apps in a timely manner, while facilitating improved development of the intervention to keep pace with a rapidly evolving mHealth environment (Collins, Murphy & Strecher, 2007).

Given the limitation of the sample size in the pilot trial of My Care Hub, future research should include a sample of patients that better reflect the general population, including those from other ethnic group and patients who do not own a smart-phone. Additionally, the future sample of patients should include more recently diagnosed patients (e.g less than 6 months) in order to further explore the full range of the potential benefits of this app.

Chapter NINE: Conclusions and Recommendations

9.1 Chapter Overview

This chapter provides summary of the research and recommendations that were made based on the findings. The research focuses on enhancing diabetes self-management education and support (DSMES) using mobile phone apps. The research question specifically focused on the necessary elements in the development of a mobile app for diabetes self-management (DSM), the levels and relationship between skills and self-efficacy in patients. It also assessed patients preferred features and contents in an app, and how a novel app could improve DSM as well as its acceptability among patients. To do so efficiently and accurately, the research was conducted in three main phases (preparatory, development and feasibility phases), where each phase built on the preceding phase providing a logical and evidenced-based process to the app development.

9.1.2 Summary of the Study Findings

This research identified that the use of health behavioural theories, user and clinical expert involvements, privacy and security considerations as well as pilot testing are very essential in the development of an app targeting diabetes self-management education and support. The applicability of the result may be extended to apps targeting behavioural modification for other chronic diseases, for example apps for cancer or hypertension. This research also confirmed a strong positive correlation between skills and self-efficacy as mediating factors for diabetes self-management. In addition, some common gaps in patients' skill set and self-efficacy were identified in the areas of: identifying and managing the impact of stress on diabetes, exercise planning to avoid hypoglycemia as well as interpreting and adjusting food intake to ensure targeted blood glucose levels are reached. Given that these findings were reported by an international audience, it may be inferred that these are common areas of difficulties experienced by many patients. Hence, it is recommended that health professionals involved in the management in diabetes lay more emphasis on these areas when providing consultations to patients.

Furthermore, this research provided evidence that multicomponent features, which could aid self-management and access to educational contents on basic steps for diabetes care and problem solving, are preferred in a diabetes app. The blended constructs of social cognitive theory and information motivation behavioural theories applied into developing a novel app intervention (My Care Hub) to provide the preferred components appeared to be effective. My Care Hub aimed to support self-management and education in people with type 1 or type 2 diabetes. The app was specifically designed for those who have type 1 diabetes with recommended blood glucose levels (BGL) of 4-8mmol/L-fasting and <10 mmol/L-2 hours postprandial, and fasting levels of 6-8 mmol/L and 2-hour post prandial levels of 6-10 mmol/L for those with type 2 diabetes. The app can promote level of skills, self-efficacy and diabetes self-management activities outcomes with a small or moderate effect size (Eta squared raged from 0.11 to 0.25). The app encouraged behavioural change through different strategies including accountability; clarity of self-management activities and impact; mindfulness of calorie consumption; motivation for self-management; and knowledge improvement and reinforcement. Overall, the results of My Care Hub app on usability, retention and engagement of users, efficacy on behavioural change and acceptability may be interpreted as a positive preliminary indicator of the potential efficacy and uptake of future versions of the app. This information could be used to design future versions, longer-term and larger scale evaluation of the app. This study also provides evidence and adds to the body of knowledge that delivering an app intervention entailing multiple components (features), can provide multiple exposure to different behavioural change strategy to consequently influence behaviour change. In addition,

the study supports previous evidence that behaviour change initiatives should have components that improve skills and build the self-efficacy of patients.

The pilot testing of the My Care Hub was conducted in North Queensland, Australia. In general, as demonstrated by the results, the app has the capability to improve self-management in Australians with type 1 or type 2 diabetes with the previously stipulated recommended blood glucose levels. Replication of the app evaluation in various settings would be beneficial. The knowledge obtained from this study could enhance future development of mobile apps to support diabetes self-management and education. This requires taking steps to identify relevant health behavioural theories to target as mediating factors to underpin the behavioural change in the intervention coupled with app features and educational contents preferred by patients. This is followed by the use of the aforementioned acquired information to develop an app and its comprehensive evaluation for usability, retention, usage, behavioural change and acceptability among target users.

9.2 RECOMMENDATIONS.

9.2.1 Patients

Attaining adequate level of skills and self-efficacy for self-management could be challenging for people with diabetes (Aljasem et al., 2001, Persell et al., 2004; Smoorenburg et al., 2019), because of the multiple areas of the required regimen (Tomky et al., 2008; Lambrinou et al., 2019), interrelationship between these regimen and overall effect on health outcomes (Herschbach et al., 1997; Tomky et al., 2008; Lambrinou et al., 2019). As noted in this study, attaining adequate levels of skills and self-efficacy on how to identify and manage the impact of stress on diabetes, how to plan exercise to avoid hypoglycaemia are common areas of regimen many patients often find difficult. Other areas include interpreting blood glucose patterns and adjusting food intake to reach targeted blood glucose levels. Assessing the services

of care team members such as diabetes educators, exercise physiologists and dieticians could assist in ameliorating such struggles. This is highly recommended in the early years of diagnosis, after which, knowledge and skills acquired could be complemented and reinforced on an ongoing basis through access to support and education in a reliable diabetes app.

9.2.2. Health Professionals

Internationally including Australia, healthcare systems are challenged with the rising rates of diabetes coupled with inability of many patients to adhere to recommended self-management strategies and the resulting complications. People with type 1 or type 2 diabetes living in rural and remote areas of Australia have a disproportionate number of complications attributed to compromised self-management (Wan et al., 2008; Australian Government Department of Health, 2015). Under-resourced diabetes education and limited access to speciality care services and community resources are common in those areas (Maguire et al., 2008; Wakerman et al., 2008; Australian Institute of Health and Welfare, 2018; 2019f) and could encourage compromised self-management. Lacking regular access to education and support may predispose patients to little or no understanding of the implications of diabetes complications and insufficient skills to handle complex self-management regimens (e.g when best to monitor blood glucose, planning exercise to avoid extreme low blood glucose, best type of foods for consumption) (Burke, Sheer & Lipman, 2014, Powers et al., 2017). Hence, health systems need to accommodate changes to bridge these gaps in order to enable patients meet their information and support needs in outpatient settings. Therefore, leveraging the potential of apps to provide these education and support is recommended. This is especially important in rural and remote areas where there is increasing rates of diabetes, its complications and yet low access to speciality healthcare services for ongoing DSMES (Wan et al., 2008; Australian Government Department of Health, 2015).

Apps have the potential to decrease barriers such as patient burden, cost and low motivation for adherence; to traditional prescriptions for behavioural change of self-management in patients. The use of apps could be a medium to reduce these barriers particularly when the apps are engaging, convenient and easy to use as found in the novel app (My Care Hub) developed in this research. My Care Hub encompasses more than a didactic, educational program and goes beyond simple dissemination of information on diabetes and its management. The pivotal objective of the novel app is to change behaviour to produce sustainable effects. These are intended to be achieved through improved patients' skills and self-efficacy to manage their diabetes through regular access to information, motivation and problem-solving support provided by the app. The research results revealed that the use of My Care Hub supports diabetes patients to self-manage their health and leads to improvements in behavioural outcomes.

9.2.3 Researchers and App Developers

Researchers and app developers should ensure that they develop diabetes app using the best available evidence. This research has identified important considerations when developing an app such as health behavioural theory, users and clinical expert involvement, ensuring data privacy and security and pilot testing. In addition, applying the elements of theoretical models such as skills and self-efficacy will foster the understanding of behavioural change that patients are going through when using the app (Norris, Engelgau & Narayan, 2001; Hilliard et al., 2016; Fitzgerald & McClelland (2016); Hekler et al., 2017; Klonoff, 2019). Co-developing apps with patients is necessary to improve their appeal, acceptability, uptake and continual engagement with the technology (McCurdi et al., 2012; Woods et al., 2018; Woods et al., 2019). Furthermore, seeking the opinions of clinical experts when developing apps will confirm the quality and relevance of the intervention components. It will also ascertain the health security of patients by ensuring that information provided by the app is not misleading. In addition, procedure must be in place for app data security and privacy, to ensure that personal health information of patients are not accessible to unauthorised persons. These aforementioned processes are then followed by testing of the developed app using a carefully phased approach, starting with a series of usability and pilot testing and moving to a definitive evaluation using long-term and larger scale studies.

In addition, it is essential for app developers and researchers to note that one of the reasons why app interventions are better and preferred interventions for DSMES than other delivery methods such as SMS is the ability to monitor desired behavioural or clinical outcomes and provide feedback (Cushing & Steele, 2010; Goyal & Cafazzo, 2013; Hartin et al., 2016). This gives the app an opportunity to go beyond providing general education regarding DSM by including more interactive levels of user interaction. The provision of health information alone is less likely to stimulate meaningful and long-term behaviour change (Molaison, 2002) but integration of self-monitoring and a two-way interaction such as providing feedback to users are significant elements of success in modifying behaviour (Payne et al., 2015; Hermsen et al., (2016). This is also supported by the results of this research.

Researchers involved in the management of diabetes may likely lack the technical ability to develop apps independently. Nevertheless, they specialise in understanding behavioural change theories and their attending constructs to carry out research related to patients' preferences, usability and pilot testings of developed apps. While on the other hand, app developers have the technical expertise relating to software development. Collaborating with each other and capitalising on each discipline's strengths will be essential to produce diabetes apps capable of providing behavioural change support and meeting the self-management needs of patients. Such apps may prove to be valuable tools for healthcare professionals in their efforts to address the current diabetes epidemic and its resulting complications.

REFERENCES

- Abras C, Maloney-Krichmar D, Preece J (2004). User-centered design. Bainbridge, W. In: Encyclopedia of Human-Computer Interaction. Thousand Oaks: Sage: 445-456.
- Ackerman L. Mobile health and fitness applications and information privacy. Report to the California Consumer Protection Foundation. Privacy Rights Clearinghouse, San Diego, CA. 2013:1-26. Available at: <u>https://privacyrights.org/sites/default/files/pdfs/mobile-</u> medical-apps-privacy-consumer-report.pdf
- Adu MD, Malabu UH, Callander EJ, Malau-Aduli AE, Malau-Aduli BS (2018a). Considerations for the development of mobile phone apps to support diabetes selfmanagement: Systematic Review. *JMIR Mhealth Uhealth*.6(6):e10115.
- Adu MD, Malabu UH, Malau-Aduli AE, Malau-Aduli BS (2018b). Users' preferences and design recommendations to promote engagement with mobile apps for diabetes self-management: Multi-national perspectives. *PLoS One*. 13(12):e0208942.
- Adu MD, Malabu UH, Malau-Aduli AE, Malau-Aduli BS (2019). Enablers and Barriers to diabetes self-management: Multi-national investigation. *PloS One*. 14(6):e0217771.
- Adu MD, Malabu UH, Malau-Aduli AEO, Malau-Aduli BS (2020a). The development of My Care Hub Mobile-phone App to Support Self-Management in Australians with Type 1 or Type 2 Diabetes (2020a). *Sci Rep*.10(1):1-10.
- Adu MD, Malabu UH, Malau-Aduli AEO, Drovandi A, Malau-Aduli BS (2020b). User retention and engagement with My Care Hub Mobile-phone App to Support Self-Management in Australians with Type 1 or Type 2 Diabetes (My Care Hub): Mixed methods study. *JMIR Uhealth Mhealth*.8(6):e17802.
- ADVANCE Collaborative Group, Patel A, MacMahon S, Chalmers J, Neal B, Billot L, et al (2008). Intensive blood glucose control and vascular outcomes in patients with type 2 diabetes. *N Engl J Med.* 358(24):2560-2572.
- Agarwal G, Gaber J, Richardson J, Mangin D, Ploeg J, Valaitis R, Reid GJ, Lalarche L, Parascandalo F, Javadi D. et al (2019a). Pilot randomised controlled trial of a complex intervention for diabetes self-management supported by volunteers, technology, and interprofessional primary health care teams. *Pilot Feasibility Stud.* 5(1):118

- Agarwal P, Mukerji G, Desveaux L, Ivers NM, Bhattacharyya O, Hensel JM, et al (2019b). Mobile app for improved self-management of type 2 diabetes: multicenter pragmatic randomised controlled trial. *JMIR Mhealth Uhealth*.7(1):e10321.
- Akter S, Ray P (2010). mHealth an Ultimate Platform to Serve the Unserved. *Yearb Med Inform*. 94-100.
- Al-Azri M, Al-Azri H, Al-Hashmi F, Al-Rasbi S, El-Shafie K, Al-Maniri A (2011). Factors Affecting the Quality of Diabetic Care in Primary Care Settings in Oman: A qualitative study on patients' perspectives. *Sultan Qaboos Uni Med J.* 11(2):207.
- Alberti KGMM, Zimmet PZ (1998). Definition, diagnosis and classification of diabetes mellitus and its complications. Part 1: diagnosis and classification of diabetes mellitus. Provisional report of a WHO consultation. *Diabet Med.* 15(7):539-553.
- Alemdag S (2018). The role of social self-efficacy on physical activity: a cross-cultural comparision. *J Educ Train Stud.* 6(5):40-46.
- Aljasem LI, Peyrot M, Wissow L, Rubin RR (2001). The impact of barriers and self-efficacy on self-care behaviors in type 2 diabetes. *Diabetes Educ*. 27(3):393-404.
- Al-Rasheedi AAS (2014). The role of educational level in glycemic control among patients with type II diabetes mellitus. *Int J Health Sci (Qassim)*. 8(2):177.
- Al-Taee M, Al-Nuaimy W, Muhsin Z, Al-Ataby A, Al-Taee A (2016). Mapping security requirements of mobile health systems into software development lifecycle. Presented at: Proceedings of the 9th International Conference on Developments in eSystems Engineering (DeSE); Liverpool, UK;87-93.
- Alvarado-Martel D, Ruiz Fernández M, Cuadrado Vigaray M, Carrillo A, Boronat M, Expósito
 Montesdeoca A, et al (2019). Identification of Psychological Factors Associated with
 Adherence to Self-Care Behaviors amongst Patients with Type 1 Diabetes. J Diabetes
 Res. 1-9.
- American Association of Diabetes Educators (AADE) (2008). AADE7 Self-Care Behaviours. *Diabetes Educ.* 34(3):445-449.
- American Diabetes Association (2002). Standards of medical care for patients with diabetes mellitus. *Diabetes Care*. 25(1):213-229.
- American Diabetes Association (2010). Standards of medical care in diabetes. *Diabetes Care.*;33(Supplement 1):S11-S61.
- American Diabetes Association (2015). Standards of medical care in diabetes-abridged for primary care providers. *Clin Diabetes*. 33(2):97-111.

- American Diabetes Association (2016). Standards of medical care in diabetes-abridged for primary care providers. *Clin Diabetes*. 34(1):3-21.
- American Diabetes Association (2017a). Pharmacologic approaches to glycemic treatment. *Diabetes Care*. 40(Supplement 1):S64-S74.
- American Diabetes Association (2017b). Classification and diagnosis of diabetes. *Diabetes Care*. 40(Supplement 1):S11-S24.
- American Diabetes Association (2017c). Standards of medical care in diabetes—abridged for primary care providers. *Clinical Diabetes*. 35(1):5.
- American Diabetes Association (2019). Lifestyle management: standards of medical care in diabetes—2019. *Diabetes Care*.42(Supplement 1):S46-S60.
- Ancker JS, Witteman HO, Hafeez B, Provencher T, Van de Graaf M, Wei E (2015). "You get reminded you're a sick person": Personal data tracking and patients with multiple chronic conditions. *J Med Internet Res.* 17(8).
- Annesi JJ (1998). Effects of computer feedback on adherence to exercise. *Percept Mot Skills*. 87(2):723-730.
- Aranda-Jan CB, Mohutsiwa-Dibe N, Loukanova S (2014). Systematic review on what works, what does not work and why of implementation of mobile health (mHealth) projects in Africa. *BMC Public Health*. 14(1):188.
- Araújo-Soares V, Hankonen N, Presseau J, Rodrigues A, Sniehotta FF (2019). Developing behavior change interventions for self-management in chronic illness: An integrative overview. *Eur Psychol.* 24(1):7.
- Arnhold M, Quade M, Kirch W (2014). Mobile applications for diabetics: a systematic review and expert-based usability evaluation considering the special requirements of diabetes patients age 50 years or older. *J Med Internet Res.* 16(4):e104.
- Aromataris E, Fernandez R, Godfrey CM, Holly C, Khalil H, Tungpunkom P (2015). Summarizing systematic reviews: methodological development, conduct and reporting of an umbrella review approach. *Int J Evid Based Health.* 13(3):132-140.
- Aronson R, Brown RE, Jiandani D, Walker A, Orzech N, Mbuagbaw L (2018). Assessment of self-management in patients with diabetes using the novel LMC Skills, Confidence and Preparedness Index (SCPI). *Diabetes Res Clinc Pract*.137:128-136.
- Arsand E (2009). User-Involved Design of mobile self-help tools for people with diabetes. The Few Touch Digital Diabetes Diary. Available at: <u>https://munin.uit.no/handle/10037/2762</u>

- Årsand E, Demiris G (2008). User-centered methods for designing patient-centric self-help tools. *Inform Health Social Care*. 33(3):158-169.
- Årsand E, Frøisland DH, Skrøvseth SO, Chomutare T, Tatara N, Hartvigsen G, et al (2012). Mobile health applications to assist patients with diabetes: lessons learned and design implications. *J Diabetes Sci Technol*. 6(5):1197-1206.
- Arsand E, Tatara N, Østengen G, Hartvigsen G (2010). Mobile phone-based self-management tools for type 2 diabetes: the few touch application. *J Diabetes Sci Technol*. 4(2): 328-336.
- Arsand E, Tufano JT, Ralston JD, Hjortdahl P (2008). Designing mobile dietary management support technologies for people with diabetes. *J Telemed Telecare*. 14(7):329-332.
- Ashford S, Edmunds J, French DP (2010). What is the best way to change self-efficacy to promote lifestyle and recreational physical activity? A systematic review with meta-analysis. *Br J Health Psychol*. 15(2):265-288.
- Atak N, Gurkan T, Kose K (2008). The effect of education on knowledge, self-management behaviours and self-efficacy of patients with type 2 diabetes. *Aust J Adv Nur*. 26(2):666-674.
- Australian Bureau of Statistics (2015). National health survey: First results, 2014–15. ABS Cat No 43640 55001. Available at: <u>https://www.ausstats.abs.gov.au/Ausstats/subscriber.nsf/0/CDA852A349B4CEE6CA</u> <u>257F150009FC53/\$File/national%20health%20survey%20first%20results,%202014-</u> <u>15.pdf</u>
- Australian Diabetes Educators Association (2015). Role and scope of practice for credentialled diabetes educators in Australia. Available at: <u>https://www.adea.com.au/Tenant/C0000017/Credentialling/Mentoring%20Program/R</u> <u>ole-and-Scope-of-Practice-for-Credentialled-Diabetes-Educators-in-Australia-Final1.pdf</u>
- Australian Government Department of Agriculture. Regional Postcode delivery classification system (2019a). Accessed 25th December, 2019. Available from: <u>https://www.agriculture.gov.au/import/online-services/delivery-</u> postcode/summary#queensland
- Australian Government Department of Agriculture, Water and Environment (2019). RegionalPostcodedeliveryclassificationsystem.Availableat:

https://www.agriculture.gov.au/import/online-services/deliverypostcode/summary#queensland

- Australian Government Department of Health (2015). Australian National Diabetes Strategy 2016–2020. Available at: https://www.health.gov.au/sites/default/files/documents/2019/09/australian-nationaldiabetes-strategy-2016-2020_1.pdf
- Australian Government Department of Health (2017). Burden of disease and injury: Summary

 results
 for
 Queensland.
 Available
 at:

 https://www.health.qld.gov.au/ data/assets/pdf_file/0025/684214/burden-of-disease

 report-2017.pdf_(9)
- Australian Government Federal Register of Legislation (2016). Privacy Amendment (Notifiable Data Breaches) Act 2017 No. Available at: <u>http://www.legislation.gov.au/Details/C2017A00012</u>
- Australian Government Federal Register of Legislation (2017). Public Governance, Performance and Accountability (Establishing the Australian Digital Health Agency) Rule. Available at: <u>https://www.legislation.gov.au/</u>
- Australian Institute of Health and Welfare (2013). Diabetes expenditure in Australia 2008-09. Available at: <u>https://www.aihw.gov.au/reports/diabetes/diabetes-expenditure-in-australia-2008-09/contents/table-of-contents</u> <u>Chapt</u>
- Australian Institute of Health and Welfare (2014). Cardiovascular disease, diabetes and chronic kidney disease: Australian facts: prevalence and incidence. Cardiovascular, diabetes and chronic disease series. no 2. Cat. no. CDK 2. Canberra: AIHW. Available at: <u>https://www.aihw.gov.au/getmedia/0ce5f234-0abf-41b9-a392-</u>

be5dd1e94c54/17034.pdf.aspx?inline=true

- Australian Institute of Health and Welfare (2015). Australian Burden of Disease Study: Impact and causes of illness and death in Australia. Australian Burden of Disease series no.19. Cat.no.BOD 22. Canberra. Available at: <u>https://www.aihw.gov.au/getmedia/c076f42f-61ea-4348-9c0a-d996353e838f/aihw-bod-22.pdf.aspx?inline=true</u>
- Australian Institute of Health and Welfare (2017). Potentially preventable hospitalisations in 2015–16. Sydney, NSW: National Health Performance Authority. 2015. Available at: https://www.aihw.gov.au/getmedia/14ba43a3-9ae9-4b19-beb9-e9c714e2fb3f/aihw-mhc-hpf-37-technical-note.pdf.aspx
- Australian Institute of Health and Welfare (2018). Survey of Health Care: selected findings for rural and remote Australians. Available at: https://www.aihw.gov.au/reports/rural-

remote-australians/survey-health-care-selected-findings-rural-

remote/contents/summary

- Australian Institute of Health and Welfare (2019a). Diabetes Overview in Reports and Data. Available at: https://www.aihw.gov.au/reports-data/health-conditions-disabilitydeaths/diabetes/overview
- Australian Institute of Health and Welfare (2019b). Diabetes.Cat.no.CVD 82. Canberra. Available at: https://www.aihw.gov.au/getmedia/5020b399-3e7e-4762-852e-21d99769c8be/Diabetes.pdf.aspx?inline=true
- Australian Institute of Health and Welfare (2019c). Cardiovascular disease, diabetes and chronic kidney disease: Australian facts: prevalence and incidence: Cardiovascular, diabetes and chronic kidney disease (CKD). CKD 2 (2). Available at: https://www.aihw.gov.au/reports/heart-stroke-vascular-disease/cardiovascular-diabetes-chronic-kidney-prevalence/contents/table-of-contents
- Australian Institute of Health and Welfare (2019d). Rural and remote health. Available at: <u>https://www.aihw.gov.au/reports/rural-remote-australians/rural-remote-health/contents/summary</u>
- Australian Institute of Health and Welfare (2019e). Diabetes. Available at: <u>https://www.aihw.gov.au/reports-data/health-conditions-disability-</u> <u>deaths/diabetes/overview</u>

Australian Institute of Health and Welfare (2019f). Rural and remote health. Available at:

- https://www.aihw.gov.au/reports/rural-remote-australians/rural-remotehealth/contents/access-to-health-care
- Australian Institute of Health and Welfare (2020). Diabetes. Available at: <u>https://www.aihw.gov.au/reports-data/health-conditions-disability-</u> deaths/diabetes/overview
- Australian New Zealand. Food Standards. Australian Food, Supplement and Nutrient (AUSNUT) database 2011-13. Available at:

http://www.foodstandards.gov.au/science/monitoringnutrients/ausnut/foodnutrient/Pa ges/default.aspx

Avery L, Charman SJ, Taylor L, Flynn D, Mosely K, Speight J, et al (2015). Systematic development of a theory-informed multifaceted behavioural intervention to increase physical activity of adults with type 2 diabetes in routine primary care: Movement as Medicine for Type 2 Diabetes. *Implement Sci.* 11(1):99.

- Balfe M, Doyle F, Smith D, Sreenan S, Brugha R, Hevey D, et al (2013). What is distressing about having type 1 diabetes? A qualitative study of young adults' perspectives. *BMC Endocr Disord*. 13(1):25.
- Bandura A (1994). Self-effficacy. In Ramachaudran, VS (Ed) Encyclopedia of Human Behaviour. Academic. USA; 71-81.
- Bandura A (1998). Health promotion from the perspective of social cognitive theory. *Psychology Health*.13(4):623-649.
- Bandura A (2001). Social cognitive theory: An agentic perspective. *Annu Rev Psychol*. 52(1):1-26.
- Bandura A (2004). Health promotion by social cognitive means. *Health Educ Behav*. 31(2):143-64.
- Bandura A and National Institute of Mental Health (1998). Social foundations of thought and action: A social cognitive theory. *Behav Change*. 5(1):37-38.
- Bandura A, Freeman W, Lightsey R (1999). Self-efficacy: The exercise of control. J Cog Psychother.13(2). 158-166.
- Baranowski T, Anderson C, Carmack C (1998). Mediating variable framework in physical activity interventions: How are we doing? How might we do better? *Am J Prev Med*. 15(4):266-297.
- Barisic A, Leatherdale ST, Kreiger N (2011). Importance of frequency, intensity, time and type (FITT) in physical activity assessment for epidemiological research. *Can J Public Health*. 102(3):174-175.
- Barling J, Beattie R (1988). Self-efficacy beliefs and sales performance. J Organ Behav Manage. 5(1): 41-51.
- Barlow J, Wright C, Sheasby J, Turner A, Hainsworth J (2002). Self-management approaches for people with chronic conditions: a review. *Patient Educ Couns*. 48(2):177-187.
- Barnett AG, Van Der Pols JC, Dobson AJ (2005). Regression to the mean: what it is and how to deal with it. *Int J Epidemiol*. 34(1):215-220.
- Barton AJ (2012). The regulation of mobile health applications. BMC Med. 10:46.
- Bassett MT (2005). Diabetes is epidemic. Am J Public Health. 95(9):1496.
- Basu R, Johnson ML, Kudva YC, Basu A (2014). Exercise, hypoglycemia, and type 1 diabetes. *Diabetes Technol Ther*. 16(6): 331-337.

- Beck J, Greenwood DA, Blanton L, Bollinger ST, Butcher MK, et al (2019). 2017 National standards for diabetes self-management education and support. *Diabetes Educ*. 45(1): 34-49.
- Beck J, Greenwood DA, Blanton L, Bollinger ST, Butcher MK, Condon JE, et al (2018). National standards for diabetes self-management education and support. *The Diabetes Educ*.44(1):35-50.
- Beckerle CM, Lavin MA (2013). Association of self-efficacy and self-care with glycemic control in diabetes. *Diabetes Spectr*. 26(3):172-178.
- Bender MS, Choi J, Arai S, Paul SM, Gonzalez P, Fukuoka Y (2014). Digital technology ownership, usage, and factors predicting downloading health apps among Caucasian, Filipino, Korean, and Latino Americans: the digital link to health survey. *JMIR Mhealth Uhealth*. 2(4):e43.
- Benferdia Y, Zakaria N (2014). A Systematic Literature Review of Content-Based Mobile Health. In Journal of Information Systems Research and Innovation. Available at: <u>https://seminar.utmspace.edu.my/jisri/download/Special%20Issue-Dec2014/</u> A%20systematic%20Literature.pdf
- Bennett IM, Chen J, Soroui JS, White S (2009). The contribution of health literacy to disparities in self-rated health status and preventive health behaviors in older adults. *Ann Fam Med*. 7(3):204-211.
- Ben-Zeev D (2012). Mobile technologies in the study, assessment, and treatment of schizophrenia. *Schizophr Bull*.38(3):384-385.
- Ben-Zeev D, Schueller SM, Begale M, Duffecy J, Kane JM, Mohr DC (2015). Strategies for mHealth research: lessons from 3 mobile intervention studies. *Adm Policy Ment Health*. 42(2):157-167.
- Bergin SM, Nube VL, Alford JB, Allard BP, Gurr JM, Holland EL, et al (2013). Australian diabetes foot network: practical guideline on the provision of footwear for people with diabetes. *J Foot Ankle Res.* 6(1):6.
- Berndt RD, Takenga C, Preik P, Kuehn S, Berndt L, Mayer H, et al (2014). Impact of information technology on the therapy of type-1 diabetes: a case study of children and adolescents in Germany. *J Pers Med.* 4(2):200-217.
- Bevan N (2009). International standards for usability should be more widely used. *J Usability Stud.* 4(3):1931-3357.
- Bickmore T, Schulman D, Yin L (2010). Maintaining engagement in long-term interventions with relational agents. *Appl Artif Intell*. 24(6):648-666.

- Bidargaddi N, Pituch T, Maaieh H, Short C, Strecher V (2018). Predicting which type of push notification content motivates users to engage in a self-monitoring app. *Prev Med Rep.* 11: 267-273.
- Bodenheimer T, Lorig K, Holman H, Grumbach K (2002). Patient self-management of chronic disease in primary care. *JAMA*. 288(19):2469-2475.
- Boger E, Ellis J, Latter S, Foster C, Kennedy A, Jones F, et al (2015). Self-management and self-management support outcomes: a systematic review and mixed research synthesis of stakeholder views. *PLoS One*.10(7):e0130990.
- Bolen SD, Chandar A, Falck-Ytter C, Tyler C, Perzynski AT, Gertz AM, et al (2014). Effectiveness and safety of patient activation interventions for adults with type 2 diabetes: systematic review, meta-analysis, and meta-regression. J Gen Intern Med. 29(8):1166-1176.
- Bonger Z, Shiferaw S, Tariku EZ (2018). Adherence to diabetic self-care practices and its associated factors among patients with type 2 diabetes in Addis Ababa, Ethiopia. *Patient Prefer Adherence*. 12:963.
- Booth AO, Lowis C, Hunter SJ, Dean M, Cardwell CR, McKinley MC (2016). Development and evaluation of a computer-based, self-management tool for people recently diagnosed with type 2 diabetes. *J Diabetes Res.2016:31926673*.
- Boren SA, Leo GD, Chanetsa FF, Donaldson J, Krishna S, Balas EA (2006). Evaluation of a diabetes education call center intervention. *Telemed J E Health*. 12(4):457-465.
- Borsboom D, Mellenbergh GJ, van Heerden J (2003). The theoretical status of latent variables. *Psychol Rev.* 110(2):203-219.
- Boschen MJ, Casey LM (2008). The use of mobile telephones as adjuncts to cognitive behavioral psychotherapy. *Professional Psychol: Res Pract*.39(5):546-552.
- Boulos MNK, Brewer AC, Karimkhani C, Buller DB, Dellavalle RP (2014). Mobile medical and health apps: state of the art, concerns, regulatory control and certification. *Online J Public Health Inform.* 5(3):229.
- Bowen DJ, Jrueter M, Spring B, Cofta-Woerpel L, Linnan L, Weiner D, et al (2009). How we design feasibility studies. *Am J Prev Med.* 36(5): 452-457.
- Boyle L, Grainger R, Hall RM, Krebs JD (2017). Use of and beliefs about mobile phone apps for diabetes self-management: surveys of people in a hospital diabetes clinic and diabetes health professionals in New Zealand. *JMIR Mhealth Uhealth*. 5(6):e85.

- Brackney DE (2018). Enhanced self-monitoring blood glucose in non-insulin-requiring Type 2 diabetes: A qualitative study in primary care. *J Clin Nurs*. 27(9-10):2120-2131.
- Brandell B, Ford C (2013). Diabetes professionals must seize the opportunity in mobile health. *J Diabetes Sci Technol.* 7(6):1616-1620.
- Braun V, Clarke V (2006). Using thematic analysis in psychology. *Qual Res Psychol*.3(2):77-101.
- Breland JY, Yeh VM, Yu J (2013). Adherence to evidence-based guidelines among diabetes self-management apps. *Transl Behav Med.* 3(3):277-286.
- Brooke, J (1996). SUS-A quick and dirty usability scale. In Jordan PW, Thomas B, Weerdmeseester BA, McClelland AL. (Eds) Usability evaluation in industry. Taylor Francis, London,189(194):4-7.
- Brown JB, Harris SB, Webster-Bogaert S, Wetmore S, Faulds C, Stewart M (2002). The role of patient, physician and systemic factors in the management of type 2 diabetes mellitus. *Fam Pract.* 19(4):344-349.
- Buijink AWG, Visser BJ, Marshall L (2013). Medical apps for smartphones: lack of evidence undermines quality and safety. *Evidence Based Med.* 18 (3): 90-92.
- Burby J, Brown A, Committee WS (2007). Web analytics definitions. Web Analytics Association, Washington DC. Available at: <u>http://www.webanalytiker.dk/wp-</u> <u>content/logo/blog/WAA-Standards-Analytics-Definitions.pdf</u>
- Burge MR (2001). Lack of compliance with home blood glucose monitoring predicts hospitalization in diabetes. *Diabetes Care*.24(8):1502-1503.
- Burge MR, Mitchell S, Sawyer A, Schade DS (2008). Continuous glucose monitoring: the future of diabetes management. *Diabetes Spectr*. 21(2):112-119.
- Burke SD, Sherr D, Lipman RD (2014). Partnering with diabetes educators to improve patient outcomes. *Diabetes Metab Syndr Obes*.7:45.
- Bus S, Armstrong D, Van Deursen R, Lewis JE, Caravaggi C, Cavanagh P, et al (2016). IWGDF guidance on footwear and offloading interventions to prevent and heal foot ulcers in patients with diabetes. *Diabetes Metab Res Rev.* 32:25-36.
- Byers D, Garth K, Manley D, Chlebowy D (2016). Facilitators and barriers to Type 2 diabetes self-management among rural African American adults. J Health Dispar Res Pract. 9(1):164-174.
- Caban JJ, Gotz D (2015). Visual analytics in healthcare–opportunities and research challenges. *J Am Med Inform Assoc*.22(2):260-262.

- Cagle CS, Appel S, Skelly AH, Carter-Edwards L (2002). Mid-life African-American women with type 2 diabetes: influence on work and the multicaregiver role. *Ethn Dis.* 12(4):555-566.
- Campbell DJ, Manns BJ, Hemmelgarn BR, Sanmartin C, Edwards A, King-Shier K (2017). Understanding financial barriers to care in patients with diabetes: an exploratory qualitative study. *Diabetes Educ*. 43(1):78-86.
- Campbell NC, Murray E, Darbyshire J, Emery J, Farmer A, Griffiths F, et al (2007). Designing and evaluating complex interventions to improve health care. *BMJ*. 334(7591):455-459.
- Carroll JK, Moorhead A, Bond R, LeBlanc WG, Petrella RJ, Fiscella K (2017). Who uses mobile phone health apps and does use matter? A secondary data analytics approach. J Med Internet Res. 19(4).
- Carter PA (1998). Self-care agency: the concept and how it is measured. *J Nurs Meas*.6(2):195-207.
- Catalyst N (2017). What Is Patient-Centered Care? NEJM Catalyst.3(1).
- Cerin E, MacKinnon DP (2009). A commentary on current practice in mediating variable analyses in behavioural nutrition and physical activity. *Public Health Nutr*. 12(8):1182-1188.
- Chao J, Nau DP, Aikens JE, Taylor SD (2005). The mediating role of health beliefs in the relationship between depressive symptoms and medication adherence in persons with diabetes. *Res Soc Adm Pharm.* 1(4):508-525.
- Charpentier G, Benhamou P, Dardari D, Clergeot A, Franc S, Schaepelynck-Belicar P, et al (2011). The Diabeo software enabling individualised insulin dose adjustments combined with telemedicine support improves HbA1c in poorly controlled type 1 diabetic patients: a 6-month, randomised, open-label, parallel-group, multicenter trial (TeleDiab 1 Study). *Diabetes Care*. 34(3):533-539.
- Chen Y, Sloan FA, Yashkin AP (2015). Adherence to diabetes guidelines for screening, physical activity and medication and onset of complications and death. *J Diabetes Complications*. 29(8):1228-1233.
- Cheng L, Leung DY-p, Sit JW-H, Li X-m, Wu Y-n, Yang M-y, et al (2016). Factors associated with diet barriers in patients with poorly controlled type 2 diabetes. *Patient Pref Adher*. 10:37-44.
- Chhanabhai P, Holt A (2007). Consumers are ready to accept the transition to online and electronic records if they can be assured of the security measures. *MedGenMed*. 9(1):8.

- Chiauzzi E, Rodarte C, DasMahapatra P (2015). Patient-centered activity monitoring in the self-management of chronic health conditions. *BMC Med.* 13:77.
- Chib A, Lin SH (2018). Theoretical advancements in mHealth: A systematic review of mobile apps. *J Health Commun*.23(10-11):909-955.
- Chlebowy DO, Hood S, LaJoie AS (2010). Facilitators and barriers to self-management of type 2 diabetes among urban African American adults. *Diabetes Educ*. 36(6):897-905.
- Chlebowy DO, Hood S, LaJoie AS (2013). Gender differences in diabetes self-management among African American adults. *West J Nurs Res.* 35(6):703-721.
- Cho H, Yen P, Dowling D, Merill JA, Schnall R (2018). A multi-level usability evaluation of mobile health applications: A case study. *J Biomed Inform*. 86:79-89.
- Cho J (2016). The impact of post-adoption beliefs on the continued use of health apps. *Int J Med Inform.* 87: 75-83.
- Cho J, Park D, Lee HE (2014). Cognitive factors of using health apps: systematic analysis of relationships among health consciousness, health information orientation, eHealth literacy, and health app use efficacy. *J Med Internet Res*.16(5).
- Chomutare T, Fernadez-Luque L, Arsand L, Hartvigsen G (2011). Features of mobile diabetes applications: review of the literature and analysis of current applications compared against evidence-based guidelines. *JMIR*. 13(3): e65.
- Clark NM, Becker MH, Janz NK, Lorig K, Rakowski W, Anderson L (1991). Self-management of chronic disease by older adults: a review and questions for research. *J Aging Health*. 3(1):3-27.
- Colagiuri S, Brnabic A, Gomez M, Fitzgerald B, Buckley A, Colagiuri R (2009). DiabCost Australia type 1: assessing the burden of type 1 diabetes in Australia. Diabetes Australia, ACT, Canberra, Australia.
- Colagiuri S, Colagiuri R, Grainger D, Graham-Clarke P, Davey P, Fitzgerald P, et al (2002). DiabCost Australia-assessing the burden of Type 2 diabetes in Australia. Diabetologia. 45(Suppl 2): 1-8.
- Colberg SR (2012). Physical activity: the forgotten tool for type 2 diabetes management. *Front Endocrinol (Lausanne)*. 3:70.
- Cole-Lewis H, Ezeanochie N, Turgiss J (2019). Understanding Health Behavior Technology Engagement: Pathway to Measuring Digital Behavior Change Interventions. *JMIR Form Res.* 3(4):e14052.
- Cole-Lewis H, Kershaw T (2010). Text messaging as a tool for behaviour change in disease prevention and management. *Epidemiol Rev.* 32(1):56-69.

- Collins LM, Murphy SA, Strecher V (2007). The multiphase optimization strategy (MOST) and the sequential multiple assignment randomised trial (SMART): new methods for more potent eHealth interventions. *Am J Prev Med.* 32(5):S112-S8.
- Commons RJ, Robinson CH, Gawler D, Davis JS, Price RN (2015). High burden of diabetic foot infections in the top end of Australia: an emerging health crisis (DEFINE study). *Diabetes Res Clin Pract.* 110(2):147-157.
- Conroy DE, Yang C-H, Maher JP (2014). Behavior change techniques in top-ranked mobile apps for physical activity. *Am J Prev Med*. 46(6):649-652.
- Conway N, Campbell I, Forbes P, Cunningham S (2016), Wake D. mHealth applications for diabetes: user preference and implications for app development. *Health Informatics J*. 22(4):1111-1120.
- Cooper H, Booth K, Gill G (2003). Patients' perspectives on diabetes health care education. *Health Educ Res.* 18(2):191-206.
- Cotter AP, Durant N, Agne AA, Cherrington AL (2014). Internet interventions to support lifestyle modification for diabetes management: a systematic review of the evidence. J Diabetes Complications. 28(2):243-251.
- Cowan LT, van Wagenen SA, Brown BA, Hedin RJ, Seino-Stephen Y, Happ P, et al (2013). Apps of steel: are exercise apps providing consumers with realistic expectations? A content analysis of exercise apps for presence of behavior change theory. *Health Educ Behav*.133-139.
- Coyle ME, Francis K, Chapman Y (2013). Self-management activities in diabetes care: a systematic review. *Aust Health Rev.* 37(4):513-522.
- Craig M, Twigg S, Donaghue Ka, Cheung N, Cameron F, Conn J, et al (2011). National evidence-based clinical care guidelines for type 1 diabetes in children, adolescents and adults. Canberra: Australian Government Department of Health and Ageing. Available at: https://diabetessociety.com.au/documents/Type1guidelines14Nov2011.pdf
- Craig P, Dieppe P, Macintyre S, Michie S, Nazareth I, Petticrew M (2008). Developing and evaluating complex interventions: the new Medical Research Council guidance. BMJ.337:a1655.
- Craighead WE, Nemeroff CB (2004). The concise Corsini encyclopedia of psychology and behavioral science: John Wiley & Sons; new Jersey USA:1482.

- Crane D, Garnett C, Brown J, West R, Michie S (2017). Factors influencing usability of a smartphone app to reduce excessive alcohol consumption: think aloud and interview studies. *Front Public Health*. 5:39.
- Creer TL, Renne CM, Christian WP (1976). Behavioral contributions to rehabilitation and childhood asthma. *Rehabil Lit.* 37(8):226-247.
- Creswell JW, Plano Clark VL, Gutmann ML, Hanson WE (2003). Advanced mixed methods research designs. In: Tashakkori A, Teddie C, editors. Handbook of mixed methods in social and behavioural research. Sage, Thousand Oaks, California; 3: 209-240
- Cui M, Wu X, Mao J, Wang X, Nie M (2016). T2DM Self-Management via Smartphone Applications: A Systematic Review and Meta-Analysis. *PLoS One*. 11(11):e0166718.
- Cummings E, Hauser J, Cameron-Tucker H, Fitzpatrick P, Jessup M, Walters EH, et al (2011). Enhancing self-efficacy for self-management in people with cystic fibrosis. *Stud Health Technol Inform.* 169:33-37.
- Cushing CC, Steele RG (2010). A meta-analytic review of eHealth interventions for pediatric health promoting and maintaining behaviors. *J Pediatr Psychol*. 35(9):937-49.
- d Brunisholz KD, Briot P, Hamilton S, Joy EA, Lomax M, Barton N, et al (2014). Diabetes self-management education improves quality of care and clinical outcomes determined by a diabetes bundle measure. *J Multidiscip Health*. 7:533-542.
- d'Emden MC, Shaw JE, Colman PG, Colagiuri S, Twigg SM, Jones G, et al (2012). The role of HbA. *Med J Australia*. 197(4):220-221.
- Danaher BG, Brendryen H, Seeley JR, Tyler MS, Wooley T (2015). From black box to toolbox: outlining device functionality, engagement activities, and the pervasive information architecture of mHealth interventions. *Internet Interv.* 2(1): 91-101.
- Dao J, Spooner C, Lo W, Harris MF (2019). Factors influencing self-management in patients with type 2 diabetes in general practice: a qualitative study. Aust J Prim Health. 25(2):176-184.
- Davis FD (1989). Perceived usefulness, perceived ease of use, and user acceptance of information technology. *MIS Quarterly*. 319-340.
- Davis R, Campbell R, Hildon S, Hobbs L, Michie S (2015). Theories of behaviour and behaviour change across the social and behavioural sciences: a scoping review. *Health Psychol Rev.* 9(3):323-344.
- Day TL, Bench SD, Griffiths PD (2015). The role of pilot testing for a randomised control trial of a complex intervention in critical care. *J Res Nurs*. 20(3),167-178.

- De Bruijn GJ (2010). Understanding college students' fruit consumption. Integrating habit strength in the theory of planned behaviour. *Appetite*. 54(1):16-22.
- de la Torre Diez I, López-Coronado M, Vaca C, Aguado JS, de Castro C (2015). Cost-utility and cost-effectiveness studies of telemedicine, electronic and mobile health systems in the literature: a systematic review. *Telemed J E Health*. 21(2):81-85.
- De Vries H, Kremers S, Smeets T, Brug J, Eijmael K (2008). The effectiveness of tailored feedback and action plans in an intervention addressing multiple health behaviours. *Am J Health Promot*. 22(6):417-424.
- Deakin TA, McShane CE, Cade JE, Williams R (2005). Group based training for selfmanagement strategies in people with type 2 diabetes mellitus. *Cochrane Database Syst Rev.* 2.
- Deed G, Ackermann E, Arthur I, Barlow J, Jagadeesan S, Kawol D, et al (2016). General Practice Management of Type 2 Diabetes: 2016–18. Royal Australian College of General Practitioners (RACGP). Available at: <u>https://www.racgp.org.au/FSDEDEV/media/documents/Clinical%20Resources/Guide</u> <u>lines/Diabetes/General-practice-management-of-type-2-diabetes_1.pdf</u>

Delamater A (2006). Improving patient adherence. Clin Diabetes. 24(2):71-77.

- Demidowich AP, Lu K, Tamler R, Bloomgarden Z (2012). An evaluation of diabetes selfmanagement applications for Android smartphones. J Telemed Telecare.18(4):235-238.
- Dennison L, Morrison L, Conway G, Yardley L (2013). Opportunities and challenges for smartphone applications in supporting health behavior change: qualitative study. *J Med Internet Res.* 15(4):e86.
- Diabetes Australia (2015). Blood glucose monitoring. Available at: https://www.diabetesaustralia.com.au/blood-glucose-monitoring
- Dick JJ, Nundy S, Solomon MC, Bishop KN, Chin MH, Peek ME (2011). Feasibility and usability of a text message-based program for diabetes self-management in an urban African-American population. *J Diabetes Sci Technol.* 5(5):1246-1254.
- DiMatteo MR (2004). Social support and patient adherence to medical treatment: a metaanalysis. *Health Psychol*. 23(2):207-18.
- Direito A, Dale LP, Shields E, Dobson R, Whittaker R, Maddison R (2014). Do physical activity and dietary smartphone applications incorporate evidence-based behaviour change techniques? *BMC Public Health*. 14(1):646.

- Dobson R, Whittaker R, Murphy R, Khanolkar M, Miller S, Naylor J, et al (2017). The use of mobile health to deliver self-management support to young people with type 1 diabetes: a cross-sectional survey. *JMIR Diabetes*. 2(1):e4.
- dos Santos, RdOj, De Oliveira JHC, Rocha JB, Giraldi JME (2015). Eye tracking in neuromarketing: a research agenda for marketing studies. *Int J Psychol Stud.* 7(1): 32.
- Dou K, Yu P, Deng N, Liu F, Guan Y, Li Z, et al (2017). Patients' Acceptance of Smartphone Health Technology for Chronic Disease Management: A Theoretical Model and Empirical Test. JMIR Mhealth Uhealth. 5(12):e177.
- Duke SAS, Colagiuri S, Colagiuri R (2009). Individual patient education for people with type 2 diabetes mellitus. *Cochrane Database Syst Rev.* 1: CD005268
- Dunstan DW, Kingwell BA, Larsen R, Healy GN, Cerin E, Hamilton MT, et al (2012). Breaking up prolonged sitting reduces postprandial glucose and insulin responses. *Diabetes Care*. 35(5):976-983.
- Dunton GF, Atienza AA (2009). The need for time-intensive information in healthful eating and physical activity research: a timely topic. *J Am Diet Assoc*. 109(1):30-35.
- Eldridge SM, Lancaster GA, Campbell MJ, Thabane L, Hopewell S, Coleman CL, et al (2016). Defining feasibility and pilot studies in preparation for randomised controlled trials: development of a conceptual framework. *PLoS One*. 11(3):e0150205.
- El-Gayar O, Timsina P, Nawar N, Eid W (2013). Mobile applications for diabetes selfmanagement: status and potential. *J Diabetes Sci Technol*. 7(1):247-262.
- Evans JM, Newton RW, Ruta DA, MacDonald TM, Stevenson RJ, Morris AD (1999) Frequency of blood glucose monitoring in relation to glycaemic control: observational study with diabetes database. *BMJ*. 319(7202):83-86.
- Eysenbach G (2002). Issues in evaluating health websites in an internet-based randomised controlled trial. *J Med Internet Res.* 4(3):e17.h
- Eysenbach G (2005). The law of attrition. J Med Internet Res. 7(1):e11.
- Fan L, Sidani S (2009). Effectiveness of diabetes self-management education intervention elements: a meta-analysis. *Can J Diabetes*.33(1):18-26.
- Fang S,Yu, Z, Tao Y (2013). The relationship between coping strategy, self-efficacy, and diabetes self-management among people living in rural areas. World Health Digest Medical Periodical. 10:79-80.

- Faridi Z, Liberti L, Shuval K, Northrup V, Ali A, Katz DL (2008). Evaluating the impact of mobile telephone technology on type 2 diabetic patients' self-management: the NICHE pilot study. *J Eval Clin Pract.* 14(3): 465-469.
- Feeley N, Cossette S, Cote J, Heon M, Stremler R, Mortorella G, et al (2009). The importance of piloting an RCT intervention. *Can J Nurs Res.* 41(2), 84-99.
- Fife E, Orjuela J (2012). The privacy calculus: Mobile apps and user perceptions of privacy and security. *Int J Eng Bus Manag.* 4:4-11.
- Fishbein H, Palumbo P (1995). Acute metabolic complications in diabetes. *Diabetes Am.* 2:283-291.
- Fisher EB, Boothroyd RI, Coufal MM, Baumann LC, Mbanya JC, Rotheram-Borus MJ, et al (2012). Peer support for self-management of diabetes improved outcomes in international settings. *Health Aff (Millwood)*. 31(1):130-139.
- Fisher JD, Fisher WA (1992). Changing AIDS-risk behavior. Psychol Bull. 111(3):455. (81)
- Fisher JD, Fisher WA, Misovich SJ, Kimble DL, Malloy TE (1996). Changing AIDS risk behavior: effects of an intervention emphasizing AIDS risk reduction information, motivation, and behavioral skills in a college student population. *Health Psychol*. 15(2):114.
- Fisher WA, Fisher JD, Harman J (2003). The information-motivation-behavioral skills model: A general social psychological approach to understanding and promoting health behavior. Soc Psychol Found Health Illness. 22:82-106.
- Fisher WA, Kohut T, Schachner H, Stenger P (2011). Understanding self-monitoring of blood glucose among individuals with type 1 and type 2 diabetes. *Diabetes Educ.* 37(1): 85-94.
- Fitzgerald M, McClelland T (2016). What makes a mobile app successful in supporting health behaviouur change. *Health Educ*. 76 (3): 373-381.
- Flesch R (1948). A new readability yardstick. J App Psch. 32: 221-233.
- Fogg BJ (2002). Persuasive technology: using computers to change what we think and do. *Ubiquity*. 5.
- Franc S, Dardari D, Boucherie B, Riveline J, Biedzinski M, Petit C, et al (2009). Real-life application and validation of flexible intensive insulin-therapy algorithms in type 1 diabetes patients. *Diabetes Metab.* 35(6):463-468.
- Franz MJ, Powers MA, Leontos C, Holzmeister LA, Kulkarni K, Monk A, et al (2010). The evidence for medical nutrition therapy for type 1 and type 2 diabetes in adults. J American Diet Assoc. 110(12):1852-1889.
- Free C, Phillips G, Galli L, Watson L, Felix L, Edwards P, et al (2013). The effectiveness of mobile-health technology-based health behaviour change or disease management interventions for health care consumers: a systematic review. *PLoS Med.* 10(1):e1001362.
- Freyne J, Yin J, Brindal E, Hendrie G.A, Berkovsky S, Noakes M (2017). Push notifications in diet apps: influencing engagement times and tasks. *Int J Hum Comput Interact.* 33(10): 833-845.
- Funnell MM, Brown TL, Childs BP, Haas LB, Hosey GM, Jensen B, et al (2009). National standards for diabetes self-management education. *Diabetes Care*. 32 (Suppl 1):S87-S94.
- Gæde P, Lund-Andersen H, Parving H-H, Pedersen O (2008). Effect of a multifactorial intervention on mortality in type 2 diabetes. *N Engl J Med.* 358(6):580-591.
- Gagnon MP, Ngangue P, Payne-Gagnon J, Desmartis M (2015). m-Health adoption by healthcare professionals: a systematic review. *J Am Med Inform Assoc*. 23(1):212-220.
- Gallagher R, Armari E, White H, Hollams D (2013). Multi-component weight-loss interventions for people with cardiovascular disease and/or type 2 diabetes mellitus: A systematic review. *Eur J Cardiovasc Nurs*. 12(4):320-329.
- Gao J, Wang J, Zhu Y, Yu J (2013). Validation of an information–motivation–behavioral skills model of self-care among Chinese adults with type 2 diabetes. *BMC Public Health*. 13(1):100.
- Garabedian LF, Ross-Degnan D, Wharam JF (2015). Mobile phone and smartphone technologies for diabetes care and self-management. *Curr Diab Rep*.15(12):109.
- Gary TL, Genkinger JM, Guallar E, Peyrot M, Brancati FL (2003). Meta-analysis of randomized educational and behavioural interventions in type 2 diabetes. *Diabetes Educ*. 29(3):488-501.
- Gazmararian JA, Williams MV, Peel J, Baker DW (2003). Health literacy and knowledge of chronic disease. *Patient Educ Couns*. 51(3):267-275.
- Gazmararian JA, Ziemer DC, Barnes C (2009). Perception of barriers to self-care management among diabetic patients. *Diab Educ*. 35(5):778-788.
- Georgsson M, Staggers N (2016). An evaluation of patients' experienced usability of a diabetes mHealth system using a multi-method approach. *J Biomed Inform*. 59:115-129.

- Ghazanfar H, Rizvi SW, Khurram A, Orooj F, Qaiser I (2016). Impact of insulin pump on quality of life of diabetic patients. *Indian J Endocrinol Metab.* 20(4):506.
- Ghimire S (2017). Barriers to diet and exercise among Nepalese type 2 diabetic patients. *Int Sch Res Notices*. 2017(3).
- Giroux D, Bacon S, King DK, Dulin P, Gonzalez V (2014). Examining perceptions of a smartphone-based intervention system for alcohol use disorders. *Telemed J E Health*. 20(10):923-929.
- Glanz K, Bishop DB (2010). The role of behavioral science theory in development and implementation of public health interventions. *Annu Rev Public Health*. 31:399-418.
- Glanz K, Rimer BK, Lewis FM (2008). Health behavior and health education: theory, research, and practice. 4th ed. Wiley & Sons, San Fransisco.
- Glasgow RE, Emmons KM (2007). How can we increase translation of research into practice? Types of evidence needed. *Annu Rev Public Health*. 28:413-433.
- Glasgow RE, Fisher L, Skaff M, Mullan J, Toobert DJ (2007). Problem solving and diabetes self-management: investigation in a large, multiracial sample. *Diabetes Care*. 30(1):33-37.
- Gleeson-Kreig JM (2006). Self-monitoring of physical activity. Diabetes Educ. 32(1):69-77.
- Goldberg L, Lide B, Lowry S, Massett HA, O'Connell T, Preece J, et al (2011). Usability and accessibility in consumer health informatics current trends and future challenges. *Am J Prev Med.* 40 (5 Suppl 2):S187-S197.
- Goyal S, Cafazzo J (2013). Mobile phone health apps for diabetes management: current evidence and future developments. *QJM*.*1*06(12):1067-1069.
- Goyal S, Nunn CA, Rotondi M, Couperthwaite AB, Reiser S, Simone A, et al (2017). A mobile app for the self-management of type 1 diabetes among adolescents: a randomised controlled trial. *JMIR mHealth and uHealth*. 5(6):e82.
- Grady PA, Gough LL (2014). Self-management: a comprehensive approach to management of chronic conditions. *Am J Public Health*. 104(8):e25-e31.
- Greenwood DA, Gee PM, Fatkin KJ, Peeples M (2017). A Systematic Review of Reviews Evaluating Technology-Enabled Diabetes Self-Management Education and Support. J Diabetes Sci Technol. 11(5):1015-1027.
- Grobosch S, Kuske S, Linnenkamp U, Ernstmann N, Stephan A, Genz J, et al (2018). What information needs do people with recently diagnosed diabetes mellitus have and what

are the associated factors? A cross-sectional study in Germany. *BMJ open*. 8(10):e017895.

- Guertler D, Vandelanotte C, Kirwan M, Duncan MJ (2015). Engagement and nonusage attrition with a free physical activity promotion program: the case of 10,000 steps Australia. *JMIR*. 17(7): e176.
- Guest G, Bunce A, Johnson L (2006). How many interviews are enough? An experiment with data saturation and variability. *Field Methds*. 18(1):59-82.
- Gürses S, Troncoso C, Diaz C (2011). Engineering privacy by design. Available at: <u>https://software.imdea.org/~carmela.troncoso/ papers/Gurses-APC15.pdf</u>
- Haartsen J (2017). Ericsson review. Bluetooth-The universal radio interface for ad hoc, wireless connectivity. Available at: <u>https://www8.cs.umu.se/kurser/TDBD16/VT07/bluetooth.pdf</u>
- Haas L, Maryniuk M, Beck J, Cox CE, Duker P, Edwards L, et al (2012). National standards for diabetes self-management education and support. *Diabetes Educ*. 38(5):619-629.
- Haas L, Maryniuk M, Beck J, Cox CE, Duker P, Edwards L, et al (2014). National standards for diabetes self-management education and support. *Diabetes Care.* 37 (Suppl 1):S144-S153.
- Haire-Joshu D, Glasgow RE, Tibbs TL (1999). Smoking and diabetes. *Diabetes Care*. 22(11):1887-1898.
- Hardeman W, Sutton S, Griffin S, Johnston M, White A, Wareham NJ, et al (2005). A causal modelling approach to the development of theory-based behaviour change programmes for trial evaluation. *Health Educ Res.* 20(6):676-687.
- Harris M, Flack J, Georgiou A, McKenzie S, Penn D, Burns J (2006). Monitoring change in diabetes care using diabetes registers: experience from Divisions of General Practice. *Aust Fam Physician*. 35(1-2):77.
- Harrison S, Stadler M, Ismail K, Amiel S, Herrmann-Werner A (2014). Are patients with diabetes mellitus satisfied with technologies used to assist with diabetes management and coping?: A structured review. *Diabetes Technol Ther*. 16(11):771-783.
- Harrison V, Proudfoot J, Wee PP, Parker G, Pavlovic DH, Manicavasagar V (2011). Mobile mental health: review of the emerging field and proof of concept study. *J Ment Health*. 20(6):509-524.

- Hartin PJ, Nugent C, McClean S, Cleland I, Tschanz JT et al (2016). The empowering role of mobile apps in behaviour change interventions: The grey matters randomized controlled trial. *JMIR mHealth and uHealth*. 4(3):e93.
- Heale R, Forbes D (2013). Understanding triangulation in research. Evid Based Nurs. 16(4):98.
- Heerwegh, D (2003). Explaining response latencies and changing answers using client-side paradata from a web survey. *Soc Sci Comp Rev.* 21(3):360-373.
- Heinrich E, de Nooijer J, Schaper NC, Schoonus-Spit MH, Janssen MA, de Vries NK (2012). Evaluation of the web-based Diabetes Interactive Education Programme (DIEP) for patients with type 2 diabetes. *Patient Educ Couns.* 86(2):172-178.
- Heinrich E, Schaper N, de Vries N (2015). Self-management interventions for type 2 diabetes: a systematic review. *Eur Diab Nurs*. 7(7):71-76.
- Heinrich E, Schaper NC, de Vries NK (2010). Self-management interventions for type 2 diabetes: a systematic review. *Eur Diab Nurs*. 7(2):71-76.
- Heisler M (2007). Overview of peer support models to improve diabetes self-management and clinical outcomes. *Diabetes Spect*. 20(4):214-221.
- Heisler M, Bouknight RR, Hayward RA, Smith DM, Kerr EA (2002). The relative importance of physician communication, participatory decision making, and patient understanding in diabetes self-management. *J Gen Intern Med.* 17(4):243-252.
- Heikler EB, Michie S, Rivera D, Collins LM, Pavel M et al (2016). Advancing models and theories for digital behaviour change interventions. *Am J Prev Med.* 51(5): 825-832.
- Henderson J, Valenti L, Bayram C, Miller G (2013). Self-monitoring blood glucose: Noninsulin-treated type 2 diabetes in Australian general practice. Aust Fam Physician. 42(9):646.
- Hermann JM, Meusers M, Bachran R, Kuhnle-Krahl U, Jorch N, Hofer SE, et al (2017). Self-reported regular alcohol consumption in adolescents and emerging adults with type 1 diabetes: A neglected risk factor for diabetic ketoacidosis? Multicenter analysis of 29 630 patients from the DPV registry. *Pediatr Diabetes*. 18(8):817-823.
- Hermsen S, Frost JH, Renes RJ, Kerkhof P. Using feedback through digital technology to distrupt and change habitual behaviour: A critical review of current literature. *Comput in Human Behav*.57: 61-74.
- Hernandez-Tejada MA, Campbell JA, Walker RJ, Smalls BL, Davis KS, Egede LE (2012). Diabetes empowerment, medication adherence and self-care behaviors in adults with type 2 diabetes. *Diabetes Technol Ther*. 14(7):630-634.

- Heron KE, Smyth JM (2010). Ecological momentary interventions: incorporating mobile technology into psychosocial and health behaviour treatments. *Br J Health Psychol*. 15(1):1-39.
- Herrejon K Hartke JL, Scherer J, Chapman-Novakofski K (2009). The creation and impact evaluation of "Your guide to diet and diabetes," an interactive web-based diabetes tutorial. *Diabetes Technol Ther.* 11(3):171-179.
- Herschbach P, Duran G, Waadt S, Zettler A, Amm C, Marten-Mittag B, et al (1997). Psychometric properties of the Questionnaire on Stress in patients with Diabetes Revised (QSD-R). *Health Psychol.* 16(2):171-174.
- Hewson C, Buchanan T (2017). Ethics guidelines for internet-mediated research. The British Psychological Society. 2017. Available from: <u>https://www.bps.org.uk/news-and-policy/ethics-guidelines-internet-mediated-research-2017</u>
- Hibbard JH, Greene J (2013). What the evidence shows about patient activation: better health outcomes and care experiences; fewer data on costs. *Health Aff (Millwood)*.32(2):207-214.
- Hill-Briggs F (2003). Problem solving in diabetes self-management: a model of chronic illness self-management behavior. *Ann of Behav Med*. 25(3):182-193.
- Hill-Briggs F, Echemendia RJ (2001). Association of metabolic control with problem-solving skills. *Diabetes Care*. 24(5):959.
- Hilliard ME, Powell PW, Anderson BJ. Evidence-based behavioural interventions to promote diabetes management in children, adolescents and families (2017). *Am Psychol*.71(7): 590-601.
- Hirani SP, Rixon L, Beynon M, Cartwright M, Cleanthous S, Selva A, et al (2017). Quantifying beliefs regarding telehealth: development of the whole systems demonstrator service user technology acceptability questionnaire. *J Telemed Telecare*. 23(4):460-469.
- Hoddinott, P (2015). A new era for intervention development studies. *Pilot Fasibility Stud.* 26(1):36.
- Hoegh RT, Nielsen CM, Overgaard M, Pedersen MB, Stage J (2006). The impact of usability reports and user test observations on developers' understanding of usability data: An exploratory study. *Int J Hum Comput Interact*. 21(2):173-196.
- Hoffmann TC, Oxman AD, Ioannidis JP, Moher D, Lasserson TJ, Tovey DI, et al (2017). Enhancing the usability of systematic reviews by improving the consideration and description of interventions. *BMJ*. 358:j2998.

- Holloway A, Watson HE (2002). Role of self-efficacy and behaviour change. Inter J Nurs Pract. 8(2):106-115.
- Holman H, Lorig K (2000). Patients as partners in managing chronic disease: partnership is a prerequisite for effective and efficient health care. *BMJ*. 7234:526-527.
- Holmen H, Torbjørnsen A, Wahl AK, Jenum AK, Småstuen MC, Årsand E, et al (2014). A mobile health intervention for self-management and lifestyle change for persons with type 2 diabetes, part 2: one-year results from the Norwegian randomized controlled trial-Renewing Health. *JMIR Mhealth Uhealth*. 2(4):e57.
- Holmen H, Wahl AK, Småstuen MC, Ribu L (2017). Tailored communication within mobile apps for diabetes self-management: a systematic review. *J Med Internet Res*.19(6).
- Holtz B, Lauckner C (2012). Diabetes management via mobile phones: a systematic review. *Telemed J E Health*. 18(3): 175-184.
- Hood KK, Hilliard M, Piatt G, Ievers-Landis CE (2015). Effective strategies for encouraging behavior change in people with diabetes. *Diabetes Manag (Lond)*.5(6):499-510.
- Höök K (2004). From Brows to Trust. User-centred design and evaluation of affective interfaces, In From brows to trust, part of the Human Coluter Interaction Series, 7th series: Springer Press: 127-160.
- Hornbæk K, Law E (2007). Meta-analysis of correlations among usability measures. Presented at: the SIGCHI conference on Human factors in computing systems. New York, USA; April 28-May 03, 2007: 617-626.
- Horvath KJ, Smolenski D, Amico KR (2014). An empirical test of the information-motivationbehavioral skills model of ART adherence in a sample of HIV-positive persons primarily in out-of-HIV-care settings. *AIDS Care*. 26(2):142-151.
- Hou C, Carter B, Hewitt J, Francisa T, Mayor S (2016). Do mobile phone applications improve glycemic control (HbA1c) in the self-management of diabetes? A systematic review, meta-analysis, and GRADE of 14 randomized trials. *Diabetes Care*. 39(11):2089-2095.
- Hsieh HF, Shannon SE (2005). Three approaches to qualitative content analysis. *Qual Health Res.* 15(9):1277-1288.
- Huang Z, Lum E, Jimenez G, Semwal M, Sloot P, Car J (2019). Medication management support in diabetes: a systematic assessment of diabetes self-management apps. BMC Med. 17(1):127.
- Huang, M, Zhao R, Li S, Jiang X (2014). Self-management behavior in patients with type 2 diabetes: a cross-sectional survey in western urban China. *PLoS One*. 9(4): 0095138.

- Hunt CW (2015). Technology and diabetes self-management: an integrative review. *World J Diabetes*. 6(2):225-233.
- Hunt LM, Pugh J, Valenzuela M (1998). How patients adapt diabetes self-care recommendations in everyday life. *J Fam Pract*. 46(3):207-216.
- Hussain R, Maple M, Mapedzahama V, Reddy P (2015). The Fly-in Fly-out and Drive-in Drive-out model of health care service provision for rural and remote Australia: benefits and disadvantages. *Rural Remote Health*.15(3):3068.
- IBM. IBM SPSS Statistics for Windows, Version 23.0; Armonk, New York; 2018. Available at: <u>https://www.ibm.com/support/pages/release-notes-ibm-spss-statistics-23</u>
- International Diabetes Federation (2010). Global Diabetes Plan 2011-2021. Available at: <u>file:///C:/Users/jc447532/Downloads/Global_Diabetes_Plan_Final.pdf (17)</u>
- International Diabetes Federation (2017). IDF Diabetes Atlas 8th Edn Brussels, Belgium. Available at: <u>file:///C:/Users/jc447532/Downloads/IDF DA 8e-EN-final.pdf</u>
- International Diabetes Federation (2019). IDF Diabetes Atlas. 9th Edn Brussels, Belgium. Available at:

https://diabetesatlas.org/upload/resources/2019/IDF_Atlas_9th_Edition_2019.pdf

- International Organisation for Standardization (2010). Egornomics of human system interaction-Part 210: Human center design for interactive systems. Available at: https://infostore.saiglobal.com/preview/is/en/2010/i.s.eniso9241-210-2010.pdf?sku=1441363
- Inzucchi SE, Bergenstal RM, Buse JB, Diamant M, Ferrannini E, Nauck M, et al (2015).
 Management of hyperglycaemia in type 2 diabetes, 2015: a patient-centred approach.
 Update to a position statement of the American Diabetes Association and the European Association for the Study of Diabetes. *Diabetologia*. 58(3):429-442.
- Iribarren SJ, Cato K, Falzon L, Stone PW (2017). What is the economic evidence for mHealth? A systematic review of economic evaluations of mHealth solutions. *PLoS One*. 12(2):e0170581.
- Istepanian RS, Zitouni K, Harry D, Moutosammy N, Sungoor A, Tang B, et al (2009). Evaluation of a mobile phone telemonitoring system for glycaemic control in patients with diabetes. *J Telemed Telecare*. 15(3):125-128.

Izahar S, Lean QY, Hameed MA, Murugiah MK, Patel RP, Al-Worafi YM, et al (2017). Content analysis of mobile health applications on diabetes mellitus. *Front Endocrinol* (*Lausanne*). 2017;8:318.

- Jahangiry L, Shojaeizadeh D, Montazeri A, Najfi M, Mohammad K, Yaseri M. (2014) Adherence and attrition in a web-based lifestyle intervention for people with metabolic syndrome. *Iran J Public Health*. 43(9):1248-1258.
- Jarvis J, Skinner T, Carey M, Davies M (2010). How can structured self-management patient education improve outcomes in people with type 2 diabetes? *Diabetes Obes Metab*. 12(1):12-19.
- Jensen K, Larsen L (2007). Evaluating the usefulness of mobile services based on captured usage data from longitudinal field trials. Presented at: Proceedings of the 4th international conference on mobile technology, applications, and systems and the 1st international symposium on Computer human interaction in mobile technology; Singapore; 10-12.
- Jeon E, Park H (2018). Development of the IMB model and an evidence-based diabetes selfmanagement mobile application. *Healthc Inform Res.* 24(2):125-138.
- Jiang X, Wang J, Lu Y, Jiang H, Li M (2019). Self-efficacy-focused education in persons with diabetes: a systematic review and meta-analysis. *Psychol Res Behav Manag*.12:67.
- Joe J, Demiris G (2013). Older adults and mobile phones for health: a review. J Biomed Inform.46(5):947-954.
- Johnston-Brooks CH, Lewis MA, Garg S (2002). Self-efficacy impacts self-care and HbA1c in young adults with Type I diabetes. *Psychosom Med.* 64(1):43-51.
- Jones L, Crabb S, Turnbull D, Oxlad M (2014). Barriers and facilitators to effective type 2 diabetes management in a rural context: a qualitative study with diabetic patients and health professionals. *J Health Psychol*. 19(3):441-453.
- Jonkman NH, Schuurmans MJ, Jaarsma T, Shortridge-Baggett LM, Hoes AW, Trappenburg JC (2016). Self-management interventions: proposal and validation of a new operational definition. *J Clin Epidemiol*. 80:34-42.
- Juarascio AS, Goldstein SP, Manasse SM, Forman EM, Butryn ML (2015). Perceptions of the feasibility and acceptability of a smartphone application for the treatment of binge eating disorders: Qualitative feedback from a user population and clinicians. *Int J Med Inform.* 84(10):808-816.
- Kahn SE, Cooper ME, Del Prato S (2014). Pathophysiology and treatment of type 2 diabetes: perspectives on the past, present, and future. *Lancet*. 383(9922):1068-1083.
- Källander K, Tibenderana J, Akpogheneta O, Strachan D, Hill Z, Asbroek A, et al (2013). Mobile Health (mHealth) Approaches and Lessons for Increased Performance and

Retention of Community Health Workers in Low-and Middle-Income Countries: a review. *J Med Internet Res.* 15:e17.

- Kalra S, Sridhar G, Balhara YPS, Sahay RK, Bantwal G, Baruah MP, et al (2013). National recommendations: Psychosocial management of diabetes in India. *Indian J Endocrinol Metab.* 17(3):376.
- Kasenda B, Von Elm E, You J, Blumle A, Tomonaga Y, Scaccilotto R. et al (2014). Prevalence, characteristics, and publication of discontinued randomised trials. *JAMA*. 311(10):1045-1052.
- Kaufman N, Khurana I, Holmen H, Torbjørnsen A, Wahl A, Jenum A, et al (2016). Using digital health technology to prevent and treat diabetes. *Diabetes Technol Ther*. 18(S1):S-56-S-68.
- Kay M, Santos J, Takane M (2011). mHealth: New horizons for health through mobile technologies. In World Health Organization Global Observatory for eHealth series.
 Volume 3. Available at: <u>http://www.who.int/ehealth/mhealth_summit.pdf.</u>
- Kayyali R, Peletidi A, Ismail M, Hashim Z, Bandeira P, Bonnah J (2017). Awareness and use of mHealth apps: a study from England. *Pharmacy (Basel)*. 5(2):33.
- Kelders SM, Kok RN, Oddbarrd HC, Van Gemert-Pijnen JE (2012). Persuasive system design does matter: a systematic review of adherence to web-based interventions. *JMIR*. 14(6): e152.
- Kennedy A, Rogers A, Chew-Graham C, Blakeman T, Bowen R, Gardener C, et al (2014).Implementation of a self-management support approach (WISE) across a health system: a process evaluation explaining what did and did not work for organisations, clinicians and patients. *Implement Sci.* 9(1):129.
- Kennedy M, Dunning T (2017). Diabetes education: Essential but underfunded in Australia. *Diabetes Prim Care Aust.* 2:10-14.
- Kennedy P (2009). How to combine multiple research methods: Practical triangulation. *Jonny Holland Magazine*. Available at: <u>http://johnnyholland.org/2009/08/practical-</u> <u>triangulation/</u>
- Kessels RP (2003). Patients' memory for medical information. J R Soc Med.96(5):219-222.
- Kim EK, Kwak SH, Baek S, Lee SL, Jang HC, Park KS, Cho YM (2016). Feasibility of a patient-centered, smartphone-based, diabetes care system: a pilot study. *Diabetes Metab J*. 40(3):192-201.

- Kim H, Choi W, Baek EK, Kim YA, Yang SJ, Choi IY, et al (2014). Efficacy of the smartphone-based glucose management application stratified by user satisfaction. *Diabetes Metab J.* 38(3):204-210.
- Kim HC (2015). Acceptability engineering: the study of user acceptance of innovative technologies. *J Appl. Res Technol.* 13(2):230-237.
- Kim HS, Choi W, Baek EK, Kim YA, Yang SJ, Choi IY, et al (2014). Efficacy of the smartphone-based glucose management application stratified by user satisfaction. *Diabetes Metab J.* 38(3): 204-210.
- Kim YJ, Rhee SY, Byun JK, Park SY, Hong SM, Chin SO, et al (2015). A smartphone application significantly improved diabetes self-care activities with high user satisfaction. *Diabetes Metab J*. 39(3):207-217.
- Kindig DA, Panzer AM, Nielsen-Bohlman L (2004). Health literacy: a prescription to end confusion. Committee of health Literacy (eds). The National Academies, Washington DC; 2004: 59-81.
- Kinmonth AL, Woodcock A, Griffin S, Spiegal N, Campbell MJ (1998). Randomised controlled trial of patient centred care of diabetes in general practice: impact on current wellbeing and future disease risk. The diabetes care from diagnosis research team. *BMJ*. 317(7167):1202-1208.
- Kinney RL, Lemon SC, Person SD, Pagoto SL, Saczynski JS (2015). The association between patient activation and medication adherence, hospitalization, and emergency room utilization in patients with chronic illnesses: a systematic review. *Patient Educ Couns*. 98(5):545-552.
- Kirk JK, Stegner J (2010). Self-monitoring of blood glucose: practical aspects. *J Diabetes Sci Technol*. 4(2):435-439.
- Kirwan M, Vandelanotte C, Fenning A, Duncan MJ (2013). Diabetes self-management smartphone application for adults with type 1 diabetes: randomized controlled trial. *J Med Internet Res.* 15(11):e235.
- Kitsiou S, Pare G, Jaana M, Gerber B (2017). Effectiveness of mHealth interventions for patients with diabetes: an overview of systematic reviews. *PLoS One*. 12(3).
- Klasnja P, Prat W (2012). Healthcare in the pocket: mapping the space of mobile-phone health interventions. *J Biomed Inform*. 45(1):184-198.
- Klein MC, Manzoor A, Mollee JS (2017). Active2Gether: A personalised m-health intervention to encourage physical activity. *Sensors*. 17(6):1436.

- Klonoff DC, Bergenstal R, Blonde L, Boren SA, Church TS, Gaffaney J, et al (2008). Consensus report of the coalition for clinical research—self-monitoring of blood glucose. J Diabetes Sci Technol. 2(6):1030-1053.
- Klonoff DC (2019). Behavioural theory: The missing ingedient for digital health tools to change behaviour and incease adherence. *J Diab Sci Technol*. 13(2).
- Knuth J, Baltzer T, Berger TE, Leise H, Pankratz CK (2019). Usability Testing: Raising the Bar for Scientific Data Portals. *American Geophysical Union Fall Meeting*. Available at: <u>https://ui.adsabs.harvard.edu/abs/2019AGUFMIN23A..04K/abstract</u>
- Koch T, Kralik D, Taylor J (2000). Men living with diabetes: minimizing the intrusiveness of the disease. *J Clinc Nurs*. 9(2):247-254.
- Kohatsu ND, Robinson JG, Torner JC (2004). Evidence-based public health: an evolving concept. *Am J Prev Med.* 27(5):417-421.
- Kollmann A, Riedl M, Kastner P, Schreier G, Ludvik B (2007). Feasibility of a mobile phonebased data service for functional insulin treatment of type 1 diabetes mellitus patients. *J Med Internet Res.* 9(5):e36.
- Kontos E, Blake KD, Chou W-YS, Prestin A (2014). Predictors of eHealth usage: insights on the digital divide from the Health Information National Trends Survey 2012. J Med Internet Res. 16(7):e172.
- Koot D, Goh PSC, Lim RSM, Tian Y, Yau TY, Tan NC, Finkelstein FA (2019). A mobile lifestyle management program (GlycoLeap) for People With Type 2 Diabetes: Single-Arm Feasibility Study. *JMIR Mhealth Uhealth*. 7(5): e12965.
- Kordonouri O, Riddell MC (2019). Use of apps for physical activity in type 1 diabetes: current status and requirements for future development. *Ther Adv Endocrinol Metab*. 10:2042018819839298.
- Krishna S, Boren SA, Balas EA (2009). Healthcare via cell phones: a systematic review. *Telemed J E-Health*. 15(3):231-240.
- Kumar P, Lee H (2012). Security issues in healthcare applications using wireless medical sensor networks: a survey. Sensors (Basel). 12(1):55-91.
- Kuntz JL, Safford MM, Singh JA, Phansalkar S, Slight SP, Her QL, et al (2014). Patientcentered interventions to improve medication management and adherence: a qualitative review of research findings. *Patient Educ Couns*. 97(3):310-26.
- Lalić NM, Lalić K, Jotić A, Stanojević D, Živojinović D, Janićijević A, et al (2017). The impact of structured self-monitoring of blood glucose combined with intensive education on HbA1c levels, hospitalizations, and quality-of-life parameters in insulin-treated patients

with diabetes at primary care in Serbia: the multicenter SPA-EDU study. *J Diabetes Sci Technol*. 11(4):746-752.

- Lalmas M, O'Brien H, Yom-Tov E (2014). Measuring user engagement. Synthesis Lectures on Information Concepts, Retrieval, and Services. Morgan and Claypool, San Rafael, USA:1-132.
- Lambrinou E, Hansen TB, Beulens JW (2019). Lifestyle factors, self-management and patients empowerments in diabetes care. *Eur Jour Prev Card. (Suppl 2)*: 55-63.
- Lancaster GA, Dodd S, Williamson, PR (2004). Design and analysis of pilot studies: recommendations for good practice. *J Eval Clin Pract*. 10(2):307-312.
- Lenhart A, Duggan M, Perrin A, Stepler R, Rainie H, Parker K (2015). Teens, social media & technology overview. Pew Research Center [Internet & American Life Project]; 2015. Available at: <u>https://www.slideshare.net/eraser/teens-social-media-technology-overview-2015</u>.
- Leon AC, Davis LL, Kraemer HC (2011). The role and interpretation of pilot studies in clinical research. *J Psychiatr Res.* 45(5): 626-629.
- Lewis BA, Marcus BH, Pate RR, Dunn AL (2002). Psychosocial mediators of physical activity behaviour among adults and children. *Am J Prev Med*. 23(2):26-35.
- Lhachimi SK, Bala MM, Vanagas G (2016). Evidence-based public health. *BioMed Res* Int.5681409.
- Li A, Sun Y, Yang X, Guo J (2019). Exploring the relationship between perceived ease of use and continuance usage of a mobile terminal: Mobility as a moderator. *Sustainability*. 11(4):1128.
- Li J, Fernando C (2016). Smartphone-based personalised blood glucose prediction. *ICT Express.* 2(4):150-154.
- Liang X, Wang Q, Yang X, Cao J, Chen J, Mo X, et al (2011). Effect of mobile phone intervention for diabetes on glycaemic control: a meta-analysis. *Diabet Med*. 28(4):455-463.
- Lie SS, Karisen B, Niemiec CP, Graue M Oftedal B (2018). Written reflection in an eHealth intervention for adults with type 2 diabetes mellitus: a qualitative study. *Patient Prefer Adherence*. 12:311-320.
- Lim S, Xue L, Yen CC, Chang L, Chan HC, Tai BC, et al (2011). A study on Singaporean women's acceptance of using mobile phones to seek health information. *Int J Med Inform.* 80(12):e189-e202.

- Lin JC, Chen JM (2007). The automatic defense mechanism for malicious injection attack. *7th IEEE International Conference on Computer and Information Technology*: 9878947.
- Lithgow K, Edwards A, Rabi D (2017). Smartphone app use for diabetes management: evaluating patient perspectives. *JMIR Diabetes*. 2(1):e2.
- Locke EA (1997). Self-efficacy: The exercise of control. Pers Psychol.50(3):801.
- Lorig K, Ritter PL, Villa FJ, Armas J (2009). Community-based peer-led diabetes selfmanagement. *Diabetes Educ*. 35(4):641-651.
- Lorig KR, Holman HR (2003). Self-management education: history, definition, outcomes, and mechanisms. *Ann Behav Med.* 26(1):1-7.
- Lund AM (2001). Measuring usability with the use questionnaire. *Usability Interface*. 8(2):3-6.
- Lustria MLA, Smith SA, Hinnant CC (2011). Exploring digital divides: an examination of eHealth technology use in health information seeking, communication and personal health information management in the USA. *Health informatics J.* 17(3):224-243.
- Luxton DD, McCann RA, Bush NE, Mishkind MC, Reger GM (2011). mHealth for mental health: Integrating smartphone technology in behavioural healthcare. *Professional Psychol: Res Pract.* 42(6):505.
- Lyons EJ, Lewis ZH, Mayrsohn BG, Rowland JL (2014). Behavior change techniques implemented in electronic lifestyle activity monitors: a systematic content analysis. *J Med Internet Res.* 16(8):e192.
- Mace FC, Belfiore PJ, Shea MC (1989). Operant theory and research on self-regulation. In Zimmerman BJ, Schunk DH.(eds) Self-regulated learning and academic achievement. Springer series in cognitive development. Spinger, New York: 27-50.
- Macefield R (2009). How to specify the participant group size for usability studies: a practitioner's guide. *J Usability Stud.5*(1)34-45.
- MacKinnon DP, Lockwood CM, Hoffman JM, West SG, Sheets V (2002). A comparison of methods to test mediation and other intervening variable effects. *Psychol Methods*. 7(1):83.
- Maguire R, McCann L, Miller M, Kearney N (2008). Nurse's perceptions and experiences of using of a mobile-phone-based Advanced Symptom Management System (ASyMS©) to monitor and manage chemotherapy-related toxicity. *Eur J Oncol Nurs*. 12(4):380-386.

- Maillet NA, D'Eramo Melkus G, Spollett G (1996). Using focus groups to characterise the health beliefs and practices of black women with non-insulin-dependent diabetes. *Diabetes Educ.* 22(1):39-46.
- Mallenius S, Rossi M, Tuunainen VK (2007). Factors affecting the adoption and use of mobile devices and services by elderly people–results from a pilot study. In Proceedings of 6th Annual Global Mobility Roundtable. Los Angeles; 31:12.
- Manroa P, Krupa Doshi M (2016). Self-monitoring of blood glucose: Advice for providers and patients. *Cleve Clin J Med.* 83(5):355.
- Manski-Nankervis J-A, Furler J, Blackberry I, Young D, O'Neal D, Patterson E (2014). Roles and relationships between health professionals involved in insulin initiation for people with type 2 diabetes in the general practice setting: a qualitative study drawing on relational coordination theory. *BMC Fam Pract*. 15(1):20.
- Martinez M, Park SB, Maison I, Mody V, Soh LS, Parihar HS (2017). iOS Appstore-based phone apps for diabetes management: potential for use in medication adherence. *JMIR Diabetes*. 2(2):e12.
- Martínez-Pérez B, De La Torre-Díez I, López-Coronado M (2013). Mobile health applications for the most prevalent conditions by the World Health Organization: review and analysis. *J Med Internet Res.* 15(6):e120.
- Mayberry LS, Osborn CY (2014). Empirical validation of the information–motivation– behavioral skills model of diabetes medication adherence: a framework for intervention. *Diabetes Care*. 37(5):1246-1253.
- Mazzuca SA (1982). Does patient education in chronic disease have therapeutic value? J Chronic Dis. 35(7); 521-529.
- Mbuagbaw L, Aronson R, Walker A, Brown RE, Orzech N (2017). The LMC Skills, Confidence & Preparedness Index (SCPI): development and evaluation of a novel tool for assessing self-management in patients with diabetes. *Health Qual Life Outcomes*. 15(1):27.
- McAuley E, Szabo A, Gothe N, Olson EA (2011). Self-efficacy: implications for physical activity, function, and functional limitations in older adults. *Am J Lifestyle Med*.5(4):361-369.
- McBain H, Mulligan K, Haddad M, Flood C, Jones J, Simpson A (2016). Self management interventions for type 2 diabetes in adult people with severe mental illness. *Cochrane Database Syst Rev.* 4.

- McCurdie T, Taneva S, Casselman M, Yeung M, McDaniel C, Ho W, et al (2012). mHealth consumer apps: the case for user-centered design. *Biomed Instrum Technol*. 46(suppl 2):49-56.
- McEwen BS (1998). Protective and damaging effects of stress mediators. *N Engl J Med.* 338(3):171-179.
- Mensing C, Boucher J, Cypress M, Weinger K, Mulcahy K, Barta P, et al (2006). National standards for diabetes self-management education. *Diabetes Care*.29(suppl 1):s78-s85.
- Merrett F (2006). Reflections on the Hawthorne effect. Educ Psychol. 26(1):143-146.
- Michie S, Johnston M, Francis J, Hardeman W, Eccles M (2008). From theory to intervention: mapping theoretically derived behavioural determinants to behaviour change techniques. *Appl Psychol.* 57(4),660-680.
- Middelweerd A, Molle JS, van der Wal CN, Te Velde S.J (2014). Apps to promote physical activity among adults: a review and content analysis. *Int J Behav Nutr Phys Act.* 11(1): 97.
- Miller G (2012). The smartphone psychology manifesto. Perspect Psychol Sci. 7(3):221-237.
- Miller W, Rollnick S (2013). Motivational interviewing: Helping people change. In Rollnick S, Miller W.R, Moyers T.B (eds) Guilford Press; New York:1-482.
- Mishali M, Omer H, Heymann A (2011). The importance of measuring self-efficacy in patients with diabetes. *Fam Pract.* 28(1):82-87.
- Misra S, Lewis TL, Aungst TD (2013). Medical application use and the need for further research and assessment for clinical practice: creation and integration of standards for best practice to alleviate poor application design. *JAMA Dermatol*. 149(6):661-662.
- MobiThinking (2014). Global mobile statistics part A: Mobile subcribers; handset market share; mobile operatiors. Available at:
- Mogre V, Johnson NA, Tzelepis F, Shaw JE, Paul C (2019). A systematic review of adherence to diabetes self-care behaviours: Evidence from low-and middle-income countries. J Adv Nurs. 75(12):3374-3389.
- Mohebi S, Azadbakht L, Feizi A, Sharifirad G, Kargar M (2013). Review the key role of selfefficacy in diabetes care. *J Educ Health Promot.* 2:36.
- Mohr DC, Dufecy J, Jin L, Ludman EJ, Lewis A, Begale M, McCarthy M Jr (2010). Multimodal e-mental health treatment for depression: a feasibility trial. *JMIR*. 12(5): e48.

- Mohr DC, Schueller SM, Montague E, Burns MN, Rashidi P (2014). The behavioral intervention technology model: an integrated conceptual and technological framework for eHealth and mHealth interventions. *JMIR*. 16(6): e146.
- Molaison EF (2002). Stages of change in clinical nutrition practice. *Nutr Clin Care*. 5(5):251-257.
- Moore GF, Audrey S, Baker M, Bond L, Bonell C, Hardeman W. et a (2015). Process evaluation of complex interventions: Medical Research Council guidance. *BMJ*. 350: h1258.
- Morera EP, de la Torre Diez I, Garcia-Zapirain B, Lopez-Coronado M, Arambarri J (2016). Security recommendations for mHealth apps: Elaboration of a developer's guide. *J Med Syst.* 40(6);152.
- Moroney L (2017). The Definitive guide to firebase: build Android apps on Google's mobile platform. Apress, Berkely, CA; 1-271.
- Morowati-Sharifabad M, Rouhani Tonekaboni N, Baghianimoghadam M (2007). Predictors of self-care behaviours among diabetic patients referred to Yazd diabetes research Centre based on extended health belief model. *SSU J*. 15(3):85-96.
- Morse J (2003). Procedures and practice of mixed method design: maintaining control, rigor and complexity. In Teddlie C, Tashakkori A. (eds) Sage handbook of mixed methods in social & behavioral research: Sage, London;
- Mulcahy K, Maryniuk M, Peeples M, Peyrot M, Tomky D, Weaver T, et al (2003). Diabetes self-management education core outcomes measures. *Diabetes Educ*. 29(5):768-803.
- Munafo MR (2016). Opening up addiction science. Addiction. 111(3): 387-388.
- Murray E, Hekler E, Andersson G, Collins L, Doherty A, Hollis C. & Wyatt JC (2016). 'Evaluating Digital Health Interventions'. *Am J Prev Med.* 51(5):843-851.
- Murray E, Khadjesari Z, White I, Kalaitzaki E, Godfrey C, McCambridge J, et al (2009). Methodological challenges in online trials. *J Med Internet Res.* 11(2):e9
- Musselwhite K, Cuff L, McGregor L, King KM (2007). The telephone interview is an effective method of data collection in clinical nursing research: A discussion paper. *Int J Nurs Stud.* 44(6):1064-1070.
- Nagelkerk J, Reick K, Meengs L (2006). Perceived barriers and effective strategies to diabetes self-management. *J Adv Nurs.* 54(2):151-158.
- National Rural Health Alliance International (2011). Diabetes in rural Australia. Fact Sheet 21. Available at: <u>https://www.ruralhealth.org.au/sites/default/files/publications/fact-sheet-21-diabetes-rural-australia.pdf</u>

- Newington L, Metcalfe A (2014). Factors influencing recruitment to research: qualitative study of the experiences and perceptions of research teams. *BMC Med Res Methodol*. 14(1):10.
- Newman S, Steed L, Mulligan K (2004). Self-management interventions for chronic illness. *Lancet*. 364(9444):1523-1537.
- Ng E, de Colombani P (2015). Framework for selecting best practices in public health: a systematic literature review. *J Public Health Res*.4(3).
- Noar SM, Zimmerman RS (2005). Health behavior theory and cumulative knowledge regarding health behaviors: are we moving in the right direction? *Health Educ Res*. 20(3):275-290.
- Norris SL, Engelgau MM, Narayan KV (2001). Effectiveness of self-management training in type 2 diabetes: a systematic review of randomized controlled trials. *Diabetes Care*. 24(3):561-587.
- Norris SL, Lau J, Smith SJ, Schimid CH, Engelgau MM (2002). Self-management education for adults with type 2 diabetes: a meta-analysis of the effect on glycemic control. *Diabetes Care.* 25 (7):1159-1171.
- Nundy S, Mishra A, Hogan P, Lee SM, Solomon MC, Peek ME (2014). How do mobile phone diabetes programs drive behavior change? Evidence from a mixed methods observational cohort study. *Diabetes Educ*. 40(6):806-819.
- O'Brien H, Cairns P (2016). Theoretical perspectives on user engagement. In Why Engagement Matters: Cross-disciplinary perpsectives of user engagement in digital media. Springer, New York; 1-26.
- O'Brien HL, Cairns P, Hall M (2018). A practical approach to measuring user engagement with the refined user engagement scale (UES) and new UES short form. *Inte J Hum Comput Stud.* 112: 28-39.
- O'Brien HL, Toms EG (2008). What is user engagement? A conceptual framework for defining user engagement with technology. *J Am Soc Inf Sci Technol*. 59(6):938-955.
- Office of the Australian Information Commissioner (2014) Guide to developing an APP privacy policy. Available at: <u>https://www.oaic.gov.au/agencies-and-organisations/guides/guide-to-developing-an-app-privacy-policy</u>
- O'Hara M, Hynes L, O'donnell M, Nery N, Byrne M, Heller S, et al (2017). A systematic review of interventions to improve outcomes for young adults with Type 1 diabetes. *Diabet Med.* 34(6):753-769.
- O'Leary Z (2014). The essential guide to doing your research project, Second Edition. Sage.

- Oldenburg B, Taylor B, O'Niel A, Cocker F, Cameron LD (2015). Using new technologies to improve the prevention and management of chronic conditions in populations. *Ann Rev Public Health* 36: 483-505.
- Orsama A, Lähteenmäki J, Harno K, Kulju M, Wintergerst E, Schachner H, et al (2013). Active assistance technology reduces glycosylated hemoglobin and weight in individuals with type 2 diabetes: results of a theory-based randomised trial. *Diabetes Technol Ther*. 15(8):662-669.
- Osawaru, ER, Habeeb RAA (2014). Highlight of security challenges in big data. *Int J Inform Syst Eng.* 2(1): 2289-2265.
- Osborn CY, Rivet Amico K, Fisher WA, Egede LE, Fisher JD (2010). An informationmotivation-behavioral skills analysis of diet and exercise behavior in Puerto Ricans with diabetes. *J Health Psychol*. 15(8):1201-1213.
- Overland J, Yue DK, Mira M (2001). Use of Medicare services related to diabetes care: the impact of rural isolation. *Aust J Rural Health*. 9(6):311-316.
- Palinkas LA, Horwitz SM, Green CA, Wisdom JP, Duan N, Hoagwood K (2015). Purposeful sampling for qualitative data collection and analysis in mixed method implementation research. *Admin Policy Ment Health*. 42(5):533-544.
- Pandit AU, Bailey SC, Curtis LM, Seligman HK, Davis TC, Parker RM, et al (2014). Diseaserelated distress, self-care and clinical outcomes among low-income patients with diabetes. *J Epidemiol Community Health*. 68(6): 557-564.
- Papatheodorou K, Banach M, Edmonds M, Papanas N, Papazoglou D (2015). Complications of diabetes. *J Diab Res.* 1-5.
- Parker SJ, Jessel S, Richardson JE, Reid MC (2013). Older adults are mobile too! Identifying the barriers and facilitators to older adults' use of mHealth for pain management. *BMC Geriatr.* 13(1):43.
- Patel VL, Yoskowitz NA, Arocha JF, Shortliffe. EH (2009). Cognitive and learning sciences in biomedical and health instructional design: A review with lessons for biomedical informatics education. *J Biomed inform.* 42(1):176-197.
- Paterson B, Thorne S (2000). Developmental evolution of expertise in diabetes selfmanagement. *Clin Nurs Res.* 9(4):402-419.
- Patrick K, Griswold WG, Raab F, Intille SS (2008). Health and the mobile phone. *Am J Prev Med.* 35(2):177-181.

- Payne HE, Lister C, West JH, Berhardt JM (2015). Behavioral functionality of mobile apps in health interventions: a systematic review of the literature. *JMIR Mhealth Uhealth*. 3(1): e20.
- Payne KFB, Wharrad H, Watts K (2012). Smartphone and medical related App use among medical students and junior doctors in the United Kingdom (UK): a regional survey. BMC Med Inform Decis Mak. 12(1):121.
- Perera N, Agboola S (2019). Are formal self-care interventions for healthy people effective? A systematic review of the evidence. *BMJ Global Health*. 4(Suppl 10).
- Perlmuter LC, Flanagan BP, Shah PH, Singh SP (2008). Glycemic Control and Hypoglycemia: Is the loser the winner? *Diabetes Care*. 31(10): 2072-2076.
- Perrin BM, Swerissen H, Payne C (2009). The association between foot-care self efficacy beliefs and actual foot-care behaviour in people with peripheral neuropathy: a cross-sectional study. *J Foot Ankle Res.* 2(1):3.
- Persell SD, Keating NL, Landrum MB, Landon BE, Ayanian JZ, Borbas C, et al (2004). Relationship of diabetes-specific knowledge to self-management activities, ambulatory preventive care, and metabolic outcomes. *Prev Med.* 39(4):746-752.
- Perski O, Blandford A, West R, Michie S (2016). Conceptualising engagement with digital behaviour change interventions: a systematic review using principles from critical interpretive synthesis. *Transl Behav Med.* 7(2):254-267.
- Petersen M, Hempler NF (2017). Development and testing of a mobile application to support diabetes self-management for people with newly diagnosed type 2 diabetes: a design thinking case study. *BMC Med Inform Decis Mak.* 17(1):91.
- Peveler RC, Bryden KS, Neil HAW, Fairburn CG, Mayou RA, Dunger DB, et al (2005). The relationship of disordered eating habits and attitudes to clinical outcomes in young adult females with type 1 diabetes. *Diabetes Care*. 28(1):84-88.
- Peyrot M, Rubin R, Lauritzen T, Snoek F, Matthews D, Skovlund S (2005). Pschological problems and barriers to improved diabetes management: results of the Cross- national Diabetes Attitudes, Wishes and Needs (DAWN) Study. *Diab Med.* 22(10): 1379-1385.
- Pham Q, Wiljer D, Cafazzo JA (2016). Beyond the randomised controlled trial: a review of alternatives in mHealth clinical trial methods. *JMIR Mhealth Uhealth*. 4(3):e107.
- Phillips A (2009). Health status differentials across rural and remote Australia. *Aust J Rural Health*. 17(1):2-9.

- Piette JD (2007). Interactive behavior change technology to support diabetes self-management: where do we stand? *Diabetes Care*. 30(10); 2425-2432.
- Poddar KH, Hosig KW, Anderson ES, Nickols-Richardson SM, Duncan SE (2010). Web-based nutrition education intervention improves self-efficacy and self-regulation related to increased dairy intake in college students. *J Am Diet Assoc*. 110(11):1723-1727.
- Polonsky WH, Fisher L, Schikman CH, Hinnen DA, Parkin CG, Jelsovsky Z, et al (2011). Structured self-monitoring of blood glucose significantly reduces A1C levels in poorly controlled, noninsulin-treated type 2 diabetes: results from the Structured Testing Program study. *Diabetes Care*. 34(2):262-267.
- Pomey M-P, Ghadiri DP, Karazivan P, Fernandez N, Clavel N (2015). Patients as partners: a qualitative study of patients' engagement in their health care. *PLoS One*. 10(4):e0122499.
- Poolsup N, Suksomboon N, Rattanasookchit S (2009). Meta-analysis of the benefits of selfmonitoring of blood glucose on glycemic control in type 2 diabetes patients: an update. *Diabetes Technol Ther*. 11(12):775-784.
- Poon EG, Haas JS, Puopolo AL, Gandhi TK, Burdick E, Bates DW, et al (2004). Communication factors in the follow-up of abnormal mammograms. J Gen Intern Med.19(4):316-323.
- Poushter J (2016). Smartphone ownership and internet usage continues to climb in emerging economies. *Pew Research Center*. 22:1-44.
- Pouwer F, Kupper N, Adriaanse MC (2010). Does emotional stress cause type 2 diabetes mellitus? A review from the European Depression in Diabetes (EDID) Research Consortium. *Discov Med.* 9(45):112-118.
- Povey RC, Clark-Carter D (2007). Diabetes and healthy eating. Diabetes Educ. 33(6):931-959.
- Powers MA, Bardsley J, Cypress M, Duker P, Funnell MM, Hess Fischl A, et al (2015a).
 Diabetes Self-Management Education and Support in Type 2 Diabetes: A Joint Position
 Statement of the American Diabetes Association, the American Association of Diabetes
 Educators, and the Academy of Nutrition and Dietetics. *J Acad Nutr Diet*. 115(8):1323-1334.
- Powers MA, Bardsley J, Cypress M, Duker P, Funnel MM, Fischi AH, et al (2015b). Diabetes Self-management Education and Support in Type 2 Diabetes: A Joint Position Statement of the American Diabetes Association, the American Association of Diabetes Educators, and the Academy of Nutrition and Dietetics. *Diabetes Care*. 38: 1372-1382.

- Powers MA, Bardsley J, Cypress M, Duker P, Funnell MM, Fischl AH, et al (2017). Diabetes self-management education and support in type 2 diabetes: a joint position statement of the American Diabetes Association, the American Association of Diabetes Educators, and the Academy of Nutrition and Dietetics. *Diabetes Educ.* 43(1):40-53.
- Preuveneers D, Berbers Y (2008). Mobile phones assisting with health self-care: a diabetes case study. Presented at: Proceedings of the 10th international conference on Human computer interaction with mobile devices and services. Amsterdam, Netherlands; 2-5.
- Preziosa A, Grassi A, Gaggioli A, Riva G (2009). Therapeutic applications of the mobile phone. *Br J Guid Counc*. 37(3):313-325.
- Prochaska JO, Velicer WF (1997). The Transtheoretical Model of Health Behavior Change. *Am J Health Promot*. 12(1):38-48.
- Proudfoot J, Klein B, Barak A, Carlbring P, Cuijpers P, Lange A, et al (2011). Establishing guidelines for executing and reporting internet intervention research. *Cogn Behav Ther*. 40(2):82-97.
- Qiu S-h, Sun Z-l, Cai X, Liu L, Yang B (2012). Improving patients' adherence to physical activity in diabetes mellitus: a review. *Diabetes Metab J*. 36(1):1-5.
- Queensland Health Report (2018). The health of Queenslanders 2018. Report of the Chief Health Officer, Queensland Government Brisbane 2018. Available at: <u>https://www.health.qld.gov.au/ data/assets/pdf file/0032/732794/cho-report-2018-full.pdf</u>
- Quinn C, Clough S, Minor J, Lender D, Okafor M, Gruber-Baldini A (2008). WellDoc mobile diabetes management randomised controlled trial: change in clinical and behavioural outcomes and patient and physician satisfaction. *Diabetes Technol Ther*. 10(3):160-168.
- Quinn CC, Clough SS, Minor JM, Lender D, Okafor MC, Gruber-Baldini A (2008).
 WellDoc[™] mobile diabetes management randomised controlled trial: change in clinical and behavioral outcomes and patient and physician satisfaction. *Diabetes Technol Ther*. 10(3):160-168.
- Quinn CC, Gruber-Baldini AL, Shardell M, Weed K, Clough SS, Peeples M, et al (2009). Mobile diabetes intervention study: Testing a personalised treatment/behavioural communication intervention for blood glucose control. *Contemp Clin Trials*. 30(4):334-346.

- Quinn CC, Shardell MD, Terrin ML, Barr EA, Ballew SH, Gruber-Baldini AL (2011). Clusterrandomized trial of a mobile phone personalized behavioral intervention for blood glucose control. *Diabetes Care*. 34(9):1934-1942.
- Quinn CC,Shardell, MD, Terrin M.L, Barr EA, Parkm D, Shaikh F, et al (2016). Mobile diabetes intervention for glycemic control in 45-to 64-year-old persons with type 2 diabetes. *J Appl Gerontol*. 35(2):227-243.
- Rao A, Hou P, Golnik T, Flaherty J, Vu S (2010). Evolution of data management tools for managing self-monitoring of blood glucose results: a survey of iPhone applications. J Diabetes Sci Technol. 4(4):949-957.
- Ricci-Cabello I, Ruiz-Pérez I, Rojas-García A, Pastor G, Rodríguez-Barranco M, Gonçalves DC (2014). Characteristics and effectiveness of diabetes self-management educational programs targeted to racial/ethnic minority groups: a systematic review, meta-analysis and meta-regression. *BMC Endocr Disord*. 14(1):60.
- Riegel B, Moser DK, Anker SD, Appel LJ, Dunbar SB, Grady KL, et al (2009). State of the science: promoting self-care in persons with heart failure: a scientific statement from the American Heart Association. *Circulation*. 120(12):1141-1163.
- Riley WT, Rivera DE, Atienza AA, Nilsen W, Allison SM, Mermelstein R (2011). Health behavior models in the age of mobile interventions: are our theories up to the task? *Transl Behav Med.* 1(1):53-71.
- Ritchie J, Spencer L, O'Connor W (2003). Carrying out qualitative analysis. In Ritchie, Lewis J. (eds) Qualitative research practice: A guide for social science students and researchers. SAGE, London; 219-262.
- Rizkalla SW, Bellisle F, Slama G (2002). Health benefits of low glycaemic index foods, such as pulses, in diabetic patients and healthy individuals. *Br J Nutr.* 88(S3):255-262.
- Room J, Hannink E, Dawes H, Barker K (2017). What interventions are used to improve exercise adherence in older people and what behavioural techniques are they based on? A systematic review. *BMJ open*. 7(12):e019221.
- Rosenthal MJ, Fajardo M, Gilmore S, Morley JE, Naliboff BD (1998). Hospitalization and mortality of diabetes in older adults: a 3-year prospective study. *Diabetes Care*. 21(2):231-235.
- Rossi MC, Nicolucci A, Di Bartolo P, Bruttomesso D, Girelli A, Ampudia FJ, et al (2010a). Diabetes Interactive Diary: a new telemedicine system enabling flexible diet and insulin therapy while improving quality of life: an open-label, international, multicenter, randomized study. *Diabetes Care*. 33(1):109-115.

- Rossi MC, Nicolucci A, Lucisano G, Pellegrini F, Di Bartolo P, Miselli V, et al (2013). Impact of the "Diabetes Interactive Diary" telemedicine system on metabolic control, risk of hypoglycemia, and quality of life: a randomised clinical trial in type 1 diabetes. *Diabetes Technol Ther*. 15(8):670-679.
- Rossi MC, Perozzi C, Consorti C, Almonti T, Foglini P, Giostra N, et al (2010b). An interactive diary for diet management (DAI): a new telemedicine system able to promote body weight reduction, nutritional education, and consumption of fresh local produce. *Diabetes Technol Ther.* 12(8):641-647.
- Rossi MCE, Nicolucci A, Pellegrini F, Bruttomesso D, Bartolo PD, Marelli G, et al (2009). Interactive diary for diabetes: A useful and easy-to-use new telemedicine system to support the decision-making process in type 1 diabetes. *Diabetes Technol Ther*. 11(1):19-24.
- Ryan P, Sawin KJ (2009). The individual and family self-management theory: Background and perspectives on context, process, and outcomes. *Nurs Outlook*. 57(4):217-225. e6.
- Ryan RM, Deci EL (2000). Self-determination theory and the facilitation of intrinsic motivation, social development, and well-being. *Am Psychol.* 55(1):68-78.
- Sainsbury E, Shi Y, Flack J, Colagiuri S (2018). Burden of diabetes in Australia: it's time for more action. Preliminary report. July 2018. Available at: <u>https://www.sydney.edu.au/content/dam/corporate/documents/faculty-of-medicine-</u> <u>and-health/research/centres-institutes-groups/burden-of-diabetes-its-time-for-more-</u> action-report.pdf
- Saleh F, Mumu SJ, Ara F, Hafez MA, Ali L (2014). Non-adherence to self-care practices & medication and health related quality of life among patients with type 2 diabetes: a cross-sectional study. *BMC Public Health*. 214(1):431.
- Sarkar U, Fisher L, Schillinger D (2006). Is self-efficacy associated with diabetes selfmanagement across race/ethnicity and health literacy? *Diabetes Care*. 29(4):823-829
- Scheibe M, Reichelt J, Bellmann M, Kirch W (2015). Acceptance factors of mobile apps for diabetes by patients aged 50 or older: a qualitative study. *Med* 2 0.4(1):e1.
- Schillinger D, Grumbach K, Piette J, Wang F, Osmond D, Daher C, et al (2002). Association of health literacy with diabetes outcomes. *JAMA*. 288(4):475-482.
- Schmocker KS, Zwahlen FS, Denecke K (2018). Mobile App for Simplifying Life With Diabetes: Technical Description and Usability Study of GlucoMan. JMIR Diabetes.3(1):e6.

- Schoenberg NE, Drungle SC (2001). Barriers to non-insulin dependent diabetes mellitus (NIDDM) self-care practices among older women. *J Aging Health*. 13(4):443-66.
- Schoeppe S, Alley S, Van Lippevelde W, Bray NA, Williams SL, Duncan MJ, et al (2016). Efficacy of interventions that use apps to improve diet, physical activity and sedentary behaviour: a systematic review. *Int J Behav Nutr Phys Act.* 13(1):127.
- Schreier M (2012). Qualitative content analysis. In Flick U. (ed) The SAGE handbook of qualitative data analysis. Sage, Thousand Oaks, Carlifornia.
- Schueller SM, Neary M, O'Loughlin K, Adkins EC (2018). Discovery of and interest in health apps among those with mental health needs: survey and focus group study. *J Med Internet Res*.20(6):e10141.
- Seaquist ER, Anderson J, Childs B, Cryer P, Dagogo-Jack S, Fish L, et al (2013). Hypoglycaemia and diabetes: a report of a workgroup of the American Diabetes Association and the Endocrine Society. *Diabetes Care*. 36(5):1384-1395.
- Sekhon M, Cartwright M, Francis JJ (2017). Acceptability of healthcare interventions: an overview of reviews and development of a theoretical framework. *BMC Health Serv Res.* 17(1):88.
- Shamseer L, Moher D, Clarke M, Ghersi D, Liberati A, Petticrew M, et al (2015). Preferred reporting items for systematic review and meta-analysis protocols (PRISMA-P) 2015: elaboration and explanation. *BMJ*. 349:g7647.
- Sherifali D, Jones H, Mullan Y (2013). Diabetes self-management: what are we really talking about? Can J Diabetes. 37(1):2-3
- Sherifali D, Viscardi V, Bai J-W, Ali RMU (2016). Evaluating the effect of a diabetes health coach in individuals with type 2 diabetes. *Can J Diabetes*. 40(1):84-94.
- Shibuta T, Waki K, Tomizawa N, Igarashi A, Yamamoto-Mitani N, Yamaguchi S, et al (2017).
 Willingness of patients with diabetes to use an ICT-based self-management tool: a cross-sectional study. *BMJ Open Diabetes Res Care*. 5(1):e000322.
- Short C, Rebar C, Plotnikoff R, Vandelanotte C (2015). Designing engaging online behaviour change interventions: A proposed model of user engagement. *Eur Health Psychol*. 17(1).
- Short CE, DeSmet A, Woods C, Willains SL, Maher C, Middleweerd A, et al (2018). Measuring engagement in ehealth and mhealth behavior change interventions: Viewpoint of methodologies. *JMIR*. 20(11): e292.
- Shrivastava SR, Shrivastava PS, Ramasamy J (2013). Role of self-care in management of diabetes mellitus. *J Diabetes*.12(1):14.

- Sieverink F, Kelders S, van Gemert-Pijnen JE (2017). Clarifying the concept of adherence to eHealth technology: systematic review on when usage becomes adherence. *JMIR*. 19(12): e402.
- Silva BMC, Rodrigues JJPC, de la Torre Diez I, López-Coronado M, Saleem K (2015). Mobilehealth: A review of current state in 2015. *J BioMed Inform*. 265-272.
- Siminerio LM (2006). Implementing diabetes self-management training programs: breaking through the barriers in primary care. *Endocr Pract*.12(Supplement 1):124-130.
- Siminerio L, Ruppert KM, Gabbay R (2013). Who can provide diabetes self-management support in primary care? Findings from a randomised controlled trial. *Diabetes Educ*. 39(5):705-713.
- Siriwardena LN, Wickramasinghe WS, Perera KD, Marasinghe RB, Katulanda P, Hewapathirana R (2012). A review of telemedicine interventions in diabetes care. J Telemed Telecare. 18(3):164-168.
- Six JM, Macefield R (2016). How to determine the right number of participants for usability studies. Available at: <u>https://www.uxmatters.com/mt/archives/2016/01/how-to-determine-the-right-number-of-participants-for-usability-studies.php</u>
- Sligo FX, Jameson AM (2000). The knowledge—behavior gap in use of health information. J Am Soc Inf Sci. 51(9):858-869.
- Smith A, Harris C (2018). Type 1 Diabetes: Management strategies. Am Fam Physician. 98(3).
- Smith KB, Humphreys JS, Wilson MG (2008). Addressing the health disadvantage of rural populations: how does epidemiological evidence inform rural health policies and research? *Aust J Rural Health*.16(2):56-66.
- Speight J, Browne JL, Holmes-Truscott E, Hendrieckx C, Pouwer F (2011). Diabetes MILES-Australia reference group. Diabetes MILES-Australia 2011 survey report. Diabetes Australia-Victoria, Melboune. Available at: <u>https://www.ndss.com.au/wpcontent/uploads/resources/report-miles-youth-2011.pdf</u>
- Stableford S, Mettger W (2007). Plain language: a strategic response to the health literacy challenge. *J Public Health Policy*. 28(1):71-93.
- Starfield B, Hyde J, Gérvas J, Heath I (2008). The concept of prevention: a good idea gone astray? *J Epidemiol Community Health*. 62(7):580-583.
- Statista (2019a). Smartphone users wordwide from 2014 to 2020. Available at: https://statinvestor.com/data/33977/number-of-smartphone-users-worldwide/

- Statista (2019b). Share of mobile phone ussers that use a smartphone in Australia from 2014 to 2019. Available at: <u>https://www.statista.com/statistics/257041/smartphone-user-</u> <u>penetration-in-australia/</u>
- Steed L, Barnard M, Hurel S, Jenkins C, Newman S (2014). How does change occur following a theoretically based self-management intervention for type 2 diabetes. *Psychol Health Med.* 19(5):536-546.
- Steinsbekk A, Rygg L, Lisulo M, Rise MB, Fretheim A (2012). Group based diabetes selfmanagement education compared to routine treatment for people with type 2 diabetes mellitus. A systematic review with meta-analysis. *BMC Health Serv Res.* 12(1):213.
- Stern MJ (2008). The use of client-side paradata in analyzing the effects of visual layout on changing responses in web surveys. Field Methds. 20(4): 377-398.
- Stetson B, Boren S, Leventhal H, Schlundt D, Glasgow R, Fisher E, et al (2010). Embracing the evidence on problem solving in diabetes self-management education and support. *Self Care.* 1(3):83-99.
- Stoyanov SR, Hides L, Kavanagh DJ, Zelenko O, Tjondronegro D, Manin M (2015). Mobile app rating scale: a new tool for assessing the quality of health mobile apps. *JMIR Mhealth Uhealth*. 3(1):e27.
- Struber JC (2004). Recruiting and retaining allied health professionals in rural Australia: why is it so difficult? *J Allied Health*. 2(2):2.
- Taj F, Klein MC, van Halteren A (2019). Digital health behaviour change technology: bibliometric and scoping review of two decades of research. JMIR Mhealth and Uhealth 7(12): e13311.
- Takenga C, Berndt R, Musongya O, Kitero J, Katoke R, Molo K, et al (2014). An ICT-Based Diabetes Management System Tested for Health Care Delivery in the African Context. *Int J Telemed Appl.* 2014:437307.
- Taki S, Lymer S, Russell CG, Campbell K, Laws R, Ong KL, Elliot R, Denney-Wilson E (2017). Assessing user engagement of an mHealth intervention: development and implementation of the growing healthy app engagement index. *JMIR Mhealth Uhealth*. 5(6): e89.
- Tang TS, Gillard ML, Funnell MM, Nwankwo R, Parker E, Spurlock D, et al (2005). Developing a new generation of ongoing diabetes self-management support interventions. *Diabetes Educ.* 31(1):91-97.
- Tatara N, Årsand E, Skrøvseth SO, Hartvigsen G (2013). Long-term engagement with a mobile self-management system for people with type 2 diabetes. *JMIR Mhealth Uhealth*. 1(1).

- Taylor S, Fatima Y, Solomon S (2017). Factors affecting the self-monitoring of blood glucose levels in Aboriginal patients: Findings from a remote community. *Austr Indigenous Health Bull*.17:1-8.
- Taylor SJ, Pinnock H, Epiphaniou E, Pearce G, Parke HL, Schwappach A, et al (2014). A rapid synthesis of the evidence on interventions supporting self-management for people with long-term conditions: PRISMS–Practical systematic Review of Self-Management Support for long-term conditions. *Health Service Delivery Research*. 2(53): 1-622.
- Thakkar J, Barry T, Thiagalingam A, Redfern J, McEwan AL, Rodgers A, et al (2016). Design Considerations in Development of a Mobile Health Intervention Program: The TEXT ME and TEXTMEDS Experience. JMIR Mhealth Uhealth.4(4):e127.
- The International Expert Committee (2009). International Expert Committee report on the role of the A1C assay in the diagnosis of diabetes. *Diabetes Care*. 32(7):1327-1334.
- The Royal Australian College of General Practitioners (2016). General practice management of type 2 diabetes: 2016-18. East Melbourne, Victoria. Available at: https://www.racgp.org.au/FSDEDEV/media/documents/Clinical%20Resources/Guide lines/Diabetes/General-practice-management-of-type-2-diabetes_1.pdf
- Thomas NJ, Jones SE, Weedon MN, Shields BM, Oram RA, Hattersley AT (2018). Frequency and phenotype of type 1 diabetes in the first six decades of life: a cross-sectional, genetically stratified survival analysis from UK Biobank. *Lancet Diabetes Endocrinol*. 6(2):122-9.
- Thomas SL, Wakerman J, Humphreys JS (2014). What core primary health care services should be available to Australians living in rural and remote communities? *BMC Fam Pract.* 15(1):143.
- Toljamo M, Hentinen M (2001). Adherence to self-care and glycaemic control among people with insulin-dependent diabetes mellitus. *J Adv Nurs*. 34(6):780-786.
- Tomky D, Tomky D, Cypress M, Dang D, Maryniuk M, Peyrot M, et al (2008). American Association for Diabetes Educator's position statement. *Diabetes Educ*. 34(3):445-459.
- Tomky D, Weaver T, Mulcahy K,Peeples, M (2000). Diabetes education outcomes: what educators are doing. *Diabetes Educ*. 26(6):951-954.
- Tong A, Sainsbury P, Craig J (2007). Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. *Int J Qual in Health Care*. 19(6):349-357.

- Toobert DJ, Hampson SE, Glasgow RE (2000). The summary of diabetes self-care activities questionnaire. Diabetes Care. 23(7): 943-950.
- Torbjørnsen A, Småstuen MC, Jenum, AK, Årsand E, Ribu L (2018). Acceptability of an mHealth app intervention for persons with type 2 diabetes and its associations with initial self-management: randomised controlled trial. *JMIR Mhealth Uhealth*. 6(5):e125.
- Torous J, Nicholas J, Larsen ME, Firth J, Christensen H (2018). Clinical review of user engagement with mental health smartphone apps: evidence, theory and improvements. *Evid Based Ment Health*. 21(3):116-119.
- Tougas ME, Hayden JA, McGrath PJ, Huguet A, Rozario S (2015). A systematic review exploring the social cognitive theory of self-regulation as a framework for chronic health condition interventions. *PLoS One*. 10(8).
- Tran J, Tran R, White JR (2012). Smartphone-based glucose monitors and applications in the management of diabetes: an overview of 10 salient "apps" and a novel smartphoneconnected blood glucose monitor. *Clinic Diabetes*. 30(4):173-178.
- Tshiananga JKT, Kocher S, Weber C, Erny-Albrecht K, Berndt K, Neeser K (2012). The effect of nurse-led diabetes self-management education on glycosylated hemoglobin and cardiovascular risk factors: a meta-analysis. *Diabetes Educ*. 38(1):108-123.
- Tuckson RV, Edmunds M, Hodgkins ML (2017). Telehealth. NEJM; 377:1585-1592.
- Turner BC, Jenkins E, Kerr D, Sherwin RS, Cavan DA (2001). The effect of evening alcohol consumption on next-morning glucose control in type 1 diabetes. *Diabetes Care*. 24(11):1888-1893.
- Ubaid Ur Rehman M, Aleem M, Islam MA, Ahmed S (2019). Smart applications for diabetes management: A comprehensive survey and ranking. *Health Informatics J*. 1460458219869159.
- UK Propspective Diabetes Study Group (1998). Tight blood pressure control and risk of macrovascular and microvascular complications in type 2 diabetes: UKPDS 38. BMJ. 317(7160):703-713.
- United State Department of Health and Human services (2000). Final Rule, 45 CFR Parts 160, and 164: Standards for Privacy of Individually Identifiable Health Information. Federal Register. 65 (25): 1-369. Available at: <u>https://www.gpo.gov/fdsys/pkg/FR-2000-12-28/pdf/00-32678.pdf</u>
- United States Government. Plain Language Action and Information Network (2010). What is plain language? Available at: http://www.plainlanguage.gov/whatisPL/index.cfm

- Van Calster B, Wynants L, Timmerman D, Steyerberg EW, Collins GS (2019). Predictive analytics in health care: how can we know it works? J Am Med Inform Assoc.26(12):1651-1654.
- Van Dyk L (2014). A review of telehealth service implementation frameworks. *Int J Environ Res Public Health*.11(2):1279-1298.
- Van Keer R-L, Deschepper R, Francke AL, Huyghens L, Bilsen J (2015). Conflicts between healthcare professionals and families of a multi-ethnic patient population during critical care: an ethnographic study. *Crit Care*. 19(1):441.
- van Netten JJ, Lazzarini PA, Armstrong DG, Bus SA, Fitridge R, Harding K, et al (2018). Diabetic Foot Australia guideline on footwear for people with diabetes. *J Foot Ankle Res.* 11(1):2.
- van Netten JJ, Seng L, Lazzarini PA, Warnock J, Ploderer B (2019). Reasons for (non-) adherence to self-care in people with a diabetic foot ulcer. *Wound Repair Regen*. 27(5):530-539.
- van Rest J, Boonstra D, Everts M, van Rijn M, van Paassen R (2014). Designing privacy-bydesign. In: Preneel B, Ikonomou D. (eds) Privacy Technologies and Policy. Annual Privacy Forum. Lecture notes in Computer Science book series (8319). Springer, Berlin; 55-72.
- Van Smoorenburg AN, Hertroijs DF, Dekkers T, Elissen AM, Melles M (2019). Patients' perspective on self-management in daily life. *BMC Health Res.* 19: 605.
- VanAnh VO, Auroy L, Sarradon-Eck, A (2019). Patients perceptions of mhealth Apps: Meta-Ethnographic review of Qualitative stidues. JMIR mHealth uHealth. 7(7):e13817.
- Vehi J, Regincos Isern J, Parcerisas A, Calm R, Contreras I (2019). Impact of use frequency of a mobile diabetes management app on blood glucose control: Evaluation study. *JMIR Mhealth Uhealth*. 7(3): e11933.
- Venkatesh V, Davis FD (2000). A theoretical extension of the technology acceptance model: Four longitudinal field studies. *Manag Sci.* 46(2):186-204.
- Ventura AD, Browne JL, Holmes-Truscott E, Hendrieckx C, Pouwer F, Speight J (2016).
 Diabetes MILES-Australia 2016 survey report. Diabetes Victoria, Melboune.
 Available at: <u>https://acbrd.org.au/wp-content/uploads/2018/06/C292_DV_miles-</u>
 <u>2_web_final.pdf</u>
- Vernooij RW, Willson M, Gagliardi AR (2015). Characterizing patient-oriented tools that could be packaged with guidelines to promote self-management and guideline adoption: a meta-review. *Implement Sci.* 11(1):52.

- Viswanathan V, Madhavan S, Rajasekar S, Chamukuttan S, Ambady R (2005). Amputation prevention initiative in South India: positive impact of foot care education. *Diab Care*. 28(5):1019-1021.
- Vojdani A (2014). A potential link between environmental triggers and autoimmunity. *Autoimmune Dis.* 2014.
- Volentine RE, Owens A, Tenopir C, Frame M (2017). Usability Testing to Improve Research Data Services. *QQML*.4(1):59-68.
- Wakerman J, Humphreys JS, Wells R, Kuipers P, Entwistle P, Jones J (2008). Primary health care delivery models in rural and remote Australia–a systematic review. *BMC Health Serv Res.* 8(1):276.
- Waki K, Fujita H, Uchimura Y, Aramaki E, Omae K, Kadowaki T, et al (2012). DialBetics: smartphone-based self-management for type 2 diabetes patients. J Diabetes Sci Technol. 6(4):983-985.
- Waki K, Fujita H, Uchimura Y, Omae K, Aramaki E, Kato S, et al (2014). DialBetics: A Novel Smartphone-based Self-management Support System for Type 2 Diabetes Patients. J Diabetes Sci Technol. 8(2):209-215.
- Wallace S, Clark M, White J (2012). 'It's on my iPhone': attitudes to the use of mobile computing devices in medical education, a mixed-methods study. *BMJ Open*. Aug;2(4):e001099.
- Wan Q, Harris M, Davies G, Jayasinghe U, Flack J, Georgiou A, et al (2008). Cardiovascular risk management and its impact in Australian general practice patients with type 2 diabetes in urban and rural areas. *Int J Clin Pract*. 62(1):53-58.
- Wang X, Lan W, Liu J, Gao L, Wei Q, Zhang J, et al (2012). Evaluation analysis of influencing factors of type II diabetes patients self-management in Shanxi Province. J Qiqihar Med College. 33:1065-1069.
- Webb T, Joseph J, Yardley L, Michie S (2010). Using the internet to promote health behavior change: a systematic review and meta-analysis of the impact of theoretical basis, use of behavior change techniques, and mode of delivery on efficacy. *J Med Internet Res.* 12(1):e4.
- Welschen LM, Bloemendal E, Nijpels G, Dekker JM, Heine RJ, Stalman WA, et al (2005). Self-monitoring of blood glucose in patients with type 2 diabetes who are not using insulin: a systematic review. *Diabetes Care*. 28(6):1510-1517.
- West R, Michie S (2016). A guide to development and evaluation of digital behaviour interventions in healthcare: Silverback, Bream,UK;1-82.

- Westergaard RP, Genz A, PanicoK., Surkan PJ, Keruly J, Hutton HE, et al (2017). Acceptability of a mobile health intervention to enhance HIV care coordination for patients with substance use disorders. *Addict Sci Clin Pract*. 2017;12(1):11.
- Whitehead L, Seaton P (2016). The effectiveness of self-management mobile phone and tablet apps in long-term condition management: a systematic review. *J Med Internet Res*.18(5):e97.
- Whitlock L, McLaughlin A (2012). Identifying usability problems of blood glucose tracking apps for older users. *Hum Factors*.115-119.
- Wiebe EN, Lamb A, Hardy M, Sharek D (2014). Measuring engagement in video game-based environments: Investigation of the User Engagement Scale. *Comput Human Behav*. 32:123-132.
- Wilkinson A. Whitehead L, Ritchie L (2014). Factors influencing the ability to self-manage diabetes for adults living with type 1 or 2 diabetes. *Int J Nurs Stud.* 51(1):111-122.
- Wilkowska W, Ziefle M (2012). Privacy and data security in E-health: Requirements from the user's perspective. *Health Informatics J.* 18(3),191-201.
- Williams SL, French DP (2011). What are the most effective intervention techniques for changing physical activity self-efficacy and physical activity behaviour—and are they the same? *Health Educ Res.* 26(2):308-322.
- Wilson EA, Wolf MS, Curtis LM, Clayman ML, Cameron KA, Eigen KV, et al (2010). Literacy, cognitive ability, and the retention of health-related information about colorectal cancer screening. *J Health Commun.* 15(S2):116-125.
- Wilson LS, Maeder AJ (2015). Recent directions in Telemedicine: Review of Trends in Research and Practice. Health Inform Res. 21(4): 213-222.
- Winters-Miner LA (2014). Seven ways predictive analytics can improve healthcare. Available at: <u>https://www.elsevier.com/connect/seven-ways-predictive-analytics-can-improve-healthcare</u>
- Wohlers E, Sirard J, Barden C, Moon J (2009). Smart phones are useful for food intake physical activity surveys. Presented at Engineering in Medicine Biology Society, Annual International Conference of the IEEE; Minneapolis, USA; 3-9.
- Woldaregay A, Issom D, Henriksen A, Marttila H, Mikalsen M, Pfuhl G, et al (2018). Motivational factors for user engagement with mHealth Apps. *Stud Health Technol inform.* 249:151-157.
- Wolfsdorf JI, Allgrove J, Craig ME, Edge J, Glaser N, Jain V et a. (2014). Diabetic ketoacidosis and hyperglycaemic hyperosmolar state. *Pediatr Diabetes*. Suppl 20:154-179.

- Woods L, Cummings E, Duff J, Walker K (2018). Conceptual design and iterative development of a mHealth app by clinicians, patients and their families. *Stud Health Technol Inform*. 252:170-175.
- Woods LS, Duff J, Roehrer E, Walker K, Cummings E (2019). Patients' Experiences of Using a Consumer mHealth App for Self-Management of Heart Failure: Mixed-Methods Study. JMIR Hum Factors. 6(2):e13009.
- World Health Organisation (1999). Definition, diagnosis and classification of diabetes mellitus and its complications: report of a WHO consultation. Part 1, Diagnosis and classification of diabetes mellitus. Geneva: World Health Organisation. Available at: <u>https://apps.who.int/iris/handle/10665/66040</u>
- World Health Organisation (2003). Adherence to long-term therapies: evidence for action. World Health Organisation, Geneva, Switzerland. Available at: <u>https://www.who.int/chp/knowledge/publications/adherence_full_report.pdf?ua=1</u>
- World Health Organisation (2005). Preventing chronic diseases: a vital investment: World Health Organisation global report 2005. Available at: http://apps.who.int/iris/handle/106665/43314
- World Health Organisation (2011a). Use of glycated haemoglobin (HbA1c) in diagnosis of diabetes mellitus: abbreviated report of a WHO consultation. World Health Organisation. Available at: <u>https://www.ncbi.nlm.nih.gov/books/NBK304267/</u>
- World Health Organisation (2011b). mHealth: new horizons for health through mobile technologies. *mHealth: new horizons for health through mobile technologies*. Available at: https://www.who.int/goe/publications/goe_mhealth_web.pdf
- World Health Organisation (2016). Global report on diabetes. Geneva: World Health Organisation. Available at: http://www.who.int/diabetes/global-report.
- World Health Organisation (2018). The top 10 causes of death. Available at: <u>http://origin.who.int/mediacentre/factsheets/fs310/en/.</u>
- Wu SFV, Courtney M, Edwards H, McDowell J, Shortridge-Baggett LM, Chang PJ (2007). Self-efficacy, outcome expectations and self-care behaviour in people with type 2 diabetes in Taiwan. J Clin Nurs. 16(11c):250-257.
- Yao J, Wang H, Yin X, Yin J, Guo X, Sun Q (2019). The association between self-efficacy and self-management behaviors among Chinese patients with type 2 diabetes. *PLoS One*. 14(11):e0224869.

- Yardley L, Morrison L, Bradbury K, Muller I (2015). The person-based approach to intervention development: application to digital health-related behaviour change interventions. *J Med Internet Res.* 17(1):e30.
- Yardley L, Spring BJ, Riper H, Morrison LG, Crane DH, Curtis K, et al (2016). Understanding and promoting effective engagement with digital behavior change interventions. Am J Prev Med. 51(5):833-842.
- Ye Q, Khan U, Boren SA, Simoes EJ, Kim MS (2018). An Analysis of diabetes mobile applications features compared to AADE7TM: Addressing self-management behaviors in people with diabetes. *J Diabetes Sci Technology*. 12(4): 808-816.
- Yoon KH, Kim H-S (2008). A short message service by cellular phone in type 2 diabetic patients for 12 months. *Diabetes Res Clin Pract.* 79(2):256-261.
- Yu CH, Parsons JA, Mamdani M, Lebovic G, Hall S, Newton D, et al (2014). A web-based intervention to support self-management of patients with type 2 diabetes mellitus: effect on self-efficacy, self-care and diabetes distress. *BMC Med Inform Decis Mak*. 14:117.
- Zaid B, Jamaludin R, Wafaa B (2012). A comparative study of usability methods for mobile applications. *Int J Sci Eng Res.* 3(8):1-4.
- Zhao J, Freeman B, Li M (2016). Can mobile phone apps influence people's health behavior change? An evidence review. *JMIR*. 18(11): e287.
- Zimmet PZ (2017). Diabetes and its drivers: the largest epidemic in human history? *Clin Diabetes Endocrinol*.3(1):1-8.
- Zurita L, Nøhr C. Medinfo (2004). Patient opinion-EHR assessment from the users' perspective. *Stud Health Technol Inform.* 107(pt 2): 1333-1336.

APPENDICES

Appendix 2.1: Some of the developmental considerations in the reviewed studies in Chapter 2b

Author, Year	Name of app / Origin (existing or developed for the study)	Users involvement	Clinical expert involvement	Functions and functionalities of the app
Kirwan et al., 2013	Glucose buddy / existing	Not reported	Not reported	Manual logs of BG, diet (food items in gms), physical activities, insulin dosages and other medications into app. Analytic function, which can present data in a customised graphical format. Allows users' to forward data via email. Also have a reminder function.
Quinn et al., 2011	Not specific, just called Smart Phone app / developed for the study	Not reported	Mobile app was designed by endocrinologists and Credentialed Diabetes Educators (Quinn et al., 2008)	Manual logs of BG, CHO intake, medications and other diabetes management information into app. In response to specific logged data, users can receive automated real-time educational, behavioural and motivational messages. Additionally, App could be used to forward data to physician.
Charpentier et al., 2011	Diabeo / developed for the study	Not reported	Not reported	Manual logs of BG, CHO counts, planned activity into app to calculate insulin dose or CHO adjustment. Data transfer to provider through GPRS and Secured website.
Quinn et al., 2016	See Quinn et al., 2011	Not reported	See Quinn et al., 2011	See Quinn et al., 2011

Waki et al., 2014	DialBetics / not reported	Not reported	Not reported	Automated logs of BG, BP, weight and pedometer count for physical activity level into app. Dietary evaluation of inputted meal through a feedback from a dietician (not automated; feedback takes 2 days). Analysis of BG and BP data with graphic outputs of measurements. Reminders for self-monitoring. Also, have a natural language processing method such that patient voice/text messages about meal and exercise that is not counted by the pedometer are sent to the server and the voice input is converted to text and matched with text in the app. Automated feedback advice on lifestyle modification in response to logged data. Alert for missed or late readings.
Orsama et al., 2013	Monica / developed for the study	Not reported	Not reported	Manual logs of BP, BG, weight and physical activity into app. Analytic function in form of graphs. Participants received automated real time education, behavioural skills and motivational messages in response to logged data.
Kim et al., 2014	Henceforth app / developed for the study	Not reported	Not reported	Manual logs of BG and BP into app with automated transfer to participants hospital.
Rossi et al., 2010a	Diabetes Interactive Diary / developed for the study	Not reported	Not reported	Manual logs of BG and insulin dose into app. Analysis to provide automated suggestion for insulin bolus dose and CHO count. General pre-stored data on; BG values, individualised correction factor: CHO ratio set by physician, food intake and physical activity. Logged data could be sent on an average of 1-3 weeks as short messages to physician for review purpose to provide feedback to the patient via text messages. (Rossi et al., 2009)

Holmen et al., 2014	The Few Touch App / developed for the study	12 - 15 end users were involved in the design process. Through incorporation of focus group meetings, semi- structured interviews, usability testing, questionnaires, and paper prototyping. Results generated the design requirements and answers to research questions (Arsand et al., 2010)	Not reported	Automated log of BG readings from meter into app. Manual log of physical activity and food intake. Analytic function in form of visual graph, trend reports and feedbacks through colour coding (below normal, normal and above normal). Provided general information on disease. It also have a personal goal setting system and automated data transfer of logged data into a secure server.
Rossi et al., 2013	See Rossi at al., 2010a	See Rossi et al., 2010a	See Rossi et al., 2010a	See Rossi et al., 2010a
Istepanian et al., 2009	Not specific just generally called m- health system / not reported	Not reported	Not reported	Automated log of BG from meter into App. Alert to remind users when measurement are due. Data transfer from mobile phone to a hospital server.

BG: Blood glucose; CHO: carbohydrate; GPRS: General Packet Radio Service; BP: Blood pressure
Appendix 2.2: Other developmental considerations and key clinical outcome-glycosylated hemoglobin in the reviewed studies in Chapter 2b

Author, Year	Data security and privacy consideration	Pilot testing	Health behavioural theory consideration	Key Clinical Outcome (HbA1 _c) ¹
Kirwan et al., 2013	Not reported	Not reported	Not reported	The intervention group showed a significantly improved HbA1 _c from baseline to 9-month follow up compared to the control group.
Quinn et al., 2011	Data security in the mobile phone app was not reported. But for the web-based system; BG data were captured in real time into a HIPAA ³ -compliant secured web-based system, where it is processed to receive personalised feedback. (Quinn et al., 2008, Quinn & Gruber- Baldin, 2009)	Pilot tested for 3 months on 30 participants with type 2 diabetes with the aim to assess the impact of the software on HbA1 _c , and assessed provider's and patients satisfaction with the technology (Quinn et al., 2008)	Not reported	One of the intervention group (maximal treatment) showed a significantly improved HbA1 _c from baseline to 12-month follow up compared to other intervention and control groups.
Charpentier et al., 2011	Data logged into app were automatically uploaded into a secured website, where they are available to the investigators at any time	Four-month observational study of 35 type 1 diabetes patients. The aim was to confirm the use of personalised flexible intensive insulin therapy results in good control of the postprandial state. (Frac et al., 2009)	Not reported	The full intervention group showed a significantly improved HbA1 $_{\rm c}$ from baseline to 6 month follow up compared to the control group.
Quinn et al., 2016	See Quinn et al., 2011	See Quinn et al., 2011	See Quinn et al., 2011	Both the intervention and control group had reduced improvement in HbA1 _c after 12 months follow up.

Waki et al., 2014	Not clear. Reported that participants measured data were transmitted to a server, following each new measurement that patient profile was updated which controlled access to patient's data and recorded access history.	One month pilot testing on 11 participants with type 2 diabetes. The aim was to assess the safety, usability and impact of mobile app on HbA1 _c and the effect on home BP ⁴ monitoring as a way of managing the complications related to diabetes. (Waki et al., 2012)	Not reported	The intervention group showed a significantly improved HbA1c from baseline to 3-month follow up compared to the control group.
Orsama et al., 2013	Not reported	Not reported	Information- Motivation behavioural skill model was used as the basis for the formulation of automated personalised feedback message contents	The intervention group showed a significantly improved HbA1 _c from baseline to 10-month follow up compared to the control group.
Kim et al., 2014	Not reported	Not reported	Not reported	Both the intervention and control groups had a non- significantly improved HbA1 _c from baseline to 3- month follow up.

Rossi et al., 2010a	Not reported	The first pilot study was done with a questionnaire to assess the feasibility and acceptability of the app. A second pilot study was done on 41 patients using DID under routine clinical practice condition were evaluated after a medium of 9 months of follow up. The aim is to investigate effectiveness of the app on metabolic control. The second pilot study was on 50 people with type 1 diabetes aged 18-65 years, with the aim of investigating the feasibility and acceptability of the app (Rossi et al., 2009)	Not reported	Both the intervention and control groups had a non- significantly improved HbA1 _c from baseline to 6- month follow up.
Holmen et al., 2014	Not reported	12 people with type 2 diabetes participated in the testing of the app during an average test period of 167 days (Arsand et al., 2009)	Not reported.	Both the intervention and control groups had a non- significantly improved HbA1 _c from baseline to 12- month follow up.
Rossi et al., 2013	See Rossi et al., 2010a	See Rossi et al., 2010a	See Rossi et al., 2010a	Both the intervention and control groups had a non- significantly improved HbA1 _c from baseline to 6- month follow up.
Istepanian et al., 2009	Not reported	Not reported	Not reported	HbA1 _c in the intervention and control groups remain unchanged after 9-month follow up.

HbA1c: glycosylated hemoglobin; BG: Blood glucose; HIPAA: Health Insurance Portability and Accountability Act

Appendix 3.1: Interview guide utilised in Chapter 3

Prompts:

General prompt on key areas of diabetes self-management. This includes regular participation in physical exercise, eating the ideal diet, regular monitoring of blood glucose, problem solving such as correcting hypoglycemia and hyperglycemia, healthy coping with the condition, reducing risk to help prevent or minimise diabetes complications (for example limiting alcohol intake, smoking cessation, regular eye examination, foot care, oral health).

For those who have type 1 diabetes, in addition to the above, other areas of diabetes management were buttressed including, adjusting insulin and diet to avoid hypoglycemia/hyperglycemia, adjusting insulin/exercise to avoid hypoglycemia.

Enablers

- 1. Can you identify some strategies you find particularly helpful to facilitate or support you to perform your diabetes management activities?
- 2. Are there things (people or services) that have made it easier for you to manage your diabetes?
- 3. What are the factors that have contributed to success in your self-management
- 4. Anything else to add?

Barriers

5. What areas (if any) of your diabetes management do you find difficult?

Probe: What makes this /these (mention the area(s) stated) difficult?

- 6. On a day to day basis, is there anything you struggle with or find difficult in your diabetes management?
- 7. Are there factors that have served as obstacles to managing your diabetes?
- 8. What do you believe are the barriers to your self-management?
- 9. Anything else to add?

Appendix 4.1: The quantitative instrument utilised in Chapter 4

Preface:

Welcome to this survey and thank you in advance for your interest to participate. The survey aims to examine the use and preference of features in mobile apps to manage diabetes. The result of this study will help in gaining better understanding of the important features and components to consider when developing future apps for diabetes self-management.

Your responses, including your demographics will remain anonymous. At the end of the survey, you will be asked if you are interest in participating in additional telephone interview, if you choose to provide your phone number, your responses to the phone interview may not be anonymous to the researcher. However, you will not be identified in any way in the research publications and the results from this study.

Eligibility:

- Only people who have type 1 or type 2 diabetes and
- 18 years and above

Instructions to complete the survey:

Please read the following questions carefully. Your candid responses are essential to ensure the reliability of this study

1. What is your gender?

 \Box Male \Box Female \Box Others (Pls. specify)

2. How old are you? _____(years)

3. What is the highest level of education you have completed?

 \Box High school or equivalent \Box Technical or vocational education

 \Box College \Box Bachelor degree \Box Post graduate degree

 \Box No formal education \Box Others

4. What is your employment status?

□ Employed	□ Unemployed	\Box Retired
------------	--------------	----------------

5. What continent do you presently live in or reside?

 \Box America \Box Asia \Box Australia \Box Europe

□ Africa □ Antarctica

6. What type of diabetes do you have?

 \Box Type 1 \Box Type 2 \Box Gestational

 \Box I don't have diabetes \Box Others (Please specify)

7. How long have you been diagnosed with diabetes (years)?

 $\square < 1$ $\square 1-5$ $\square 6-10$ $\square 11-15$ $\square \ge 15$ years \square Never

Section B

1. Do you currently have apps on your mobile phones that provides support related to your diabetes management?

Yes \Box (please go to question 2)

No \Box (please go to question 5)

- 2. If yes, what are the features in the apps available on your smart phone which you use to manage your diabetes (tick all that applies to you).
 - \Box Blood pressure tracker
 - \Box Blood glucose tracker
 - □ Reminder (e. g to take medication or attend a health appointment)
 - □ Fitness / exercise monitor
 - □ Body weight tracker
 - \Box Food calorie counter
 - \Box Feature to transfer electronic health information to your doctor
 - □ Others (Please Specify.....)

3. Which of these features in apps do you use regularly? Only tick the ones you use at least 4 times in a week

- \Box Blood pressure tracker
- \Box Blood glucose tracker
- □ Reminder (e. g to take medication or attend a health appointment)
- \Box Fitness/ exercise monitor
- □ Body weight tracker

- \Box Food calorie counter
- □ Feature to transfer electronic health information to your doctor
- □ Others (Please Specify.....)
- 4. Are there specific reasons why you don't use other features regularly? Please state them:
- 5. If No to question 1 of this section, which of the following is a reason for not using apps to support your diabetes management (tick all that applies to you).
 - \Box I am not interested
 - \Box I am not aware
 - \Box Lack of smart phone
 - \Box It is expensive
 - \Box Limited access to the internet
 - □ Others (please specify.....)
- 6. Please rate each of these app features by indicating the extent to which you think they are useful to support your diabetes self-management. Pick only one number for each of the app feature. Score 10 being the most useful and 1 being the least useful.

Health Applications			Percei							
	Mos	t useful								Least useful
	10	9	8	7	6	5	4	3	2	1
Blood pressure tracker										
Blood glucose tracker										
Food nutrient composition										
Fitness / exercise monitor										
Body weight monitor										
Data transfer to health care team (e. g doctor)										

Task reminder app (e.g to take					
medication, appointment with					
doctor)					

- 7. If offered the option of using a new mobile diabetes application (app) to support your diabetes management? Please choose the features you would like the app to have. (Tick all that applies to you).
 - \Box Provide access to nutritional information on nutrient contents of foods
 - \Box Log blood glucose levels
 - □ Physical exercise tracker
 - □ Task reminders (e.g reminder to take my medication)
 - □ General information on diabetes; its management and complication prevention
 - □ Graphical display of logged data
 - □ Logs and display of body weight maintenance
 - □ Feature that allows me to network with other people having diabetes

8. Are there other suggestions or improvements that is not listed which you think are very important in apps that support the management of diabetes

Would you be interested to be contacted for an individual interview session? The session is to hear your experience on managing diabetes and your opinion on how mobile phone apps could be improved to manage diabetes and some of the important components to include in future apps for diabetes management. If yes, please indicate and provide your phone number and best time to contact you.

 \Box Yes

Phone number......Best time to contact.....

 \Box No

Appendix 4.2: Interview guide utilised in Chapter 4

Introduction of study and interviewer:

Hello, my name isI'm calling in regards to an online questionnaire that you participated in a while ago relating to diabetes management and the use of mobile phone application (app). You left a phone number to indicate your interest in participating in a follow up interview to the survey and a text was sent to you few days ago. The interview will take about 10 minutes, is it ok to proceed with the interview now or you would like me to call back at a later time?

Thank you for your willingness to participate and be interviewed. Please note that I will be recording this interview, and it will be transcribed and analysed. You are free to choose what to disclose for any of your answers to the questions or refuse to answer a particular question. The aim of the interview is to gather information on how apps could be improved to self-manage diabetes. An app is any program downloadable to a smart phone, which is used to support any aspect of your diabetes management to foster improved health outcomes. It could be an app for lifestyle monitoring such as those for tracking diet and exercise or for monitoring your clinical outcomes such as blood glucose or blood pressure.

We have observed that often times, when a lot of people with diabetes use apps to manage any aspect of their health, their interest in using the app decreases over time. I would like to explore your opinions on what to include in apps to prevent such decrease in usage over time.

Questions:

- 1. Do you currently use an app to support your diabetes care?
 - a. If yes, what are the ones you use/what aspect of your diabetes care do they support?
 - b. If no, have you in time past used any?

311

c. If yes to question b above, what are your reasons for stopping? (if no, reasons for non-usage)?

2. What do you think could be the reason why people with diabetes use an app for a period of time with a subsequent decrease / stoppage in its use over time?

3. What do you think would be a motivator to encourage you to continually use an app for your diabetes management?

4. Is there any feature or function you would consider very important for inclusion in a good app which could stimulate your interest to use it regularly?

5. What are possible improvements in app features that might motivate you to use it regularly?

6. Are there other aspects/components/features you would like to see in future apps, which could encourage you to use it more frequently and regularly? What are your expectations of future apps to manage diabetes?

For all questions involving app features, there were specific probes on features for blood glucose monitoring, activity tracking, reminder, features on nutrition to foster healthy eating, etc.)

Educational topics desired in diabetes apps

 When it comes to diabetes and its management, what sort of education or information would you desire to be included in future apps to support diabetes management? (Specifically probe on healthy eating, preventing diabetes complications, risk reduction, engaging in physical activities, adherence to medication intake, responding to challenges of diabetes clinical outcomes and self-management).

Appendix 5.1: Usability instrument utilised in Chapter 5

Introductions

Welcome to this survey and thank you in advance for your interest to participate. This survey aims to examine the usability (how easy and pleasant the features) of a newly developed mobile app (My Care Hub) to support diabetes self-management. The outcome of this survey will be of great value to determine the area of improvement, which may be required in the app. The survey should take approximately 10 minutes. Your responses including your demographic information will remain anonymous. Participation is voluntary.

Instructions on completing the survey

Please read and consider the following questions carefully. Your candid responses are essential to ensure the reliability of this study.

Section A: Demographic

Below are questions about your background. Please pick one response for the multiple-choice questions.

1.	What is	your	gender?
----	---------	------	---------

 \Box Male \Box Female \Box Other (Please specify)

2. How old are you? (years)

3. What is the highest level of education you have completed?

 \Box High school or equivalent \Box Technical or vocational education

 \Box College \Box Bachelor degree \Box Post graduate degree

 \Box No formal education \Box Other

4. What type of diabetes do you have?

 $\Box Type 1 \qquad \Box Type 2 \qquad \Box I don't have diabetes (please move to$

section B) \Box Other (Please specify)

5. How long have you been diagnosed with diabetes (years)?

 $\square < 1$ $\square 1-5$ $\square 6-10$ $\square 11-15$ $\square \ge 15$ years \square Never

6. What is the present fasting blood glucose range recommended by your health care provider? Please input your response in mmol/L. For example 4 to 6 mmol/L

.....

7. What is the present 2 hours post prandial blood glucose range recommended by your healthcare provider? Please input your response in mmol/L. For example 6 to 10mmol/L.

.....

Section B

The next set of questions are related to your experiences with the use of My Care Hub App over the last 7 days. Please answer each question based on your experience with the app. Your candid responses are essential to ensure the reliability of this information.

Functionality

1. **Performance**: How accurate/ fast do My Care Hub features (functions) and components (buttons/ menu) work?

1. App is broken; no/ insufficient/inaccurate response (e.g crashes/bugs/broken features e.t.c)

- 2. Some functions work, but lagging or contains major technical problems
- 3. App works overall. Some technical problems need fixing/Slow at time
- 4. Mostly functional with minor/negligible problems
- 5. Perfect/timely response; no technical bugs found
- 2. **Ease of use:** How easy is it to learn how to use My Care Hub; how clear are the menu labels/icons?
 - 1. Menu labels/icons are confusing; complicated
 - 2. Useable after a lot of time/effort
 - 3. Usable after some time/effort
 - 4. Easy to learn how to use the app
 - 5. Able to use app immediately; intuitive; simple
- 3. **Navigation:** Is moving between My Care Hub screens logical /accurate/appropriate/uninterrupted; are all necessary screen link present?

1. Different sections within My Care Hub app seem logically disconnected and random/confusing/navigation is difficult

- 2. Usable after a lot of time/effort
- 3. Usable after some time/effort
- 4. Easy to use or missing a negligible link
- 5. Perfectly logical, easy, clear and intuitive screen flow throughout or offers shortcut
- 4. **Gestural design:** Are interaction (taps/scrolls) consistent and spontaneous across all components/ screens of My Care Hub?
 - 1. Completely inconsistent/confusing
 - 2. Often inconsistent/confusing
 - 3. OK with some inconsistencies/confusing elements
 - 4. Mostly inconsistent/spontaneous with negligible problems
 - 5. Perfectly consistent and spontaneous

Aesthetics

5. **Layout:** Is arrangement and size of buttons/icons/menus/content on the apps' screen appropriate?

1. Very bad design, cluttered, some options impossible to select/locate/see/read device display not optimised

2. Bad design, random, unclear, some options difficult to select/locate/see/read

3. Satisfactory, few problems with selecting/locating/seeing/reading items or with minor screen size problems

4. Mostly clear, able to select/locate/see/ read items

5. Professional, simple, clear, orderly, logically organised, device display optimised. Every design component has a purpose

6. **General Graphics:** How high is the quality/resolution of graphics used for buttons/icons/menus/content of My Care Hub?

1. Graphic appear amateur, very poor visual design-disproportionate, completely stylistically inconsistent

2. Low quality resolution graphics; low quality visual design-disproportionate, stylistically inconsistent

3. Moderate quality graphics and visual design (generally consistent in style)

4. High quality/resolution graphics and visual design-mostly proportionate, stylistically consistent

5. Very high quality/resolution graphics and visual design-proportionate, stylistically consistent throughout

- 7. **Visual appeal:** How good does My Care Hub look?
 - 1. No visual appeal, unpleasant to look at, poorly designed
 - 2. Little visual appeal-poorly designed, bad use of colour.
 - 3. Some visual appeal-average, neither pleasant nor unpleasant
 - 4. High level of visual appeal-seamless graphics-consistent and professionally designed

5. As above + attractive, memorable, stands out; use of colour enhances app features/menu

8. **Graphic visual appeal of the analytics:** is the visual display of the log of graphs clear and understandable? Note, this refer to the analytic function called "view insight" present in My Care Hub?

1. Graphic appear amateur, very poor visual design-disproportionate, completely stylistically inconsistent

2. Low quality/resolution graphics; low quality visual design-disproportionate, stylistically inconsistent

3. Moderate quality graphics and visual design (generally consistent in style)

4. High quality/resolution graphics and visual design-mostly proportionate, stylistically consistent

5. Very high quality/resolution graphics and visual design-proportionate, stylistically consistent throughout

9. What is your overall rating of My Care Hub?

- 1. * One of the worst apps have used
- 2. **
- 3. *** Average
- 4. ****
- 5. **** One of the best apps have used

Additional Questions for participants who have diabetes

Satisfaction with My Care Hub for Diabetes Self-Management

- 1. **Usefulness**: Features of My care Hub are useful for diabetes self-management.
 - 1. Strongly Disagree
 - 2. Disagree
 - 3. Neither agree nor disagree
 - 4. Agree
 - 5. Strongly Agree
- 2. **Target group:** Is My Care Hub features appropriate for people with type 1 or type 2 diabetes?
 - 1. Completely inappropriate/unclear/ confusing
 - 2. Mostly inappropriate/unclear/confusing
 - 3. Acceptable but not targeted. May be inappropriate/unclear/confusing
 - 4. Well targeted, with negligible issues
 - 5. Perfectly targeted, no issues found
- **3. Quality of information:** Is My Care Hub educational content well written, understandable and appropriate for people with type 1 or type 2 diabetes? This refers to the feature called "Educational tips" in My Care Hub and the messages under it.
 - 1. Irrelevant/inappropriate/incoherent/incorrect
 - 2. Poor. Barely relevant/appropriate/may be incorrect
 - 3. Moderately relevant/appropriate/coherent/correct
 - 4. Relevant/appropriate/coherent/correct
 - 5. Highly relevant, appropriate, coherent and correct

- 4. **App feedback**: Is the feedback messages in My Care Hub understandable and appropriate for your type of diabetes? This refers to the feedback (pop-up) messages displayed when blood glucose data is inputted into the app.
 - 1. Irrelevant/inappropriate/incoherent/incorrect
 - 2. Poor. Barely relevant/appropriate/may be incorrect
 - 3. Moderately relevant/appropriate/coherent/correct
 - 4. Relevant/appropriate/coherent/correct
 - 5. Highly relevant, appropriate, coherent and correct
- 5. Awareness: My Care Hub educational information is useful for improving knowledge/awareness about the importance of diabetes self-management
 - 1. Strongly Disagree
 - 2. Disagree
 - 3. Neither agree nor disagree
 - 4. Agree
 - 5. Strongly Agree
- 6. **Motivation:** The use of My Care Hub is likely to increase motivation of people with diabetes to engage in their self-management activities.
 - 1. Strongly Disagree
 - 2. Disagree
 - 3. Neither agree nor disagree
 - 4. Agree
 - 5. Strongly Agree
- 7. **Intention to use:** If I have continual access to this app. I predict that I could use it.
 - 1. Strongly Disagree
 - 2. Disagree
 - 3. Neither agree nor disagree

- 4. Agree
- 5. Strongly Agree
- 8. **Recommendation:** Would you recommend this app to people with type 1 diabetes (T1D) or Type 2 Diabetes (T2D)
 - 1. I would not recommend this app to anyone with T1D or T2D
 - 2. I could recommend this app to few people with T1D or T2D
 - 3. I could recommend My Care Hub to several people with T1D or T2D
 - 4. I could recommend My Care Hub to many people with T1D or T2D
 - 5. I could recommend My Care Hub to anyone with T1D or T2D

General impressions

- 1. Is there anything about My Care Hub that you like in particular?
- 2. Anything that you did not like. Anything that annoyed you?
- 3. Are there other observations you noted during your use of My Care Hub or you have some suggestions, please state it.



Appendix 6.1: Other Screen Shots for My Care Hub that were published in Chapter 6

Sample Feedback Messages 8 Message 3 A Limiting or avoiding additional sugar in foods BG Level seems too Low. could be very helpful for BG within target range. diabetes management Early symptoms of hypo are headache, blurry vision, fatigue, Keeping blood glucose levels irritability, hunger.. 1 1 within target reduces the risk of complications. OK 0.0 Kg ave Wt 0.0 mins ave Activity LOG WEIGHT LOG ACTIVITY OK

Sample Push Notification

Appendix 6.2: Semi-structured interview guide utilised in Chapter 6

Prologue

1. Introduce myself and the reason for calling etc.

Set the focus of the interview and explain the app and the previous 3 weeks' study. Use the following script:

"Thank you for agreeing to take part in this interview and for using My Care Hub app in the past 3 weeks. I want to understand what you think about the app. The app is aimed to provide support and education to foster diabetes self-management by providing a platform to document and monitor self-management activities, give feedback on inputted blood glucose levels per time, provide information within the app on how to self-manage diabetes. In this interview, we aim to know your opinion and experience with the app during the course of usage. There are no right or wrong answers to my questions."

- 2. Give opportunity for questions (if any),
- 3. Explain that the interview will be recorded:

"I would like to record what you say as that saves me having to scribble when you're talking and give opportunity for me to concentrate on what you're saying. The interview will be transcribed and your identity will be anonymised in the published work. Is that okay with you?

Questions.

Before you go into details about your experience with the app, I want to ask you some background information about your health and personal self-management

Opening questions

• Can you tell me, what type of diabetes you have and how long you have been diagnosed?

• Can you tell me how or the various steps you take to manage your diabetes? *e.g Therapeutic regimen, mediation intake, frequency of exercise, diet management?*

(Please note, if participant has mentioned the type of diabetes from interview start, please refer to it in particular when asking these questions, instead of using the Phrase Your type of diabetes''

Experience of using My Care Hub

- Did you use My Care Hub?
 - How often
 - If the app was not used, why?
- Describe your experience of learning how to use the app
 - Did you have any problem installing/using the app?
- Tell me about your experience with using My Care Hub app over the past 3 weeks?
 - How did you find navigating the pages and finding information?
 - How did you find inputting your data into the app?
 - Was there anything you find particularly hard to use?
 - If you have problems with the app, what were they?
 - How did you find using My Care Hub for your diabetes self-management?

Views about features provided

- What are your views about the feature where you input your blood glucose? -How did you find inputting blood glucose into that feature?
- What are you views about the automated feedback messages received immediately after inputting your blood glucose into the app?
 Were they useful?
 What was your response to it?
- Did you use the analytic feature to monitor trends? -If not, why?

-If yes, what are your views about it?

- Did it help you in your diabetes management? How?
- What are you views about the diabetes educational tips embedded in the app? i.e the feature containing information on what diabetes is, carbohydrates in foods, healthy eating, exercise, medication usage, monitoring of blood glucose

e.t.c

-What do you think of them?-Did you find them useful?- Is there any other useful information that you would have liked to be included in the app?

• What are your views about the daily educational messages that are intermittently

displayed in the app (push notifications)?

-Was there anything you dislike about this messages in particular? - Did you find them useful?

- What feature(s) in the app did you find most useful? Why?
- What feature(s) in the app did you find least useful? Why?
- What do you think about the overall content and functionality of the App?
- Was there anything you disliked in the app?

Perceived ongoing benefit /Impact/intention for continue usage

- What ongoing benefit do you think there might be for patients with your type of diabetes who continue to use My Care Hub app?
- Is there anything you would do differently regarding your self-management following your participation in the study?
- What factors/features in the app interest you most which could help you engage regularly with the app.
- Were there benefits you derived from using the app during the intervention period?

- What are your views about the educational component of the app?
- Would you continue to use My Care Hub, if it were available?
- Would you recommend the app to other people with your type of diabetes?

Recommendations

- Do you have suggestions on how the app could be improved?
- Do you think any features should be removed from the app?

-If yes, please explain

- Was there anything you wanted or expected to see in the app but didn't?
- What would you ask the app developers to change? E.g colour, layout, icons, font size, any feature.

Is there anything we have not talked about that you think it is important to share before we end the interview?